

WestminsterResearch

http://www.westminster.ac.uk/westminsterresearch

Mitochondrial Function as a Tool for Assessing Function, Quality and Adulteration in Medicinal Herbal Teas Woodley, S.

This is a PhD thesis awarded by the University of Westminster.

© Mr Steven Woodley, 2024.

https://doi.org/10.34737/wwwvy

The WestminsterResearch online digital archive at the University of Westminster aims to make the research output of the University available to a wider audience. Copyright and Moral Rights remain with the authors and/or copyright owners.

University of Westminster

Mitochondrial Function as a Tool for Assessing Function, Quality and Adulteration in Medicinal Herbal Teas

Steven Woodley

May 2024

A thesis submitted in partial fulfilment of the requirements of the University of Westminster for the degree of Doctor of Philosophy

Abstract:

Introduction: A review of current and historical herbal quality assessment methods revealed adulteration with low potency herbal material to be challenging. Mitochondrial functional analysis presents a novel opportunity to address this deficiency. Liquorice (*Glycyrrhiza glabra* L., *Fabaceae*) was selected as an initial herb to profile due to its ubiquity in herbal medicine and commercial teas.

Methods: MCF-7 and MCF-10A cells were seeded and treated for 24 hours with an original liquorice tea extract (OLTE; Three Licorice Tea; Pukka, UK) and subjected to mitochondrial functional assays. Use of pre-extracted material was selected as a method of functional adulteration and the effects of a pre-extracted liquorice tea extract (XLTE) was then compared to OLTE assays with significant results. HPTLC was used for comparison to existing industry methods.

Results: 2000 µg / mL OLTE induced a 14.88% (p < 0.05) rise in reactive oxygen species (ROS) in MCF7 cells after 3 hours followed by a 14.73% (p < 0.05) drop at 72 hours. Oxygen consumption (OCR) rate dropped 72.6% (p < 0.01) and 85.16% (p < 0.01) in MCF10A cells treated with 500 and 2000 µg / mL OLTE respectively. The effects of mitochondrial modulating drugs were reduced in MCF7 cells pre-treated with 2000 µg / mL OLTE (oligomycin: 29.94%, p < 0.01; FCCP: 43.2%, p < 0.01; antimycin A & rotenone: 32.09%, p < 0.01). Further investigation revealed improved survival in 2000 µg / mL OLTE pre-treated MCF10A cells given cisplatin (50 µM: 11.79%, p < 0.05; 100 µM: 8.186%, p < 0.001), but had no effect on MCF7 cells. XLTE also demonstrated a 18.7% (p < 0.0001) drop in ROS in MCF7 cells treated with 2000 µg / mL at 72 hours but no effect at 3 hours. XLTE also decreased basal OCR in MCF10A cells at all doses (10 µg / mL: 21.12%, p < 0.05; 500 µg / mL: 53.52%, p < 0.0001; 2000 µg / mL: 56.41%, p < 0.0001), but no effect on drug resistance. HPTLC comparison revealed that differentiating OLTE from XLTE at identical strength was challenging.

Discussion: Possible mechanisms are discussed, proposing electron capture for the reduction in ROS and OCR without other modulations being observed, and efflux transporter expression explaining drug resistance. Drug resistance was the most consistent quality marker and is related to clinical use. Implications, further tests and generation of a mitochondrial functional testing pipeline are presented.

Contents:

Abstract	1
Contents	2
Index of Figures	12
Index of Tables	20
Acknowledgements	21
Declaration of Contributions	22
List of Abbreviations	23

Intr	oduc	tion		28
1.	The	lssues	Associated with Quality in Medicinal Herbal Teas	28
	1.1.	The I	Modern Herbal Tea Industry	28
	1.2.	A Bri	ef History of Herbal Adulteration and Quality Control	31
		1.2.1.	Adulteration and Quality Testing in the Ancient World	31
		1.2.2.	The Scientific Revolution	33
		1.2.3.	The Industrial Age	35
2.	Liter	ature F	Review of Existing Methods of Quality Control in	
	Med	icinal F	lerbal Teas	38
	2.1.	Sear	ch Methods	38
	2.2.	Chro	matography	42
		2.2.1.	Liquid Chromatography	43
		2.2.2.	Gas Chromatography	46
		2.2.3.	Thin Layer Chromatography	48
	2.3.	Spec	troscopy / Spectrometry	52
	:	2.3.1.	Mass Spectrometry	52

		2.3.2.	Nuclear I	Magnetic Resonance	56
		2.3.3.	Ultraviole	et-Visible Spectrophotometry	59
		2.3.4.	Infrared S	Spectroscopy	62
		2.3.5.	Optical E	mission & Atomic Absorption Spectroscopy	64
	2.4.	Biolo	gical Metho	ods	68
		2.4.1.	DNA Bar	coding and Genetic Analysis	68
		2.4.2.	Bioassay	'S	75
	2.5.	The L	imitations	of Existing Methods of Herbal Quality Control	93
	2.6.	Func	ional Adul	teration and the Problem of Functional Testing	95
		2.6.1.	Functiona	al Adulteration	95
		2.6.2.	Adulterat	ion and Substitution	99
		2.6.3.	Functiona	al Testing	100
3.	Mit	ochondı	ia as a Po	tential Focal Point for Functional Testing	102
	3.1.	The F	Role of Mite	ochondria in Health and Disease	102
		3.1.1.	Respirati	on	102
		3.1.2.	Reactive	Oxygen Species	105
		3.1.3.	Immune	Responses	109
		3.1.4.	Apoptosi	S	111
		3.1.5.	Hormesis	s and Adaptation to Stress	113
		3.1.6.	Ageing a	nd Degeneration	118
		3.1	.6.1. Ne	eurodegeneration	118
		3.2	.6.2. Ca	ardiovascular Disease	120
		3.2	.6.3. Ca	ancer	122
		3.1.7.	Drug Res	sistance	124
	3.2.	The F	Role of Mite	ochondria in Traditional Medical Systems	125
		3.2.1.	Hormesis	s in Traditional Medicine	128
		3.2.2.	Exercises	S	128

	:	3.2.3.	Dieta	ry Interventions	129
	;	3.2.4.	Manu	al Therapies	130
	;	3.2.5.	Herba	al Medicine	132
		3.2	.5.1.	Antioxidants	134
		3.2	.5.2.	Adaptogens	136
4.	Liqu	orice			139
	4.1.	Histor	у		141
	4.2.	Main	Constit	uents	142
		4.2.1.	Glycy	rrhizin, also known as Glycyrrhizic Acid	142
		4.2.2.	Glycy	rrhetinic Acid, also known as enoxolone	145
		4.2.3.	Glabr	idin and Glabrene	146
	4	4.2.4.	Liquir	itin, its precursors, isomers and apiosides	147
		4.2.5.	Glycy	coumarin	148
		4.2.6.	Licocl	halcones	149
	4.3.	Use ir	n West	ern Herbal Medicine	149
	4.4.	Use ir	ו Ayur	edic and Unani Medicine	150
	4.5.	Use ir	n Chine	ese Herbal Medicine	150
	4.6.	Risks	Assoc	iated with Liquorice	152
	4.7.	Qualit	y Evalı	uation of Liquorice	153
	4.8.	Adulte	eration	of Liquorice	155
	4.9.	Curre	nt Mito	chondrial Research into Liquorice	158
5.	Rese	earch A	im, Ob	jectives, Questions and Hypothesis	160
	5.1.	Resea	arch Ai	m	160
	5.2.	Resea	arch Ol	bjectives	160
	5.3.	Resea	arch Q	uestions	161
	5.4.	Hypot	hesis		161

Materials and Methods

Ivia	terials	s and methods	162			
6.	Cells	s and Culture				
	6.1.	Cell Lines	162			
	6.2.	Cell Culture	163			
7.	Herb	al Tea	164			
	7.1.	Original Liquorice Tea Extract Preparation	164			
	7.2.	Pre-extracted Liquorice Tea Extract Preparation	165			
	7.3.	Treatment Protocols	167			
8.	Mitoo	chondrial Function Assays	169			
	8.1.	Cell Viability	169			
	8.2.	Detection and Quantification of Reactive Oxygen Species	171			
	8.3.	Detection and Quantification of Mitochondrial Membrane				
		Potential	174			
	8.4.	Detection and Quantification of Calcium lons	175			
	8.5.	Observation of Mitochondrial Morphology via Confocal				
		Microscopy	176			
	8.6.	Mitochondrial Respiration	176			
	8	3.6.1. Protein Assays for SeaHorse Normalisation	179			
	8.7.	Statistical Analysis	180			
9.	Drug	Resistance Assays	181			
	9.1.	Seahorse Drug Resistance Assay	181			
	9.2.	Cisplatin Resistance Assay	182			
	9.3.	Statistical Analysis	183			
10.	Adul	teration Assays	184			
	10.1.	Adulteration Detection with Mitochondrial Functional Analysis	184			
	10.2.	Visual Inspection	184			
	10.3.	High Performance Thin Layer Chromatography	184			

Results

11.	 Results from Basic Mitochondrial Assays on Original Liquorice Tea Extract 					
	11.1.	Effects of Original Liquorice Tea Extract on Cell Viability	187			
	11.2.	Effects of Original Liquorice Tea Extract on Reactive Oxygen	188			
	11.3.	Effects of Original Liquorice Tea Extract on Mitochondrial Membrane Potential	190			
	11.4.	Effects of Original Liquorice Tea Extract on Calcium Ions	193			
	11.5.	Effects of Original Liquorice Tea Extract on Mitochondrial				
		Morphology	194			
	11.6.	Summary of Results from Basic Mitochondrial Assays on				
		Original Liquorice Tea	195			
12.	12. Results from Respiratory Measurements of Original Liquorice					
	Tea Extract in MCF7 Cells					
	12.1.	Effects of Original Liquorice Tea Extract on the Mitochondrial				
		Respiration and its Measurement in MCF7 Cells	196			
	12.2.	Error Checking the Results of Original Liquorice Tea Extract on Mitochondrial Respiration in MCF7 Cells	203			
	12.3.	Interpretation of the Results of Original Liquorice Tea Extract on Mitochondrial Respiration in MCF7 Cells as Protection				
		Against Mitochondrial Modulating Drugs	205			
	12.4.	Summary of Results from Original Liquorice Tea on Respiration in MCF7 Cells	209			
13.	Resu	Its from Respiratory Measurements of Original Liquorice				
	Tea E	Extract in MCF10A Cells	210			
	13.1.	Effects of Original Liquorice Tea Extract on Mitochondrial				
		Respiration and its Measurement in MCF10A Cells	210			

187

	13.2.	Interpretation of the Results of Original Liquorice Tea Extract on Mitochondrial Respiration in MCF10A Cells as Protection	
		Against Mitochondrial Modulating Drugs	216
	13.3.	Summary of Results from Original Liquorice Tea on	
		Respiration in MCF10A Cells	218
14.	Resu	Its from Original Liquorice Tea Extract on Drug Resistance	219
	14.1.	Effects of Original Liquorice Tea Extract on Oligomycin,	
		FCCP, Antimycin A and Rotenone Induced Perturbations in ROS in MCF7 Cells	219
	14.2.	Effects of Original Liquorice Tea Extract on Cisplatin Induced Reduction of Cell Viability in MCF7 Cells	221
	14.3.	Effects of Original Liquorice Tea Extract on Cisplatin Induced Reduction of Cell Viability in MCF10A Cells	222
	14.4.	Summary of Results from Original Liquorice Tea Extract on Drug Resistance	225
15.	Resu	Its from Basic Mitochondrial Assays on Pre-extracted	
	Liquo	price Tea Extract	226
	15.1.	Effects of Pre-extracted Liquorice Tea Extract on ROS in MCF7 Cells	226
	15.2.	Summary of Results from Basic Mitochondrial Assays of Pre-extracted Liquorice Tea	228
16.	Resu	Its from Respiratory Measurements of Pre-extracted	
	Liquo	price Tea Extract in MCF7 Cells	229
	16.1.	Effects of Pre-extracted Liquorice Tea Extract on Mitochondrial Respiration and its Measurement in MCF7	
	40.0		229
	16.2.	Interpretation of the Results of Pre-extracted Liquorice Tea Extract on Mitochondrial Respiration in MCF7 Cells as	
		Protection Against Mitochondrial Modulating Drugs	235

	16.3.	Sumi	mary of Results from Respiratory Measurements of	
		Pre-e	extracted Liquorice Tea in MCF7 Cells	237
17.	Resu	lts fro	m Respiratory Measurements of Pre-extracted	
	Liquo	orice T	ea Extract in MCF10A Cells	238
	17.1.	Effec	ts of Pre-extracted Liquorice Tea Extract on	
		Mitoc	hondrial Respiration and its Measurement in MCF10A	
		Cells		238
	17.2.	Interp	pretation of the Results of Pre-extracted Liquorice Tea	
		Extra	ct on Mitochondrial Respiration in MCF10A Cells as	
		Prote	ction Against Mitochondrial Modulating Drugs	244
	17.3.	Sumi	mary of Results from Respiratory Measurements of	
		Pre-e	extracted Liquorice Tea in MCF10A Cells	246
18. Results from Pre-extracted Liquorice Tea Extract on Drug				
	Resis	stance	in MCF10A Cells	247
	18.1.	Effec	ts of Adulterated Liquorice Tea Extract on Cisplatin	
		Induc	ced Reduction of Cell Viability in MCF10A Cells	247
	18.2.	Sum	mary of Results from Pre-extracted Liquorice Tea Extract	
		in Dr	ug Resistance	249
19.	Resu	lts fro	m Other Methods of Adulteration Detection	250
	19.1.	Visua	al Inspection	250
	19.2.	High	Performance Thin Layer Chromatography	251
	19	.2.1.	HPTLC Comparison of Species	254
	19.2.2.		HPTLC Comparison of HPTLC Comparison of	
			Extraction Method	255
	19	.2.3.	HPTLC Comparison of Source	256
	19	.2.4.	HPTLC Comparison of Original and Pre-extracted	
			Material	257
	19.3.	Sum	mary of Results from Other Methods of Adulteration	
		Dete	ction	260

Discussion

20.	Discussion of the Results 2				
	20.1.	The I	Effects of Liquorice Tea Extracts on Reactive Oxygen		
		Speci	ies	261	
	20).1.1.	Increases in Reactive Oxygen Species	261	
	20).1.2.	Decreases in Reactive Oxygen Species	265	
	20).1.3.	A Combined Explanation for Modulations in Reactive Oxygen Species	268	
	20.2.	The Meas	Effects of Liquorice Tea Extracts on Respiratory surements	270	
	20).2.1.	Possible Explanations from Clinical Uses of Liquorice	273	
	20).2.2.	Possible Explanations from Mechanistic Research	274	
	20).2.3.	Testing the Theoretical Mechanism	278	
	20.3.	Drug	Resistance Assays	281	
	20).3.1.	Detoxifying Effects of Liquorice	282	
	20	.3.2.	Development of the Drug Resistance Assays	286	
	20).3.3.	Cisplatin Resistance Assay Results	287	
21.	Adult	teratio	n Detection	290	
	21.1.	Ability	y of Mitochondrial Analysis to Detect Functional		
		Adulte	eration	290	
	21	.1.1.	Modulations in ROS	290	
	21	.1.2.	Respiratory Measurements	292	
	21	.1.3.	Cisplatin Resistance	294	
	21	.1.4.	A Question of Dose	296	
	21.2.	Comp	parison with Other Methods of Adulteration Detection	296	
	21	.2.1.	Visual Inspection	297	

261

	21	.2.2.	HPTLC Analysis	297
	21.3.	Relati	onship to the Hypothesis	299
	21.4.	Issue	s Associated with Mitochondrial Functional Analysis to	
		Detec	t Adulteration	299
22.	Limit	ations	of this Study	302
	22.1.	Metho	od of Adulteration	302
	22.2.	Mixing	g Adulterated and Original Extracts	304
	22.3.	Other	Herbs and Blends	305
23.	Impli	cations	s of the Results for Other Research	306
	23.1.	Implic	ations for Adulteration Research	306
	23.2.	Implic	ations for Research into Liquorice	307
	23.3.	Implic	ations for Chemotherapy Research	309
24.	Poter	ntial Fu	iture Work	314
	24.1.	Furthe	er Profiling of Liquorice	314
	24	.1.1.	Other Cell Lines	314
	24	.1.2.	Actions on Other Disease Models	317
	24	.1.3.	Comparison Between Species	320
	24.2.	Other	Forms of Adulteration	321
	24	.2.1.	Storage Conditions	322
	24	.2.2.	Wild Harvested vs. Cultivated Liquorice	324
	24	.2.3.	Mixing of Adulterated and Genuine Samples	325
	24	.2.4.	Real-World Sampling	326
	24.3.	Buildi	ng a Pipeline	326
	24.4.	Mitocl	hondrial Profiling of Other Herbs	329
	24	.4.1.	Cinnamon	329
	24	.4.2.	Ginger	331
	24	.4.3.	Теа	333

	24.4.4.	Ginseng	334
	24.4.5.	Ginkgo	334
	24.5. Mitoc	chondrial Profiling of Blends and Formulas	336
	24.5.1.	Cinnamon and Liquorice	337
	24.5.2.	Liquorice and Ginger	337
	24.5.3.	Peony and Liquorice	337
	24.5.4.	More Complex Blends	338
25.	Conclusio	on	341
26.	Reference	es	344

Index of Figures:

Figure 1: Illustrations from Chinese Pharmacopoeias	32
Figure 2: Illustration comparing seeds under magnification from Nehemiah Grew's <i>Anatomy of Plants</i>	34
Figure 3: Fraunhofer's observation of dark lines within the sun's spectrum	37
Figure 4: PRISMA Flow Diagram of the Literature Review Process	40
Figure 5: Overview of High Performance Liquid Chromatography	43
Figure 6: Example of an HPLC Chromatogram of G. glabra	44
Figure 7: Overview of Gas Chromatography	47
Figure 8: Overview of Thin Layer Chromatography	49
Figure 9: Example of an HPTLC Chromatogram	50
Figure 10: Functional Diagram of a Mass Spectrometer	53
Figure 11: Mass Spectra of Glycyrrhizin, the main constituent of	
Glycyrrhiza spp.	53
Figure 12: Principle of NMR Spectroscopy	57
Figure 13: Example of an NMR readout of G. glabra	58
Figure 14: Principle of Ultraviolet-Visible Spectrophotometry	60
Figure 15: Comparison of Underivatized HPTLC Chromatograms of	
Glycyrrhiza spp. Under Normal and Ultraviolet Light	61
Figure 16: Effects of Infrared Radiation on Vibrational Modes of Molecules	63
Figure 17: Fourier Transform Infrared Spectra of G. glabra	64
Figure 18: Principle of Optical Emission Spectroscopy	66
Figure 19: Principle of Atomic Absorption Spectroscopy	67
Figure 20: DNA Barcoding Workflow	69
Figure 21: Forsythia fructus and Flos Syzygia	96
Figure 22: The Krebs Cycle, Citric Acid Cycle or Tricarboxylic Acid Cycle	103

Figure 23: Oxidative Phosphorylation	104
Figure 24: Sources of ROS and antioxidant defence mechanisms	106
Figure 25: Simplified Diagram of the Healing Cycle, Salugenesis	111
Figure 26: The Role of Mitochondria in Apoptosis	112
Figure 27: Traditional Model of ROS in Cells	113
Figure 28: Hormesis Dose-Response Curve	114
Figure 29: Mitochondria, the Endoplasmic Reticulum and Ca ²⁺	117
Figure 30: The Role of ROS and Melatonin in Sleep / Wake Cycles and t Oxidative Effects of Sleep Deprivation	he 120
Figure 31: Mitochondrial mechanisms in ischaemia-reperfusion injury	121
Figure 32: Hallmarks of Cancer	123
Figure 33: Theoretical Model of the Simultaneous Pro- and Anti- Oxidant Effects of Phytochemicals on Cancerous and Normal Cells	135
Figure 34: Liquorice (G. uralensis); Raw, Dried and Sliced Roots	139
Figure 35: Glycyrrhizin	142
Figure 36: Diagram of a Micelle and its use as a Drug Delivery System	144
Figure 37: Glycyrrhetinic Acid and its Similarity to Hydrocortisone	145
Figure 38: Glabridin	146
Figure 39: Glabrene	146
Figure 40: Liquiritin	147
Figure 41: Liquiritigenin	147
Figure 42: Isoliquiritigenin	148
Figure 43: Liquiritin apioside	148
Figure 44: Glycycoumarin	148
Figure 45: Licochalcone A	149
Figure 46: Macroscopic Identifying Characteristics of Liquorice	153
Figure 47: Microscopic Identifying Characteristics of Liquorice	154

Figure 48: HPTLC Profiles of G. glabra	155
Figure 49: PubMed Search Results by Year for "Licorice AND	
mitochondria"	158
Figure 50: MTT Mechanism of Action	170
Figure 51: DCFDA Mechanism of Action	172
Figure 52: TMRE Mechanism of Action	174
Figure 53: A Typical Seahorse MitoStress Assay Profile	177
Figure 54: Results of MTT Assay Showing Changes in Cell Viability After	
Treatment with Original Liquorice Tea Extract	187
Figure 55: Results of DCFDA Assay Showing Changes in ROS 3 Hours	
After Treatment with Original Liquorice Tea Extract	188
Figure 56: Results of DCFDA Assay Showing Changes in ROS 24 Hours	
After Treatment with Original Liquorice Tea Extract	189
Figure 57: Results of DCFDA Assay Showing Changes in ROS 72 Hours	
After Treatment with Original Liquorice Tea Extract	190
Figure 58: Results of TMRE Assay Showing Changes in $\Delta \Psi m$ 3 Hours	
After Treatment with Original Liquorice Tea Extract	191
Figure 59: Results of TMRE Assay Showing Changes in $\Delta\Psi$ m 24 Hours	
After Treatment with Original Liquorice Tea Extract	191
Figure 60: Results of TMRE Assay Showing Changes in $\Delta\Psi m$ 72 Hours	
After Treatment with Original Liquorice Tea Extract	192
Figure 61: Results of Rhod-2 Assay Showing Changes in Ca ²⁺ 24 Hours	
After Treatment with Original Liquorice Tea Extract	193
Figure 62: Mitotracker Red Stained Confocal Images of MCF7 Cells After	
Treatment with Original Liquorice Tea Extract	194
Figure 63: Results of SeaHorse MitoStress Assay Showing Changes in	
OCR of MCF7 Cells after Treatment with Original Liquorice Tea	
Extract	197

Figure 64: Effect of Original Liquorice Tea Extract 24 Hour Treatment on	
Basal OCR of MCF7 Cells Shown as Percentage in Relation to	
Control	198
Figure 65: Results of SeaHorse MitoStress Assay Showing Changes in	
OCR of MCF7 Cells after Treatment with Original Liquorice Tea	
Extract as Percentage of Basal OCR	199
Figure 66: Effect of Original Liquorice Tea Extract 24 Hour Treatment on	
ATP linked OCR of MCF7 Cells	200
Figure 67: Effect of 24 hour Original Liquorice Tea Extract Treatment on	
Mitochondrial Proton Leak in MCF7 Cells	201
Figure 68: Effect of Original Liquorice Tea Extract 24 hour Treatment on	
Mitochondrial Maximal Capacity in MCF7 Cells	202
Figure 69: Effect of Original Liquorice Tea Extract 24 hour Treatment on	
Mitochondrial Spare Capacity in MCF7 Cells	203
Figure 70: Results of SeaHorse MitoStress Assay Testing For Plate	
Positioning or Technical Error	204
Figure 71: Effect of Original Liquorice Tea Extract 24 Hour Treatment	
Followed by Oligomycin Administration on OCR of MCF7 Cells	206
Figure 72: Effect of Original Liquorice Tea Extract 24 Hour Treatment	
Followed by FCCP Administration on OCR of MCF7 Cells	207
Figure 73: Effect of Original Liquorice Tea Extract 24 Hour Treatment	
Followed by Antimycin A and Rotenone Administration on OCR	
of MCF7 Cells	208
Figure 74: Results of SeaHorse MitoStress Assay Showing Changes in	
OCR of MCF10A Cells After Treatment with Original Liquorice	
Tea Extract	211
Figure 75: Effect of Original Liquorice Tea Extract 24 Hour Treatment on	
Basal OCR of MCF10A Cells Shown as Percentage in Relation	
to Control	212

Figure 76: Results of SeaHorse MitoStress Assay Showing Changes in	
OCR of MCF10A Cells After Treatment with Original Liquorice	
Tea Extract as Percentage of Basal OCR	213
Figure 77: Effect of Original Liquorice Tea Extract 24 Hour Treatment on	
ATP linked OCR of MCF10A Cells	214
Figure 78: Effect of Original Liquorice Tea Extract 24 Hour Treatment on	
Mitochondrial Proton Leak in MCF10A Cells	215
Figure 79: Effect of Original Liquorice Tea Extract 24 Hour Treatment on	
Mitochondrial Maximal Capacity in MCF10A Cells	215
Figure 80: Effect of Original Liquorice Tea Extract 24 Hour Treatment on	
Mitochondrial Spare Capacity in MCF10A Cells	216
Figure 81: Effect of Original Liquorice Tea Extract 24 Hour Treatment	
Followed by Oligomycin on OCR of MCF10A Cells	217
Figure 82: Effect of Original Liquorice Tea Extract 24 Hour Treatment	
Followed by FCCP on OCR of MCF7 Cells	217
Figure 83: Effect of Original Liquorice Tea Extract 24 Hour Treatment	
Followed by Antimycin A and Rotenone on OCR of MCF10A	
Cells	218
Figure 84: Effect of Original Liquorice Tea Extract and Mitochondrial	
Modulating Drugs on ROS in MCF7 cells	220
Figure 85: Effect of Original Liquorice Tea Extract Pre-treatment on	
Cytotoxic Effects of Cisplatin in MCF7 Cells	221
Figure 86: Difference Between Original Liquorice Tea Extract	
Pre-treatment and No Pre-treatment on the Cytotoxic Effect of	
Cisplatin in MCF7 Cells	222
Figure 87: Effect of Original Liquorice Tea Extract Pre-treatment on	
Cytotoxic Effects of Cisplatin in MCF-10 Cells	223
Figure 88: Difference Between Original Liquorice Tea Extract	
Pre-treatment and No Pre-treatment on the Cytotoxic Effect of	
Cisplatin in MCF10A Cells	224

Figure 89: Results of DCFDA Assay Showing Changes in ROS 3 Hours	
After Treatment with Pre-extracted Liquorice Tea Extract	227
Figure 90: Results of DCFDA Assay Showing Changes in ROS 72 Hours	
After Treatment with Pre-extracted Liquorice Tea Extract	228
Figure 91: Results of SeaHorse MitoStress Assay Showing Changes in	
OCR of MCF7 Cells after treatment with Pre-extracted	
Liquorice Tea Extract	230
Figure 92: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment on Basal OCR of MCF7 Cells	231
Figure 93: Results of SeaHorse MitoStress Assay Showing Changes in	
OCR of MCF7 Cells after Treatment with Pre-extracted	
Liquorice Tea Extract as a Percentage of Basal OCR Shown as	
Percentage in Relation to Control	232
Figure 94: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment on	
ATP Linked OCT of MCF7	233
Figure 95: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment on	
Mitochondrial Proton Leak in MCF7 Cells	234
Figure 96: Effect of Pre-extracted Liquorice Tea 24 hour treatment on	
Mitochondrial Maximal Capacity in MCF7 Cells	234
Figure 97: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment on	
Mitochondrial Spare Capacity in MCF7 Cells	235
Figure 98: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment	
followed by Oligomycin Administration on OCR of MCF7 Cells	236
Figure 99: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment	
Followed by FCCP on OCR of MCF7 Cells	236
Figure 100: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment	
Followed by Antimycin A and Rotenone on OCR of MCF7 Cells	237
Figure 101: Results of Seahorse MitoStress Assay Showing Changes in	
OCR of MCF10A cells After Treatment with Pre-extracted	
Liquorice Tea Extract	239

 Basal OCR of MCF10A Cells Shown as Percentage in Relation to Control Figure 103: Results of Seahorse MitoStress Assay Showing Changes in OCR of MCF10A cells After Treatment with Pre-extracted Liquorice Tea Extract as a Percentage of Basal OCR Figure 104: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment on ATP Linked OCR of MCF10A Cells Figure 105: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment on Mitochondrial Proton Leak in MCF10A Cells Figure 106: Effect of Liquorice Tea 24 Hour Treatment on Mitochondrial Capacity in MCF10A Cells Figure 107: Effect of Liquorice Tea 24 Hour Treatment on Mitochondrial Spare Capacity in MCF10A Cells Figure 108: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Oligomycin on OCR of MCF10A Cells Figure 109: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by FCCP on OCR of MCF10A Cells Figure 110: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Antimycin A and Rotenone on OCR of MCF10A Cells Figure 111: Effect of Pre-extracted Liquorice Tea Extract Pre-treatment on Cytotoxic Effect of Cisplatin in MCF10A Cells Figure 112: Difference Between Pre-extracted Liquorice Tea Extract Pre-treatment on Cytotoxic Effect of Cisplatin in MCF10A Cells Figure 113: Visual Comparisons of the Original and Pre-extracted Liquorice Tea Extract Figure 113: Visual Comparisons of the Original and Pre-extracted Liquorice Tea Extract
 to Control Figure 103: Results of Seahorse MitoStress Assay Showing Changes in OCR of MCF10A cells After Treatment with Pre-extracted Liquorice Tea Extract as a Percentage of Basal OCR Figure 104: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment on ATP Linked OCR of MCF10A Cells Figure 105: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment on Mitochondrial Proton Leak in MCF10A Cells Figure 106: Effect of Liquorice Tea 24 Hour Treatment on Mitochondrial Capacity in MCF10A Cells Figure 107: Effect of Liquorice Tea 24 Hour Treatment on Mitochondrial Spare Capacity in MCF10A Cells Figure 108: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Oligomycin on OCR of MCF10A Cells Figure 109: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by FCCP on OCR of MCF10A Cells Figure 110: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Antimycin A and Rotenone on OCR of MCF10A Cells Figure 111: Effect of Pre-extracted Liquorice Tea Extract Pre-treatment on Cytotoxic Effect of Cisplatin in MCF10A Cells Figure 112: Difference Between Pre-extracted Liquorice Tea Extract Pre-treatment on Cytotoxic Effect of Cisplatin in MCF10A Cells Figure 113: Visual Comparisons of the Original and Pre-extracted Liquorice Tea Extract Figure 114: HPIL C Bocklin After Development Linder Visual Linder
 Figure 103: Results of Seahorse MitoStress Assay Showing Changes in OCR of MCF10A cells After Treatment with Pre-extracted Liquorice Tea Extract as a Percentage of Basal OCR Figure 104: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment on ATP Linked OCR of MCF10A Cells Figure 105: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment on Mitochondrial Proton Leak in MCF10A Cells Figure 106: Effect of Liquorice Tea 24 Hour Treatment on Mitochondrial Maximal Capacity in MCF10A Cells Figure 107: Effect of Liquorice Tea 24 Hour Treatment on Mitochondrial Spare Capacity in MCF10A Cells Figure 108: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Oligomycin on OCR of MCF10A Cells Figure 109: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by FCCP on OCR of MCF10A Cells Figure 110: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Antimycin A and Rotenone on OCR of MCF10A Cells Figure 111: Effect of Pre-extracted Liquorice Tea Extract Pre-treatment on Cytotoxic Effect of Cisplatin in MCF10A Cells Figure 112: Difference Between Pre-extracted Liquorice Tea Extract Pre-treatment and No Pre-treatment on the Cytotoxic effect of Cisplatin in MCF10A Cells Figure 113: Visual Comparisons of the Original and Pre-extracted Liquorice Tea Extracts Figure 114: HETL C Beoute After Development Under Visual Light
 OCR of MCF10A cells After Treatment with Pre-extracted Liquorice Tea Extract as a Percentage of Basal OCR Figure 104: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment on ATP Linked OCR of MCF10A Cells Figure 105: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment on Mitochondrial Proton Leak in MCF10A Cells Figure 106: Effect of Liquorice Tea 24 Hour Treatment on Mitochondrial Maximal Capacity in MCF10A Cells Figure 107: Effect of Liquorice Tea 24 Hour Treatment on Mitochondrial Spare Capacity in MCF10A Cells Figure 108: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Oligomycin on OCR of MCF10A Cells Figure 109: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by FCCP on OCR of MCF10A Cells Figure 110: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Antimycin A and Rotenone on OCR of MCF10A Cells Figure 111: Effect of Pre-extracted Liquorice Tea Extract Pre-treatment on Cytotoxic Effect of Cisplatin in MCF10A Cells Figure 112: Difference Between Pre-extracted Liquorice Tea Extract Pre-treatment and No Pre-treatment on the Cytotoxic effect of Cisplatin in MCF10A Cells Figure 113: Visual Comparisons of the Original and Pre-extracted Liquorice Tea Extracts
 Liquorice Tea Extract as a Percentage of Basal OCR Figure 104: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment on ATP Linked OCR of MCF10A Cells Figure 105: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment on Mitochondrial Proton Leak in MCF10A Cells Figure 106: Effect of Liquorice Tea 24 Hour Treatment on Mitochondrial Maximal Capacity in MCF10A Cells Figure 107: Effect of Liquorice Tea 24 Hour Treatment on Mitochondrial Spare Capacity in MCF10A Cells Figure 108: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Oligomycin on OCR of MCF10A Cells Figure 109: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by FCCP on OCR of MCF7 Cells Figure 110: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Antimycin A and Rotenone on OCR of MCF10A Cells Figure 111: Effect of Pre-extracted Liquorice Tea Extract Pre-treatment on Cytotoxic Effect of Cisplatin in MCF10A Cells Figure 112: Difference Between Pre-extracted Liquorice Tea Extract Pre-treatment and No Pre-treatment on the Cytotoxic effect of Cisplatin in MCF10A Cells Figure 113: Visual Comparisons of the Original and Pre-extracted Liquorice Tea Extracts Figure 114: HETL C Provide After Development Linder Viscol Light
 Figure 104: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment on ATP Linked OCR of MCF10A Cells Figure 105: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment on Mitochondrial Proton Leak in MCF10A Cells Figure 106: Effect of Liquorice Tea 24 Hour Treatment on Mitochondrial Maximal Capacity in MCF10A Cells Figure 107: Effect of Liquorice Tea 24 Hour Treatment on Mitochondrial Spare Capacity in MCF10A Cells Figure 108: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Oligomycin on OCR of MCF10A Cells Figure 109: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by FCCP on OCR of MCF10A Cells Figure 110: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Antimycin A and Rotenone on OCR of MCF10A Cells Figure 111: Effect of Pre-extracted Liquorice Tea Extract Pre-treatment on Cytotoxic Effect of Cisplatin in MCF10A Cells Figure 112: Difference Between Pre-extracted Liquorice Tea Extract Pre-treatment and No Pre-treatment on the Cytotoxic effect of Cisplatin in MCF10A Cells Figure 113: Visual Comparisons of the Original and Pre-extracted Liquorice Tea Extracts Figure 114: HETLC Provide After Development Linder Visual Light
 ATP Linked OCR of MCF10A Cells Figure 105: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment on Mitochondrial Proton Leak in MCF10A Cells Figure 106: Effect of Liquorice Tea 24 Hour Treatment on Mitochondrial Maximal Capacity in MCF10A Cells Figure 107: Effect of Liquorice Tea 24 Hour Treatment on Mitochondrial Spare Capacity in MCF10A Cells Figure 108: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Oligomycin on OCR of MCF10A Cells Figure 109: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by FCCP on OCR of MCF10A Cells Figure 110: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Antimycin A and Rotenone on OCR of MCF10A Cells Figure 111: Effect of Pre-extracted Liquorice Tea Extract Pre-treatment on Cytotoxic Effect of Cisplatin in MCF10A Cells Figure 112: Difference Between Pre-extracted Liquorice Tea Extract Pre-treatment and No Pre-treatment on the Cytotoxic effect of Cisplatin in MCF10A Cells Figure 113: Visual Comparisons of the Original and Pre-extracted Liquorice Tea Extract Figure 114: HETLC Results After Development Linder Visual Light
 Figure 105: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment on Mitochondrial Proton Leak in MCF10A Cells Figure 106: Effect of Liquorice Tea 24 Hour Treatment on Mitochondrial Maximal Capacity in MCF10A Cells Figure 107: Effect of Liquorice Tea 24 Hour Treatment on Mitochondrial Spare Capacity in MCF10A Cells Figure 108: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Oligomycin on OCR of MCF10A Cells Figure 109: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by FCCP on OCR of MCF7 Cells Figure 110: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Antimycin A and Rotenone on OCR of MCF10A Cells Figure 111: Effect of Pre-extracted Liquorice Tea Extract Pre-treatment on Cytotoxic Effect of Cisplatin in MCF10A Cells Figure 112: Difference Between Pre-extracted Liquorice Tea Extract Pre-treatment and No Pre-treatment on the Cytotoxic effect of Cisplatin in MCF10A Cells Figure 113: Visual Comparisons of the Original and Pre-extracted Liquorice Tea Extracts Figure 114: HBTL C Results After Development Linder Visual Light
Mitochondrial Proton Leak in MCF10A Cells Figure 106: Effect of Liquorice Tea 24 Hour Treatment on Mitochondrial Maximal Capacity in MCF10A Cells Figure 107: Effect of Liquorice Tea 24 Hour Treatment on Mitochondrial Spare Capacity in MCF10A Cells Figure 108: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Oligomycin on OCR of MCF10A Cells Figure 109: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by FCCP on OCR of MCF7 Cells Figure 110: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Antimycin A and Rotenone on OCR of MCF10A Cells Figure 111: Effect of Pre-extracted Liquorice Tea Extract Pre-treatment on Cytotoxic Effect of Cisplatin in MCF10A Cells Figure 112: Difference Between Pre-extracted Liquorice Tea Extract Pre-treatment and No Pre-treatment on the Cytotoxic effect of Cisplatin in MCF10A Cells Figure 113: Visual Comparisons of the Original and Pre-extracted Liquorice Tea Extracts Figure 114: HPTLC Results After Development Linder Visual Light
 Figure 106: Effect of Liquorice Tea 24 Hour Treatment on Mitochondrial Maximal Capacity in MCF10A Cells Figure 107: Effect of Liquorice Tea 24 Hour Treatment on Mitochondrial Spare Capacity in MCF10A Cells Figure 108: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Oligomycin on OCR of MCF10A Cells Figure 109: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by FCCP on OCR of MCF7 Cells Figure 110: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Antimycin A and Rotenone on OCR of MCF10A Cells Figure 111: Effect of Pre-extracted Liquorice Tea Extract Pre-treatment on Cytotoxic Effect of Cisplatin in MCF10A Cells Figure 112: Difference Between Pre-extracted Liquorice Tea Extract Pre-treatment and No Pre-treatment on the Cytotoxic effect of Cisplatin in MCF10A Cells Figure 113: Visual Comparisons of the Original and Pre-extracted Liquorice Tea Extracts Figure 114: HPTL C Results After Development Linder Visual Light
 Maximal Capacity in MCF10A Cells Figure 107: Effect of Liquorice Tea 24 Hour Treatment on Mitochondrial Spare Capacity in MCF10A Cells Figure 108: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Oligomycin on OCR of MCF10A Cells Figure 109: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by FCCP on OCR of MCF7 Cells Figure 110: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Antimycin A and Rotenone on OCR of MCF10A Cells Figure 111: Effect of Pre-extracted Liquorice Tea Extract Pre-treatment on Cytotoxic Effect of Cisplatin in MCF10A Cells Figure 112: Difference Between Pre-extracted Liquorice Tea Extract Pre-treatment and No Pre-treatment on the Cytotoxic effect of Cisplatin in MCF10A Cells Figure 113: Visual Comparisons of the Original and Pre-extracted Liquorice Tea Extracts
 Figure 107: Effect of Liquorice Tea 24 Hour Treatment on Mitochondrial Spare Capacity in MCF10A Cells Figure 108: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Oligomycin on OCR of MCF10A Cells Figure 109: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by FCCP on OCR of MCF7 Cells Figure 110: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Antimycin A and Rotenone on OCR of MCF10A Cells Figure 111: Effect of Pre-extracted Liquorice Tea Extract Pre-treatment on Cytotoxic Effect of Cisplatin in MCF10A Cells Figure 112: Difference Between Pre-extracted Liquorice Tea Extract Pre-treatment and No Pre-treatment on the Cytotoxic effect of Cisplatin in MCF10A Cells Figure 113: Visual Comparisons of the Original and Pre-extracted Liquorice Tea Extracts
 Spare Capacity in MCF10A Cells Figure 108: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Oligomycin on OCR of MCF10A Cells Figure 109: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by FCCP on OCR of MCF7 Cells Figure 110: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Antimycin A and Rotenone on OCR of MCF10A Cells Figure 111: Effect of Pre-extracted Liquorice Tea Extract Pre-treatment on Cytotoxic Effect of Cisplatin in MCF10A Cells Figure 112: Difference Between Pre-extracted Liquorice Tea Extract Pre-treatment and No Pre-treatment on the Cytotoxic effect of Cisplatin in MCF10A Cells Figure 113: Visual Comparisons of the Original and Pre-extracted Liquorice Tea Extracts
 Figure 108: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Oligomycin on OCR of MCF10A Cells Figure 109: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by FCCP on OCR of MCF7 Cells Figure 110: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Antimycin A and Rotenone on OCR of MCF10A Cells Figure 111: Effect of Pre-extracted Liquorice Tea Extract Pre-treatment on Cytotoxic Effect of Cisplatin in MCF10A Cells Figure 112: Difference Between Pre-extracted Liquorice Tea Extract Pre-treatment and No Pre-treatment on the Cytotoxic effect of Cisplatin in MCF10A Cells Figure 113: Visual Comparisons of the Original and Pre-extracted Liquorice Tea Extracts Figure 114: HBTLC Results After Development Linder Visual Light
 Followed by Oligomycin on OCR of MCF10A Cells Figure 109: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by FCCP on OCR of MCF7 Cells Figure 110: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Antimycin A and Rotenone on OCR of MCF10A Cells Figure 111: Effect of Pre-extracted Liquorice Tea Extract Pre-treatment on Cytotoxic Effect of Cisplatin in MCF10A Cells Figure 112: Difference Between Pre-extracted Liquorice Tea Extract Pre-treatment and No Pre-treatment on the Cytotoxic effect of Cisplatin in MCF10A Cells Figure 113: Visual Comparisons of the Original and Pre-extracted Liquorice Tea Extracts Figure 114: HPTLC Paculta After Development Linder Visual Light
 Figure 109: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by FCCP on OCR of MCF7 Cells Figure 110: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Antimycin A and Rotenone on OCR of MCF10A Cells Figure 111: Effect of Pre-extracted Liquorice Tea Extract Pre-treatment on Cytotoxic Effect of Cisplatin in MCF10A Cells Figure 112: Difference Between Pre-extracted Liquorice Tea Extract Pre-treatment and No Pre-treatment on the Cytotoxic effect of Cisplatin in MCF10A Cells Figure 113: Visual Comparisons of the Original and Pre-extracted Liquorice Tea Extracts
 Followed by FCCP on OCR of MCF7 Cells Figure 110: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Antimycin A and Rotenone on OCR of MCF10A Cells Figure 111: Effect of Pre-extracted Liquorice Tea Extract Pre-treatment on Cytotoxic Effect of Cisplatin in MCF10A Cells Figure 112: Difference Between Pre-extracted Liquorice Tea Extract Pre-treatment and No Pre-treatment on the Cytotoxic effect of Cisplatin in MCF10A Cells Figure 113: Visual Comparisons of the Original and Pre-extracted Liquorice Tea Extracts
 Figure 110: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Antimycin A and Rotenone on OCR of MCF10A Cells Figure 111: Effect of Pre-extracted Liquorice Tea Extract Pre-treatment on Cytotoxic Effect of Cisplatin in MCF10A Cells Figure 112: Difference Between Pre-extracted Liquorice Tea Extract Pre-treatment and No Pre-treatment on the Cytotoxic effect of Cisplatin in MCF10A Cells Figure 113: Visual Comparisons of the Original and Pre-extracted Liquorice Tea Extracts Figure 114: HETLC Results After Development Linder Visual Light
 Followed by Antimycin A and Rotenone on OCR of MCF10A Cells Figure 111: Effect of Pre-extracted Liquorice Tea Extract Pre-treatment on Cytotoxic Effect of Cisplatin in MCF10A Cells Figure 112: Difference Between Pre-extracted Liquorice Tea Extract Pre-treatment and No Pre-treatment on the Cytotoxic effect of Cisplatin in MCF10A Cells Figure 113: Visual Comparisons of the Original and Pre-extracted Liquorice Tea Extracts Figure 114: HPTLC Pacults After Development Linder Visual Light
Cells Figure 111: Effect of Pre-extracted Liquorice Tea Extract Pre-treatment on Cytotoxic Effect of Cisplatin in MCF10A Cells Figure 112: Difference Between Pre-extracted Liquorice Tea Extract Pre-treatment and No Pre-treatment on the Cytotoxic effect of Cisplatin in MCF10A Cells Figure 113: Visual Comparisons of the Original and Pre-extracted Liquorice Tea Extracts Figure 114: HPTLC Posults After Development Linder Visual Light
 Figure 111: Effect of Pre-extracted Liquorice Tea Extract Pre-treatment on Cytotoxic Effect of Cisplatin in MCF10A Cells Figure 112: Difference Between Pre-extracted Liquorice Tea Extract Pre-treatment and No Pre-treatment on the Cytotoxic effect of Cisplatin in MCF10A Cells Figure 113: Visual Comparisons of the Original and Pre-extracted Liquorice Tea Extracts Figure 114: HPTLC Results After Development Linder Visual Light
Cytotoxic Effect of Cisplatin in MCF10A Cells Figure 112: Difference Between Pre-extracted Liquorice Tea Extract Pre-treatment and No Pre-treatment on the Cytotoxic effect of Cisplatin in MCF10A Cells Figure 113: Visual Comparisons of the Original and Pre-extracted Liquorice Tea Extracts Figure 114: HPTLC Results After Development Linder Visual Light
 Figure 112: Difference Between Pre-extracted Liquorice Tea Extract Pre-treatment and No Pre-treatment on the Cytotoxic effect of Cisplatin in MCF10A Cells Figure 113: Visual Comparisons of the Original and Pre-extracted Liquorice Tea Extracts Figure 114: HPTLC Results After Development Linder Visual Light
Pre-treatment and No Pre-treatment on the Cytotoxic effect of Cisplatin in MCF10A Cells Figure 113: Visual Comparisons of the Original and Pre-extracted Liquorice Tea Extracts
Cisplatin in MCF10A Cells Figure 113: Visual Comparisons of the Original and Pre-extracted Liquorice Tea Extracts Figure 114: HPTLC Results After Development Linder Visual Light
Figure 113: Visual Comparisons of the Original and Pre-extracted Liquorice Tea Extracts
Liquorice Tea Extracts
Figure 114: HDTLC Desults After Development Linder Visual Light
Figure 114. THE LC Results Alter Development Onder Visual Light
Figure 115: HPTLC Results After Development Under UV 254 nm
Figure 116: HPTLC Results After Development Under UV 366 nm

Figure 117: HPTLC Results After Derivatization Under Visible Light	253
Figure 118: HPTLC Results After Derivatization Under UV 366 nm	253
Figure 119: HPTLC Comparison of G. glabra and G. uralensis	254
Figure 120: HPTLC Comparison of Ethanol and Water Extracts of <i>G. glabra</i> After Development and Viewed Under UV 366 nm	255
Figure 121: HPTLC Comparison of <i>G. glabra</i> from a Commercial Tea Brand and a Herbal Medicine Clinic After Derivatization under UV 366 nm	256
Figure 122: HPTLC comparison of Original and Pre-extracted Liquorice Tea Extracts at Natural Strength if Extraction was Taken	230
Directly from an Pre-extracted Tea Bag	257
Figure 123: HPTLC Comparison of Original and Pre-extracted Liquorice Tea Extracts at the Same Dosage of Freeze Dried Material	258
Figure 124: HPTLC Comparison of Original and Pre-extractedLiquorice Tea Extracts at the Same Dosage of Freeze Dried Material after Development Under UV 366 Showing Probable	
Glycyrrhizin Elution Zone	259
Figure 125: Theoretical Mechanisms to account for Lower Oxygen Consumption in High Dose Liquorice Treated Cells	277
Figure 126: Seahorse Glycolytic Rate Test Example	280
Figure 127: Comparison of Original Liquorice Tea Extract and Pre-extracted Liquorice Tea Extract on ROS Production in	
MCF7 Cells after 3 Hours	291
Figure 128: Comparison of Original Liquorice Tea Extract and Pre-extracted Liquorice Tea Extract on ROS Production in MCF7 cells After 72 Hours	292
Figure 129: Comparison of Original and Pre-extracted Liquorice Tea Extract on Cell Viability When Used as Pre-treatment for 24 Hours Before 24 Hour Exposure to Cisplatin in MCE10A cells	005
Hours before 24 Hour Exposure to Cispiatin in MCFTOA Cells	295

Index of Tables:

Table 1: Popularity of Methods Used in Quality Control of Herbal Teas	41
Table 2: Uses of Liquid Chromatography in the Analysis of Herbal Teas	44
Table 3: Uses of Gas Chromatography in the Analysis of Herbal Teas	47
Table 4: Uses of Mass Spectrometry in the Analysis of Herbal Teas	54
Table 5: Results of Genetic Analysis on Teas	70 - 72
Table 6: Biological Assays Used in Quality Assessment of Herbal Teas	
and their Purpose	76 - 92
Table 7: Multivalency of the Word Qi 氣 in Chinese	127
Table 8: Papers Discussing Mitochondria and Specific Herbs	271

Acknowledgements:

A PhD is a challenging time, with this one made even more challenging due to the events of the last few years. Fortunately, this has been alleviated by the people surrounding me whose support managed to turn a potentially difficult few years into some of the most rewarding and productive of my life. It is to these people that I dedicate this thesis.

I would like to acknowledge and thank the Research Centre for Optimal Health for their support during my PhD. In particular, Prof. Jimmy Bell and Dr. Anthony Booker for providing me with the direction and space to allow this project to flourish; Dr. Rhys Mould and Dr. Meliz Sahuri-Arisoylu for their patience and instruction in lab procedures which helped my skills in life sciences to grow; Grace Penelli for her assistance in preparing assays during the final stages of this project; Prof. Louise Thomas for providing support with university processes when times were tough; Ifigenia Kalampouka for proofreading and general support. I would also like to thank Pukka Herbs for the funding that made this endeavour possible.

I would also like to thank my family and friends for their support throughout this time; my clients for their understanding of my reduced and frequently changing working times; and in particular my partner, Hazeena Azeez, who has had to sacrifice more than any other in order for me to complete this work.

Declaration of Contributions:

The majority of the work in this thesis was performed by the author. Any assistance and collaborations are detailed below:

Initial Samples and MTT Assay:

The initial samples were prepared and MTT assays were performed by Dr Meliz Sahuri-Arisoylu due to laboratory restrictions during Covid-19. Subsequent samples, MTT assays and the analysis of all acquired data was performed by the author.

Plate Seeding of Adulteration Assays:

The plates for assays relating to the adulterated teas were performed by Grace Penelli. The assays and analysis of the data was performed by the author.

HPTLC Assay:

Dr Anthony Booker assisted in the preparation of solvents, acquisition of some of the control samples, and derivatization of the plates for the HPTLC assay. The preparation of all samples and analysis of the data was performed by the author.

Photographs:

The photographs of liquorice, cloves and forsythia in the introduction were taken by Hazeena Azeez using samples acquired by the author.

List of Abbreviations:

ΔΨm	Mitochondrial membrane potential
ABC	ATP-binding cassette
ABTS	2,2'-azino-bis(3-ethylbenzothiazoline-6-sulfonic acid)
ADP	Adenosine diphosphate
AMP	Adenosine monophosphate
AMPK	AMP-activated protein kinase
ANOVA	Analysis of variance
ATP	Adenosine triphosphate
Bak	Bcl-2 antagonist killer
Bax	Bcl-2 associated X
BCA	Bicinchoninic
Bcl-2	B cell lymphoma 2
BCRP	Breast Cancer Resistance Protein
BOLD	Barcode of Life Data
BSA	Bovine Serum Albumin
C	Carbon atom
Ca ²⁺	Calcium ion
CAT	Catalase
СНМ	Chinese herbal medicine
Cisplatin	Cis-diamminedichloroplatinum(II)
CO ₂	Carbon dioxide
CYP450	Cytochrome P450
DCF	2',7'-dichlorofluorescein
DCFDA	2',7'-dichlorodihydrofluorescein diacetate
DMEM	Dulbecco's modified Eagle's medium
DNA	Deoxyribonucleic acid
DPPH	2,2-diphenyl-1-picryl-hydrazyl-hydrate

eATP	Extracellular ATP
EGF	Epidermal growth factor
ELISA	Enzyme-linked immunosorbent assay
EMA	European Medicines Agency
ETC	Electron transport chain
ER	Endoplasmic reticulum
ERK	Extracellular signal-regulated kinase
FAD	Flavin adenine dinucleotide
FADH ₂	Flavin adenine dinucleotide plus hydrogen
FBS	Foetal Bovine Serum
FCCP	Carbonyl cyanide-4 (trifluoromethoxy) phenylhydrazone
Fe ²⁺	Ferrous ion
FRAP	Ferric Reducing Antioxidant Power
GC	Gas chromatography
GPx	Glutathione peroxidase
GSH	Glutathione
GR	Glutathione reductase
Н	Hydrogen atom
H⁺	Hydrogen ion
H₂O	Water
H_2O_2	Hydrogen peroxide
HAT	Hydrogen atom transfer
HIF-1α	Hypoxia inducible factor 1α
HO-1	Heme oxygenase-1
HPLC	High performance liquid chromatography
HPTLC	High performance thin layer chromatography
IL	
IMM	Inner mitochondrial membrane
IRI	Ischaemia-reperfusion injury
IRS	Infrared spectrometry

Keap1	Kelch-like ECH-associated protein 1
LC	Liquid chromatography
МАРК	Mitogen-activated protein kinase
MCF10A	Michigan Cancer Foundation 10A
MCF7	Michigan Cancer Foundation 7
MCU	Mitochondrial Calcium Uniporter
MDA	Malondialdehyde
MEM	Modified Eagle's medium
MRP	Multidrug Resistance-associated Protein
mPTP	Mitochondrial permeability transition pore
MS	Mass spectroscopy
mtDNA	Mitochondrial DNA
MTT	(3-(4,5-dimethylthiazol-2-yl)-2,5-diphenyltetrazolium bromide)
Na⁺	Sodium ion
NAD	Nicotinamide adenine dinucleotide
NADH	Nicotinamide adenine dinucleotide plus hydrogen
NCBI	National Center for Biotechnology Information
NCCIH	National Center for Complementary and Integrative Health
NF-κB	Nuclear factor kappa-light-chain-enhancer of activated B cells
NLRP	Nucleotide-binding oligomerization domain, leucine rich repeat and
	pyrin domain containing
NMR	Nuclear magnetic resonance
NO	Nitric oxide
NOX	NADPH oxidases
Nrf2	Nuclear factor erythroid 2-related factor 2
0	Oxygen atom
O ₂	Molecular oxygen
O ₂ • ⁻	Superoxide radical
OGD/R	Oxygen-glucose deprivation / reperfusion

•ОН	Hydroxyl radical
•00H	Peroxyl radical
OES	Optical emission spectroscopy
OLTE	Original Liquorice Tea Extraction
OMM	Outer mitochondrial membrane
OXPHOS	Oxidative phosphorylation
Ρ	Phosphate
PAH	Polycyclic aromatic hydrocarbons
PBS	Phosphate-buffered saline
PC12	Rat adrenal medulla pheochromocytoma
P-gP	P-glycoprotein
PGC-1α	PPARγ coactivator-1α
PPARγ	Peroxisome proliferator activated receptor-gamma
PRX	Peroxiredoxins
Rhod-2	Rhod-2 acetoxymethyl ester
RIPA	Radioimmunoprecipitation assay
RNS	Reactive nitrogen species
ROS	Reactive oxygen species
SHMT2	Serine hydroxymethyltransferase 2
SIRT	Silent information regulator
SOD	Superoxide dismutase
TAC	Total anthocyanin content
ТСМ	Traditional Chinese Medicine
TLC	Thin layer chromatography
TNF-α	Tumour necrosis factor-α
TPC	Total phenolic content
TRX	Thioredoxin
UV	Ultraviolet
UVS	Ultraviolet-visible spectrophotometry

UPLC Ultra-high performance liquid chromatography

WHM	Western herbal medicine
WST-1	2-(2-methoxy-4-nitrophenyl)-3-(4-nitrophenyl)-5-(2,4-disulfophenyl)-2
	H-tetrazolium, monosodium salt
XLTE	Pre-Extracted Liquorice Tea Extraction

Introduction

Chapter I

1. The Issues Associated with Quality in Medicinal Herbal Teas

1.1. The Modern Herbal Tea Industry

The global trade in medicinal herbs was estimated to be worth US\$60 billion in 2017 involving 29,000 herbal substances and growing by 15% each year (Srirama et al., 2017). The Covid-19 pandemic saw an unprecedented rise in demand across all regions resulting in an evaluation of US\$151.91 billion in 2021 and estimated to grow to US\$165.66 billion in 2022 and US\$347.50 billion by 2029 (Fortune Business Insights, 2022). This is thought to be due to people taking a greater interest in their health and immunity during the pandemic. Pharmaceutical and nutraceutical products account for the majority of this share, at 56.49% with food and beverages occupying only 12.17%. Herbal tea accounts for US\$3.29 billion in 2021 (Data Bridge Market Research, 2022) but does not include the US\$207.1 billion global market in conventional tea (*Camellia sinensis* (L.) Kuntze, *Theaceae*) over 2020 (Statista, 2023), or a US\$30.2 billion market in Traditional Chinese Medicine (TCM) in 2023 (Persistence Market Research, 2023), which comprised 18.5% of the global herbal medicines market and are predominantly also extracted in boiling water.

Decoctions and infusions are some of the oldest methods of extracting the medicinal properties from herbs and still one of the most popular today (Wang, Sasse & Sheridan, 2019). Both are aqueous extraction techniques and only differ in that decoctions simmer the water while the herbs are being soaked, while infusions are made by pouring freshly boiled water over the herbs (Bjorn, 2021). Infusions are most popular in the Western world due to their convenience and

familiarity, essentially being the same as making a cup of tea, but are best suited for soft materials such as leaves and flowers, with root and stem material needing to be finely cut to ensure a reasonable extraction. Decoctions have a long history in Chinese medicine for their ability to extract constituents from tough plant materials such as roots, barks and fungi (Wilde, 2020) but take time to prepare. Although the increasing pace of life in Asia has seen a decline in people actually decocting herbal medicines themselves, many Chinese Herbal Medicine (CHM) companies have developed convenient instant powders and granules which are still mostly derived from dehydrated decoctions (Sun Ten, 2015).

Aqueous extractions have maintained their popularity throughout history for their ability to extract polar compounds using a solvent that is safe, even in large doses. One of the reasons herbal medicines have grown in popularity in recent years is their perception as being "more natural" (Welz, Emberger-Klein & Menrad, 2018), often erroneously equated with being more safe (Sax, 2015). They often seem familiar from family traditions where water extractions would have been easily available to any household. In contrast, other solvents such as chloroform, acetone, ether or n-Hexane which can extract non-polar compounds, are usually toxic and cannot be entirely eliminated requiring maximum safe limits to be set and determined by analysis (ICH, 2019) making them the domain of professional laboratories preparing supplements. Alcohol is the main exception and retains popularity in Western herbal medicine (WHM) to make tinctures (Abubakar & Haque, 2020), but is also sometimes used in Chinese medicine where herbs may be soaked over weeks or months in a jar of high strength rice wine or other spirit (Flaws, 1994). Alcohol can also be exploited for use in decoctions and is used in CHM by pre-treating certain herbs with a wine-washing or wine-frying technique to free non-polar constituents before being added to the decoction (Sionneau, 1995; Dong et al., 2006).

Despite the popular perception of herbal medicines being natural and safe, a great deal of adulteration has been reported. While much of it is accidental, at the level of foraging or purchasing at markets, there is evidence to suggest that it is also deliberate and economically motivated, including the addition of drugs to enhance the effects of supposedly "natural" supplements (Booker, Jalil et al.,

2016), dyes that make the colour of vibrant herbs look more potent (Müller-Maatsch, Schweiggert & Carle, 2016; Booker et al., 2018), sometimes using toxic colour pigments such as lead chromate in turmeric (Curcuma longa L., Zingiberaceae) to cover blemishes and increase weight on sale at the risk of poisoning millions of end consumers (Cowell et al., 2017; Newby 2023), and substitution with cheaper herbal material which may have similar chemical profiles and so evade detection by all but the most sophisticated tests (Booker et al., 2018; Frommenwiler et al., 2019). In some supplements, as many as 94% of commercial products may be adulterated, such as Booker, Frommenwiler et al. (2016) found when testing 35 Ginkgo biloba L. Ginkgoaceae supplements. Other estimates have found as much as 68% of herbal products having some form of adulteration (Newmaster at al., 2013). Raw herbs and teas are less vulnerable to adulteration with estimates ranging from 12% (Kumar et al., 2018) to 35% (Stoekle et al., 2011), and in some cases as high as 80% (Srirama et al., 2017). Although not technically a herb but frequently included among traditional herbs as a natural remedy, a recent survey of honey from the UK found 100% of tested samples were adulterated (European Commission, 2023). This is not a new problem and adulteration of raw herbal medicines for economic gain has been a persistent problem ever since medicinal herbs have been traded.

In this introduction, the history of herbal adulteration and its methods of detection will first be summarised, followed by a literature review of contemporary methods of detection and their limitations. The concept of functional adulteration will be introduced and the role of mitochondria in health, disease and traditional medicine will then be outlined to demonstrate the potential relevance of these organelles when evaluating functional potency of herbal medicines. Finally, the initial herb to be evaluated, liquorice (*Glycyrrhiza glabra* L., *G. uralensis* Fisch. ex DC. and *G. inflata* Batalin, *Fabaceae*) will be introduced, with its constituents, uses, risks, current quality evaluation, adulteration and known effects on mitochondria.

1.2. A Brief History of Herbal Adulteration and Quality Control

1.2.1. Adulteration and Quality Testing in the Ancient World

Quality control of medicinal herbal teas has been an issue for as long as humans have been using plants to heal (Foster, 2011; Woodley, 2021). C. 50 CE, in Dioscorides' De Materia Medica, one of the earliest written examples of a systematic pharmacopoeia, the issue of adulteration through both accidental and fraudulent practices was acknowledged, including 40 examples of specific tests on how to detect them (Riddle, 1985, pp.74-76). The majority were organoleptic, detecting adulteration through the senses, but several employed the technologies of the time in chemico-physical tests such as dropping balsam (probably Commiphora opobalsamum (L.) Engl., Burseraceae) into a woollen cloth and then washing it out which will come clean if the balsam is pure, while its adulterants stain. His contemporary, Pliny the Elder, also wrote of widespread adulteration of food, wine and medicinal remedies in the markets with methods of detecting them (Bush, 2002). In the 2nd century CE the Roman physician Galen also complained about the quality of supplies purchased in the marketplace and suggested using animal testing to evaluate the quality of the antivenom formula Theriac (Thorndike, 1922). This issue was not unique to the WHM tradition with Tao Hongjing in 5th century China complaining that as medicine became more specialised, physicians were no longer able to tell what really mattered to determine a herb's therapeutic efficacy and were often tricked by market sellers into buying medicines that looked impressive but were grown, harvested or stored in poor conditions making them less effective (Liu, 2021). Tao attempted to remedy the situation by expounding upon the currently used pharmacopoeia with detailed commentaries to educate practitioners regarding the attributes to look for in potent medicinal herbal materials. These guides continued to be published throughout China's history with later volumes being compiled by government sponsored teams, including sections on processing, illustrations that may be hand painted for superior accuracy, or mass published versions using block-printing (Liang & Zhao, 2017) (figure 1). *Materia medica* texts like these are still in use today using colour pictures to assist herbal pharmacists with the identification of their raw materials (Leon & Lin, 2017) but they are of little use once the herbs are processed into powders or extracts.





Figure 1: Illustrations from Chinese Pharmacopoeias

Block printed illustrations (Tang & Cao, 1249, left; work in public domain) and full colour pictures (right, illustration from the *Ben Cao Pin Hui Jing Yao*, 本草品彙精要, 1505; photographed by Eric Brand, reproduced with permission).

The link between adulteration and economic gain became an important issue in the saffron (*Crocus sativus* L., *Iridaceae*) trade where saffron could easily be worth its weight in gold. During the 5th to 15th centuries, an armed guard was employed to inspect supplies entering Venice and in Nuremberg in 1358 this became encoded in the first known food law demanding the death penalty for adulteration of saffron, often by being burned alive with your own adulterated produce (Farhan et al, 2020). In 9th century Baghdad, the medicinal properties of aromatic herbs and spices were recognised and a emerging class of professional

pharmacists developed, writing new pharmacopoeias such as those of Yuhanna Ibn Masawayh (يوحنا بن ماسويه, Romanised: Johannes Mesue), which focused on aromatics and detecting their adulteration (Tschanz, 2003). These pharmacists were held to higher standards than regular spice traders and inspected by government appointed officials, al-Muhtasib (محتسب), to ensure their quality and to curb excessive profiteering.

1.2.2. The Scientific Revolution

The scientific revolution brought in new technologies to enable the identification of correct herbal species, most notably microscopy and taxonomy.

Magnification had been documented by Euclid in his treatise on optics ca. 300 B.C., but it took a further 1900 years before the invention of the compound microscope (Bardell, 2004). The exact inventor remains a mystery but its ability to reveal the details of cellular structure opened up a new world to biologists and botanists. Nehemiah Grew (1682) gave the first vivid descriptions and drawings of magnified plant anatomy in his Anatomy of Plants which demonstrated the complexity of plant life and led to the possibility of species being differentiated by their minutiae (figure 2). Improved design and the development of methods such as phase-contrast, differential interference contrast, confocal and super-resolution microscopy, along with advances in staining agents, including fluorescence staining have led to the ability to detect ever increasing detail (Somssich, 2022). It remains one of the most common methods of detecting adulteration in plant materials, in part due to its simplicity that enables people from non-scientific backgrounds involved in the growing, harvesting and trade of herbal materials to understand the images presented in books on herbal materials. As a result magnified images of plant materials are a staple in books on herbal quality (Leon & Lin, 2017).



Figure 2: Illustration comparing seeds under magnification from Nehemiah Grew's *Anatomy of Plants* (1682, p.532; work in the public domain)
Linnaeus' taxonomy, first presented in his Systema Naturae (1735) and extended to plants in his Species Plantarum (1753), also represented a major milestone in herbal quality control. Systematic classification of plants had been attempted before, first by Aristotle's student Theophrastus (370-285 B.C.) who developed a natural system of dividing plants according to the habitat or form (Woodland, 2003). This system was followed, with some improvements and additional classifications, such as the recognition of monocots and dicots by Albertus Magnus (1200-1280), but all following a natural system of classifying according morphological characteristics. By contrast, the Linnaean system was an artificial system constructed by selecting a few characteristics, primarily sexual ones in plants, and categorising them according to these characteristics (Bremer, 2008). All organisms were sorted via a hierarchical system of kingdom, class, order, genus, and species using a binomial nomenclature. Since then, the original five level system has been expanded to include domain, phylum or division, and family to make eight levels, and the sexual system has been largely replaced with phylogenetic characteristics used by the Angiosperm Phylogeny Group (APG IV, 2016), but the Linnaean system intensified research into plants and the binomial nomenclature enables botanists to identify and categorise plants referred to under many different colloquial names according to precise features and place them in relation to others. For those interested in herbal quality assessment, this means not only identifying correct species, but also common adulterants and the unique features that make them similar enough to get confused with the correct species, yet different enough to detect when this has happened.

1.2.3. The Industrial Age

The industrial age saw a similar leap in the technologies that could be used to identify a plant species through the use of chemometrics.

Chromatography originated with Mikhail Tsvet in 1903 who percolated an alcohol and ether extract through a column of calcium carbonate and discovered that the different pigments separated into different bands (Chromatography Today, 2014). Previously it was thought that there were only two kinds of pigments in

plants, chlorophyll and xanthophyll, but his work showed two different kinds of chlorophyll and an additional eight other pigments. Although his work was largely overshadowed by the political events of the early 20th century, it was revived by Kuhn and Lederer (1931) and then further developed by Martin and Synge (1941) who experimented with different solvents, mobile and stationary phases to devise partition chromatography. This earned them a Nobel Prize in chemistry and led to many different specialised forms of this technology.

Another area of technology which enjoyed a renaissance during the 20th century was spectroscopy. Observations on the interaction of light and matter had been recorded since Roman times and developed into a science with Newton's observations in his work on *Opticks* (1730). Further observations by Fraunhofer during the 19th century using a prism spectroscope revealed dark bands in the spectrum of light (figure 3) leading to an understanding that the spectrum is discrete and not continuous (Sack, 2018). A subsequent superior design of spectroscope by Kirchhoff and Bunsen then enabled precise measurements of wavelengths, and using this new precision, they were able to demonstrate that the black bands created by the sun were the same as the line produced by salts introduced to a flame (Sella, 2013). In a series of painstaking experiments, they introduced as many substances as possible to a flame and filtered the resulting light, discovering that this method could identify microgram quantities of elements.

The turn of the 20th century saw the beginning of quantum theory when, in 1900, Max Planck proposed that atoms and molecules can emit or absorb energy in discrete quantities (Planck, 1967). Bohr's (1913) discovery of atomic structure and Einstein's (1917) proposition, that electrons can absorb and emit radiation of a specific frequency that corresponds to the energy required to make their electrons shift from their ground state to an excited state, laid the theoretical foundations for modern quantum theory and new applications for spectroscopy. This enabled the identification of elements within unknown compounds to be identified and, combined with the separating of different compounds with chromatographic analysis, allowed for complete chemometric fingerprinting to be developed.



Figure 3: Fraunhofer's observation of dark lines within the sun's spectrum (created in Biorender, adapted from Fraunhofer, 1815, p.230; work in the public domain).

The latter part of the 20th century saw another revolution in fingerprinting medicinal plants using genetics. Just a few years after Jeffreys, Wilson & Thein (1985) published the first paper on human genetic fingerprinting, the method was being applied to plant, animal and microorganism identification (Ryskov et al, 1988). These early efforts relied predominantly on restriction fragment length polymorphism technique which was time consuming and so mainly applied to economically important crops (Nybom, Weising & Rotter, 2014). The advent of polymerase chain reaction based technologies by Saiki et al. (1988) soon gave rise to a host of techniques that were simpler, faster and required smaller samples of DNA, expanding and diversifying the potential applications of genetic markers. High-throughput sequencing machines have enhanced the speed of genetic profiling even further, allowing biologists to develop huge databases of genetic information used to identify and authenticate plants of commercial, medicinal and ecological interest.

Although organoleptic testing and microscopic examination is still performed by herbalists, the majority of scientific literature on the detection of adulteration and quality evaluation is based on modern technologies. To assess the efficacy and drawbacks of these methods, a literature review of the most frequently employed techniques in the scientific literature will be undertaken.

Chapter II

2. Literature Review of Existing Methods of Quality Control in Medicinal Herbal Teas

2.1. Search Methods

PubMed and Google Scholar databases were searched for literature on the existing methods of examining herbal teas for quality control. The search terms used were:

PubMed:

((chromatography) OR (spectrometry) OR (spectroscopy) OR (NMR) OR (ultraviolet) OR (infrared) OR ((DNA) OR (genetic) barcoding) OR (bioassay)) AND ((herbal tea) OR (tisane) OR (decoction)) AND ((adulteration) OR (guality) OR (contamination))

With "Full Text" and "English" filters applied, obtaining 616 results on 26th April 2024.

Google Scholar was searched with the term:

allintitle: "herbal tea" (chromatography OR HPLC OR HPTLC OR GC OR spectrometry OR spectroscopy OR MS OR NMR OR ultraviolet OR infrared OR "DNA barcoding" OR "genetic barcoding" OR bioassay)

Obtaining 76 results on 28th April 2024 for a total of 692 papers.

PRISMA literature review process was then conducted (figure 4) and 15 duplicates removed.

Exclusion criteria for screening was that the papers:

- Contained no authors, title, date of publication or abstract (n=1)
- Full text not available in English (n=34)

Inclusion criteria for eligibility was that the publications were:

- Research papers (n=23 excluded)
- Investigating teas, decoctions or water extractions (n=26 excluded)
- Investigating quality control (n=196 excluded)

A total of 397 papers were included in the analysis.

An earlier discussion of 245 papers published before 2020 has been published in Woodley et al. (2021). The new information provided here, reviewing an additional 152 papers with further discussion, extended tables and additional figures, combined with the open access nature of the earlier work, meant that an editor for the journal decided there would be no copyright issues in this update, provided the earlier work was referenced.

PRISMA flow diagram



Figure 4: PRISMA Flow Diagram of the Literature Review Process.

Created with Biorender.com, adapted from Moher et al. (2009).

Table 1 reveals a breakdown of the methods and their popularity:

Method	Number of Studies	Percentage
Liquid Chromatography (LC)	303	76%
Gas Chromatography (GC)	34	9%
Thin Layer Chromatography (TLC)	16	4%
Mass Spectrometry (MS)	240	60%
Nuclear Magnetic Resonance (NMR)	17	4%
Ultraviolet-Visible Spectrophotometry (UVS)	88	22%
Infrared Spectroscopy (IRS)	16	4%
Optical Emission Spectrometry (OES)	8	2%
Atomic Absorption Spectrometry (AAS)	7	2%
Deoxyribonucleic acid (DNA) Barcoding	26	7%
Biological Assays	65	16%

Table 1: Popularity of Methods Used in Quality Control of Herbal Teas

From this we can easily see that chromatography is the most popular technique with liquid chromatography being the most favoured method overall, followed by mass spectrometry. These two techniques are far ahead of all others. This is mainly due to the fact that both of these techniques can be used to separate as well as analyse which is a very useful quality when analysing complex mixtures as is generally found in herbal mixtures such as herbal tea blends. Each individual method shall now be described with its advantages and disadvantages and a summary of what the literature revealed.

2.2. Chromatography

Chromatographic methods are some of the most popular for identifying adulteration of herbs due to the fact that they can separate complex compounds into their constituent components, creating a chemical fingerprint. All chromatographic methods achieve this through the same principle of separating a mixture by distributing its components between two phases: a mobile phase, which carries the components through a medium, and a stationary phase that remains fixed in place causing the various constituents to separate as they migrate at different speeds (Liu, 2011, pp.129-130). The fingerprint that this creates can then be compared against an authenticated sample or a suspected adulterant.

This reveals one of the main limitations of chromatography: without a reference standard to compare the peaks to, it is limited in its ability to identify or analyse the structure of the eluted compounds, for which spectroscopic methods are normally required. This is clearly illustrated by Xiao et al. (2022) who used LC to find specific constituents of Three Whites Decoction (San Bai Tang, consisting of Atractylodes macrocephala Koidz, Asteraceae; Paeonia lactiflora Pall, Paeoniaceae; Poria cocos Fr. (1822), Polyporaceae and Glycyrrhiza uralensis Fisch. ex DC., Fabaceae) and correlate these peaks with their antioxidant capacity, determined by a 2,2-diphenyl-1-picryl-hydrazyl-hydrate (DPPH) assay, but were only able to list the relative antioxidant capacities of peaks and not identify what those peaks were. This is also the strength of chromatography, for many forms of spectroscopy require some prior separation before analysis, making it a common preparatory step to further analysis. The combination of chromatographic and spectroscopic techniques is referred to as a hyphenated technique and is very popular in natural product research for its ability to simultaneously separate and analyse.

There are many different types of chromatography, usually named after one of their phases. Liquid and gas chromatography are named after the mobile phase, while paper and thin layer chromatography refers to the type of stationary phase used. Of particular interest in the analysis of herbal medicine are "High Performance" varieties of liquid and thin-layer chromatography which use automated techniques to get more accurate, reproducible readings. Each technique has its own advantages and disadvantages.

2.2.1. Liquid Chromatography

This is one the most popular techniques found in the literature with a total of 303 papers included in the review. While LC simply refers the state of the mobile phase, the majority of papers analysing herbs used "High Performance Liquid Chromatography" (HPLC) or "Ultra-High Performance Liquid Chromatography" (UPLC) where the liquid mobile phase is pumped through a solid adsorbent stationary phase in a column under high pressure enabling sufficient resolution to be used quantitatively (figure 5), while LC is primarily used as a preparative technique for other forms of analysis. The only difference between the HPLC and UPLC is the amount of pressure used and the resolution achieved, UPLC being a later development of HPLC with superior pressure, resolution and speed (Dyad Labs, 2018).



Figure 5: Overview of HPLC (created with BioRender.com)

HPLC and UPLC account for 256 of the 303 papers (84%) with only 56 (18%) using the standard form (9 using both). It is especially useful for quality

control because of the process being largely automated which makes the results highly reproducible. They are usually presented as a line graph where the peaks can be compared to show the presence of various compounds and their relative quantity making it ideal for generating a chemical "fingerprint" against which other samples can be compared (figure 6).





Glycyrrhizic acid is shown as a peak of interest.

G. glabra will be used for all chromatograms and spectrograph examples due to the relevance of this herb to the remaining thesis, and to enable comparison.

This ability to detect specific compounds by their elution peaks is reflected in the overwhelming majority of the papers using LC to identify quality markers in order to standardise market products.

Purpose	Number of Studies
To identify quality markers	231
To detect pyrrolizidine alkaloids	23
To detect adulterant species	15

To detect mycotoxins	15	
To detect pesticide residues	11	
To detect tropane alkaloids	9	
To detect adulterant drugs	5	
To detect heavy metals	3	
To polycyclic aromatic hydrocarbons (PAH)	3	
To detect adulterant toxins	3	
To detect adulterant parts of correct plant species	1	

Table 2: Uses of Liquid Chromatography in the Analysis of Herbal Teas

The primary use of LC is to separate out compounds for further analysis with spectroscopic techniques for which HPLC features ultraviolet (UV) detection as an integral part of the system to detect when different compounds are eluting from the column. However, 254 (84%) of the studies using LC also hyphenated it with another spectroscopic technique such as MS, NMR or IRS. MS was the most popular with 204 papers (67%) combining it with LC while only 11 used NMR and 9 of those used it in conjunction with MS to provide additional data. 6 papers used IRS spectroscopy with LC, and in all cases this was used alone, without any other spectroscopic technique.

LC seems the obvious choice in analysing herbal teas since they are naturally delivered in liquid form. It therefore does not require the solution to be subjected to additional processing, preserving less stable compounds while GC is superior at analysing more volatile compounds that are unchanged by heat (Ng, 2017). It is especially useful for analysing samples containing salts or carrying a charge which cannot be analysed with GC (Painter, 2018). It also cannot analyse inert substances that do not dissolve in the usual solvents used making it unable to detect the addition of substances like sand, silica and maltodextrin (Gafner et al., 2023).

Illustrating these limitations, Karioti et al. (2014) used procyanidin content to develop detailed reports of the chemical composition of *Tiliae flos* (*Tilia cordata* Miller, *T. platyphyllos* Scop. & *T. x vulgaris* Heyne, *Tiliaceae*) but could not record the volatile content, despite these being important markers of quality. Yi et al. (2007) also had difficulty detecting volatile carvacrol and Z-butylidenephthalide in a quality control test for *Dang Gui Bu Xue Tang* (a combination of *Astragalus membranaceus* (Fisch.) Bge., *Fabaceae* and *Angelica sinensis* (oliv.) Diels, *Apiaceae*) despite these being important components of the effects of *A. sinensis*.

2.2.2. Gas Chromatography

GC works along the same principles as LC, except that the mobile phase is vaporised and carried by an inert gas such as helium, hydrogen, nitrogen or argon (Drawell, 2023) into the stationary phase which is located inside an oven where the temperature can be controlled (Turner, 2020) (figure 7). The stationary phase is usually a liquid coated on a solid support contained within a glass column which elutes the components of the mobile phase at different rates depending on their different interactions with the stationary phase. The resulting extraction is then passed to a detector which displays the results as a chromatogram. Various types of detector can be used, such as flame ionisation detectors, electron capture detectors, and mass spectrometers, enabling the simple chromatogram, similar to that obtained by LC (figure 6) to be combined with three-dimensional information on the analytes to be obtained.





Of the 34 studies included in the review that used GC, 27 (79%) of them connected it with MS. Of the remaining 7 studies, two used a flame ionisation detector (Farajzadeh et al. 2020; Ugoeze et al., 2023), two used an ion mobility spectrometer (Wang et al., 2021; Li et al., 2024), one used an e-nose (Qin et al., 2024) and one used both an electron capture detector and a flame photometric detector together (Bicci et al., 2003). Only one study used results based on a chromatograph without any additional detector (Naithani & Kakkar, 2004).

Purpose	Number of Studies
To identify quality markers	20
To detect pesticide residues	9
To detect heavy metals	5
To detect PAH	3
To detect adulterant drugs	1

Table 3: Uses of Gas Chromatography in the Analysis of Herbal Teas

One of the most notable trends in the literature of gas chromatography is that there are more papers dedicated to detecting pesticides than with any other technique. Out of 17 papers that searched for pesticides, 9 used GC (53%), with a similar proportion of studies detecting PAH also using GC (3 out of 6, or 50%). This comprises 18% of the total GC papers, compared to the other 3 which used LC, comprising only 1% of the total LC papers. This is because it can be both selective and sensitive with simultaneous detection of many residues at lower concentrations than other techniques (van der Hoff & van Zoonen, 1999) giving it an important specialist role in the detection of volatile compounds that includes many environmental pollutants.

GC is also useful in the analysis of storage methods because the volatile components are the most affected by storage. One example of this is Han et al. (2023), who used GC-MS to compare the effects of different drying methods on the volatile oil content of flower scented tea (*Bletilla striata*, (Thunb.) Rchb.f. (1878), *Orchidaceae*) in order to preserve the scent. They found that freeze drying without steam fixation retained the most volatile oils, compared to air drying and stove drying. This method retained the best scent and flavour profile, while also retaining the most flavonoids, polysaccharides and phenolic components, as well as having the best antioxidant and antibacterial profiles.

2.2.3. Thin Layer Chromatography

TLC uses a thin layer of material made of small particles for the stationary phase. This causes the mobile phase to rise along the surface via capillary action at a constant rate and the compounds adhere to the vacant spaces (termed adsorption) at differing rates depending on their relative solubility in the mobile phase and their affinity for the stationary phase (figure 8). These then remain in place after the mobile phase has evaporated and can be derivatized by various treatments to make all the analytes detectible (CAMAG, n.d).



Figure 8: Overview of TLC (created with BioRender.com).

TLC results can then be displayed visually in a series of compact bands as the mobile phase travels up the plate which can easily be compared to reference standards or other samples on the same plate to obtain a relative picture of quality or adulteration (figure 9). This also facilitates a high throughput with roughly half the development time of HPLC and the option of developing multiple samples at the same time. The small size of the mobile phase means that it is also relatively inexpensive on materials. This has facilitated the growth of a High Performance Thin Layer Chromatography (HPTLC) atlas of plants (HPTLC Association, 2020) which can be used for reference standards for all herbal products.





Track 1 shows glycyrrhizin used as a reference standard, tracks 2-7 show *G. glabra* and tracks 8-13 show *G. uralensis* with additional quenching zone at 0.43 Rf that can enable species identification.

TLC is a less popular method of analysing herbal teas than LC with only 16 papers included in the review utilising this method but deserves a thorough examination because its use for analysing herbal materials is expanding.

The most common reason for using TLC was to find identifying quality markers to confirm a correct species, in 10 of the papers (63%); of the others, the uses were as varied as detecting adulterant species, adulterant drugs or moulds and mycotoxins. Of those detecting adulterant species Guzelmeric et al. (2017) compared HPLC with HPTLC to detect adulteration of chamomile (*Matricaria chamomilla* L., *Asteraceae*) with other asteraceae (*Bellis sp., Anthemis sp., Tanacetum sp.* & *Chrysanthemum sp.*) and found HPTLC to be the superior method; Shen et al., (2012) analysed Chinese star anise (*Illicium verum* Hook.f., *Schisandraceae*) for adulteration with neurotoxic Japanese star anise (*Illicium anisatum* L. *Schisandraceae*); and Lam et al. (2016) used HPTLC to differentiate snow chrysanthemum (*Coreopsis tinctoria* Nutt., *Asteraceae*) from regular chrysanthemum (*Chrysanthemum* × *morifolium* (Ramat.) Hemsl., *Asteraceae*). Two studies used HPTLC to screen for adulterant drugs, including Miller & Stripp

(2007) who screened New York outlets selling Chinese herbal medicines for prescription drugs and found some in patent pills but none in teas, while Wang, Y. et al. (2023) used HPTLC combined with a bioluminescent bacterial solution for derivatization to detect hyperlipidemic drugs in a selection of herbal teas. Two more studies used HPTLC to detect mycotoxins, with Halt (1998) using TLC to find 1 of the 11 teas contaminated with *Aspergillus flavus* Link, *Trichocomaceae*, and trace amounts of ochratoxin, while Haq et al. (2024) used TLC to detect aflatoxin and ochratoxin in herbal teas, but then used HPLC to quantify these findings. These show the potential for TLC to be almost as varied as for liquid chromatography but also reveal some of its limitations.

The main limitation of TLC and HPTLC is that they are mainly qualitative and only semi-quantitative. This means that while it can determine the presence of a compound, it is difficult to determine the amount. This is demonstrated in the study by Haq et al. (2024) who used TLC to determine the presence of mycotoxins and HPLC to quantify them. Quantification with TLC can be attempted through the appearance of faded or absent bars on a chromatogram which might alert the analyst but intelligent adulterers can attempt to replace these with a purified chemical marker from a cheaper source that elutes onto the plate at the same point as the desired compound (Gafner et al., 2023). You et al. (2022) found this issue in 5 of 14 turmeric (C. longa) supplements which were adulterated with curcumin from synthetic sources and detected only by the use of ¹⁴C radiocarbon analysis via spectroscopic methods. Some of these dyes are not only ways to increase the value of poor products but can be a significant safety issue. A number of turmeric products from Bangladesh were found to be deliberately contaminated with lead chromate which mimics the brilliant orange-yellow colour or fresh turmeric but results in increased lead and chromium in the blood of people who consume it (Forsyth et al., 2019; Newby 2023) and a similar outbreak was reported in the U.S.A. among cinnamon containing drinks for children (U.S. Food and Drug Administration, 2024). While this could be easily detected through TLC, it shows that adulterers often do not consider the consequences of their actions on the end users of their products and if toxic dyes can best mimic the compounds they wish to emulate, then they will likely be used.

As can be seen from these papers, a High Performance variety of TLC has been developed similar to the High Performance version of LC. It is a relatively new technique evidenced by the fact that far fewer papers have been written using it, with only 7 included in the present search, with all but one (Ramachandran et al., 2012) being written in the last 8 years. Its high throughput speed, low cost and options for analysing the results visually or by computer readout means that it is rapidly becoming the technique of choice for verifying species and detecting adulteration in herbal medicines (Frommenwiler et al., 2019; Omicron UK, 2018). The number of studies found in the present search belies the popularity and potential of HPTLC in herbal medicines research due the fact that it was restricted to teas while most studies investigate supplements using other forms of extraction and raw herb materials. If the Pubmed search is expanded to include herbal medicine that is not in tea form using the phrase "(HPTLC) AND (herbal medicine) AND ((adulteration) OR (quality) OR (contamination))" then 196 results were returned on 28th July 2024. Some of these may include raw materials used in teas without specifically mentioning the term and there is no reason why HPTLC cannot be used with water extractions, so the literature using this method is almost certainly larger than this search implies and likely to expand in coming years.

2.3. Spectroscopy / Spectrometry

Spectroscopy is the study of how radiated energy and matter interact; spectrometry is the measurement of this interaction (Verichek Tech Services, 2016). The common principle to all methods involves projecting some form of energy at the matter being analysed which absorbs it at a specific wavelength creating an excited state and emits it when it returns to its ground state. The specific wavelengths absorbed or emitted provide detailed qualitative and quantitative information about the substance.

2.3.1. Mass Spectrometry

MS measures the mass-to-charge ratio (m/Q) of ions in order to plot a mass spectrum which reveals the elemental or isotopic signature of a molecule or

compound and enables its quantification. First the substance under investigation is ionised in order to make it susceptible to influence by a magnetic field, then the charged ions are accelerated to a known speed and deflected using a magnetic field (Clark, 2019) (figure 10).



Figure 10: Functional Diagram of a Mass Spectrometer (created with BioRender.com).

The degree to which the magnetic field deflects the ionised molecules from their course is detected and used to determine the molecular weight (figure 9).



Figure 11: Mass Spectra of Glycyrrhizin, the main constituent of *Glycyrrhiza spp.* (Sawada, Matsuda & Hirai, 2008: reproduced under licence).

The base peak corresponds to glycycrrhizin's molecular weight of 822.9 g/mol.

This process can be further enhanced by adding an additional mass analyser to filter the ions and take further measurements as they travel, known as tandem mass spectrometry, MS/MS or MS² (Mittal, 2015). The most common forms of tandem mass spectrometry involve measuring the time it takes to pass through the detector, their Time of Flight, based on the principle that heavier ions travel more slowly than lighter ions (Gilhaus, 2005), and by selecting specific ions based on the stability of their trajectories as they pass through oscillating electrical fields generated by four cylindrical rods (a quadrupole). Quadrupoles are often set up in a triplicate formation, called a Triple Quadrupole, where the first quadrupole is used to select certain ions, the central quadrupole acts as a collision chamber to fragment them and the last selects the fragments to be analysed (Schreiber, 2017). These setups are quite common in biomolecular research of complex molecules.

MS is the most popular spectroscopic method in the reviewed papers for analysing herbal teas with a total of 240 papers using some form of this method. This is mainly due to its powerful ability to identify and characterise unknown compounds which gives it an enormous edge compared to chromatographic techniques when analysing a previously undescribed plant compound, or forensically analysing an adulterant substance whose identity is uncertain.

Purpose	Number of Studies
To identify compounds as quality markers	165
To detect pyrrolizidine alkaloids	23
To detect adulterant species	15
To detect heavy metals	13
To detect mycotoxins	10
To detect pesticide residues	10
To detect adulterant toxins	7

To detect adulterant drugs	4
To detect PAH	3

 Table 4: Uses of Mass Spectrometry in the Analysis of Herbal Teas

We can see this reflected in the main usage being similar to LC, to identify and characterise marker compounds for future quality control. One of the main differences is that many of these studies are characterising a plant and identifying markers for the first time which can then be found with a simple chromatography test, or where a forensic analysis of herbs was being undertaken without prior knowledge of what the researchers were searching for. However, the cost is high, approximately \$500,000 initial layout with \$250,000 yearly maintenance fees (EI-Khoury, 2018) making them the domain of specialist laboratories dedicated to research and not for a cost-conscious quality control centre of a small herbal business.

MS is usually combined with a chromatography step to separate the compounds before analysis. 225 (94%) studies included in this paper combined it with some form of chromatography with LC being the most popular (204, or 91%), followed by GC (27, or 12%) and rarely with TLC (6, or 3%). These preparatory steps can have considerable influence on the results. A comparison of MS studies searching for mycotoxins in conventional tea (C. sinensis) demonstrates this: Reinholds et al. (2020) found that almost 97% of black teas, 88% of green teas and 100% of oolongs included in their study contained quantifiable levels of fungus with all Puerh samples containing mycotoxins despite having the lowest levels of fungal contamination, while Monbaliu et al. (2010) analysed 91 teas and found only one sample of Ceylon melange to be contaminated with no mycotoxins in the drinkable products despite also testing Puerh. The testing of Puerh is significant as this tea is fermented and known to contain high concentrations of mould (Sedova, Kiseleva, & Tutelyan, 2018; Pandey, Samota & Sanches Silva, 2022) showing that the sensitivity of MS makes it vulnerable to detecting differences in the preparatory steps as well as in the products themselves. In this example, Monbaliu et al.

(2010) used a UPLC separation technique, while Reinholds et al. (2020) used 2-dimensional LC. 2-dimensional LC has been developed to overcome the limitations of 1-dimensional techniques in the separation of compounds closely related in polarity and charge, such as may be found in food contaminants. This may explain how Reinholds et al. (2020) were able to find mycotoxins in Puerh, where Mobaliu et al. (2010) were not. Standardised methods of preparation can mitigate this, such as that proposed by Zhou, H. et al. (2021) who developed a method of extracting aflatoxins with acetonitrile and purifying with mixed fillers to develop a more sensitive method of detection with LC-MS. Hu, W. et al. (2023) also demonstrated that it may be possible to avoid this issue entirely by replacing chromatographic separation with paper spray ionisation combined with a portable mini-MS, allowing for direct analysis of complex samples.

2.3.2. Nuclear Magnetic Resonance

NMR is one of the preeminent techniques for determining the structure of organic compounds (Aryal, 2020). By placing molecules in a strong magnetic field the nuclear spin can be made to align with the magnetic force. The energy transfer required to make this happen occurs at a wavelength that corresponds to radio frequencies and when the spin is reversed it releases energy at this same frequency which can be detected as a radio signal and used to provide information about individual molecules and their functional groups (figure 12).



Figure 12: Principle of NMR Spectroscopy (created with BioRender.com).

It can provide information not just on the structure of the resonating atoms but their quantity, their neighbouring atoms and relative positions (Zinkel, 2019). By examining the peaks and cross-peaks of the readout which relate to the chemical shifts, multiplicity, coupling constants and integration characteristics of the molecules (figure 13), it is possible to build a complete 3D structure (Emery Pharma, 2018).





Top line (black) shows *G. glabra* crude extract, middle line (blue) shows *G. glabra* extracted using a targeted enrichment method and the bottom line (red) shows liquiritin, the target compound, for comparison.

NMR is capable of determining the molecular structure of an unknown compound in even greater detail than MS but so is its expense with initial costs often exceeding \$1 million although smaller, lower resolution benchtop versions are available for under \$100, 000 (Reisch, 2015). NMR is also relatively insensitive requiring 10-100 times the amount of sample than MS, LC or GC to take a measurement and will still usually only return information on 50-200 compounds with concentrations >1 μ M, compared with thousands that can be obtained by MS with concentrations as low as 10 - 100 nM, making it unsuitable for detection of trace metabolites (Emwas et al., 2019).

This is reflected in the literature with only 17 papers utilising NMR with more than half of these (10, or 59%) also using MS and often adding NMR to acquire additional information when MS was not specific enough. All but one study was searching for specific quality markers. The only paper that did not employ NMR to identify structures was exploring a novel method to identify the composition of herbal mixtures without the use of prior separative techniques (Marchetti et al., 2020), building on the work of Booker et al. (2014) who used the same technique to analyse chemical variability of turmeric (*C. longa*) along value chains. Given the expense generally involved, it does not seem a likely method to become popular in the herbal tea industry but could if the results can be replicated with the cheaper benchtop NMRs.

2.3.3. Ultraviolet-Visible Spectrophotometry

UVS works by passing wavelengths of light in the UV (100 - 400 nm) and visible (400 - 700 nm) spectrums through a sample and measuring the light received by a sensor on the other side (Raja & Barron, 2020) (figure 14). The wavelengths not transmitted indicate the photons have exactly matched the energy band gap required to promote a molecule from its ground state to an excited state and have been absorbed. This provides an absorbance spectrum readout that can be compared to published literature for qualitative identification, or used quantitatively, measuring the concentration of a sample using Beer's Law if the absorptivity of a substance is known, or calculated with a calibration curve (Venton, 2020). The light source is usually a deuterium lamp that emits light in the 170–375 nm UV spectrum, and a tungsten filament lamp, which produces light from 350-2,500 nm, for the visible range which is then filtered for specific wavelengths. Alternatively a diode-array which can measure a whole spectrum of light in a single run can also be used. It is particularly useful in natural product research for determining the level of conjugation in organic molecules due to increased conjugation being associated with a lower energy gap, thus absorbing lower energy longer wavelengths of UV and visible light (Ashenhurst, 2022).



Figure 14: Principle of Ultraviolet-Visible Spectrophotometry (created with BioRender.com)

Of the 88 studies included in the review, 84 (95%) coupled UVS with HPLC. This is a common combination and most HPLC systems feature UVS as an integral part of the system to detect compounds not visible light and enable their quantification by comparison to a reference beam (Taylor, 2015). This means that the other HPLC studies will have also used UVS but did not discuss the results in their papers. It is also integrated into most HPTLC devices where it enables detection of analytes invisible under normal light (Camag, n.d.) (figure 15).





The reference compound, glycyrrhizin (track 1) is only visible under UV 254 nm.

One of the major limitations of UVS is its reliance on published literature. Most studies (84 or 95%) compared their analytes against known constituents. Among other uses, Zhao et al. (2012) suggested UVS to detect adulteration of Chinese medicines with drugs. They successfully identified 11 antihypertensive drugs mixed into herbal medicines, but since they only used spiked samples, they had the references to hand. Herbal medicines are increasingly adulterated with designer analogues of drugs made to avoid detection, meaning LC-MS and NMR are often necessary for their identification (Patel et al., 2014; Haneef et al., 2013). Zhang, R. et al. (2020) and Özzeybek (2020) employed UVS to detect pesticide residues, which constitutes an application more appropriate to real world situations since the pesticides the farmers use can be supplied to the quality control centres and used as reference standards. Zwerger et al. (2023) also used UVS to detect pyrrolizidine alkaloids from ragwort (*Senecio spp. L., Asteraceae*), also using MS to identify them initially, but which could then be analysed for their UV spectra and detected quickly and cheaply with a supercritical fluid UPLC method coupled with UVS.

Another limitation of UVS is that the wavelength of interest can be absorbed by non-target analytes resulting in an overestimation of quantities (Gafner et al., 2023). This can be taken advantage of by fraudulent sellers who may adulterate an extract with another substance that has a similar absorption spectrum, such as adding anthocyanin-rich extracts to bilberry (*Vaccinium myrtillus* L., *Ericaceae*) or elderberry (*Sambucus* spp.) extracts.

2.3.4. Infrared Spectroscopy

The setup of IRS is similar to UVS spectroscopy, passing specific wavelengths of radiation, this time in the infrared spectrum (780 nm - 1 mm), through a sample and measuring how much is transmitted to the other side. The effect of infrared light is very different, interacting with the bonds of molecules causing them to stretch or bend in symmetric or asymmetric ways when energy is absorbed from a particular wavelength depending on the functional groups within the molecule (Reusch, 2013) (figure 16).



Figure 16: Effects of Infrared Radiation on Vibrational Modes of Molecules (created with BioRender.com, based on Soderberg, 2023)

Troughs in transmittance in the region above 6.5 μ m gives information about functional groups, while the region below 6.5 μ m is known as the fingerprint region and gives a very intricate pattern that can be used to determine a compound. IRS can take place in the near-, mid- or far-infrared regions, or use a Fourier Transform method to simultaneously beam many frequencies at once, repeated in bursts of different combinations over a short time and then use a computer to calculate the absorbance at each wavelength (figure 17). The near range, from 780 - 2526 nm is the most commonly used in quality control due to having higher energy and penetration capacity and producing less heat (Zeng et al., 2011, p.398) while the mid-range spectrum (2.5 - 25 μ m) is superior for identifying structure and functional groups (Liang et al., 2011, p.153-158).





The two lines show how the transmittance of different infrared wavelengths as *G*. *glabra* is affected after interacting with copper ions in wastewater.

Only 16 papers included in the review used IRS but these reveal a wide range of potential uses. Kokalj, Štih, Kreft (2014) found it to be low cost, rapid and simple making it suitable for routine examination of single herbs and mixtures. Ma et al. (2017) were able to detect samples of Chinese yam (*Dioscorea polystachya* Turcz., *Dioscoreaceae*) powders that had been adulterated with cheaper corn starches. Yap, Chan & Lim (2007) used IRS to differentiate asian ginseng (*Panax ginseng* C.A. Meyer, *Araliaceae*) from its American counterpart (*P. quinquefolius* L., *Araliaceae*). Lee et al. (2014) found IRS to be superior to GC for identifying the caffeine and catechin content of tea (*C. sinensis*). Chen et al. (2019) found IRS equally effective to HPLC when evaluating decoction pieces of *Rhizoma Atractylodis* (*Atractylodes macrocephala* DC., *Asteraceae*). Qin et al. (2024) employed IRS along with GC to detect age fraud in sun dried, age stored mandarin peel (*Citrus reticulata* Blanco, 1837, *Rutaceae*). Cebi, Yilmaz & Sagdic (2017) used IRS to detect sibutramine in spiked tea samples, although real world

adulteration with designer analogues that lack reference standards might be more problematic. This wide range of uses demonstrates the potential versatility of IRS for a variety of applications ranging from species identification, constituent quantification, adulteration with drugs and even age fraud.

Despite this range of uses, the use of IRS remains low in Europe. This is due to most labs having UVS fitted as standard equipment on their chromatography devices meaning an extra expense to install an infrared device, train staff in its use and to develop the reference models, for which there are few available in English and whose development requires time and expertise in chemometrics (Zeng et al., 2011, p.399 & 412). The literature is instead dominated by MS or NMR for which the literature shows there is already a vast library of data from universities, pharmaceutical companies and the Asian herbal medicine market which is far more integrated into conventional healthcare and better funded than in Europe (Liu & Salmon, 2010).

2.3.5. Optical Emission & Atomic Absorption Spectroscopy

OES and AAS have a very specific use for detecting metal content. While OES achieves this by measuring the wavelengths of light emitted by a sample, AAS measures the wavelengths absorbed by a sample (CS Analytical, 2020). Due to the way they interact with light, it can only offer elemental information on metals and for chemical structure or chirality, MS is the preferred method (Measurlabs, 2024).

In OES, a spark is generated by an electrode and the sample in an high energy inductively coupled plasma state and a spectroscope detects the unique spectrum specific to each element (Shimadzu, n.d.) (figure 18).





In AAS, the sample is nebulised and then atomised with a heat source, usually either a flame or a graphite furnace. Then, specific wavelengths of light that correspond to the atomic absorption spectrum of particular elements is projected through the sample and the amount of light absorbed to transition electrons from their ground state to an excited state can determine the concentration of that particular element (figure 19).



Figure 19: Principle of Atomic Absorption Spectroscopy (created with BioRender.com).

These techniques were the least common included in this review, totalling only 13 papers. 8 used OES and 7 used AAS (2 using both) with all employing this technique to evaluate toxic levels of metals in herbal teas. Rubio et al. (2012) used OES and found levels of aluminium in mint (*Mentha* sp. L., *Lamiaceae*) that warranted imposing consumption limits, while Popović et al. (2022) used AAS to find levels of manganese in *Betulae folium* (*Betula pendula* Roth and/or *Betula pubescens* Ehrh., *Betulaceae*) and *Frangulae cortex* (*Rhamnus frangula* L. syn. *Frangula alnus* Miller, *Rhamnaceae*) in Serbian markets that could potentially pose some health hazards to consumers. Malik et al. (2013) used both techniques to determine that hibiscus (*Hibiscus sabdariffa* L., *Malvaceae*) contained enough aluminium to warrant a consumption limit of no more than 1 L of hibiscus per day.

2.4. Biological Methods

Biological methods differ from those above in that they use some kind of biological material to test for authenticity and quality in herbs. The most widespread is certainly genetic testing but there were also several examples of using biological assays to assess the effects of herbs and extracts on cell cultures and even live animals to determine if the material was having the effect it was supposed to.

2.4.1. DNA Barcoding and Genetic Analysis

DNA barcoding takes a small section of genetic code from an unidentified organism and compares it to a reference library of DNA sections such as the Barcode of Life Data (BOLD) systems database (iBOL, 2020), Catalogue of Life database (Bánki et al., 2022) or the National Institute for Health's Genbank (Benson et al., 2012) (figure 20). It is often quicker and more precise than traditional taxonomic classification but the reliability of data is only as good as the database and, in the rush to categorise as many species as possible, there are many errors that may be present (Lathe, 2008). Countries where there is a strong tradition of herbal medicine have created large genetic databases dedicated to these materials, such as the Medicinal Materials DNA Barcode Database of Traditional Chinese Medicines (Lou et al., 2010; Wong et al., 2018), but where there is less of a vested interest in employing curators to error check and correct herbal plant materials, the number of vouchered species is likely to be less and the database of poorer quality.



Figure 20: DNA Barcoding Workflow (created with BioRender.com)

26 studies included in the review used genetics. As expected, all were being used to identify the correct species, almost half (12, 46%) to check for adulterant species and identify them, the rest to confirm the species when using another technique for identification. 11 papers looked at adulteration in commercial supplies, revealing just how frequently adulterated tea samples can be:

Author	Aim	Result
Chan et al. (2024)	Investigate the quality of 33 commercial "five flowers tea" products, containing <i>Lonicera japonica</i> Thunb. <i>Caprifoliaceae</i> , <i>Plumeria rubra</i> L. <i>Apocynaceae</i> , <i>Pueraria lobata</i> (Willd.) Ohwi, <i>Fabaceae</i> , <i>Bombax ceiba</i> L. <i>Malvaceae</i> and <i>Sophora japonica</i> L., <i>Fabaceae</i>	 Only 8 samples (24.2%) contained all of the five floral herbs. Acceptable substitutions for cheaper alternatives included 28 (84.8%) substituting <i>S. japonica</i> for <i>Agrostis capillaris</i> L., <i>Poaceae</i>, and 21 (63.6%) substituting <i>P. lobata</i> with <i>Prunella vulgaris</i>, L., <i>Lamiaceae</i>. 10 samples (29%) contained adulterants, including <i>Desmodium styracifolium</i> (Osbeck) Merr. <i>Fabaceae</i>, <i>Potenilla spp.</i> and <i>L. confusa</i> being used in place of <i>L. japonica</i>. Contamination with insects <i>Lasioderma serricorne</i>, (Fabricius, 1792), <i>Ptinidae</i> was also confirmed by macroscopic investigation and DNA barcoding.
De Castro et al. (2017)	Examine 32 herbal teas.	2 (6%) found to be adulterated.
Duan et al. (2017)	Identify species in <i>Radix</i> <i>Clerodendrum</i> tea samples used in the Dai ethnic group's medicine.	Of 27 samples, only 1 (3.7%) was authentic <i>Clerodendrum japonicum (Thunb.) Lamiaceae</i> . Most were other medicinal species but 4 were potentially toxic <i>Lantana camara L. Verbenaceae</i> .
Lu et al. (2022)	Identify adulterants in 18 batches of commercial <i>Violae herba</i> (<i>Viola</i> <i>philippica</i> Cav., <i>Violaceae</i>) decoction pieces.	Most decoction pieces, including the reference material were identified as <i>Viola prionantha</i> Bunge, clashing with the accepted pharmacopoeia which should be updated.
--------------------------	---	--
Olivar et al. (2016)	5 <i>Vitex negundo L. Lamiaceae</i> samples, often used as herbal tea in the Philippines.	Only one satisfied the database criteria for genetic authenticity.
Omelchenko et al. (2019)	Examine 6 herbal teas, 6 herbal medicines & 6 spices for adulteration.	12 (67%) products contained different materials to those labelled. 6 likely to be economically motivated.
Osathanunkul (2018)	Find adulterants in Soursop (Annona muricata L. Annonaceae) teas.	3 out of 11 (27%) samples contained incorrect species.
Wang et al. (2016)	To find a DNA signature region which can be used to identify <i>Angelica</i> <i>sinensis (Oliv.) Diels. Apiaceae</i> in decoction powders.	Of 9 decoction powders, 7 (78%) were identified as <i>Angelica pubescens Maxim. Apiaceae</i> .
Xin et al. (2015)	Authenticate 90 commercial Rhodiola crenulata (Hook.f. & Thomson)	Only 36 (40%) contained the correct species. 35 (38.9%) contained <i>R. serrata H. Ohba</i> and 9 (10%)

	<i>H.Ohba Crassulaceae</i> products from hospitals and drug stores.	<i>R. rosea L</i> . Remaining 10 (11.1%) were 3 other <i>R</i> . species.
Xin et al. (2022)	To use DNA barcoding to identify adulteration along the entire industrial chain of <i>Rhei Radix et Rhizoma</i> (<i>Rheum palmatum</i> L., <i>Polygonaceae</i>)	25% of decoction piece samples collected midstream and downstream were identified as adulterants but exact species of adulterant could not be identified. All 8 patent medicines were identified as genuine.
Xiong et al. (2022)	Identify Andrographis paniculata (Burm.f.) Nees, Acanthaceae and differentiate species from adulterants <i>Mesembryanthemum cordifolium</i> L.f., <i>Aizoaceae</i> and <i>Rhinacanthus nasutus</i> (L.) Kurz, Acanthaceae.	No commercial products were found to be adulterated, but five samples which used alcohol extracts could not be amplified.

 Table 5: Results of Genetic Analysis on Teas to Detect Adulteration

DNA barcoding has added much to the safety of herbal medicines by confirming the correct species is being used but also has certain limitations. When reports of liver damage induced by black cohosh supplements (Actaea racemosa, L., Ranunculaceae), began to circulate in 2002, but without a causal relationship being identified, it was presumed that it may be due to adulterant species (National Institute for Health, 2020). Its current use in the West is for menopausal symptoms of hot flushes and excessive sweating, for which it is approved by the European Medicines Agency (EMA, 2018), but it has a history of use in the Americas for conditions as varied as musculoskeletal pain, fever, cough and gynaecological issues. Similar cohosh plants A. heracleifolia, A. dahurica and A. cimicifuga are also used in Chinese medicine under the name Sheng Ma 升麻 and sometimes *Hei Sheng Ma* 黑升麻 (*Hei* 黑 = "black," and therefore "black cohosh" in Chinese) for releasing "heat" trapped inside the body and manifesting as fever, rashes, sores and ulcers in the upper body (Bensky et al., 2004, p. 82). These overlapping indications and the complexity of the naming conventions of Actaea species, which all have several synonyms, including Cimicifuga being used interchangeably for Actaea in nearly all circumstances, A. simplex also being known as A. cimicifuga, and Serratula chinensis S. Moore, Asteraceae being referred to as Guang Sheng Ma 广升麻 or Hei Sheng Ma 黑升麻 ("wide" or "black cohosh") in China and often exported to Hong Kong and the United States as black cohosh despite A. racemosa not growing in Asia, have caused much species confusion (Foster, 2013). DNA analysis provided a simple method which enabled species differentiation and was especially effective where HPTLC fingerprints produced similar results making chemical differentiation difficult, preventing a complete ban on black cohosh extract products for fear of safety concerns. However, no particular species was ever identified as causing the reports of liver toxicity and precautionary warning labels must be applied by law in Australia, Canada, the UK and US (Mohapatra et al., 2022). More recent studies has suggested that the A. heracleifolia used in Chinese medicine is therapeutically similar and a promising alternative to A. racemosa (Miao et al., 2019), while the hepatotoxicity reports could have been due to the fibrous roots of A. dahurica, which demonstrate subchronic hepatotoxicity not shown by the rest of the root, not being removed during processing (Yu et al., 2022). This raises a question of whether it was the incorrect species or an incorrect part which caused the reports of liver damage.

This is a major limitation of DNA barcoding (Blumenthal in Gustafson, 2015) and vigilant reporting practices will determine whether reports continue despite DNA barcoding having determined the correct species is being used.

A second limitation of DNA barcoding is it cannot be used on extracts and processed herbal material where the DNA will be degraded, incomplete or not present at all (Parveen et al., 2016). This problem affected Xin et al., (2022) when attempting to apply DNA barcoding to decoction pieces of rhubarb root (*R. palmatum*) which had been processed by steaming or charring and Xiong et al. (2022) who could not amplify the DNA from five samples of *A. paniculata* that had been alcohol extracted. This can have serious consequences, as demonstrated by the assessment of herbal supplements by the New York State Attorney General who attempted to cease sales of herbal supplements after testing revealed that only 21% contained verified DNA from the substances listed on the ingredients (James, 2015). It was soon criticised because many of the supplements sold were extracts and a good extract should only contain the compounds they are extracting and not any DNA from the original material (Morrell, 2015).

Another argument that can be levelled against DNA barcoding in the taxonomy of traditional medicines is that these methods were not the methods used by the original authors when describing the materials they intended to use. Therefore any number of species which can fit the description may have been meant and we are now forcing their morphological and ecological descriptions to fit a narrow DNA profile. This is evident in the way that many reference books of traditional herbal medicine have multiple species assigned a single entry (Bensky et al., 2004). Hence, when analysing the medicine of a traditional ethnic group for authenticity such as Duan et al. (2017) undertook, claiming only one of 27 samples tested was the official species, it assumes an elitist position of DNA barcoding over the people who actually developed and defined the medicinal material in the first place. Similarly, Lu et al. (2022) found so many samples of *Violae herba* did not correspond with the official pharmacopoeia's designation of *V. prionantha*. This was also the only species which had sufficient index components (cichoriin,

esculin, esculetin, and prionanthoside) to meet the Chinese Pharmacopoeia's requirements, suggesting that the official species needed to be revised.

2.4.2. Bioassays

Bioassays present an interesting opportunity to determine the quality of herbs by measuring their effects on a living cell culture or organism. It could be argued that this is the most important factor of quality control since people are often drinking herbal teas and consuming medicinal plants for their effects with the correct species and levels of a principal component being only a means to accomplish that effect. With herbal medicines, the components responsible for an effect may often be complex or unknown (Panossian, 2023; Zhang, W. et al. 2019) and several genetically varied species may have the same effect and be acceptable equivalents or substitutes (see section 2.6.2), so measuring the effect of a herb would be the ideal quality control mechanism.

66 papers included in the review used some form of biological assay to assess the quality of herbal teas. Many different assays were used for different purposes which are outlined in table 6.

Author	Bioassays Used	Purpose
Abourashed et al.	Neutral red cytotoxicity assay.	Assess safety of Clutia richardiana Müll.Arg., Peraceae
(2003)		decoctions by evaluating cytotoxicity.
Arthur et al. (2011)	DPPH and Ferric Reducing	Assess effect of steam pasteurisation on antioxidant
	Antioxidant Power (FRAP)	capacity of Lippia multiflora Moldenke, Verbenaceae to
	antioxidant assays.	determine if it affects the quality of the herb.
Cheng et al. (2022)	Minimum inhibitory concentration	Determine the compounds discovered by UPLC-MS that
	assays for <i>Staphylococcus aureus</i>	are most effective at inhibiting bacterial species
	and Moraxella catarrhalis	responsible for upper respiratory tract infections in
		Artemisia and Soft-Shelled Turtle Shell Decoction
		(Qinghao Biejia Tang), for use as quality markers.
Chua et al. (2018)	Folin–Ciocalteu, formation of	Compare the effects of storage duration on antioxidant
	flavonoid-aluminium complex, DPPH,	capacity of Orthosiphon aristatus (Blume) Miq.,
	2,2'-azino-bis(3-ethylbenzothiazoline-	Lamiaceae, on total phenolic content (TPC), total
	6-sulfonic acid) (ABTS), Ferrous ion	flavonoid content, antioxidant capacity, iron chelating
	(Fe ²⁺) chelation assay and Ferric (III)	and reducing capacity.
	reduction assay.	
Dong et al. (2023)	In vivo: Hot plate and acetic acid	Determine the constituents, discovered with HPLC-MS,
	writhing test in mice	with the most analgesic and antispasmodic effects in

		Peony and Liquorice (<i>Shaoyao-Gancao Tang</i>) decoction pieces and granules, for use as quality markers.
Fang et al. (2022)	Immunochromatographic strip.	Detection of various aminopyrine (a nonsteroidal anti-inflammatory drug) in adulterated herbal teas using a method of combining chromatography with antibody coated strips.
Félix-Silva et al. (2018)	Antimicrobial broth microdilution and agar dilution antimicrobial assays used to assess bacterial inhibition.	Compare antimicrobial activity of aqueous leaf extracts of <i>Jatropha gossypiifolia</i> L., <i>Euphorbiaceae</i> and <i>J.</i> <i>mollissima</i> (Pohl) Baill. used in Brazilian folk medicine and decide components to be used in quality evaluation, detected with HPLC and TLC.
Ferreres et al. (2014)	DPPH, Superoxide Anion (O2 ^{•-}) and Nitric Oxide (NO) scavenging antioxidant assays; acetylcholinesterase, butyrylcholinesterase and monoamine Oxidase-A inhibitory assays; lactate dehydrogenase release assay; 3-(4,5-dimethylthiazol-2-yl)-2,5-diphe	Assess antioxidant, cholinesterases and monoamine oxidase-A inhibition; cell injury, cell viability and protection from H ₂ 0 ₂ toxicity of <i>Grindelia robusta</i> Nutt., <i>Asteraceae</i> components, identified by HPLC, to select quality markers.

	nyltetrazolium bromide (MTT) cell	
	viability assay; Hydrogen Peroxide	
	(H ₂ O ₂)-induced toxicity assay (MTT	
	after exposure to herb and then	
	H_2O_2).	
Guan et al. (2024)	Intracellular accumulation of	Identify compounds in Fresh Ginger Decoction to Drain
	5(6)-Carboxyfluorescein diacetate to	the Heart (Shengjiang Xiexin Decoction) and its
	measure efflux by Multidrug	metabolites in rat plasma, detected with UPLC-MS, with
	resistance-associated protein (Mrp)-2	Mrp-2 inhibitory activity for use as quality markers based
	compared against MK-571 as	on previous study that suggested this was its
	positive control.	mechanism of action.
Guo et al. (2007)	Cell Counting Kit-8 (CCK-8) assay to	Evaluate the best extraction method of Peony and
	measure cell viability and Nitrite	Liquorice Decoction (Shao Yao Gan Cao Tang) by
	Reporter gene for Nuclear Factor	comparing their constituent profile using HPLC and
	kappa-light-chain-enhancer of	biological activity to find the best components to
	activated B cells (NF-кB) activity	optimise extracts for superior quality.
	assay.	
Hahm et al. (2021)	Folin–Ciocalteu and DPPH	Measure TPC and antioxidant capacity of 22 regional
		samples of ground ivy (Glechoma hederacea var.
		Longituba, L., Lamiaceae) to find consistent antioxidant

		compounds that can be used as quality markers for all regional specimens.
Han et al. (2023)	Total Flavonoids, Total Polysaccharides and TPC; DPPH; inhibition of <i>Escherichia coli</i> , <i>Pseudomonas aeruginosa</i> , and <i>Staphylococcus aureus</i> assay.	Discover the best method of preserving <i>B. striata</i> by comparing preserved samples flavonoid, polysaccharide and phenolic content, antioxidant capacity and ability to inhibit bacterial growth.
He et al. (2022)	Pancreatic lipase inhibitory activity assay with Orlistat as positive control, α-Glucosidase inhibitory activity assay with Acarbose as positive control.	Correlate the 14 main peaks obtained by UHPLC-MS/MS with <i>in vitro</i> pancreatic lipase and α-glucosidase inhibitory activity to determine the main active ingredients to use as quality markers.
Hernández-Abreu et al. (2011)	Vasorelaxant assay (ex-vivo, on endothelium-intact aortic rat rings).	Measure vasorelaxant capacity of tilianin, a component in <i>Agastache mexicana</i> (Kunth) Lint & Epling <i>Lamiaceae</i> , to determine its use as a quality control marker.
Jansen et al. (2017)	Plasmodial lactate dehydrogenase activity assay and	Determine anti-plasmodial and cytotoxicity of <i>Mezoneuron benthamianum</i> Baill., <i>Fabaceae</i> leaf constituents, identified with HPLC, NMR, UV, IR and MS

	2-(2-methoxy-4-nitrophenyl)-3-(4-nitr	for quality control of traditional antimalarial teas.
	-2H-tetrazolium, monosodium salt	
	(WST-1) cell proliferation assay.	
Jiang et al. (2024)	Lipopolysaccharide (LPS) induced	Determine the anti-inflammatory capacity of selected
	macrophage inflammation model.	quality marker compounds, identified with UPLC-MS in
		Astragalus and Cinnamon Twig Five Substance
		Decoction (Huangqi Guizhi Wuwu Tang).
Kan et al. (2022)	Dextran sulfate sodium-induced	Compare the individual ginsenoside levels in The Four
	ulcerative colitis mouse model	Gentlemen Decoction (Si Jun Zi Tang), measured with
	assayed with HE staining,	UPLC, and correlate them with the anti-inflammatory
	immunohistochemistry analysis for	effects to decide which should be used as quality
	occludin and ZO-1 antibodies stained	markers.
	with 3,3' Diaminobenzidine kit and	
	counterstained with hematoxylin,	
	enzyme-linked immunosorbent assay	
	(ELISA) for serum levels of TNF- α ,	
	IL-1β, IL-6 and NO.	
Lam et al. (2016)	DPPH antioxidant assay.	Evaluate the antioxidant capacity of 7 components of
		Snow Chrysanthemum (C. tinctoria), separated using

		HPTLC, to use as quality markers.
Li, J. et al. (2021)	Mouse myocardial infarction model with immunohistochemical analysis and HE staining.	Assays combined with HPLC analysis of components and molecular docking experiments to determine active components of "Compound <i>Longmaining</i> " decoction for use as quality markers.
LI, J. et al. (2022)	3,3'-dioctadecylindocine-low density lipoprotein uptake assay; <i>in vivo</i> : total cholesterol, triglycerides, low density lipoprotein and high density lipoprotein levels of mice fed a high fructose diet and supplemented with either acacetin, diosmetin, rutin or nothing; <i>ex vivo</i> liver weight/body weight index and the alanine transaminase and aspartate transferase of the same mice.	flowers (<i>Citrus aurantium</i> L. var. <i>amara</i> Engl., <i>Rutaceae</i>), determined by UPLC-MS, to decide constituents to be measured for quality control.
Li, T. et al. (2022)	Serum analysis of rats with adjuvant induced rheumatoid arthritis, before and after administration of Aconite	Serum analysed with UPLC-MS for metabolites of the herbal formula with potential anti-rheumatoid arthritis activity.

	Decoction (<i>Wutou Tang</i>) for 28 days.	
Li, Y. et al. (2021)	Everted intestinal sac model and <i>in vivo</i> portal vein blood samples of rats following oral administration.	Combined with HPLC-MS to determine the gut permeability and metabolisation of constituents in Angelica Decoction to Warm the Extremities (<i>Dang Gui</i> <i>Si Ni Tang</i>) to decide which compounds to use as potential quality standards.
Liang et al. (2021)	Cerulein induced acute pancreatitis mouse model.	Treatment of cerulein induced pancreatitis in mice used to validate the choice of quality markers, determined by HPLC-MS, selected for Beuplurum and Scutellaria Decoction to Order the Qi (<i>Chaiqin Chengqi Tang</i>).
Liang et al. (2022)	Alkaline phosphatase inhibition, thrombin inhibition and Na+/K+ -ATPase inhibition assays.	Determine the quality of rhubarb (<i>Rheum officinale</i> Baill., <i>R. tanguticum</i> Maxim. ex Balf., and <i>R. palmatum</i> Linn., <i>Polygonaceae</i>) based on bioactivity of major components identified with HPLC.
Liu et al. (2022)	Idiopathic pulmonary fibrosis rat model with histopathological analysis of cells using HE and Sirius red staining, and ELISA.	Detect pathological changes and levels levels of inflammatory cytokines after treatment with <i>Exocarpium</i> <i>Citri Grandis</i> (<i>Huajuhong</i> , <i>Citrus Grandis</i> (L.) Osbeck, <i>Rutaceae</i>), combined with network pharmacology and UPLC analysis of constituents to develop a method of determining grades of quality among decoction pieces.

Lu, Wang & Kong	In vitro: MTT assay; in vivo: mouse	Measure anti-inflammatory effect of whole decoctions vs
(2011)	paw oedema model.	fractions, quantified with HPLC, and single herbs in
		Coptidis Decoction to Resolve Toxicity (Huang Lian Jie
		<i>Du Tang</i>). MTT is mentioned in methods but no results
		are discussed.
Mahamoud et al. (2024)	Total organic carbon concentration,	Determine the stability of polyphenols in red rooibos (A.
	Folin-Ciocaltreau assay to measure	<i>linearis</i>) tea, measured with LC-MS, to evaluate the
	TPC and DPPH to measure ROS.	degradation of teas used in rodent experiments which
		are often prepared in batches and stored.
Micheli et al. (2014)	DPPH and FRAP antioxidant assays.	Compare the traditional decoction method and modern
		extraction of <i>Teucrium chamaedrys</i> L., <i>Lamiaceae</i> on
		phenol content, measures with HPLC, and antioxidant
		activity.
Pinela et al. (2016)	Folin–Ciocalteu and DPPH assays.	Measure the effects of radiation on TPC and antioxidant
		capacity of Dwarf Mallow (<i>Malva neglecta</i> Wallr.,
		Malvaceae) to determine if gamma radiation sterilisation
		of products affects their antioxidant capacity. UPLC-UV
		was also used to measure effects on constituents.
Qin et al. (2019)	Griess test assay and	Compare Kudzu Decoction (Ge Gen Tang) dry herb
	ELISA.	pieces and its instant granules for their constituents,

		determined by UPLC-MS, and effects on NO production
		and modulation of interleukin(IL)-1 β , IL-6 and Tumour
		Necrosis Factor- α (TNF- α) to determine
		anti-inflammatory effects.
Qu et al. (2023)	Allergic asthma rat model and	Identify the anti-asthma and anti-inflammatory effect of
	zebrafish inflammation model.	major bioactive compounds found in rat serum and
		identified with UPLC-MS, after being fed raw and wine
		steamed Schisandra chinensis (Turcz.) Baill.,
		Schisandraceae, to determine the main quality markers
		and how wine-steaming changes them.
Razmovski-Naumovski	FRAP and DPPH antioxidant assays.	Compare Angelicae sinensis Radix (A. sinensis)
et al. (2020)		decoction pieces and granules with UPLC and
		determine the superior antioxidant.
Schmeda-Hirschmann	DPPH, xanthine oxidase &	Measure antioxidant capacity, TPC and total flavonoid
et al. (2003)	erythrocyte lipid peroxidation	content of components of Boldo (Peumus boldus
	antioxidant assays; Folin–Ciocalteau	Molina, <i>Monimiaceae</i>) to determine which should be
	and aluminium chloride assays.	used for quality control.
Schunk et al. (2016)	Micronucleus test assay for	Assess the cytotoxicity due to heavy metal content,
	cytotoxicity.	determined by OES, in Brazilian herbal teas.
Seo et al. (2019)	Cytotoxicity assay, Oil Red O	Determine which compounds in Ginseng, Poria and

	staining, Triglyceride quantification	Atractylodes powder (Samryeongbaekchul-san)
	assay & Leptin immunoassay.	decoctions, identified with HPLC, have an
		anti-adipogenic effect.
Shim et al. (2012)	ELISA	Detect aflatoxins in 70 Korean herbal medicines with
		ELISA and LC-MS to determine fungal contamination.
Simoes-Pires et al.	Antiplasmodial assay.	Discover active antimalarial alkaloids with NMR in
(2014)		Argemone mexicana L. Papaveraceae for quality
		markers of antimalarial teas.
Song et al. (2004)	Anti-platelet aggregation assay and	Compare the effects of extraction time, extraction
	oestrogenic effects measured by	volume and number of repeats of the extraction on the
	alkaline phosphatase activity with	anticoagulation and oestrogenic activity of Nourish
	ELISA and cell count using MTT and	Blood with Angelica Decoction (Dang Gui Bu Xue Tang)
	protein count of MG-63 cells whose	to determine the best extraction method.
	cell growth and alkaline phosphatase	
	activity is modulated by 17β-estradiol	
	(Lajeunesse, 1994).	
Ting, Chow & Tan	Microbial plate assays.	Determine the effect of boiling on microbial
(2013)		concentrations of four herbal decoctions. Heavy metals
		were also evaluated with AAS.
Viegas et al. (2023)	Fungal cultures from tea samples	Determine the extent of fungal contamination in tea with

	grown on agar plates.	species and mycotoxins. Fungal cultures and specific
		toxins determined by DNA fingerprinting and HPLC-MS.
Viraragavan et al.	Glucose uptake assay by liquid	Compare the effects of aspalathin, a component of
(2020)	scintillation in skeletal and liver cells,	green rooibos (<i>Aspalathus linearis</i> (Burm.f.) Dahlgren,
	intracellular lipid content by modified	Fabaceae) with known effects on metabolic disorders
	Oil-Red-O assay, intestinal	against the whole herb extract by measuring its effects
	permeability of aspalathin measured	of cellular glucose and lipid concentrations and its
	by introducing aspalathin to apical	intestinal permeability, to decide if it can be used as a
	compartment of caco-2 cells and	sole quality marker or if multiple compounds are
	measuring transport to basolateral	required.
	compartment with HPLC-UVS.	
Walkowiak-Bródka et al.	DPPH antioxidant assay.	Evaluate the antioxidant capacity of 19 products
(2022)		containing Morus alba L. 1753, Moraceae and correlate
		this with levels of 1-deoxynojirimycin, an α -glucosidase
		inhibitor, detected and quantified by IRS and HPLC-UV
Wan et al. (2022)	Rat common cold infection model	Determine the effectiveness of Cinnamon with Kudzu
	with analysis of blood and	Decoction (Gui Zhi Jia Ge Gen Tang) against a
	histopathological analysis of lung	"Wind-Cold" type of common cold and find the
	samples	constituents responsible for development of quality
		standards. The decoction was analysed with HPLC-MS

		for its constituents, and the rat blood serum levels for its
		metabolites, markers of immune response.
		Histopathological analysis of lung samples analysed for
		changes to lung tissue.
Wan et al. (2023) and	Rat observational study for toxicity	Develop toxicity quality markers for Gardenia and
Wan et al. (2024)	markers (inactivity, lethargy, isolation,	Magnolia Decoction (<i>Zhi Zi Hou Po Tang</i>) by correlating
	hair contamination, diarrhoea and	weight loss, biochemical indicators, cellular inflammation
	weight loss), biochemical markers of	and damage with potential toxic compounds found in rat
	rat serum for markers of liver and	serum, detected with HPLC-MS.
	kidney injury (aspartate	
	aminotransferase, alanine	
	aminotransferase, total bilirubin, urea	
	nitrogen, and creatinine and	
	histopathological studies.	
Wang et al. (2017)	DPPH antioxidant assay.	Quality assessment of Big Red Robe Tea (Da Hong Pao
		<i>Cha</i>) by measuring antioxidant capacity and comparing
		to constituents measured by HPLC.
Wang et al. (2020)	High-fat diet / streptozotocin-induced	Determine the mechanism of compounds, determined
	diabetic mice tested with: serum and	UPLC-MS, responsible for antidiabetic effect of Tonify
	urine assays,	the Kidney and Invigorate Blood Circulation Decoction

	hematoxylin and eosin (HE) staining,	(Bu Shen Huo Xue Tang) to define quality control
	periodic acid-Schiff staining,	markers.
	immunohistochemistry staining,	
	Western Blot analysis.	
Wang et al. (2022)	Immunochromatographic strip.	Detection of pain relieving and fever reducing drug
		acetophenetidin in herbal tea blends through coating a
		strip with antibodies that react to the presence of the
		drug.
Wang, Y. et al. (2023)	Bioluminescent bioautography.	Detect adulterant hypolipidemic drugs by using a
		bioluminescent bacterial solution as a derivatizing agent
		after HPTLC separation.
Wu et al. (2020)	DPPH antioxidant assay &	Compare two Polygonum chinense L.H.Gross,
	Xylene-induced mouse ear oedema	Polygonaceae varieties to find the best to use in "Cool
	anti-inflammatory model (<i>in vivo</i>).	Tea" (Liang Cha) for superior antioxidant and
		anti-inflammatory activity. Constituents measures with
		UPLC-MS.
Wu, Yang & Chiang	Total Anthocyanin Content (TAC),	Determine the effect of heat and pH on anthocyanins in
(2018)	TPC and DPPH, ABTS & FRAP	roselle (H. sabdariffa) decoctions to determine the best
	antioxidant assays.	methods of storage to preserve their antioxidant
		properties.

Xiao et al. (2022)	DPPH antioxidant assay.	Correlate antioxidant capacity of components with peaks
		obtained by HPLC in Three Whites Decoction (San Bai
		Tang).
Xie et al. (2022)	Immunochromatographic assay.	Detection of various sulfonylureas (anti-diabetic drugs)
		in adulterated multi-herbal teas using a method of
		combining chromatography with antibody coated strips.
Xu, N. et al. (2022)	DCFDA and ELISA.	Determine antioxidant and anti-inflammatory effect
		through C-Reactive Protein inhibition of individual
		components of Pinellia, Atractylodes and Gastrodia
		Decoction (Banxia Baizhu Tianma Tang), detected with
		UPLC, for use as quality control standards.
Xu, Y. et al. (2022)	In vitro smooth muscle contraction	Determine the antispasmodic effects components of
	assay.	Peony and Liquorice Decoction (Shaoyao Gancao
		<i>Tang</i>), determined by HPLC, by measuring their effects
		on acetylcholine-induced contraction.
Yadav, Gupta & Aeri	Mouse forced swimming test,	Evaluate antidepressant effect and antioxidant effects of
(2022)	thiobarbituric acid reactive	Ashwagandha (<i>Withania somnifera</i> (L.) Dunal,
	substances and GSH assay.	Solanaceae) based formulations, compared using
		HTPLC, to determine a standardised formulation.
Yang, F. et al. (2022)	α-glucosidase inhibition assay.	Assess the difference between raw and fried cassia

		seeds (<i>Senna obtusifolia</i> (L.) H.S.Irwin & Barneby, <i>Fabaceae</i>) for their anti-diabetic effect and identify the components responsible with UPLC-MS, to use as quality indicators.
Zhang et al. (2014)	Antiviral assay	Assess antiviral activity of components of <i>Radix isatidis</i> (<i>Isatis tinctoria</i> L., <i>Brassicaceae</i> or <i>I. indigotica</i> Fort.), identified with LC, for use as quality markers.
Zhang et al. (2024)	<i>Ex vivo</i> : Rat model of heart failure by ligation of the left anterior descending branch of coronary artery and assayed with HE Masson and Sirius red staining and ROS detection (method not stated). <i>In vitro</i> : cell viability measured with CCK-8 assay after 6h treatment followed by anoxia model (Na ₂ S ₂ O ₄ for 8h), protein analysis after anoxia model with western blotting	Determine the constituents, identified in decoctions and their metabolites in rat serum using UPLC-MS, and determine the mechanism responsible for the effect of Astragalus and Aconite Decoction (<i>Qi Fu Tang</i>) in heart failure to decide on quality markers.
Zhang, C. et al. (2021)	Bacteriostasis and α-glucosidase inhibition assays.	Compare <i>Coptidis Rhizoma</i> (<i>Coptis chinensis</i> Franch., <i>Ranunculaceae</i>) decoction pieces and granules to

		determine the superior antibacterial and antidiabetic effects in relation to constituents, quantified with HPLC.
Zhang, H. et al. (2020)	ELISA.	Discover compounds responsible for antidepressant effects in Lily Bulb and Rehmannia Decoction (<i>Bai He Di</i> <i>Huang Tang</i>), quantified with HPLC, to be used for quality control.
Zhang, M. et al. (2022)	Rat mifepristone induced abortion model, ELISA, reverse transcription polymerase chain reaction (RT-PCR) and western blotting.	Evaluate the best quality markers to use to evaluate Gelatin and Mugwort Decoction (<i>Jiao Ai Tang</i>). UPLC-MS identified main components which were tested with bioassays to determine: effect on uterine bleeding quantity and histopathological sections from rat study; effects on luteinizing hormone, follicular stimulating hormone, estradiol and progesterone using ELISA; gene and protein expression levels using RT-PCR and western blotting.
Zhang, X. et al. (2022)	Immunochromatographic strip.	Detection of morphine, codeine and thebaine from opium poppies (<i>Papaver somniferum</i> L., <i>Papaveraceae</i>) in herbal teas.
Zhao et al. (2022)	LPS-induced macrophage inflammation model, Griess assay for	Evaluate anti-inflammatory and antioxidant capacity of two species of yellow flowered tea (<i>Camellia nitidissima</i>

	NO production and DCFDA ROS	C.W.Chi, <i>Theaceae</i> , and <i>C. euphlebia</i> Merr. ex Sealy)
	assay.	and a common adulterant (C. insularis Orel & Curry) to
		compare against UPLC-MS determined marker
		compounds.
Zhao et al. (2024)	Mifepristone and miso-prostol rat	Identify the components of Four Substance Decoction
	model of "Blood Stasis," measured by	with Peach Kernel and Safflower (Tao Hong Si Wu
	volume of blood on intravaginal	Tang) with UPLC-MS and evaluate which are most
	cotton ball and assayed with HE	effective in a vaginal bleeding model to determine
	staining and ELISA to measure IL-6,	quality markers.
	TNF- α and IL- β in rat serum.	
Zheng et al. (2020)	FRAP and DPPH antioxidant assays.	Find antioxidant components, quantified with HPLC,
		responsible for the effects of Clear Epidemics and
		Overcome Toxins Decoction (Qing Wen Bai Du Tang) to
		use as markers for quality control.
Zhou et al. (2019)	CCK-8 cell viability assay	Assess the antiproliferative effect of Four Gentlemen
		Decoction (Si Jun Zi Tang) and its components,
		detected with UPLC, on PC9 lung cancer cells to
		determine which compounds should be used for quality
		control when given to lung cancer patients.

 Table 6: Biological Assays Used in Quality Assessment of Herbal Teas and their Purpose

One of the most notable points that emerge from the literature is that very few suggest the use of bioassays as a method of quality assessment in itself, but usually to find a biological mechanism of action behind certain compounds which can then be used for quality control with chromatographic or spectroscopic methods. Only one paper did not use a chemometric technique alongside the bioassays to identify compounds (Wu, Yang & Chiang, 2018) and their approach was less concerned with identifying marker compounds but assessing how processing methods affected the anthocyanin content and subsequent antioxidant activity to inform manufacturing processes. A spectroscopic technique was still used because anthocyanins were measured using a pH differential method that depends on anthocyanins reversibly changing colour under low pH which can be measured with a spectrophotometer (Lee, Durst & Wrolstad, 2005) but it does not identify the precise compound. Instead it compares this feature to the reference anthocyanin, cyanidin-3-glucoside. The overwhelming number of other studies that all attempted to find a specific active ingredient responsible for the effects observed in the bioassays reveals a distinct pharmaceutical bias in the fields of quality assessment of herbal teas, despite the fact that the isolation of active compounds frequently proves the isolated ingredients to be ineffective or less effective (eg. Lu, Wang & Kong, 2011), leading modern drug discovery to look at synergies between multiple compounds and complex effects using -omics analysis, network pharmacology and systems biology (Thomford et al., 2018; Panossian, 2023).

2.5. The Limitations of Existing Methods of Herbal Quality Control

The existing methods of quality control in herbal teas focuses heavily on chemometric testing for specific marker compounds or genetic sequences. When there is attention to biological activity it is largely to test novel compounds in order to determine which ones are biologically active and can potentially be developed into drugs. These often become the primary quality markers (eg. Qin et al., 2019),

followed by compounds unique to the species and then specific levels of others in relation to other species.

This reveals a distinct tendency towards the pharmaceuticalization of herbal medicine and presents a problem when dealing with complex mixtures and herbs with unknown mechanisms. This is demonstrated by Viraragavan et al. (2020) who compared the bioactive effects of green Rooibos tea (A. linearis) and aspalathin, the primary candidate for its bioactivity, on *in vitro* metabolic models, finding that it could not account for all effects and that up to 12 compounds showed metabolic bioactivity. When expanded to entire formulas, Fan et al. (2021), in attempting to develop a "holistic quality control" method using UPLC-MS/MS on a TCM formula (Yangwei Decoction, or "Nourish the Stomach" Decoction) consisting of 11 herbs, found 226 components, and in order to make a manageable shortlist focused on only 5 main quality markers from only 4 of the ingredients known to have an effect in gastric diseases, and still confessed to being unable to explain the effective components and pharmacological actions of the formula as a whole. Similarly, Feng et al. (2022) examined another TCM formula (Danggui Jianzhong Decoction, or "Construct the Middle with Angelica" Decoction), consisting of 6 herbs but found that all 7 peaks associated with quality were from just 2 herbs, Paeoniae Radix Alba (Paeonia lactiflora Pall., Paeoniaceae) and Glycyrrhizae Radix et Rhizoma (G. glabra), neither of which are in the chief herb named in the formula (A. sinensis) and both are common in many formulas, and even constitute a formula in themselves (Shao Yao Gan Cao Decoction, or "Peony and Liquorice" Decoction; Scheid et al., 2009, p. 344-345). In the most extreme case, Cheng et al. (2022) reduced a 5 herb formula (Qinghao Biejia Tang, or "Artemisia and Soft-shelled Turtle Decoction") to a single compound, mangiferin, deemed the main antibacterial constituent responsible for the formula's antibacterial action. This is despite fact that the use of artemisinin, another compound extracted from artemisia annua L., Asteraceae, as a monotherapy for malaria has contributed to the development of resistant Plasmodium falciparum parasites (Czechowski et al., 2020) and that the traditional method of combining *A. annua* with other herbs with different mechanisms of action that may be responsible for the formula's effects and the reason A. annua has remained effective for thousands of years.

Another instance where limitations in current methods are exposed is when the correct plant is used but has been incorrectly grown, handled or stored in a way that may reduce its efficacy. These may be best tested functionally, based on what they do to a complex biological system when tested as a whole blend rather than the sum of its parts.

2.6. Functional Adulteration and the Problem of Functional Testing

2.6.1. Functional Adulteration

Functional adulteration refers to forms of adulteration that affect the functioning of the herb or blend in question. This most often refers to adding drugs to herbal supplements in order to impart efficacy to an otherwise inactive blend (Choules et al., 2020), however another form of functional adulteration is to offer a poor quality sample of the correct herb coming from a poor growing region, herbs that have been harvested or handled incorrectly post-harvesting, or a blend that has been mixed incorrectly, as a high quality sample. The result is that the herb may be the correct species and so pass visual and genetic identification and contain a chemical profile that may not appear outside of the normal range of variation on chemometric analysis, but their clinical effects may be diminished. On occasion pre-extracted herbs can be dried again and resold by unscrupulous sellers if their appearance is unchanged, as in the case in weeping forsythia capsules (Forsythia suspensa, (Thunb.) Vahl, Oleaceae; Al-Khafaji, 2016), anise (Pimpinella anisum, L., Apiaceae; Ilyas, 1980) and cloves (Syzygium aromaticum (L.) Merr. & L.M.Perry, *Myrtaceae*; Centre for the Promotion of Imports from Developing Countries, 2018). While it is most common in these hard herbal materials that do not change much in appearance after extraction (figure 20), other herbs have been adulterated in this manner, with suspected spent material being present in herbal extracts detected through a complete but weak HPTLC fingerprint reported in milk thistle seed (Silybum marianum (L.) Gaertn., Asteraceae), ginkgo (G. biloba) leaf, echinacea (Echinacea angustifolia DC., E. pallida (Nutt.) Nutt., or E. purpurea (L.) Moench, Asteraceae) root or herb,

cranberry (*Vaccinium oxycoccos* L., *Ericaceae*) fruit and St. John's wort (*Hypericum perforatum* L., *Hypericaceae*) herb (Gafner et al., 2023). There has been also been one case of pre-extracted rhodiola root (*Rhodiola rosea* L., *Crassulaceae*) being resold as dried herbs by a major supplier for which suspicion was aroused by its lack of organoleptic properties and subsequent chemometric testing found it almost completely absent of active ingredients (Etheridge, 2023).



Figure 21: *Forsythia fructus* (left) and *Flos Syzygia* (right) Both are unchanged in outward appearance by extraction and so commonly adulterated with exhausted products (photos by Hazeena Azeez, 2023)

This latter form of functional adulteration has received little attention in modern literature since it remains elusive to most modern techniques: being the correct species, it will appear the same under visual and microscopic inspection, and containing the same genetic and chemical markers. Quantities of an active ingredient may be less, but this is difficult to determine if no single active ingredient is known to be responsible for its therapeutic effect. This becomes an even greater challenge in formulas, blends, extracts and powders where the number of potential markers is increased and where marker compounds may be altered and DNA may be degraded from processing (Parveen et al., 2016). This has resulted in a publication bias (DeVito & Goldacre, 2019) against reporting this form of adulteration despite being of great concern to historical authors.

In 1st - 2nd century Rome, Pliny the Elder documented the use of smoking to make new wine appear older (Bush, 2002) and in 5th century China Tao Hongjing raised the issue of herbs grown in an incorrect terroir, or handled inappropriately in harvesting or storage (Liu, 2021), resulting in the correct plant being used but the quality and potency being negatively affected. Pliny the Elder even exclaimed that "By Hercules! - a thing truly astounding - that, in reality, a wine is more innoxious in its effects, in proportion as it enjoys a less extended renown" (Pliny the Elder, Bostock & Riley, 1857). The adulteration of wine presents a comparable issue to herbal medicine due to the issue of terroir (Kim, 2023) and there have been attempts to address this problem through chemometric means but it has met many problems (Merkytė et al., 2020). Even Pliny the Elder's issue of detecting smoked wine can be difficult today due to the legitimate use of charred and toasted barrels to impart flavour to alcoholic beverages including wine (Archibald, 2020). This introduces a question of what is a legitimate level of foreign chemicals imparted from treatment or processing before adulteration can be suspected? Also, once an aged chemotype is discovered, the marker components can be artificially added or created. For example, trans-resveratrol has been shown to convert to cis-resveratrol during the ageing of wines (Naiker et al., 2020), but this conversion can be promoted by exposure to UV radiation (Pannu & Bhatnagar, 2019) or added from other plants that are abundant in resveratrol and grow much faster than grapes, such as Japanese Knotweed (Polygonum cuspidatum Sieb.et Zucc., Polygonaceae) that grows invasively throughout Europe (Peng et al., 2013). The cis- isomer is less valued for health supplementation than the trans- isomer, so trans-resveratrol could be extracted and sold for health supplements while the remaining cis-resveratrol could be added to wines to imitate the ageing process and raise their value. Similar adulteration has been reported with other herbs such as rutin from corn added to ginkgo (G. biloba) supplements (Booker et al., 2016) making this a realistic possibility. Although wine's primary bioactive compound is alcohol which disrupts mitochondrial function, increasing ROS production and reducing antioxidant defences (Hoek, Cahill & Pastorino, 2002) and will mask the more subtle changes induced by other components making comparison between varieties difficult, research into functional adulteration may be of use to the wine industry since the samples will have to be dried in order to add to a culture media which will remove the alcohol and allow the remaining compounds to be tested for their functional activities.

In Chinese herbal medicine, the importance of region has also long been recognised as an important factor affecting the potency. The oldest classics of Chinese herbal medicine, The Divine Farmer's Materia Medica (*Shen Nong Ben Cao Jing*), compiled 25–220 C.E., states that:

"Whether they are dried in the shade or in the sun, the time and month when they are gathered and prepared, whether they are used raw or cooked, the locale where they come from, and whether they are genuine or false, and old or fresh, these also have their precedents" (Wilms 2017, p.23).

Even in this ancient text, the correct species is only one condition of several for ensuring the potency of herbs. Even after the attempts of Tao Honging to improve on functional quality evaluation with his commentaries, successive authors also raised the issue, providing the best growing regions along with other clinical information in their pharmacopoeias (Liu, X. et al., 2020). It was eventually given the term Daodi (道地, literally "the way of the earth) in the Ming dynasty (1368 - 1644 CE) and referred not only to different soil associated with specific regions but the weather conditions, hours of sunlight and the experience of the local people in cultivation, harvesting, processing and storage which lead to a superior product. Clinical and animal trials have found a superior effect to Daodi herbs compared to non-Daodi herbs in 83.3% of published animal trials and in one clinical trial on humans (Yang, X. et al., 2018). However, the problem which faced Tao Hongjing has resurfaced many times since. In one publicised example, the demand for ginseng (Panax ginseng, C.A.Mey., Araliaceae) has resulted in exhaustion from overharvesting in its native northern regions several times. During the Great Leap Forward, attempts were made to cultivate it on the southern island of Hainan but resulted in roots that were large but almost completely lacking in active ingredients (Zhao, Guo & Brand, 2012).

2.6.2. Adulteration and Substitution

To complicate matters further, when preparing formulas or blends based on a traditional prescription, it is an accepted practice to substitute herbs which are problematic due to scarcity, legality, cost or toxicity with another that has similar properties and actions (Kashari, 2021; Leon & Lin, 2017; Moorhouse et al., 2022). This can mean that there may be several potential variations of a traditional formula which are correct, each containing different ingredients but aimed at having a similar effect to the original.

For example, in the U.K., all aconite species (Aconitum spp., L., Ranunculaceae) are banned from internal consumption (Medicines & Healthcare products Regulatory Agency, 2014) but in CHM this has been an essential herb since the earliest herbal formulary was written in the Han dynasty (Zhang & Liu, 2016). Prepared aconite root, known as Fu Zi (附子, Radix Aconiti Lateralis) Preparata from the species Acontium carmichaelii, Debeaux, Ranunculaceae) is understood to powerfully invigorate Yang (the warm, energetic, virile aspects of the body under the Yin-Yang dualistic understanding of TCM), having indications such as weak pulse with cold limbs, shortness of breath and chest pain, or joint pain that is worse in cold, damp conditions, or impotence and infertility (Bensky, 2004, pp.673-4). While China has developed processing methods to alleviate the toxicity posed by the diterpene alkaloids present in this plant, to the point where it is even sometimes used as a vegetable in soups, and all medical products containing aconite are subjected to HPLC analysis to ensure safe levels are met (Singhuber et al., 2009), the U.K. has adopted a complete ban. This means that CHM practitioners in the U.K. have to find substitutions for this herb and the various products on the market have adopted different solutions for this issue.

The classic aconite containing formula *Jin Gui Shen Qi Wan* (Kidney Qi Pill from the *Essentials from the Golden Cabinet*) utilises aconite for all of these potential functions and in modern use different versions of this formula have opted for different substitutions. The indications of fatigue, weak pulse and cold limbs suggest imminent heart failure are referred to hospitals in the U.K. but the formula

is still commonly used for the remaining indications of cold-type joint pain and impotence or infertility. Hence, Aura Herbs (2023) have chosen to increase the amount of cassia bark (*Cinnamomum cassia*, (L.) J.Presl, *Lauraceae*) already present in the formula for its similarity to aconite in terms of its hot taste and warming nature with similar indications for alleviating joint pain. Meanwhile, GinSen (n.d.) have chosen to add fo-ti (*Polygonum multiflorum*, Thunb., *Polygonaceae*), goji berries (*Lycium barbarum* L. *seu chinense* Mill., *Solanaceae*) and schisandra berries (*Schisandra chinensis* (Turcz.) Baill., *Schisandraceae*) which have few warming effects and offer little assistance with joint pain but instead share aconite's indication for fertility issues which their clinic specialises in.

These will obviously have vastly different chemical profiles from both each other and from the formulas produced in China but all are acceptable variations of the same formula aimed at treating similar patterns of disease. The ideal method to assess whether they are equivalent or should be considered functionally distinct formulas would be to compare their effects on a living system to find out if they actually achieve the same desired effect.

2.6.3. Functional Testing

The methods for functionally evaluating the quality of herbs have not evolved much since 2nd century Rome when Galen proposed testing the effectiveness of antivenom purchased at the markets by separating two groups of chickens and giving them an equal dose of poison, then administering the antivenom to the active arm and comparing the death rates to the control group (Karaberopoulos, Karamanou & Androutsos, 2012). A similar method is described in 11th century Chinese literature, using human volunteers to test the quality of ginseng (*Panax ginseng* C.A.Mey., *Araliaceae*) by pitting two runners against each other, one who has taken the ginseng while the other did not, and comparing their breathing after a race (Shang & Gong, 1994).

This method of observing an effect on a living organism may be more sensitive than chemical analysis for detecting the functional potency of herbs. A

review of studies comparing Daodi to non-Daodi herbs found when chemical analysis was used to assess guality by equating higher levels of active ingredients to efficacy, 64.5% found *Daodi* herbs to be superior. However, when using animal models to assess direct effects on living systems, 83.3% of papers found Daodi herbs to be more effective (Yang, X. et al., 2018). Animal studies are ethically problematic (Ferdowsian & Gluck, 2015) and provide limited information (Bracken, 2009) making them unsuitable for routine investigation into the functional potency of herbal products. Meanwhile, formal human testing through clinical trials is even less ethically appropriate for routine examination of herbal supplies, while also being too costly and time consuming to be practical for batch testing. Unofficially though, retrospectively assessing the efficacy of each batch is often the only method herbal clinicians have to evaluate the quality of their supplies. This means that they are using herbs of unknown potency every time they purchase a new batch, potentially putting patients at risk of adverse effects through a sudden increase in potency, or reduced therapeutic efficacy from reduced potency. A method of functionally evaluating herbs that does not put human or animal lives at risk from poor quality herbal medicines would clearly be advantageous to the herbal industry, clinicians and patients.

Chapter III

3. Mitochondria as a Potential Focal Point for Functional Testing

Mitochondrial analysis presents a unique opportunity to solve the problem of functional testing. Mitochondria are integral to most biological functions, providing the energy necessary to perform most tasks as well as generating important signalling mechanisms both within (Quirós, Mottis & Auwerx, 2016) and between cells (Liu & Ho, 2018; Picard et al., 2018) through DNA, redox, hormonal, neurological and immune mediated pathways. Their functioning is altered in both normal physiological states and disease (Annesley & Fisher, 2019; Nunnari & Suomalainen, 2012) which can be measured and used to assess the functional effects of a herb or blend on a complex living system.

3.1. The Role of Mitochondria in Health and Disease

3.1.1. Respiration

Mitochondria were first identified as the site of respiration in 1949 (Kennedy & Lehninger, 1949) who observed fatty acid oxidation occurring in isolated rat liver mitochondria. This process had already been described by Hans Krebs in the 1930s who developed the model which now bears his name, also known as the citric acid cycle or tricarboxylic acid cycle, after the compound created from acetyl coenzyme A in its first step (figure 22). In this model, the pyruvate generated from glycolysis enters the mitochondria and is converted to acetyl coenzyme A, which then acts to renew a cyclical series of reactions by converting oxaloacetate into citric acid with the

addition of two carbon atoms (C), four hydrogen atoms (H) and two oxygen atom (O) units (Alberts et al., 2002). As the cycle progresses, carbon and O units are lost giving off carbon dioxide (CO_2), 2 units of adenosine triphosphate (ATP) are generated which can be used as energy, and H is used to convert nicotinamide adenine dinucleotide (NAD)+ to NAD + H (NADH) and flavin adenine dinucleotide (FAD) to FAD + H (FADH₂). These activated carrier molecules provide high energy electrons that can enter the electron transport chain (ETC) to generate further ATP through oxidative phosphorylation (OXPHOS) (Cooper, 2000).





The process of OXPHOS to generate chemical energy was then completely described by Mitchell in 1961. In this model a series of proton pumps located in the inner mitochondrial membrane (IMM) utilise the energy from the electrons donated by NADH and FADH₂ to pump protons across

the IMM to the intermembrane space creating a chemiosmotic gradient which drives ATP synthase to generate ATP from adenosine diphosphate (ADP) and phosphate (P) (figure 23). Molecular oxygen (O_2) acts as the final electron acceptor in this chain resulting in the creation of water (H_2O).



Figure 23: Oxidative Phosphorylation (created with BioRender.com)

The advantage of OXPHOS is that it is much more efficient than glycolysis, producing approximately 30 ATP from a single molecule of glucose, compared to 2 ATP generated by glycolysis or fatty acid oxidation alone with its end product being water (Alberts et al. 2002). The disadvantage is that it is relatively slow compared to glycolysis (Zheng, 2012) so the glycolytic pathway is utilised when energy needs exceed that which the mitochondria can provide, either due to excessive demand or lack of O_2 . When mitochondria are unable to use the pyruvate generated from glycolysis, it is fermented using NADH to become lactate. This was traditionally seen as a waste product which is excreted from the cells but emerging evidence suggests this may actually be an alternative fuel source

which can then be taken back up by cells which reverse the reaction to generate NADH for OXPHOS (Rabinowitz & Enerbäck, 2021).

3.1.2. Reactive Oxygen Species

Reactive Oxygen Species (ROS) are a by-product of aerobic respiration. In an ideal situation, electrons transported through the ETC combine with O_2 and 2 hydrogen ions (H⁺) to form H_2O (Figure 23) but under normal physiological conditions, an estimated 0.2-2% electrons leak from up to eleven potential sites in the mitochondria to combine with O_2 to form $O_2^{\bullet-}$ (Zhao et al., 2019; Zhang, B. et al., 2022). In order to prevent this reactive molecule from damaging the mitochondria, it is converted by superoxide dismutase (SOD) into hydrogen peroxide (H_2O_2) which can diffuse out of the mitochondria and act as a redox signalling molecule (Zhao et al. 2019). H_2O_2 is then broken down further by catalase (CAT) or glutathione peroxidase (GPx) to become H_2O and O_2 again (figure 24). GPx can then subsequently be restored through the action of glutathione (GSH) and glutathione reductase (GR). Peroxiredoxins (PRX) and thioredoxin (TRX) are other enzymes that can catalyse the reaction of H_2O_2 to H_2O in a similar fashion. If the amount of ROS exceeds that which the cell can dismutate, it may react to form several other highly reactive species (Phaniendra, Jestadi & Periyasamy, 2015). If NO reacts with $O_2^{\bullet-}$ then Reactive Nitrogen Species (RNS) may be formed, while excessive H_2O_2 can transform into hydroxyl radicals (•OH) through the Fenton reaction with ferrous ions (Fe²⁺), or the Haber-Weiss reaction with $O_2^{\bullet-}$.



Figure 24: Sources of ROS and antioxidant defence mechanisms (created with BioRender.com).

Abbreviations: SOD: superoxide dismutase; CAT: catalase; GPx: Glutathione Peroxidase; GR: Glutathione Reductase; GSH: Glutathione; GSSG: Glutathione disulphide, oxidised GSH

Although mitochondria are some of the main sources of intracellular ROS, other sources such as NADPH oxidases (NOX), xanthine oxidase and other oxidative enzymes present in the cell are also important contributors, possibly more so than mitochondria (Zhang & Wong, 2021). Endoplasmic reticulum (ER) stress can also stimulate the expression of NOXs and the pathological accumulation of misfolded proteins such as β -amyloids and oligomeric α -synuclein can also produce ROS independent of mitochondria,
while tau and huntingtin can interfere with the ETC to cause the mitochondria to produce more ROS (Abramov, et al., 2020). In addition, exogenous sources such as radiation, pollution, heavy metals, pesticides and drugs can be important contributors of ROS (Phaniendra, Jestadi & Periyasamy, 2015). Regardless of the sources of ROS, mitochondria still play an important role in acting as ROS sinks due to the high level of antioxidative enzymes they contain to manage their own ROS production.

Excessive levels of ROS result in oxidative stress that have been associated with poor health outcomes. This is due to the cumulative damage that it causes over many years and has been posited as a potential mechanism for ageing since Harman first presented the "free-radical theory of ageing" in 1956. More recent versions of this theory suggest that ROS from mitochondrial dysfunction can damage telomeres which, in turn, inhibits the biogenesis of mitochondria leading to a cycle of disruption and senescence (Gao et al., 2022).

ROS are not exclusively associated with cellular damage and degenerative disease but have recently been implicated in a number of important signalling processes that are essential for cellular health (Zhang et al., 2016). ROS influences the activation of nuclear factor kappa-light-chain-enhancer of activated B cells (NF-kB) through both canonical and non-canonical pathways, the extracellular signal-regulated kinase (ERK) pathways of the mitogen-activated protein kinase (MAPK), it promotes the dissociation of nuclear factor erythroid 2-related factor 2 (Nrf2) and kelch-like ECH-associated protein 1 (Keap1), activates phosphoinositide-3-kinase while inhibiting phosphatase and tensin homolog that negatively regulates protein kinase B, has a bidirectional relationship with calcium ion (Ca²⁺) signalling, acts upon the mitochondrial permeability transition pore (mPTP) both directly and in conjunction with Ca²⁺, interacts with protein kinases and modulates the ubiquitination/proteasome system.

These pathways are involved in many cellular processes including immune, inflammatory, proliferation, differentiation, autophagy, senescence, apoptosis and stress responses.

Some of these reactions are of particular importance in the redox homeostasis of ROS. The first of these is Nrf2 which is bound to Keap1 under normal conditions, targeting it for ubiquitination. In the presence of ROS, Keap1 dissociates from Nrf2, enabling its activation which leads to enhanced transcription of multiple antioxidant enzymes (Baird & Yamamoto, 2020), creating a negative feedback loop assisting the cell to maintain optimal levels of ROS. Exercise generated ROS also activates Peroxisome Proliferator Activated Receptor-gamma (PPAR γ) which can sensitise tissues to insulin, and Adenosine monophosphate (AMP)-activated protein kinase (AMPK) which acts as a gatekeeper to the master regulator of mitochondrial biogenesis, PPAR γ Coactivator-1 α (PGC-1 α), leading to increased numbers of mitochondria and increased mitochondrial fusion (Webb et al., 2017; Cantó & Auwerx, 2009).

It has also been suggested that, since mitochondria share a common ancestry with bacteria, that they may possess signalling pathways to communicate with our microbiome. Ballard & Towarnicki (2020) have proposed that ROS may act as a bi-directional signalling molecule, with gut dysbiosis being correlated with chronic inflammation marked by excessive levels of ROS and lower levels, within the stimulatory cell signalling range, being correlated with greater diversity of the microbiome, maintaining intestinal barrier integrity and appropriate mucosal immune responses. Additionally, many toxins released by pathogenic gut microbiota are able to affect mitochondria directly and can even cross the blood-brain barrier to affect the mitochondria of neuronal cells in the brain providing a chemical link for the postulated gut-brain axis (Zhu et al. 2022). Exercise has also been shown to have positive effects on both mitochondria and gut microbial health through shared signalling pathways (Clark & Mach, 2017).

3.1.3. Immune Responses

regulate the multiple Mitochondria immune system through mechanisms. In response to stress, mitochondria release a number of Damage Associated Molecular Patterns, including cardiolipin, n-formyl peptides, ROS and mitochondrial DNA (mtDNA), into the cytosol or extracellularly which are detected by pathogen recognition receptors that trigger the release of cytokines which recruit and activate immune cells and regulate responses to the insult (Banoth & Cassel. 2018). In addition, pro-inflammatory (M1) macrophages utilise a "broken" TCA cycle by downregulating isocitrate dehydrogenase, inhibiting the conversion of citrate into α -ketoglutarate resulting in citrate accumulation and its conversion into itaconate. In addition to some direct antimicrobial effects, itaconate also inhibits succinate dehydrogenase, complex II of the ETC, resulting in increased ROS formation that, combined with additional ROS generated from NOX2, inactivates phagocytosed microbes and stabilises Hypoxia Inducible Factor 1α (HIF- 1α) leading to upregulation of pro-inflammatory cytokine IL-1ß (Herb & Schramm, 2021). Mitochondria are also involved in antiviral defences through the activation of Mitochondrial Antiviral Signalling proteins located on the outer mitochondrial membrane (OMM) (Mills, Kelly & O'Neill, 2017). Viral ribonucleic acid within the cell activates this system leading to activation of the Nucleotide-binding oligomerization domain, Leucine rich Repeat and Pyrin domain containing (NLRP) 3 and NF-kB inflammasome responses resulting in pyroptosis of the infected cell.

Besides driving inflammatory responses, mitochondria have also been suggested as the drivers behind stopping the inflammatory process and enabling the body to begin healing. Naviaux (2023) proposes that the inflammatory response is only part of the healing cycle and, to fully recover, the organism must also undergo phases of proliferation to regrow the damaged tissue and differentiation in order to ensure the cells are fit for purpose. Changing levels of extracellular ATP (eATP) from different mitochondrial phenotypes drive the transition between each of these phases. Inability to successfully transition from one stage to the next results in the organism becoming stuck in a loop involving one or more of the phases of the healing cycle, driven by abnormal eATP signalling (figure 25). This results in chronic inflammation, disorders of excessive proliferation or incorrect differentiation, or a combination of these, which account for many health conditions prevalent today. Unfortunately there is little mention of what therapies might be able to facilitate this healing response, only that they should work in tandem with behavioural and device-based methods to shift the whole body from one phase to the next, meaning that there is little concrete evidence for this model's therapeutic efficacy. However, it may help to explain how some traditional healing methods that appear to be triggering fresh trauma could operate (eg. vigorous forms of manipulation, acupuncture, cupping or bloodletting), by restarting the healing cycle with an artificial stressor and allowing an opportunity for it to continue to completion.



Figure 25: Simplified Diagram of the Healing Cycle, *Salugenesis* (created with BioRender; adapted from Naviaux, 2023).

3.1.4. Apoptosis

Besides being central to cellular life, mitochondria are also pivotal to initiating apoptosis, the process of programmed cell death which ensures that defective cells are destroyed in a way that does not harm the organism (Bock & Tait, 2020). Failure to initiate apoptosis can result in cancer or autoimmunity whereas excessive cell death results in degeneration.

There are two pathways to initiate apoptosis, both of which involve mitochondria (figure 19). The intrinsic pathway, also known as the mitochondrial pathway, is initiated when cellular stresses exceed the mitochondria's threshold triggering members of the B cell lymphoma 2 (Bcl-2) protein family activate the apoptosis regulator proteins Bcl-2

associated X (Bax) and Bcl-2 antagonist killer (Bak) which cause the mPTP on the outer mitochondrial membrane to open and release cytochrome c from the ETC into the cytoplasm. Cytochrome c then binds with apoptotic peptidase activating factor 1 to form a complex called the apoptosome. This binds to caspase 9 which activates the executioner caspases 3 and 7 starting an irreversible process towards cell death. The various stimuli which can initiate this process include withdrawal of negative feedback such as growth factors that lead to failure of suppression of cell death programs, or positive stimulation from radiation, toxins, hypoxia, hyperthermia, viral infections or ROS (Elmore, 2007). Mitochondria are also involved in the extrinsic apoptotic pathway through the activation of Bax/Bak although it also directly activates the executioner caspases through the activity of caspases 8 and 10.





3.1.5. Hormesis and Adaptation to Stress

Besides prompting the cell to die in the presence of excessive stress, mitochondria also facilitate adaptive responses to stress. The traditional understanding of oxidative stress runs through a simple scale of low levels being healthy as the cell can easily maintain homeostasis, through to high levels causing apoptosis as described in section 3.1.4 above. A middle level is also proposed where the conditions required for cancerous mutations and tumorigenesis are met, where ROS is enough to cause damage to DNA while remaining below the apoptotic threshold (Moloney & Cotter, 2018) (figure 27).





(created with BioRender.com, adapted from Moloney & Cotter, 2018).

In contrast, the hormetic model proposes that small amounts of a potentially harmful stimulus promote cellular adaptations that lead to greater resilience against future exposure, explaining the health benefits seen in activities such as exercise and fasting (Pinches et al., 2023). Too little oxidative stimuli is also becoming recognised to cause an impaired state of cellular functioning, known as reductive stress (Xiao & Loscalzo, 2020) (figure 28).





Hormesis has a long observational history, with Mithrades (135 - 63 BC) being reported to have taken low doses of poison so that he could survive a larger dose if administered by conspirators, through to the observation of Paracelsus in the 16th century that: "All things are poison and nothing is without poison; only the dose makes a thing not a poison"

(Stringer & Rossiter, 2017). In the 19th century, this biphasic dose-response was coined the "Arndt-Schultz law" and suggested to be the mechanism behind homoeopathy until homoeopathy fell out of favour with the medical establishment (Calabrese et al. 2013). It was then forgotten until 1943, when Southam and Erlich noted that small doses of antifungal red cedar extracts promoted fungal growth at low doses and coined the term "hormesis". Since then it has been proposed as a hypothesis which states that a low dose of toxic stimulation applied to a living organism can increase growth and respiration while a high dose of the same stimuli can have an inhibitory effect (Calabrese & Baldwin, 1998). Two main mechanisms have been proposed for how hormesis may work: overcompensation and direct stimulation (Calabrese et al., 2016). Overcompensation happens when an organism experiences a disruption in homeostasis and responds with an overcompensated response that ultimately leads to greater resistance to the stressor in future. Direct stimulation refers to a situation where the same drug may interact with different receptors or pathways depending on its dose. Three potential mechanisms have been proposed for this (Calabrese, 2013): either the same receptors are stimulated or inhibited by the same substance at different doses, or different receptors mediate the response to the same substance at different doses, or a low dose may activate a receptor or signalling pathway, but a high dose does not.

When applied to mitochondria it is termed mitohormesis and states that mild mitochondrial stress activates a cascade of signalling pathways that render the cell less susceptible to future perturbations (Yun & Finkel, 2014). The most relevant of these signalling pathways are ROS which are generated from any number of situations in the mitochondria including increased energy demand from exercise (Merry & Ristow, 2016), or reduced supply from calorie restriction (Schulz et al., 2007), or an increase or perturbations in O₂ supply (Burtscher et al., 2022). Exercise places excess demand upon the mitochondria, beyond their normal capacity, while fasting or hypoxia affects the supply of substrates necessary for efficient functioning of the ETC and results in an increase in ROS. Hyperoxia also results in increased ROS because there is extra O_2 present which bonds to free electrons, creating more ROS. Although the pathways for each are different, all of these stressors function as a signal to initiate improved mitochondrial function through the upregulation of antioxidant defences, increased fission and mitophagy to remove defective mitochondria, mitochondrial biogenesis to increase mitochondrial numbers, and the fusion of healthy mitochondria into a network that can share resources and improve the functioning of existing mitochondria.

The ER is an important piece of cellular infrastructure that allows the cell to communicate stress to the mitochondria which can then respond to the increased energy demand (Rossi, Pizzo & Filadi, 2019). In times of stress, Ca²⁺ reserves that are stored in the ER are released through inositol 1,4,5-trisphosphate receptors into the cytosol near to the mitochondria that are bound in close proximity to the ER with mitochondria associated membranes (Marchi et al., 2018) (figure 29). This enables them to be exposed to much higher levels of Ca²⁺ than are present in the general cytosol, and which are taken up through voltage dependent anion channels into the intermembrane space, to be transported through a specialised mitochondrial calcium uniporter (MCU) into the mitochondrial matrix.



Figure 29: Mitochondria, the Endoplasmic Reticulum and Ca²⁺ (Marchi et al., 2018; reproduced with permission) Abbreviations: MAMs: Mitochondria Associated Membranes; MCU: Mitochondrial Calcium Uniporter; SERCA: Sarco-Endoplasmic Reticulum Calcium ATPase; VDAC: Voltage Dependent Anion Channels;

Moderate levels enable mitochondrial adaptation to the cell's metabolic needs by boosting ATP production through the activation of several enzymes in the citric acid cycle, while high levels sensitise the mitochondria to the opening of mPTP, initiating apoptosis (Bravo-Sagua et al., 2017). Several diseases involve this calcium dependent apoptotic mechanism including ischaemia-reperfusion injury (IRI), cardiomyopathy, liver and muscle diseases, cancer and neurodegenerative disorders (Finkel et al., 2015; Britti et al., 2018; Romero-Garcia & Prado-Garcia, 2019).

3.1.6. Ageing and Degeneration

The damaging effects of ROS on cellular constituents and connective tissues led to the "Free Radical Theory of Ageing" being proposed in the 1950s (Harman, 1956). Under this theory it is the gradual, cumulative damage to DNA, lipids, proteins and polysaccharides caused by ROS which leads to ageing and the various disorders that increase with age, especially in tissues with a high energy demand such as neurons and cardiovascular cells, and those with a link to damaged to DNA, such as cancer.

3.1.6.1. Neurodegeneration

Neurons have a high energy demand and an unusual shape, sometimes as much as 10, 000 times the size of an average cell, with a strict aerobic metabolism (Vergara et al., 2019), which has led them to develop unique methods to achieve mitochondrial homeostasis. Neurons have to last a lifetime, and require high amounts of energy to maintain resting potentials, generate action potentials and achieve synaptic neurotransmission, with these functions often needing to be maintained a long distance from the soma of the cell. Most of this energy is derived from mitochondrial ATP so mitochondria must be continually renewed and consumed. Canonically, this has been thought to take place in the soma but emerging evidence suggests that both biosynthesis and mitophagy may happen locally, and mitochondria may even be imported from adjacent cells (Misgeld & Schwartz, 2017). Regardless of their origin, they are docked at high energy demand areas of the cells through a variety of tightly regulated docking mechanisms in the cytoskeleton (Cardanho-Ramos, Faria-Pereira & Morais, 2017). This high energy demand will result in the generation of high levels of ROS, which must be managed effectively. A certain level is necessary for optimal function (Cobley, Fiorello & Bailey, 2018) but excessive levels lead to

dysfunction, apoptosis and neurodegenerative disorders (Abdelhamid & Nagano, 2023).

Closely related to neurological health is sleep. Loss of sleep with age appears to have a bi-directional relationship with neurodegenerative disorders (Mander, Winer & Walker, 2017), resulting in elevated levels of oxidative stress (Singh et al. 2019). Reimund's (1994) "Free Radical Flux Theory of Sleep" attempts to explain the fundamental reason for sleep. While many theories of sleep consider it useful for cognitive functions, such as consolidation and organisation of memory (Cousins & Fernández, 2019), Reimund was interested in why animals, whose great advantage over plants are that they can move in order to seek food and evade predators, spend almost a third of their life unconscious and motionless. His hypothesis was that the high energy demand of the brain, which consumes around 20% of the total O₂ supply of the body, results in high levels of ROS. This is managed by periodically decreasing the cerebral metabolic load through sleep while ROS are removed. Melatonin, which is an important regulator of nocturnal physiology, whose synthesis is inhibited by daylight (Zisapel, 2018), is also a potent antioxidant found in high concentrations within mitochondria (Rieter et al., 2018) and protects the ETC from neuronal nitric oxide synthases which cause excessive levels of highly damaging peroxynitrites (see section 3.1.2), improving mitochondrial antioxidant defences and preventing apoptosis by inhibiting the opening of the mPTP (Hardeland, 2017). Sleep also results in a short glucose fast for the brain mitochondria, providing a hormesis effect while they must rely on ketone metabolism, and allows the clearing of accumulated β -amyloids that are generated during wakefulness (Kramer & Bressan, 2018). Sleep deprivation studies in animals and autopsies of humans with familial fatal insomnia reveal signs of mitochondrial dysfunction from oxidative stress with decreased cristae density and alteration in morphology leading to increased apoptosis (Melhuish Beaupre et al., 2021) (figure 30).





(created with BioRender.com, adapted from Melhuish Beaupre et al. 2021).

3.1.6.2. Cardiovascular Disease

The cardiovascular system also has a very high energy demand and is subject to degenerative changes with age. Cardiomyocytes are some of the most mitochondria rich cells in the body, consisting of ~30-40% mitochondria (Murphy & Liu, 2023). In cardiovascular disease there is often altered mitochondrial morphology, reduced ATP production with excessive ROS production, increased apoptosis and deregulated autophagy (Chistiakov et al., 2019). In atherosclerosis, the oxidation of low density lipoproteins promotes atherogenesis by increasing leukocyte adhesion, inducing endothelial dysfunction and apoptosis, activating platelets and

increasing plaque instability which may trigger a cardiovascular event (Kattoor et al., 2017). The high reliance of cardiomyocytes on OXPHOS makes them extremely vulnerable to hypoxia and subsequent IRI during ischaemic heart disease (Ramachandra et al., 2020). In IRI, the initial ischaemia leads to hypoxia which forces the cell to rely on glycolysis, reducing the ATP required for effective contraction, while resulting in the production of lactate and accumulation of protons, which increases acidity. This acidification inhibits mPTP opening, disrupting its role in maintaining ion homeostasis causing the cell to overload with sodium ions (Na⁺) and Ca^{2+} . Upon reperfusion, the influx of O_2 and nutrients causes the mitochondria to reinitiate OXPHOS, resulting in a burst of ROS. The MCU opens absorbing the high levels of Ca2+ that have accumulated during ischaemia, overloading the mitochondria with Ca²⁺ leading to more ROS. The normalisation of pH disinhibits the mPTP, which is opened by the high levels of Ca²⁺ and ROS leading to collapse of the mitochondrial membrane potential ($\Delta \Psi m$), inducing rigour hypercontracture and cell death (figure 31).





Other forms of heart disease that involve mitochondrial dysfunction include cardiac hypertrophy and chronic heart failure. The precise mechanisms vary depending on the types and causes of heart failure and other comorbidities but they always involve a metabolic shift away from OXPHOS and a change in the substrates used by the mitochondria which leads to alterations in ROS, Ca²⁺, morphology and biogenesis, resulting in an energy deficit that is compensated for with glycolysis that is unable to restore normal cardiac function due to its relative inefficiency (Lopaschuk et al., 2021).

3.1.6.3. Cancer

Another disease that becomes more common with age is cancer. Alterations in the metabolism of cancer cells to favour glycolysis, even in the presence of O_2 (aerobic glycolysis) were observed by Otto Warburg in 1927 who noticed that O_2 did not suppress glycolysis in tumour cultures (Otto, 2016) and reached acceptance with his famous paper in 1956 (Warburg, 1956), subsequently becoming known as the "Warburg Effect". Since then, many more hallmarks of cancer that differentiate them from healthy cells have been identified with the Warburg Effect coming under a broader category of deregulated metabolism (Hanahan, 2022; Hanahan & Weinberg, 2000) (figure 32).





Despite this early recognition of metabolic differences in cancer cells, the predominant theory on its origins has been that cancer is a disease triggered by genetic mutation, but with many of these hallmark processes being controlled by mitochondria, including the generation of ROS which causes genetic mutation, the resistance of apoptosis and deregulated metabolism, and the role of mitochondria in inflammation and immune function, cancer is rapidly being reframed as primarily a disorder of mitochondrial dysfunction (Seyfried & Chinopoulos, 2021).

3.1.7. Drug Resistance

As organelles that enable the cell to adapt to stress, mitochondria have also been identified as playing a potential role in drug resistance. The hormetic mechanisms of biogenesis and mitophagy, fusion and fission, which have been outlined in section 3.1.5. above, have also been found to be involved in chemoresistance (Jin, P. et al., 2022) and chemoresistant cancer cells have been found to have altered metabolism with a greater reliance on mitochondrial ATP (Bokil & Sancho, 2019). Sublethal release of cytochrome c combined with sublethal mPTP opening have been identified as key mechanisms that create drug-tolerant persister cells (Kalkavan et al., 2022). Chemoresistant cell phenotypes have also been identified to have reduced MAMs connecting to the mitochondria to the ER (Coku et al., 2022), described in 1.5.1.5. making them less sensitive to Ca^{2+} and less likely to initiate apoptosis when confronted with the stress of chemotherapeutics.

Another mechanism of drug resistance that may be mitochondrial dependent is through the expression of efflux pumps. One of the largest families of these transporter proteins are the ATP-Binding Cassette (ABC) transporters which use mitochondrial generated ATP as their main source of energy (Giddings et al., 2021). Two of the most significant, Breast Cancer Resistance Protein (BCRP) and Multidrug Resistance-associated Protein (MRP) 1, have been observed to be localised to mitochondria in doxorubicin resistant breast cancer cells, and the accumulation of doxorubicin is decreased in proportion to mitochondrial ATP production, an effect that can be partially reversed by oligomycin, a mitochondrial ATP synthase inhibitor (Dartier et al., 2017).

Cis-diamminedichloroplatinum(II) (cisplatin) is another chemotherapy agent widely used in the treatment of several solid tumours but which also has encountered problems of chemoresistance (Cocetta, Ragazzi & Montopoli, 2020). Cisplatin acts through the induction of DNA lesions, with a greater affinity for mtDNA than nuclear DNA (Oliveira et al., 1995), inducing oxidative stress and apoptosis primarily via the intrinsic pathway, disrupting DNA repair and inducing cell cycle arrest (Ghosh, 2019). Common side effects include nephrotoxicity, hepatotoxicity and vomiting through the disruption of redox homeostasis and apoptosis in healthy cells, evidenced by the reduction of these side-effects by natural antioxidants (Zhou et al., 2022; Abd Rashid et al., 2021; Alam et al., 2017). Antioxidants such as melatonin (Baghal-Sadriforoush et al., 2022) and guercetin (Hasan et al., 2023) which have demonstrated conditional pro-oxidative effects on cancer cells, discussed further in section 3.2.5.1, increase the apoptotic effect of cisplatin, also indicating a redox pathway. These suggest that the problem of cisplatin chemoresistance is also one of mitochondrial adaptation and this has been observed in cisplatin resistant ovarian cancer cells which undergo a shift to enhanced oxidative metabolism, a reorganisation of the mitochondrial network and an increased mitochondrial turnover rate (Zampieri et al., 2020).

3.2. The Role of Mitochondria in Traditional Medical Systems

People seek traditional, complementary and alternative medicine for a variety of conditions for which mitochondria may hold a central role. Lists of conditions treated are difficult to obtain due to the Advertising Standards Agency requiring that medical claims be supported by strong and robust evidence (Advertising Standards Agency, 2021) which is often lacking in complementary and alternative therapies (Ernst, 2011). However, the Professional Register of TCM based in Ireland list the common complaints as described by patients frequently seen within TCM clinics and they include

a number of conditions mediated by mitochondria including infections and immune disorders, metabolic, cardiac and neurological conditions, stress and chemo- and radiotherapy side effects (Professional Register of TCM, 2022) whose connection to mitochondrial function have all been described above in section 3.1. Biomolecular texts of herbal medicines also focus on disorders of ageing, redox balance. metabolic regulation, immunomodulatory functions. and cancer prevention (Benzie & Wachtel-Galor, 2011) which all involve mitochondrial function. The multiple, non-specific effects of herbal treatments, said to guide the body in a general therapeutic direction, as opposed to the specific targeting of synthetic drugs (Karimi et al., 2015) may also indicate that mitochondrial modulation could be a common factor. Wallace (2008) even speculated that mitochondria may be behind fundamental traditional medical concepts such as "Qi" (气 or 氣, pronounced "Chee").

Some experimental evidence has supported this connection between mitochondria and Qi (Wong et al., 2012; Leong et al., 2018; Chen, Y., Li et al., 2020; Chen,Y., Feng et al., 2020) but a straightforward equation of the two concepts can be problematic. An update on the popular definition "vital energy" is well overdue, a definition that reflects more the vitalist doctrines of early translators of Chinese medical texts (Kendall, 2008) than the multi-valent concept that it is, which can be used to draw together many different ways of understanding the world, including cosmological, political, philosophical and physiological models and used in reference to phenomena in fields as diverse as diverse as astronomy, meteorology and other natural sciences; political relations; religious ritual; and internal or tactile sensations felt during emotion, disease states, medical and self-cultivation practices, in often overlapping metaphors (Ingraham, 2019; Stanley-Baker, 2022). Its use is deeply embedded in the Chinese language meanings as diverse as "steam, vapour, air, anger, gas" with

(Archchinese.com, 2023) and featured in hundreds of compound terms with highly diverse meanings (table 7).

Chinese	Pinyin	Literal Meaning	Interpreted Meaning
氣象	Qixiang	Qi image	Weather pattern
人氣	Renqi	Person Qi	Personality or character
生氣	Shengqi	Growing Qi	Anger
大氣	Daqi	Big Qi	Atmosphere, heavy breathing or magnanimity depending on context
名氣	Mingqi	Status Qi	Reputation or fame
士氣	Shiqi	Soldier Qi	Morale
運氣	Yunqi	Moving Qi	Luck
正氣	Zhengqi	Righteous Qi	Healthy Environment
氣味	Qiwei	Qi flavour	Odour or Scent
氣色	Qise	Qi colour or Qi Appearance	Complexion

Table 7: Multivalency of the Word *Qi* 氣 in Chinese Translations obtained from ArchChinese.com (2023)

Narrowly defining *Qi* as "vital energy" has led to it becoming tied to the fate of vitalism that has long been disproven and discarded by the scientific and medical community and remains popular only in some naturopathic and chiropractic circles (Coulter, Snider & Neil, 2019). If *Qi* is tied too closely to mitochondrial function, we run the risk of repeating that mistake and leaving *Qi* at the mercy of another scientific trend. Alternatively, we may create too much of a narrow definition that strips *Qi* of any aspects not demonstrated to be due to mitochondrial function, thereby losing its phenomenological and poetic uses which are arguably some of its most powerful (Wegmüller, 2015). However, these similarities do suggest that traditional indications of the effects of herbs on *Qi* could be a starting point for generating hypotheses to test with a scientific method, providing we remain mindful that *Qi* encompasses more than mitochondrial function alone.

3.2.1. Hormesis in Traditional Medicine

The concept of hormesis can be used to explain the beneficial effects of many modes of traditional medicine and health promoting systems. Hormesis has been noted to have particular similarity to the Chinese medical concepts of *Yin* and *Yang* (Sun et al., 2020) making this system of medicine, which comprises physical, psychological, dietary and herbal interventions, a good example of how hormesis is at work in traditional therapies.

3.2.2. Exercises

In exercise an increase in the energy demand of the cell results in increased ROS production that ought to cause an increase in oxidative stress damaging the system but actually produces the opposite effect, resulting in increased efficiency of the mitochondria and improvements in overall health (Merry & Ristow, 2016a). This is an example of the overcompensation mechanism of hormesis whereby an organism overcompensates for a disruption to homeostasis (Calabrese et al., 2016), but due to the number of different pathways that ROS activates (see 1.5.1.2), it is also an example of a single substrate activating different pathways depending on the dose. More extreme forms of exercise may push mitochondria above the hormetic threshold and result in a less efficient

network (Flockhart et al., 2021) and may explain the development of Long Covid among 4% of college athletes (Massey et al., 2022) whose activities should provide protection from a condition associated with mitochondrial health (Srinivasan et al., 2021; Cox, 2022).

Traditional medical systems do not only use physical exercises to promote health. Many also use mental exercises and breathing exercises in the form of meditation. The slowed breathing during meditation has demonstrated an acute decrease followed by a sustained increase in O_2 saturation (Bernardi et al., 2017) and mild, acute O_2 deprivation has also demonstrated a protective capacity against IRI through mitochondrial mechanisms (Boengler, Lochnit & Schulz, 2018). Meditation also increases resting levels of melatonin (Nagendra, Maruthai & Kutty, 2012), a powerful antioxidant (Reiter et al., 2016), and may help to replace sleep or pay off sleep debt (Kaul et al., 2010), suggesting a hormetic antioxidant mechanism.

3.2.3. Dietary Interventions

Fasting and ketogenic diets which limit carbohydrates and thus glucose availability, forces greater reliance on β -oxidation of fatty acids in mitochondria for fuel, and have demonstrated improvements in health outcomes from various metabolic disorders (Miller, Villamena & Volek, 2018). Both exercise and calorie restrictions increase the fusion of healthy mitochondria into networks and the fission and autophagy of defective mitochondria resulting in a more efficient system. Fasting is an important aspect of Ayurvedic medicine (Gangele & Paliwal, 2020) and intermittent fasting is practised in Unani medicine as part of the religious obligations required of all muslims during Ramadan (Visioli et al., 2022), both with many perceived health benefits (Trabelsi et al., 2022). Chinese medicine does not emphasise complete fasting but the early medical text Eliminating Grain and

Eating Qi (卻穀食氣 *Quegu shiqi*) from 168 B.C. encourages the avoidance of grain (辟穀 *Bigu*) as a technique for longevity (Harper, 1998, pp.305-9). It was later adopted as a method of eliminating malevolent ghostly parasites (三尸 *sanshi* or 三蟲 *Sanchong*) that were understood to feed upon grains and shorten the lifespan of their host (Dannaway, 2009) suggesting an early version of a ketogenic diet.

3.2.4. Manual Therapies

Even most manual systems of medicine such as acupuncture, massage and heat therapies often involve stimulating inflammatory reactions which suggest a temporary and controlled increase in oxidative stress (Checa & Aran, 2020) which are considered to be indicators of a positive outcome (Zhu, 2014) suggesting a hormetic intention. In addition, deformation of the skin through the use of needles, heat and electrical stimulation releases large amounts of eATP from keratinocytes which circulate, acting as signalling molecules (Burnstock, 2009). These are suggested to be a potential mechanism behind the reported effects of acupuncture on metabolism (Wang & Zhou, 2023), preconditioning against IRI (Wang, L. et al., 2023), circadian cycles (Wang, X. et al., 2023), pain modulation (He, J-R. et al., 2020), recovery from injury (Wang, M-J. et al., 2023), moderation of hypertension (Li et al, 2023), atherosclerosis (Wu et al., 2023) and depression (Yu et al., 2023; Zou et al., 2023). Besides mitochondria being the main source of eATP, this eATP also triggers Ca²⁺ entry into T-cells which enhances mitochondrial ATP production (Trebak & 2019) making purinergic ionotropic receptors a means of Kinet. communication between the keratinocytes deformed by needle manipulation and the subsequent visible immune response. In addition, the importance of mitochondria to maintain the high energy demand of neurological signalling (Misgeld & Schwarz, 2017; Datta & Jaiswal, 2021) might suggest that the

deliberate stimulation of nerve fibres in acupuncture which leads to adaptive neuroplasticity (Xiao et al., 2018) is also driven by mitohormetic changes.

Moxibustion, the application of heat on acupuncture points through the use of burning mugwort (Artemisia vulgaris, L., Asteraceae) on, or near, acupuncture points, or by fixing it to the top of needles, is another method used in TCM that most likely utilises hormetic mechanisms. The temperature achieved from indirect moxibustion can reach 65°C on the outer skin and 45°C on the subcutaneous layer and sometimes higher in methods involving direct skin contact (Deng & Shen, 2013). The body functions within a narrow temperature range and deviations from this induce the denaturation of proteins, including those involved in the ETC, antioxidant defences and cellular repair (Slimen et al., 2014). This will result in reduced ATP production, increased ROS, cellular damage that goes unrepaired, and ultimately apoptosis. To protect the cell from excessive heat, Heat Shock Proteins are activated by raised temperatures which act as chaperones to prevent the misfolding of proteins and refold those which have been denatured (Kurop et al., 2021). Since misfolded protein aggregation and a reduced Heat Shock Protein response is associated in many disease states (Hu et al., 2020), there is the possibility that they could be employed therapeutically and that moxibustion and similar energetic stresses like electroacupuncture may be stimulating this mechanism (Cakmak, 2009).

In cupping and scraping therapies, the haematomas produced imply the release of heme, a strongly pro-oxidant complex, which then breaks down into biliverdin and then bilirubin (Jeney et al., 2013), both potent antioxidants (Lowe, 2017). Experimental evidence has also shown an upregulation of antioxidant enzyme heme oxygenase-1 (HO-1) in internal sites distant to the bruising after the application of scraping (Kwong et al., 2009). These suggest a hormetic response to an artificially induced oxidative stress, which is followed by a decrease through both hormetic mechanisms and the direct actions of biliverdin, bilirubin and HO-1. This may explain some of the findings of reviews that report it may be effective at reducing pain and inflammation (Wood et al., 2020; Mohamed et al., 2023), and others that suggest cupping has a potential use in controlling obesity (Kang et al., 2023), implying a mitohormetic mechanism, although all call for more higher quality research.

3.2.5. Herbal Medicine

Herbal medicines have also been suggested to follow a hormetic paradigm by promoting adaptive responses at low doses but having inhibitory effects at higher doses, corresponding to "regulating" and "curative" effects (Wang et al., 2018; Sun et al., 2020). These may be present in the same formula through the use of the "Monarch, Minister, Assistant, and Guide" system used in CHM whereby the herbs in a formula are ranked in order of descending dosage, promoting stimulation of some systems through low dose administration and the inhibition of others through a high dosage.

One example of this is the CHM use of cinnamon (*C. cassia*, (L.) J.Presl syn. *C. aromaticum* Nees et *Neolitsea cassia* (L.) Kosterm., *Lauraceae*) where it has a dual function: in small doses (1.5 - 4.5 g) the bark is said to "warm the *Yang*" (Bensky et al., 2004, pp.684), providing improved energy, mobility and bodily warmth but the branchlets are also used in higher doses to treat colds and flu (ibid., pp.8-10), typically around 9 g but sometimes gradually climbing as high as 30 g. Historically it may have been standard to use as much as 42 g¹ from the start and unlike modern texts, *Gui Zhi* (桂枝, "cinnamon twig") referred to all forms of cinnamon with the the bark that is recommended in small doses today being the likely part

¹He (2013) identifies 1 *liang* as being 13.875 g in the Eastern Han dynasty and the classic formula "Cinnamon Twig Decoction" written during this period recommends using 3 *liang* of cinnamon twigs (Zhang & Liu, 2016).

used (Liu, J. et al., 2020). This can be explained by hormesis. C. cassia contains high levels of coumarins which are produced by plants for many purposes, one of which is to protect the plant from infections (Lončar et al., 2020) and mitochondria are theorised to have evolved from a bacterium that entered into a symbiotic relationship with a eukaryotic cell (Roger, Muñoz-Gómez & Kamikawa, 2017). Therefore, compounds metabolised by a plant to have a disruptive effect on infectious bacteria may also have a disruptive effect on mitochondria. This is seen in the literature with several components of cinnamon having antibacterial effects via multiple mechanisms, some of which would affect mitochondria such as the inhibition of ATPases (Vasconcelos, Croda & Simionatto, 2018). This can be seen in the warming effect of C. cassia deriving from an uncoupling effect (Li X. et al., 2021). This is when protons which generate the $\Delta \Psi m$ are disassociated from ATP synthesis, causing the excess energy from the dissipation of the proton gradient to be dispersed as heat, leading to thermogenesis (Demine, Renard & Arnould, 2019). This stimulates a hormetic response with mitochondrial biogenesis (Gannon et al., 2015), especially in adipocytes leading to browning of adipose tissue (Kwan et al., 2017) and alleviating insulin resistance (Couturier et al., 2016) when taken in doses generally considered safe (2 g / day) by the Food Standards Agency (Apekey & Khokhar, 2009). When taken in higher doses, this disruptive effect might be enough to inhibit bacteria and potentially mitochondria too, but the mitochondria have the added protection of being inside a cell so will only be exposed to the level in the cytoplasm, whereas free floating bacteria will have a far higher exposure. In addition the mitochondria may already have developed compensatory mechanisms through chronic exposure to low doses of cinnamon.

3.2.5.1. Antioxidants

Many herbs are marketed for their antioxidant potential despite there being little evidence that antioxidants can help *in vivo* (Berger et al., 2012). This situation came about when ROS were discovered to be responsible for atherosclerosis, cancer, vision loss and a number of other chronic degenerative diseases which caused an immediate marketing drive among the media, food and supplement industries who were keen to promote products rich in antioxidants (Harvard School of Public Health, 2021). Subsequent research failed to meet the anticipated expectations and may even be detrimental by preventing the ROS mediated signalling behind hormetic responses to exercise (Pingitore et al., 2015; Merry & Ristow, 2016b; Higgins, Izadi & Kaviani, 2020), although it may be useful in certain populations where exercise causes oxidative stress (Wang, H. et al., 2023). Concerns have even been raised that antioxidants could facilitate cancer progression through inhibition of ROS mediated apoptosis (Harris & DeNicola, 2020).

Antioxidants have a controversial history with chemotherapy. Many chemotherapeutic agents rely on the induction of oxidative stress, initiating apoptosis in cancer cells, presenting the theoretical possibility that antioxidants may reduce the effectiveness of chemotherapy in tumour cells (Lawenda et al., 2008) or encourage tumour growth (Harris & DeNicola, 2020), but several papers present the opposite to be true (Poljsak & Milisav, 2019; Athreya & Xavier, 2017). Publication bias may prevent more studies from demonstrating an interfering effect of antioxidants on chemotherapy (DeVito & Goldacre, 2019) which makes many clinicians err on the side of caution. However, since ROS is necessary to stimulate cancer cell proliferation, invasion and metastasis, antioxidants could slow progression (George & Abrahamse, 2020). Alternatively, some authors suggest that antioxidants may have a biphasic dose-response leading to them being both

pro- and anti-oxidant at different doses, with them potentially interacting differently with the metabolisms of cancerous and non-cancerous cells (Shin et al., 2020; Mahanta & Challa, 2022; Zhang & Zhang, 2014; Fernando, Rupasinghe & Hoskin, 2019). This model proposes a different hormetic redox curve for different cell types with the greater reliance of normal cells on OXPHOS making them capable of tolerating higher doses before having a pro-oxidant effect (figure 33).





Quercetin is one example of a phytochemical that has this dual effect on redox states, simultaneously being reported to induce apoptosis in cancer cells through induction of oxidative stress, and providing neuroprotection through a reduction in ROS. It is theorised that it may achieve this effect through the common target of modulating the mitochondrial ETC (Carrillo-Garmendia, Madrigal-Perez & Regalado-Gonzalez, 2023). In cancer cells this disrupts the ETC which is still necessary for the production of ATP, resulting in $\Delta\Psi$ m disturbance and ATP deficiency with cytotoxic effects. In neurodegenerative disorders, this same effect is able to restore the impaired ETC complexes, resulting in a pro-survival effect.

It is therefore possible for the right dose to exert an anti-oxidant effect on normal cells, protecting them from oxidative stress of chemotherapy, while having a pro-oxidant and pro-apoptotic effect on tumour cells (Prasad, Gupta & Tyagi, 2017). Using phytochemicals to prevent the side effects of chemotherapy has long been investigated, but the notion that they may enhance the efficacy of chemotherapeutic drugs is relatively new and has become known as "priming" the cells for destruction (Mould, 2019). This mechanism has been demonstrated with herbal extracts including cannabidiol (Henley, 2015) and ashwagandha (*Withania somnifera* (L.) Dunal, *Solanaceae*; Henley et al., 2017).

3.2.5.2. Adaptogens

One of the most popular classes of herbs are adaptogens which aim to enhance our non-specific responses to stress. The term adaptogen was defined by Dardymov & Brekhman (1967) as:

"non-specific remedies against a very broad spectrum of harmful factors ("stressors") of different physical, chemical and biological natures, and must be innocuous to have a broad range of therapeutic effects without causing any disturbance (other than very marginally) to the normal functioning of the organism" (cited in Tian et al., 2022).

Many medicinal herbs and phytochemicals have been investigated for their adaptogenic properties, often through their antioxidant potential, assistance with improving energy levels, boosting the immune system and enhancing mental functioning (Panossian & Wikman, 2010) but their multi-target effects have made it difficult to determine any singular mechanisms by which they can be measured (Gerontakos et al., 2020).

Early models suggested they may work by suppressing the General Adaptation Syndrome of Selye (1936) and an antioxidant capacity to reduce the damage caused by ROS generated by the stressor (Passovian, Wikman & Hagner, 1999). More recently, hormesis has been presented as a potential mechanism for the effect of adaptogens (Lamming, Wood & Sinclair, 2004). This suggests that foreign molecules produced by plants under stressful conditions can mimic the stress response in animals that feed on them and generate adaptive responses that then increase their tolerance when actual stressors are experienced.

This method of inducing hormesis with adaptogenic plants has been coined xenohormesis (Baur & Sinclair, 2008). One example of a xenohormesis inducing compound is resveratrol, which is generated by grapes in response to harsh environments and activates Silent Information Regulator (SIRT)1 in humans, associated with the health enhancing effects of calorie restriction, making resveratrol a compound of interest in metabolic hormesis (de Ligt, Timmers & Schrauwen, 2015). Another example are ß-glucans present in medicinal mushrooms that activate microbial Pattern Recognition Receptors, training the immune system for when an actual pathogen invades (van Steenwijk, Bast & de Boer, 2021). This implies that different strategies may be appropriate before and after the onset of disease, similar to the CHM concept of "tonics" that help to prevent disease but are of little use and even contraindicated once a pathogen has taken hold (Tian et al., 2022; Bensky et al., 2004, p.721). Given the role of mitochondria in many of the functions adaptogens claim to improve, it seems possible that mitochondria are the main target of adaptogenic herbs.

As well as adding to the general knowledge on these herbs, potentially providing new avenues for treatments to be developed, mitochondria may provide a means of evaluating the quality of a herbal blend, by observing their activity on these organelles and comparing future batches to see if they have the same effect. A system such as this can profile a single herb with complex ingredients such as adaptogens, or a blend of herbs which may also have multiple targets and mechanisms since the only important measure is the end result. Initial research will require some pilot work to find the ideal cell lines and specific assays to use for any particular herb or blend, but once profiled and recorded, future batches can be tested against the same standard and observed to see if they have the desired effect. Potency can also be measured by comparing the degree to which the herb or blend creates the known effect as the reference standard. The advantage of using biological assays instead of chemical testing is that knowledge of the active ingredients or their proportion, which may be numerous, have complex interactions with each other, and vary according to batch or harvest, is not necessary. Instead the effect of the herbal tea on its target tissues is measured directly and a statement of quality pronounced based on its final effects rather than the current methods of measuring quantities of a specific compound. The first herb to be selected was liquorice due to its ubiquity in TCM, and commercial supplements and herbal teas blends sold in the UK based on Chinese and Ayurvedic medicine, including Pukka (Rolfe, 2023) and Yogi Tea (2023), and Aura Nutrition (2023a), making it an important herb to test to ascertain its effects in any subsequent blends that contain liquorice.

Chapter IV

4. Liquorice



Figure 34: Liquorice (*G. uralensis*); Raw, Dried and Sliced Roots (Photo by Hazeena Azeez, 2023)

Liquorice (*G. glabra* L., *G. uralensis* Fisch. ex DC. & occasionally *G. inflata* Batalin) was chosen as the first herb to test for its capacity to modulate mitochondrial activity. It has a long history of medicinal use in every documented herbal medical tradition in the world and is still one of the most commonly consumed herbal medicines today (Brinckmann, 2020). It has a range of pharmacological properties and industrial uses, is considered an adaptogen, and present in many commercially available medicinal herbal teas in the UK (Rolfe, 2023; Yogi Tea, 2023; Aura Nutrition, 2023a). This means that as well as being

highly likely to have observable mitochondrial effects, these effects need to be documented before any future studies of more complex blends is undertaken in case the modulations detected in future blends can be attributed to liquorice alone.

The name liquorice derives from its Greek genus name *Glycyrrhiza* meaning "sweet root" and is renowned for its sweet taste, reputedly 50 times sweeter than refined sugar (Isbrucker & Burdock, 2006). Its name in other languages has similar translations: in Ayurveda it is called *Yashtimadhu*, यष्टिमधु, meaning "sweet stick" (Pole, 2012, p.220), and in Chinese it is called *Gan Cao*, 甘 草, meaning "Sweet Herb" (Wilms, 2017, p.40).

As well as being one of the most widely consumed medicinal herbs today (Brinckmann, 2020), liquorice also has many industrial uses. Its sweet taste makes it valuable as a sweetener in the food industry, and it has been manufactured into confectionery since the 19th century. It is used in pharmaceuticals where an extract of glycyrrhizin is used as a hepatoprotective agent in much of asia (Hayashi & Sudo, 2009) and the surfactant properties of its saponin content facilitate the absorption of some drugs such as anthraquinone glycosides (Öztürk et al., 2017). It is used in cosmetics for its anti-inflammatory and skin whitening properties, and more recently as an emulsifier (Ralla et al., 2020) and shampoo (Azadbakht et al., 2018) for its effect on steroid dehydrogenases, potentially stimulating hair growth in androgenetic alopecia (Dhariwala & Ravikumar, 2019), while also suppressing body odour causing Staphylococcus bacteria (Hara, Matsui & Shimizu, 2014). The tobacco industry is a major consumer of liguorice which is used to sweeten, enhance and balance the smoke flavour, improve absorption of other flavouring agents, reduce roughness that causes dryness in mouth and throat, and prevent desiccation to improve shelf life (Carmines, Lemus & Gaworski, 2005). Recently it has also attracted attention as a potential alternative for antibacterial agents and artificial growth stimulants in animal feeds and aquaculture (Ding et al., 2022). In 2007, its global market value was US\$42 million which increased to US\$848.9 million in 2022 with an estimated growth of 5.3% predicted to reach US\$1, 418. 4 million in 2032 (Market Future Insights, 2022).

4.1. History

The earliest evidence of liquorice being used by humans dates the ancient Assyrian and Egyptian empires (Ding et al., 2022). It is mentioned in the ancient Assyrian stele, the code of Hammurabi, and clay tablets excavated from the library of Assyrian King Ashurbanipal. Large amounts were also found preserved in the tomb of Egyptian pharaoh Tutankhamun, sealed c. 1330 B.C.E. and there is mention of liquorice in Egyptian papyri (American Botanical Council, 2022). In India it is first described in Ayurvedic sources (सुश्रुतसंहिता, Sushrata Samhita, Compendium and चरक संहिता, Charaka Samhita, Charaka's Sushruta's Compendium, both 1st Millenium BC), and then in Europe by Theophrastus in the 3rd - 4th century B.C.E., who said the Greeks learned of its therapeutic uses from the Scythians (Fiore et al., 2005). In China, it is mentioned in the earliest known medical writings, "Recipes for Fifty-two Ailments" (Wushier Bingfang, 五十二病方) found in a tomb sealed in 168 B.C.E. (Harper, 1998, p.221-304) and also in the earliest pharmacopoeia of Chinese medicine, The Divine Farmer's Materia Medica Classic (Shen Nong Ben Cao Jing, 神農本草經, 2nd Century C.E.) and takes a prominent role in in the earliest canonical work of Chinese herbal medicine, the Discussion of Cold Damage and Miscellaneous Diseases (Shang Han Za Bing Lun, 傷寒雜病論) compiled by Zhang Zhongjing (張仲景), c. 220 C.E., where it is used in almost half of the formulas presented (Jiang et al., 2020).

Liquorice has remained a popular herb through most of history. Its popularity appears to have been limited in Anglo-Saxon England where the only surviving herbarium from this period, *The Old English Herbarium Manuscript V*, includes a single line entry on liquorice, referring to it as "herb glycyrida" with no attempt to provide a common name suggesting it was peripheral in this herbal tradition (Pollington, 2008, p.135) and notably omitting its main contemporary indication for coughs (ibid., p.269). However, soldiers returning from the crusades brought some specimens back to England where it was cultivated in Pontefract, before demand for confectionery made its import a more economical option (Butler, 2022, p.709-710). Its therapeutic uses were also imported from the Islamic world with the indications for respiratory, gastrointestinal, cardiac and genitourinary

disorders mentioned by Avicenna in the 10th century (Fiore et al., 2005) matching those given by Culpeper (1653) who recommends it for "dry cough or hoarseness, wheezing or shortness of breath, and for all the griefs of the breast and lungs, phthisic or consumptions ... in all pains of the reins [kidneys], the stranguary, and heat of urine." By the time of Grieve's *Modern Herbal* (1931) it was no longer used in urinary disorders or chest pain but was "an ingredient in almost all popular cough medicines on account of its valuable soothing properties" and was recognised that its sweetness is safe in diabetic patients for the alleviation of thirst.

4.2. Main Constituents

The non-active constituents of liquorice are reported to be 20% moisture, 3–16% sugars, 30% starch, and 6% cinder (Esmaili & Karami, 2022). The active constituents of liquorice have been well characterised (Pastorino et al., 2018; El-Saber Batiha et al., 2020) and include saponins, flavonoids and phenolic compounds. The most notable of these are:





Figure 35: Glycyrrhizin (National Center for Biotechnology Information [NCBI], 2023a)

Glycyrrhizin (figure 35) is a pentacyclic triterpenoid saponin compound that is reputed to be 50x sweeter than sucrose (Isbrucker &
Burdock, 2006). All species should contain over 2% glycyrrhizin (Hsueh et al., 2022, p.193) with the dried root usually containing between 5-10% (Rizzato et al., 2017). It is also the compound responsible for the antiinflammatory therapeutic actions of liquorice through its non-selective inhibition of 11 β -hydroxysteroid dehydrogenase, which converts active cortisol into inactive cortisone, resulting in accumulation of glucocorticoids with anti-inflammatory and mineralocorticoid effects (Lim, 2016). This same action is responsible for the toxic effects of overconsumption of liquorice since both cortisol and aldosterone bind to the same receptors, resulting in mineralocorticoid-related hypokalemia and hypertension (Nazari, Rameshrad & Hosseinzadeh, 2017).

Recently glycyrrhizin has gained attention for its antiviral properties. Its ability to bind to ACE2 receptors, which also have an important role in SARS-CoV2 infection, have led to its potential role as a competitive inhibitor to initial infection (Chrzanowski, Chrzanowska & Graboń, 2021). It has also been researched for its ability to inhibit replication in a large number of viruses including hepatitis B, C and E, herpes, Epstin-Barr, Acquired Immuno-Deficiency Syndrome, influenza and others (Zuo et al., 2023) but the large dose required for an independent antiviral effect have led to it being disregarded as a primary therapy and limited to use as an adjuvant, assisting in the antiviral activity of other drugs while reducing comorbidities through its anti-inflammatory and steroid enhancing effects (Sun et al., 2021).

Glycyrrhizin has also attracted interest in its potential as a drug delivery system (Butler, 2022). This is because the molecule is amphiphilic, with its sugar end being hydrophilic and the steroid end, hydrophobic. At sufficient concentration in an aqueous solution, this causes it to aggregate in a ball shape called a micelle, with the sugar end pointing outwards, bonding to the water, and the hydrophobic end in the centre. This can be exploited by incorporating some of the molecules of a lipophilic drug into the micelle, enhancing its solubility and bioavailability (figure 36). Glycyrrhizin also appears to influence the permeability and elasticity of cell membranes, further enhancing the ability of glycyrrhizin to deliver a drug to its intended target (Selyutina et al., 2016; Selyutina & Polyakov 2019), showing promise in cancer therapy (Su et al., 2017) and enhancing the delivery of Atorvastatin (Kong et al., 2018).



Figure 36: Diagram of a Micelle and its use as a Drug Delivery System (created with BioRender.com)

Besides medical usages and food usages, glycyrrhizin is also used in the cosmetic industry as a flavouring and an emulsifier. It has been demonstrated to form nano-sized emulsion droplets at neutral pH which were stable across a broad range of pH values, ionic strengths and temperature ranges (Ralla et al., 2020).

4.2.2. Glycyrrhetinic acid, also known as enoxolone



Figure 37: Glycyrrhetinic acid (left) and its similarity to Hydrocortisone (right) (NBCI, 2023b; NCBI 2023c)

Glycyrrhetinic acid (figure 37) is the aglycone metabolite of glycyrrhizin and found within the extract of the plant itself (Graebin, 2018) but also obtained by enzymatic hydrolysis by the intestinal flora after consumption (Hattori et al., 1983). It is considered to be the active part of glycyrrhizin, able to act on cortisone due to its structural similarity to hydrocortisone (Aronson, 2018). It can have two stereoisomers, 18a and 18β -glycyrrhetinic acid with its 18β form being the most often studied. 18β -glycyrrhetinic acid is of interest in dermatology for its anti-inflammatory, antioxidant and antimicrobial properties combined with its low toxicity to normal cell lines (Kowalska & Kalinowska-Lis, 2019). Like glycyrrhizin, it has also received interest as a drug delivery system but with a particular focus on cancer, especially liver cancer due to its ability to bind with receptors on the surface of hepatocytes, leading to the scaffold being used to create semisynthetic compounds aimed at targeting cancers (Hussain et al., 2021; Wu et al., 2018; Speciale et al., 2022). However, the removal of the glycone component also makes glycyrrhetinic acid lose the amphiphilic properties of glycyrrhizin and become almost completely insoluble in aqueous solutions (Cayman Chemical, 2022).

4.2.3. Glabridin and Glabrene



Figure 38: Glabridin (NCBI, 2023h)

Figure 39: Glabrene (NCBI, 2023i)

Glabridin (figure 38) and glabrene (figure 39) are prenylated isoflavonoids which have demonstrated both agonist and antagonistic activities with different oestrogen receptors (Simons et al., 2011). Glabridin is present almost exclusively in *G. glabra*, comprising between 0.08 and 0.35% of the root's dried weight (Hayashi et al., 2003), and can be used as a marker compound to differentiate species (Rizzato et al., 2017; Avula et al., 2022). It has been primarily studied in relation to its anti-inflammatory, cardioprotective, oestrogenic and metabolic effects, but has also been reported to have antioxidant, neuroprotective, anti-osteoporotic and skin-whitening effects (Simmler, Pauli & Chen, 2013). However, it has poor water solubility (Tian, Yan & Row, 2008), so is unlikely to be extracted in teas.

Glabrene has been primarily researched for its potential oestrogenic activity (Tamir et al., 2001), protective effects against osteoporosis (Liu, J. et al., 2021; Somjen, Katzburg et al., 2004) and cardioprotective effects (Somjen, Knoll et al., 2004) but also acts as an tyrosinase inhibitor making it of potential interest to cosmetics as a skin whitening agent (Cerulli et al., 2022).

4.2.4. Liquiritin, its precursors, isomers and apiosides



Figure 40: Liquiritin (NCBI, 2023d) Figure 41: Liquiritigenin (NCBI, 2023e)

Liquiritin (figure 40) is a flavonoid glycoside with numerous pharmacological effects including anti-Alzheimer's disease, antidepressant, antitumor. anti-inflammatory, cardiovascular protective, antitussive. hepatoprotective and skin protective effects (Qin et al., 2022). Its aglycone component, liquiritigenin (figure 41), also has reported neuroprotective, antioxidant, anti-inflammatory, antibacterial, anti-asthmatic, anti-diabetic, anti-oesteoposis, anti-neoplastic and oestrogen receptor signalling activities (Ramalingam et al., 2018). Isoliquiritigenin (figure 42), the precursor and isomer chalcone to liquiritigenin with a similar pharmacological profile, has recently come to the attention of the media for a potential role in prevention of pancreatic cancer (Tetzlaff-Deas, 2023; Zhang, Yung & Ko, 2022; Zhang, Z. et al., 2022) and has previously been identified as a tyrosinase inhibitor with potential use as a skin whitening agent in cosmetics (Cerulli et al., 2022). Liquiritin, liquiritigenin and their isomers isoliguiritin and isoliquiritigenin, are found predominantly in G. uralensis although liquiritin apioside (figure 43) and isoliquiritin apioside are found equally distributed among species (Kondo et al., 2007) or predominantly in G. glabra (Avula et al., 2022). Together these comprise approximately 1% of liquorice aqueous extracts making the most abundant phenolics (Rizatto et al., 2017).





Figure 42: Isoliquiritigenin (NCBI, 2023f)

Figure 43: Liquiritin apioside (NCBI, 2023g)

4.2.5. Glycycoumarin



Figure 44: Glycycoumarin (NCBI, 2023j)

Glycycoumarin (figure 44) is a coumarin compound which has demonstrated anti-spasmodic, antimicrobial, antitumor and hepatoprotective activities (Tang et al., 2022). It is found exclusively in *G. uralensis* where it can be used as a marker compound (Avula et al., 2022) although different coumarins exist in other species.

4.2.6. Licochalcones



Figure 45: Licochalcone A (NCBI, 2023k)

Licochalcones are a chalcone group found in substantial quantities only in *G. inflata* so can be used as marker compounds for this species (Kondo et al., 2007; Rizzato et al., 2017). Smaller quantities of licochalcones A and B have also been reported in *G. glabra* and either absent or in trace amounts in *G. uralensis* (Avula et al., 2022; Fu et al., 2013; Cerulli et al., 2022). Licochalcone A (figure 45) has attracted particular attention for reported anticancer, anti-inflammatory, antioxidant, antibacterial, anti-parasitic, bone protective, blood glucose and lipid regulatory, neuroprotective and skin protective effects (Li, M-T. et al., 2022). Its anticancer effects are suspected to be due to activation of the mitochondrial apoptosis pathway.

Minor compounds include licocoumarin A, licoarylcoumarin, licopyranocoumarin, prenyllicoflavone A, glisoflavone, semilicoisoflavone B, licoriphenone, kanzonol R, shinpterocarpin, 1-methoxyficifolinol, furfuraldehyde and tetramethyl pyrazine (El-Saber Batiha et al., 2020).

4.3. Use in Western Herbal Medicine

In WHM, liquorice is primarily used for digestive discomfort, acid reflux and coughs (European Medicines Agency (EMA), 2012). Its main actions are described as expectorant, demulcent, anti-inflammatory, anti-hepatotoxic,

antispasmodic and mild laxative (Hoffman, 2003, pp.554-5). Less common actions include anti-ulcer, immunomodulatory, anti-allergy, chemopreventive, antitumor, neuroprotective, hepatoprotective, antimicrobial. antidepressant, cognitive enhancing, able to help with management of dyslipidemia and insulin resistance, and in the management of polycystic ovarian syndrome (Bone & Mills, 2013, p.726-30). Its interaction with steroid hormones has also given it indications in the treatment of Addisons's disease and steroid withdrawal (Brown, 2014). Its actions on coughs, viruses and steroid metabolism has also encouraged its use in the treatment of mild to moderate cases of Covid-19 (Silveira et al., 2020; Bailly & Vergoten, 2020; Diomede et al., 2021) making it the most commonly prescribed herbal medicine for this condition in a survey of herbal medicine practitioners in the UK (Frost et al., 2021).

4.4. Use in Traditional Ayurvedic and Unani Medicine

In Ayurvedic medicine, liquorice (*G. glabra*) is used as a tonic, able to balance all three *doshas* (the three primary energies), strengthen the kidneys and calm the nervous system, in addition to the common WHM indications of cough and digestive disorders (Pole, 2012, p.220). It is also suggested as an aphrodisiac in the *Kama Sutra* and its sequel, the *Ananga Ranga* (Aronson, 2018). In the Unani tradition, liquorice was recommended by Rhazes in the 9-10th century for cough and by Avicenna in the 10th century for making the voice clear, reducing hoarseness of the throat, disorders of the kidneys and bladder, palpitations, skin ulcers and as a remedy for gastrointestinal disorders (Fiores et al., 2005) but most of these were later dropped for many centuries and it retained only expectorant, demulcent and antiinflammatory actions until modern research suggested it be used for peptic ulcers (Saad & Said, 2011, p.264).

4.5. Use in Chinese Herbal Medicine

In CHM, liquorice (usually *G. uralensis* but *G. glabra* and sometimes *G. inflata* are considered viable species) is the most commonly used herb, present in between a quarter to half of all Chinese medical formulas (Wang et al., 2013; Guo

et al., 2014) and is said that "nine out of ten formulas contain liquorice" (*Shifang Jiucao*, 十方九草) (Jiang et al., 2020). It is one of the only herbs prescribed as a single herb without any additional supporting herbs in the classic Han dynasty formulary text, *Discussion of Cold Damage* (*Shang Han Lun*, 傷寒論, c. 220 C.E.), where a decoction of liquorice (*Gan Cao Tang*, 甘草湯) is indicated for sore throat that has persisted for 2-3 days (Liu, 2016, p.739). Although it retains the WHM indications for coughs and digestive discomfort, its primary indication is as a *Qi* tonic and it contains a few additional indications unique to CHM such as resolving toxicity, harmonising the effects of other herbs and acting as an envoy, enabling other herbs to effect all twelve traditional organ systems (Bensky et al., 2004, pp.732-4; Chen & Chen, 2000, pp.866-71). This has earned it the title of "Nations Elder" (*Guo Lao*, 國老), a term used to refer to the Emperor's teachers, the only people in the traditional hierarchy that were masters to the Emperor (Li, 2022, p.36).

In traditional theory, sweet taste and yellow colour are associated with the neutral, central element of Earth (Marshall, 2020) making liquorice "the king of herbs with regard to the neutral earth element" (Liu, 2020, p.487) and providing it with the power to treat both "Hot" and "Cold" disorders, and resolve toxicity since the earliest Chinese medical writings (Wilms, 2017, p.40), by bringing either extreme back to the centre. Its effects on the cardiovascular system have also been recognised since the Han dynasty with a large dose being used as the chief herb to treat irregular pulse and palpitations in the formula "Prepared Liquorice Decoction" (Zhi Gan Cao Tang, 炙甘草湯) that remains one of the most popular formulas in Taiwan today for improving myocardial contractility, alleviating oxidative stress and reducing inflammation in ischaemic heart disease (Wang, L. et al., 2022). This prepared form involves stir-frying with honey to increase its ability to augment the TCM organs of Heart and Spleen, strengthen the digestion (中焦 Zhong Jiao) and harmonise the properties of other herbs (Bensky et al., 2004, p.734). Liquorice is also often combined with fresh ginger (Zingiber officinale, Roscoe, Zingiberaceae) and jujube berries (Ziziphus jujuba, Mill., Rhamnaceae), which also have actions of settling the stomach, improving digestion by enhancing "Spleen Qi" and moderating the harshness of other herbs (Bensky et al., 2004, p.30 & 730), working in a triad as assistants and envoys to the main herbs in the formula, to smooth the diverse actions of the other ingredients into a harmonious whole that is appended to many herbal remedies (Fruehauf, 1995; Scheid et al., 2009, p.272).

4.6. Risks Associated with Liquorice

Liquorice has a well documented history of toxicity when consumed in large amounts over a long term causing a loss of potassium and an elevation in blood sodium levels leading to mineralocorticoid excess with potentially fatal consequences due to hypertension (Corsi et al., 1983; Omar et al., 2012; Deutch et al., 2019). This may be exacerbated in cases of anorexia nervosa where even ordinarily safe levels can cause hypokalemia and low plasma aldosterone (Fetissov & Crook, 2011). Some incidents have made international news (Associated Press, 2020) and prompted warnings from the NHS (NHS, 2022) making it even more imperative to understand the mechanisms of this herb in as much detail as possible. This is mainly due to the action of glycyrrhizin which has resulted in the development and sale of deglycyrrhizinated liquorice as a safer alternative to liquorice in the treatment of peptic ulcer and intestinal spasms (Tewari & Trembalowicz, 1968).

Liquorice also has the potential to interact with several drugs. Corticosteroids can be potentiated by liquorice, and interactions with the oral contraceptive pill where the pill may increase sensitivity to glycyrrhizin and increase the risks of hypokalemia and hypertension due to the oestrogen modulating activity of liquorice have also been reported (Fugh-Berman, 2000). Interactions with CYP450 enzymes and P-gp mediated transport have also been suggested, which could increase efficacy of chemotherapy but also increase toxicity to normal cells, decrease bioavailability of some drugs, enhance blood–brain barrier permeability and neurotoxicity, and decrease renal elimination (Nazari, Rameshrad & Hosseinzadeh, 2017).

4.7. Quality Evaluation of Liquorice

Liquorice quality is traditionally evaluated through macroscopic inspection and the three species can be differentiated through their gross morphological characteristics (figure 46; Zhao & Chen, 2014, pp.107-8). *G. uralensis* has a reddish-brown cork that is unevenly tight with obvious lenticels with a firm, slightly fibrous but powdery texture. *G. inflata* has a rough, greyish-brown or tan skin with a hard, woody and fibrous texture. *G. glabra* has a smooth, greyish-brown exterior with thin lenticels that may not be obvious and a relatively firm texture. The centre of all species should contain a radial lines called a "chrysanthemum heart" (*Ju Hua Xin*, 菊花心) but this ring is not unique to liquorice though and is found in the centre of other roots such as astragalus (*Astragalus membranaceus*, Fisch. ex Bunge, *Fabaceae*) (ibid., p.129). Inferior quality specimens usually have rough, greyish-brown surface, loose texture with a dark yellow cross-section (Bensky et al., 2004, p.734).



Figure 46: Macroscopic Characteristics of Liquorice.

(a) and (b) are roots and rhizomes of *G. glabra*, (c) and (d) are *G. uralensis* (Raman et al., 2022; reproduced with permission)

Cross sectional microscopic analysis can also be used for identification (Hong Kong Baptist University, 2023; Shah, 2020). Liquorice contains an outer cork layer with many rows of yellow-brown cells, a narrow secondary cortex, 1-3 cells wide, and a thick, curved phloem with cracks. Xylem rays are 3~5 rows wide in bundles and the root and rhizome centre has no pith. Fibres are often grouped and parenchyma cells often contain starch and calcium oxalate crystals. Differences between species can be observed including variations in cork thickness, medullary ray width, vessel element size and arrangement, and pore frequency (Raman et al., 2022) (figure 47).



Figure 47: Microscopic Characteristics of *G. glabra* (left) and *G. uralensis* (right). Abbreviations: ck: cork; ph: phloem; xy: xylem; px: primary xylem; mr: medullary rays (Raman et al., 2022; reproduced with permission).

Chemometric analysis can also differentiate between species. Some compounds are unique to each species and others which are present in several have differing levels which can be used for identification (Avula et al., 2022). These include glabridin which is unique to *G. glabra* and glycycoumarin which is unique to *G. uralensis* while *G. inflata* is the only species to contain a significant amount of licochalcones. In addition, different forms of liquiritin are abundant in different species with *G. glabra* containing the most liquiritin apioside and isoliquiritin apioside, while *G. uralensis* contains the most liquiritin, neoliquiritin and isoliquiritin. These differences can be seen using HPTLC which reveals some unique quenching zones for *G. uralensis* compared to *G. glabra* (figure 48).





Left shows underivatized plates under UV 254 nm (top), and UV 366 nm (bottom). Right shows derivatized plates under UV 366 nm and visible light. Track 1 shows glycyrrhizic acid (0.5 mg/mL). Tracks 2-7 show *G. glabra*, tracks 8-15 show *G. uralensis* with additional quenching zone.

4.8. Adulteration of Liquorice

With its high global market value increasing rapidly due to its expanding uses, the chances of liquorice becoming adulterated for economic gain are also likely to increase. Before the advent of chemometric testing, adulteration of the powdered herbs with sucrose has been reported since it can imitate the sweet taste of liquorice while increasing its bulk to command a higher price by weight (Lee, 2018). This is still common today with the majority of "liquorice" confectionery on the market being primarily sugar, molasses, cane syrup, treacle or low sugar sweeteners combined with anise (*P. anisum*) oil, due to anise having a similar smell and taste to liquorice (National Center for Complementary and Integrative Health [NCCIH], 2020; O'Kelly, 2021). These could all be detected quite easily through chemometric or spectroscopic means (Process NMR Associates, LLC, 2007).

Liquorice confectionery has been the main source of adulteration throughout most of history because it has already been extracted and boiled down, giving the possibility of adding other ingredients during this process. This was first described and published in the 1820s when Frederick Accum recorded the practice of mixing gum arabic, used for shoe polish, with a small amount of genuine liquorice juice and reconstituting the mixture into cylinders to be sold as liquorice confectionery (Accum, 1822, p.332). Other common adulterants in liquorice confectionery from the 19th century came from the copper in the cooking pots being scraped off and ending up in the mixture (Hassall, 1876, pp.607-8). These could also be easily detected today with chemometric analysis.

In the raw herb trade, stones have been reported inside bales of liquorice root and powdered olive stones mixed with powdered herbs (Jiwaji University, n.d.). The former may simply increase the weight for the unsuspecting buyer but can also contain toxic mineral compounds such as lead which may be preferred as they are heavier but will also negatively affect the safety of the herb for the end user. The toxicity of heavy metals involves disruption of mitochondrial function (Sun et al., 2022) and so could be a potential target for adulteration detection by mitochondrial analysis, but spectroscopic analysis would be more sensitive and could detect the presence of heavy metals at doses that would not produce a marked biological effect. The use of olive stones in powdered herbs will reduce the efficacy of the herbal powder and could potentially be a candidate for functional evaluation but may also be detected with the presence of foreign marker compounds.

A common source of adulteration with raw herbs is incorrect species being gathered, either through incorrect identification during foraging or harvesting, or purchased incorrectly at markets. Russian, Iranian and Manchurian variants (G. glabra var. glandulifera, G. glabra var. Violacea and G. uralensis) have been listed as adulterants for G. glabra (Pharmacognosy, 2022) but these species also have similar constituents and are listed as accepted variants for medicinal use in other sources (EMA, 2013; Shah, 2020). G. glabra and G. uralansis are often considered interchangeable in many herbal sources of both WHM and CHM traditions (American Botanical Council, 2000; EMA, 2012; Bensky et al., 2004, p.732). Taverniera cuneifolia (Roth) Arn., Fabaceae can also sometimes be misidentified but has a very similar phytochemical profile with an appreciable glycyrrhizin content (Zore et al., 2008; Nagar, Rane & Dwivedi, 2022) and considered by some herbalists to be an acceptable substitute (Bone & Mills, 2012, p.720). It would be easier to detect the presence of this different species with genetic analysis on whole plant samples, or chemometric analysis which does show a clear interspecies fingerprint despite common zones but mitochondrial analysis could potentially be used to detect mixed samples.

Liquorice is usually wild harvested due to the cultivated roots lacking the same levels of constituents (Brinckmann, 2020; Wang, H. et al., 2021). The herbs used in this study are also wild harvested from Georgia, Kazakhstan and Spain (Rolfe, 2023). The superiority of wild liquorice compared to cultivated varieties also introduces the prospect of deliberate adulteration of the wild harvested crop with a cultivated one. As demand grows and the natural habitat of wild liquorice declines (Khaitov et al., 2022), the likelihood of over-foraging increases and suppliers may attempt to adulterate wild harvested crops with cultivated ones. This would represent exactly the form of functional adulteration described by Tao Hongjing in the 5th Century China (Liu, 2021). Therefore, a practical functional adulteration test could involve examining the difference in mitochondrial modulating activity between wild and cultivated liquorice and whether a mixture of the two can be detected. This would be difficult to detect by genetic analysis and simple chemometric analysis but could establish whether the adulterated crop is capable of producing the same biological effect as genuine wild-harvested liquorice.

Finally, there is the possibility of exhausted herbs being re-marketed for the herbal industry. Exhausted liquorice has been used historically in China for external use by reboiling it into a coarse cloth to make a medicinal towel but not for internal consumption (Ding et al., 2022). However, approximately 6-7 million tonnes of liquorice compost is produced every year in China alone, for which no industrialised applications have been successfully applied (Santulli et al., 2020). Even if a potential use for this waste is found, it is unlikely to be as profitable as resale for medicinal use, and if mixed with authentic herbal material, or used in extract form to increase the volume of the extracted material, it would be almost undetectable but may affect the potency, making this another form of functional adulteration that mitochondrial analysis may be able to detect.

4.9. Current Mitochondrial Research into Liquorice

Being such an essential herb in CHM, WHM, confectionery, food and cosmetics, liquorice has been very well researched. A search on Pubmed, conducted on 18th August 2023 reveals almost 6,000 papers returned when using the term "liquorice" alone. 102 are returned for "liquorice AND mitochondria," with the earliest result dating from 1973, seeing a steady increase from 2000 onwards suggesting this is an area of growing interest (figure 49).



Figure 49: PubMed search results by year for "Licorice AND mitochondria" Search performed on 18th August 2023

The majority of these focus on isolated extracts rather than the entirety of the herb. When "Seahorse Flux Analyzer" is added to the search then no studies are found suggesting that despite many of the active ingredients working via mitochondrial modulation, no complete profile of whole liquorice root extraction on mitochondrial activity has been undertaken.

A brief summary of these results reveals a few core themes. Almost half (40) of these are looking at cancer, either by inducing apoptosis via the mitochondrial pathway, especially using licochalcones (Yuan et al. 2013; Kang et al., 2017; Wang et al., 2019; Hong et al., 2019; Oh et al., 2018; Lin et al., 2019; Kim et al., 2015), glycyrrhizin (Lin et al., 2018; Chueh et al., 2012), glycyrrhetinic acid (Luo et al., 2021; Wang, D., Wong et al., 2014; Yang, J. et al., 2012; Lee et al., 2010; Sharma et al., 2012) and isoliquirtigenin (Hirchaud et al., 2013; Ma et al., 2001; Jung et al., 2006; Chen et al., 2019; Hsu et al., 2009), although the recent interest in glycyrretinic acid as a scaffold with which to target cancers has also led to the possibility it may be able to direct delivery of other compounds to mitochondria (Jin et al., 2022; Zhang et al., 2018). Others looked at the attenuation of toxic side effects from chemotherapy and other xenobiotics which were discussed in Section 4.6. Another important category is those investigating neuroprotective effects against apoptosis initiated by endogenous neurotoxic factors such as glutamate (Wang, D., Guo et al., 2014; Lee et al., 2018; Yang, E. et al., 2012), corticosterone (Zhou et al., 2017; Li, X. et al., 2020) and 7-ketocholesterol (Kim et al., 2009). There were also studies into protection from mitochondrial dysfunction caused by metabolic disorders in renal cells (Yokozawa et al., 2005; Cheng, Qiu & Wang, 2020), liver damage in rats (Sil, Ray & Chakraborti, 2015; Sil & Chakraborti, 2016), peripheral neuropathy (Ciarlo et al., 2021) and pancreatic islets (Rahimifard et al., 2014), as well as a mitohormesis effect of isoliquiritigenin which inhibited ATP production in mice with Type 2 diabetes mellitus that triggered mitochondrial biogenesis to restore balance (Yang, L. et al., 2022). Another looked at reduction of mitochondrial ROS mediated inflammatory damage in ulcerative colitis (Kong et al., 2023). The common theme in almost all these studies is the amelioration of ROS generated through mitochondrial dysfunction leading to apoptosis.

Chapter V

5. Research Aim, Objectives, Questions and Hypothesis

5.1. Research Aim

The research aim of this project is:

To profile the ways in which a medicinal herbal tea (liquorice) modulates mitochondrial activity and determine whether this can be used to detect a functionally adulterated extract

5.2. Research Objectives

To achieve this aim, the research objectives are:

- To develop a body of research data into the mitochondrial effects of liquorice tea (*G. glabra* L.)
- To discover unique characteristics which can be used to develop reference standards and validate the testing protocol
- To refine the methodology so that it is rapid and efficient
- To develop a pipeline methodology that can discover the mitochondrial fingerprint of novel herbs and apply this to functional quality assessment

5.3. Research Questions

To develop these objectives, some specific questions to be answered are:

- What unique effects does liquorice have on mitochondria that can be used as a reference standard to enable identification and measure potency?
- What kind of adulteration can be used that would be challenging to detect via chemical or genetic testing but detectable with mitochondrial analysis?
- Can a standard set of tests be developed for a pipeline that will generate a profile?

5.4. Hypothesis

The null hypothesis (H0) is:

A medicinal herbal tea will not induce specific mitochondrial modulations that enable a measurement of functional potency and identification of adulteration.

And the alternative hypothesis (H1) is:

A medicinal herbal tea will induce specific mitochondrial modulations that enable a measurement of functional potency and identification of adulteration.

Materials and Methods

Chapter VI

6. Cells and Culture

6.1. Cell Lines

Michigan Cancer Foundation 7 (MCF7) human breast cancer cell lines were grown in modified Eagle's medium (MEM) (Sigma, UK), supplemented with 10% filter sterilised Foetal Bovine Serum (FBS), 1% L-glutamine and 1% Penicillin/Streptomycin. Michigan Cancer Foundation 10A (MCF10A) non-cancerous human breast cell line were grown in Dulbecco's modified Eagle's medium (DMEM) (Sigma, UK), supplemented with 5% filter sterilised Horse Serum, 1% Penicillin/Streptomycin, 20 ng/mL epidermal growth factor (EGF) (Sigma, UK), 0.5 mg/mL hydrocortisone, 100 ng/mL cholera toxin (Sigma, UK) and 10 µg / mL insulin (Sigma, UK). All cell lines were maintained at 37°C in a humidified 5% CO_2 atmosphere.

MCF7 and MCF10A cell lines were chosen as well characterised cancerous and non-cancerous analogues of breast cells with different respiratory mechanisms and sensitivities that may respond differently to herbal teas. Cancer cells metabolise differently, favouring glycolysis even in the presence of oxygen, discussed in section 3.1.6.3, and MCF7 cells are oestrogen-sensitive (Comşa, Cîmpean & Raica, 2015) which may mean that they will respond to phytoestrogens found in medicinal herbs, of which liquorice (*G. glabra*) has some of the highest of any known herbs (Jungbauer & Medjakovic, 2014). Both of these cell lines were also available to the lab despite the constraints Covid-19 exerted on supply chains. If a herb targets specific organs, systems, or tissues then cell lines from these origins would be preferred. Fortunately liquorice (*G. glabra*), has a wide range of therapeutic uses, from bacterial and viral infections, cough, sore throat, digestive problems, menopausal symptoms and topical application to the skin, with indications in traditional medical systems for lung, liver, circulatory, and kidney diseases (NCCIH, 2020) and said in Chinese medicine to be one of the only herbs to enter all 12 systems (Bensky et al., 2004: 732). This means we may expect to see a response from a wide range of cells.

6.2. Cell Culture

All cell culture work was performed using aseptic technique in a laminar flow hood. Cells were grown in sterile T75 flasks and passaged based on visual inspection. For passage, media was removed from cells, washed with sterile phosphate-buffered saline (PBS) and then detached from the surface of the flask using 1 mL TrypLE for 7 minutes (MCF7) or 20 minutes (MCF10A). Following detachment, the cells were resuspended in fresh media and diluted into new flasks. Only passages between 5 and 25 were used for experimentation.

For seeding cells for experimentation, cells were detached as described above, centrifuged at 1000 rpm for 5 minutes, the media and TrypLE removed, then resuspended in fresh media and counted using a Countess automated cell counter (Thermofisher, UK). The resulting suspension was then diluted to the appropriate seeding density for the experiment and added to the wells with the appropriate volume of media for the plate being used (2.5×10^4 for 96-well plates and 3×10^4 for SeaHorse assay). The seeding for the original liquorice tea extract was performed by the researcher, and for the exhausted liquorice tea extract it was performed by the research associate.

Chapter VII

7. Herbal Tea

7.1. Original Liquorice Tea Extract Preparation

The Original Liquorice Tea Extract (OLTE) was made from "Three Licorice" tea (Pukka, UK) which consists of three liquorice (*G. glabra*) cultivars that are wild harvested from organic soils in Kazakhstan, Georgia and Egypt or Spain (Pukka Herbs, 2020a; Rolfe, 2023). As an industry product, this was considered an ideal sample for testing the potential industrial applications of this method of quality and potency evaluation. The reasons for selecting liquorice as the initial herb to test are described in section 4.

In order to maintain a consistency as close to that of a cup of tea made by a typical consumer, the herbs were prepared following Pukka's Standard Operating Procedure for organoleptic testing of herbal teas (Zoani, 2019; personal communication by email from Marion Mackonochie, 1st October 2020). This suggests using 1 tea bag in 250 mL of boiled water, left to steep for 15 minutes and being agitated after 2 minutes. For consistency across experiments the amount was scaled up to result in the following preparation method:

- 5 teabags were added to 1.25 L of boiling tap water. Tap water was chosen to make the result as close to an actual cup of tea as possible.
- The tea bags were left to steep for 15 minutes and agitated after 2 minutes. The tea bags were then removed and the liquid left to cool.
- Once cool, 25 mL aliquots were measured into 50 mL Falcon tubes and frozen at -20°C in preparation for freeze drying.
- A desired quantity was then freeze dried and kept at -20°C for future use.

The 5 bags of liquorice tea in a single batch produced 2.83 g of freeze dried powder. This was calculated to be 566 mg per bag. Each bag contained 1.5 g of raw material, equating to 377.33 mg / g of raw material.

When ready to use:

- The dried material was weighed and 566 mg (1 bag equivalent) of freeze dried material was measured
- This was rehydrated using 10 mL filter sterilised distilled water to create a solution 56.6 mg / mL solution
- This stock solution was then frozen at -20°C in 2 mL aliquots for future use.

On the day of the experiment:

- The stock solution was thawed and 35.4 µL was added to 1mL of media to make a solution with 2000 µg / mL freeze dried liquorice.
- Further serial dilutions were made from this initial dilution to obtain lower concentrations. The most common were:
 - \circ 250 µL of 2000 µg / mL solution in 1 mL media to obtain 500 µg / mL
 - $\circ~$ 20 μL of 500 μg / mL solution in 1 mL media to obtain 10 μg / mL
- Remaining stock was then refrozen at -20°C and stored in aliquots for future use.

7.2. Pre-extracted Liquorice Tea Extract Preparation

The supplier of the original liquorice tea (Pukka Herbs, UK) was contacted to find any samples which had failed their inspection process and a PhD student at another university (UCL), also investigating quality control methods using handheld Raman spectroscopy devices, was contacted to find samples that had failed quality tests but neither had any samples available. Instead it was decided to adulterate our own sample with a method that would be undetectable to genetic analysis and difficult to detect by chemical analysis using pre-extracted herbs to make an Pre-Extracted Liquorice Tea Extract (XLTE). Pre-extraction has already been identified as a potential problem in herbal medicine and a promising test for functional adulteration. This has been fully detailed in sections 2.6.1 and 4.8 but to summarise: pre-extracted resold herbs are a recognised form of adulteration and reported in several instances of medicinal and culinary herbs and spices (Ilyas, 1980; Al-Khafaji, 2016; Centre for the Promotion of Imports from Developing Countries, 2018; Etheridge, 2023). Liquorice is one of the most widely consumed medicinal herbs around the world (Brinckmann, 2020) with up to 7 million tonnes of liquorice waste being produced each year in China alone, which lack alternative industrial applications (Santulli et al., 2020). Combined with a dwindling natural habitat (Khaitov et al., 2022) and the superiority of the wild harvested crop over cultivated ones (Wang, H. et al., 2021), there is a considerable economic motive to exploit this waste for fraudulent financial gain. In addition, these samples will share the identical genes and similar HPTLC profiles (Gafner et al., 2023), enabling them to evade the two most common tests currently performed in herbal quality evaluation today.

The method used to achieve this was to perform the extraction procedure outlined in section 7.1 once, and then leave the tea bags dry over 72 hours before performing the same extraction procedure again. This second batch was then frozen, freeze dried and stored in aliquots using the same method as before.

Due to problems with the freeze drying process which caused some of the extract to boil over and be lost, a precise total count of the exhausted extract was not possible. This was instead calculated from the yield produced by a single 20 mL tube which yielded 9.2 mg of freeze dried material. Based on a cup having 1.5 g of raw material in each bag extracted in 250 ml water, this meant that the adulterated tea yielded 115 mg per bag, or 76.6667 mg / g.

Compared to the original extraction, which yielded 377.33 mg / g, the total yield was 20.32% of the original, which itself could be used as a test for raw herbs, but it also suggests that companies selling extracts could use this procedure to increase their yields by 20% providing sufficient motivation to consider this a plausible form of adulteration.

7.3. Herbal Tea Treatment Protocols

Cells were treated for 24 hours with varying dosages of liquorice tea, diluted in media. Previous lab experience with herbal extracts and published literature showed 24 hours to be a reasonable starting point (Husni et al., 2015; Ling et al., 2016; Lombardi, Carrera & Cacabelos, 2017; Ziemlewska, Zagórska-Dziok & Nizioł-Łukaszewska, 2021). Once the results from 24 hour exposure were determined, the ROS and $\Delta\Psi$ m assays (described in sections 11.2 and 11.3 respectively) were repeated with a shorter (3 hour) and longer (72 hour) duration to determine the effects of acute and chronic exposure.

Initial dosages were decided by submitting a range of titrations of 1, 10, 100, 250, 500 and 1000 μ g / mL to a cell viability assay (described in section 8.1) in order to find a toxic dose. Once no toxic effect was found at these doses an additional 2000 μ g / mL was added. This also produced no toxic effects so low (10 μ g / mL), medium (500 μ g / mL) and high (2000 μ g / mL) dosages were selected for subsequent assays. All treatments were performed by the researcher.

This experimental approach to finding an appropriate dosage was required because a review of the literature found no standardised dosage of herbs used when performing *in vitro* testing and the literature often appearing biassed towards obtaining the results desired. Chen et al. (2014) used doses of 0, 0.1, 0.3, 1, 3, and 10 μ g / mL when testing the cytotoxicity of a Chinese medicine formula *Jia Wei Xiao Yao San* with Tamoxifen concluding that there is no interaction between the Chinese formula and the drug but there may potentially be interference at a molecular level. Conversely, Languon et al. (2018) used doses of 62.5 - 1000 μ g / mL to assess efficacy and cytotoxicity of Ghanian herbal medicines against cancer cell lines. The low dose in the first study may be deliberately designed to not show interactions or toxicity of the most popular formula in Taiwan for the treatment of breast cancer (Lai, Wu & Wang, 2012), especially when the product was made by Sun Ten, a famous herbal granule company whose products are certified by the Taiwanese Department of Health and covered by National Health Insurance (Tso, 2009), and the funding for the study funding comes from several hospitals and a

government ministry. These suggest possible vested interests in demonstrating a popular brand of medicine as safe and effective which could have introduced bias. The Ghanian study may conversely be deliberately overdosing in order to demonstrate cytotoxicity on cancer cell lines to prove effectiveness, especially when they used no non-cancerous cell lines for comparison. This dosing bias is even more evident in Bailon-Moscoso et al. (2017) who used 100 μ g / mL for their cell viability assays and 1000 μ g / mL in their ROS assay in their investigations of "Horchata", a blend of Ecuadorian herbs. This variability between studies and the possibility of their concentrations being selected based on the desired outcome demanded that we develop our own titration system to find the optimal concentrations.

Chapter VIII

8. Mitochondrial Function Assays

8.1. Cell Viability

Cell viability assays are used to determine the effects of a substance on the proliferation or death of cells by measuring the number of viable cells remaining at the end of the experiment. Regardless of the experiment being carried out, the number of viable cells remaining at the end will affect the measurements so it is essential to determine the cytotoxic or proliferative effects of treatments first to know whether the changes are simply a change in cell number (Adan, Kiraz & Baran, 2016). The standard methodology behind most of these is to introduce a reagent to the cell culture whose substrate can be converted into a coloured or fluorescent product only by viable cells which can then be measured by a plate reader. The signal generated when examined by a plate reader is proportional to the number of living cells left in the assay.

There are several colorimetric assays that use tetrazolium dyes to assess cell viability. Gordon, Brown & Reynolds (2018) describe some of the main ones, with the full chemical name of the tetrazolium dye they use to be:

- MTT (3-(4,5-dimethylthiazol-2-yl)-2,5-diphenyltetrazolium bromide)
- MTS (3-(4,5-dimethylthiazol-2-yl)-5-(3-carboxymethoxyphenyl)-2-(4-sulfophenyl)-2H-tetrazolium)
- XTT (2,3-bis-(2-methoxy-4-nitro-5-sulfophenyl)-2H-tetrazolium-5-carboxanilide)
- WST (2-(2-methoxy-4-nitrophenyl)-3-(4-nitrophenyl)-5-(2,4-disulfophenyl)-2H-tetrazolium, monosodium salt)

Each has their own advantages and disadvantages but the most common, and the one chosen for this study, was MTT. In the MTT assay, a yellow tetrazolium dye is introduced to a culture that penetrates living cells to be converted into purple formazan by reduced NAD(P)H-dependent oxidoreductase enzymes inside living, metabolically active cells (Riss, Moravec & Miles, 2013; figure 50). After incubating for 1-4 hours the resulting insoluble formazan which builds up inside the cells and on their surface is dissolved using a solvent and the colour transformation is evaluated by recording changes in light absorbance at 570 nm and subtracting this from a background absorbance reading at 630 nm using a plate reading spectrophotometer.



Figure 50: MTT Mechanism of Action (created with BioRender.com)

The advantage of the MTT assay is that it is an easy, cost-effective protocol with the potential for a high throughput (Gordon, Brown & Reynolds, 2018). This makes it ideal for getting an initial gauge of changes in viability prior to investigating with more specific methods that gather more detail.

The main limitation with the MTT assay is that it relies on a secondary measure of cell viability, the mitochondrial NAD(P)H-dependent cellular oxidoreductase enzymes, meaning that experiments which affect these enzymes or the metabolic activity of cells may distort the results (Zhang & Cox, 1996; Liu & Dalgleish, 2009). MTT assays are therefore usually the first test in a series in order to observe any obvious changes in viability, after which further tests can determine whether the result is due to a change in metabolic function, enzyme production, or genuine cell viability.

For this project, an MTT Assay kit (Sigma, UK) was used. Cells were seeded onto 96 well plates at a density of 2.5 x 10^4 and treated as described in section 7.3. Following treatment, cells were washed with PBS and the media replaced with phenol red-free media, then treated with 10 µL MTT solution (50 mg MTT in 10 ml PBS). After 3 hours the media was removed and the formazan crystals dissolved with 100 µL DMSO or 2-propanol and shaken for 5 minutes. Measurements were then taken using a microplate reader (SPECTROstar Nano, BMG Labtech, Germany) by subtracting absorbance at 570 nm with absorbance at 690 nm and presented as a percentage of the untreated control.

8.2. Detection and Quantification of Reactive Oxygen Species

2',7'-Dichlorodihydrofluorescein diacetate (DCFDA) is one of the most widely used techniques for directly measuring the redox state of a cell (Eruslanov & Kusmartsev, 2010). It works through intracellular esterases which cleave the two ester bonds from the original molecule to produce H_2DCF that then accumulates intracellularly and oxidises to form highly fluorescent 2',7'-dichlorofluorescein (DCF). This can then be measured by detecting the increase in fluorescence at

535 nm when excited at 485 nm to provide a measure of generalised oxidative stress (figure 51). It is unable to provide a direct measure of any particular reactive species since neither H_2O_2 nor $O_2^{\bullet-}$ can oxidise H_2DCF directly but must be decomposed to radicals, while other substances in the cell may also produce the reaction including other radicals as well as cytochrome c, responsible for activating the caspase cascade that initiates apoptosis, making assessment of ROS during apoptosis using DCFDA especially problematic (Lawrence et al., 2003).



Figure 51: DCFDA Mechanism of Action (created with BioRender.com)

MitoSOX is another important assay used for detecting ROS, especially $O_2^{\bullet^-}$, which contains a positively charged derivative of dihydroethidium (DHE) that rapidly accumulates in mitochondria where it is oxidised by $O_2^{\bullet^-}$ to become 2-hydroxyethidium (2-OH-E⁺), which then binds to DNA producing a red fluorescence (Kauffman et al., 2016). This is more specific than DCFDA but can still suffer from overlapping fluorescence from ethidium (E⁺) which is not formed from $O_2^{\bullet^-}$ and may be generated in larger quantities than the radicalised form (Nazarewicz, Bikineyeva & Dikalov, 2013; Zielonka & Kalyanaraman, 2010).

Although both assays have their drawbacks, a critical approach to measuring ROS combined with data from other tests and controls can give some valuable insights into the degree of oxidative stress that a cell culture is undergoing.

The DCFDA Cellular ROS assay kit (Abcam, UK) was chosen for this study because it is able to provide a generalised measure of oxidative stress. Following treatment, the cells were washed with 1x buffer solution and incubated with 20 μ M DCFDA at 37°C 5% CO₂ in the dark for 45 minutes. Measurements were then taken using a fluorescence microplate reader (FLUOstar Optima, BMG Labtech, Germany) using excitation at 485 nm and emission at 535 nm.

8.3. Detection and Quantification of Mitochondrial Membrane Potential

Several fluorescent lipophilic cationic dyes dyes can be used to measure $\Delta\Psi m$ including:

- TMRM & TMRE (tetramethylrhodamine methyl and ethyl ester)
- Rhod123 (Rhodamine 123)
- DiOC6(3) (3,3'-dihexyloxacarbocyanine iodide)
- JC-1 (5,5',6,6'-tetrachloro-1,1',3,3'-tetraethylbenzimidazolylcarbocyanine iodide)

The basic principle of these is all the same: they accumulate within the mitochondria in inverse proportion to $\Delta \psi m$ making more polarised mitochondria accumulate more dye and depolarised mitochondria accumulate less (Perry et al., 2011) (figure 52). This can then be detected by measuring the fluorescence or imaged with a camera. Each dye has its particular advantages and disadvantages but the overall limitations of these methods are that changes in mitochondrial morphology, localization, or mass might also affect fluorescence measurements and so controls that assess whether these changes are also happening are advisable to conduct alongside the $\Delta \psi m$ probes, although these tests are also not without their limitations and may affect $\Delta \psi m$ and respiration themselves (Buckman et al., 2001).



Figure 52: TMRE Mechanism of Action (created with BioRender.com)

TMRE Mitochondrial Membrane Potential assay kit (Abcam, UK) was selected for detecting and measuring modulations in $\Delta \Psi m$. TMRE loads specifically into polarised mitochondria (Perry et al., 2011) and is especially

suitable for detecting changes in $\Delta\Psi$ m caused by oxidative stress in live cells (Tian et al., 2021). Following treatment, the media was removed and replaced with 20 nM TMRE in fresh media. The cells were then incubated at 37°C 5% CO₂ in the dark for 35 minutes before being washed with PBS and their media replaced with 100 µL PBS. Measurements were taken using a fluorescence microplate reader (FLUOstar Optima, BMG Labtech, Germany) using excitation at 549 nm and emission at 575 nm.

8.4. Detection and Quantification of Calcium lons

Rhod-2 acetoxymethyl ester (Rhod-2) can be used to detect Ca²⁺ levels inside mitochondria by increases in fluorescence when it binds with Ca²⁺, measured at excitation at 557 nm and emission at 581 nm (Cayman Chemical, 2021). It is not mitochondrially specific so another spectrally distinct dye such as MitoTracker Green should also be used to differentiate mitochondrial calcium from other sources on imaging (Maxwell et al., 2018). Combined with the previous assays, additional information can then be ascertained on the mechanism behind modulations in ROS, $\Delta\Psi$ m or apoptosis.

Rhod-2 dye (Thermofisher, UK) was used in this study. For MCF7 cells, following treatment, the media was removed and replaced with 1 μ M Rhod-2 in fresh media. The cells were then incubated at 37°C 5% CO₂ in the dark for 30 minutes before being washed with PBS and their media replaced with 100 μ L PBS. Measurements were taken using a fluorescence microplate reader (FLUOstar Optima, BMG Labtech, Germany) using excitation at 552 nm and emission at 581 nm. For MCF10A cells, this protocol had to be adjusted due to a failure to get any readings at this concentration. After experimentation with different concentrations, incubation periods and seeding densities, the results were obtained using a concentration of 4 μ M Rhod-2 in fresh media and the seeding density reduced to 1 x 10⁴ per well.

8.5. Observation of Mitochondrial Morphology via Confocal Microscopy

Cells were seeded and treated as per section 7.3, then washed with PBS and stained with 35 nM MitoTracker Red (Thermofisher, UK) and 4 drops of NucBlue (Thermofisher, UK) for 40 minutes at $37^{\circ}C$ 5% CO₂. Images were taken using EVOS FL Auto 2 (Thermofisher, UK) using a 40X 0.65 NA plan fluorite objective.

8.6. Measurement of Mitochondrial Respiration

To measure mitochondrial respiration, oxygen consumption rate (OCR) via the SeaHorse XF_e24 Flux Analyzer (Agilent, USA) was chosen. This measures OCR in real time by isolating a monolayer of cells covered with an extremely small volume of media (about 2 μ L) and placing a probe 200 microns above the monolayer (Agilent Technologies, 2020). The probe measures the concentrations of dissolved O₂ in the transient microchamber every 2-5 minutes and calculates the OCR. It then lifts, allowing the media to mix with the microchamber, restoring cell values to the baseline. This provides a direct measure of cellular respiration but other cellular processes also consume O₂ including NADPH-oxidase and oxidoreductase enzymes (Herst & Berridge, 2007) so these processes have to be controlled for. Up to four drugs can be injected during the assay, which can be used to inhibit particular respiratory processes, including the entire ETC, enabling this limitation to be controlled for and other parameters to be measured too.

The SeaHorse MitoStress Assay (Agilent Technologies, 2019) utilises four drugs with known effects on the ETC in order to measure how much O_2 is consumed by those processes through observing the change in OCR when these drugs are applied. These drugs are:

1) **Oligomycin**: inhibits ATP Synthase causing a drop in OCR which enables quantification of ATP linked respiration.

- 2) Carbonyl cyanide-4 (trifluoromethoxy) phenylhydrazone (FCCP): an uncoupling agent that disrupts ΔΨm and collapses the proton gradient, disinhibiting electron flow through the ETC. O₂ consumption is then increased to offset the proton leak, drawing upon the spare capacity of the mitochondria to reach their maximal OCR (Marchetti, Fovez et al., 2020). By subtracting the basal rate from the maximal OCR after FCCP administration, the spare respiratory capacity can be calculated.
- 3) Antimycin A and rotenone: Antimycin A inhibits Complex III and rotenone inhibits Complex I of the ETC which causes mitochondrial respiration to shut down completely, revealing non-mitochondrial respiration. This can then be deducted from the other measurements to reveal mitochondrial OCR alone.

This results in a predictable wave (figure 53) whose modulations can be used to determine the mitochondrial effects of any prior treatments.



Figure 53: A Typical Seahorse MitoStress Assay Profile (created with BioRender.com, based on Agilent Technologies, 2022).

MCF7 and MCF10A cells were seeded onto SeaHorse plates at a density of 3×10^4 using 100 µL of media to ensure a monolayer, which was then topped up to 250 µL after adherence, 3 hours later. 24 hours after seeding they were treated as described in section 7.3 but using the maximum capacity of the well, 500 µL. For each assay, prepared Seahorse media was made appropriate to the cell line by using 50 mL of Seahorse XF Media (Agilent, USA) and adding 1% sodium pyruvate, 1% L-Glutamine and the amount of glucose required for the individual cell line (equivalent to 1 g / L for MCF7 and 3.152 g / L for MCF10A) and balancing the pH and filter sterilising, if required. The cells were then washed twice and replaced with 500 µL of the prepared Seahorse media and incubated for a further 45 minutes to 1 hour in a non-CO₂ incubator at 37°C while the drugs were prepared in the injection ports of the Seahorse Sensor Plate. The drugs were prepared in the following proportions, as outlined in the instructions (Agilent Technologies, 2019):

- 1) **Oligomycin** (Sigma, UK): 56 μL of 10 μg / mL solution in the prepared Seahorse media was added to injection port A
- 2) FCCP (Sigma, UK): 62 μL of 5 μM solution in the prepared Seahorse media was added to injection port B. This was later increased to 6 μM when the predicted response was present in the control group but seemed to have a weaker response than expected, suggesting the drug had lost some of its potency.
- Antimycin A (Sigma, UK) and rotenone (Sigma, UK): 69 μL of a solution containing 10 μM of each drug in the prepared Seahorse media was added to injection port C.

The sensor plate was then calibrated using a utility plate filled with 1 mL calibrant fluid (Agilent, USA) per well which was entered into the Seahorse Flux Analyzer running Agilent Wave software. Once calibrated, the utility plate was replaced with the cell plate and left to run.
Once completed the OCR measurements were then corrected for total protein concentration using the Bradford or Pierce[™] bicinchoninic (BCA) assays (see section 8.6.1). The protein estimations of each well were then entered into Agilent Wave software to normalise the recorded OCR pmol/min based on µg protein for each well.

8.6.1. Protein Assays for SeaHorse Normalisation

Following the SeaHorse Mito Stress Test, a protein assay was performed to quantify the total protein content and correct for variations in cell density. Initially a Bradford protein assay was conducted but after the first set of results this was changed to Pierce[™] BCA assay due potential incompatibility of the Bradford assay with the surfactants in radioimmunoprecipitation assay (RIPA) buffer potentially introducing background noise, and a superior working range of the Pierce[™] BCA assay (ThermoFisher Scientific, n.d.; CiteQ Biologics, n.d.).

For the Bradford assay, each well was washed twice with ice cold PBS before being lysed with 50 μ L ice cold RIPA buffer. Each well was scraped and the contents transferred to individual ice cooled Eppendorf tubes before being centrifuged at 1600 G for 15 minutes at 4°C. During this time a series of protein standards were created using FBS against which a standard curve could be created to allow calculation of the unknown samples via linear regression (performed using Google Sheets). 45 μ L of supernatant was then removed and placed into a new set of tubes, whereupon 5 μ L was added to the wells of a 96-well plate in triplicate. 200 μ L of Bradford Reagent was added to each well and the absorbance measured at 595 nm (SPECTROstar Nano, BMG Labtech, Germany).

For the Pierce[™] BCA assay, each well was washed twice with ice cold PBS before being lysed with 30 µL ice cold RIPA buffer. Each well was scraped and the contents transferred to individual ice cooled Eppendorf tubes before being centrifuged at 1600 G for 20 minutes at 4°C. During this time a series of protein standards were created using Bovine Serum Albumin (BSA) against which a

standard curve could be created to allow calculation of the unknown samples via linear regression (performed using Google Sheets). 20 μ L of water and 5 μ L supernatant was added to the wells of a 96-well plate in triplicate. 200 μ L of Working Reagent was then added to each well, the plate covered in foil and shaken for 30 seconds, incubated at 37°C with no CO₂ for 30 minutes and allowed to cool to room temperature. The absorbance was then measured at 562 nm (SPECTROstar Nano, BMG Labtech, Germany).

8.7. Statistical Analysis

Technical replicates were averaged to achieve a result for each biological replicate and compared as a percentage change from the control to eliminate interday variability. Statistical analysis was carried out using GraphPad Prism (version 9.5). A one-way analysis of variance (ANOVA) was performed with Dunnett's post-hoc analysis to compare all groups to a control, or a Tukey Test to compare treatment groups to each other (Frost, 2019). These were chosen because three or more groups were being compared, the data was not measuring points in the same population and so is unpaired, and can be assumed to have a normal distribution making it parametric. Significance was taken when p < 0.05.

Chapter IX

9. Drug Resistance Assays

9.1. Seahorse Drug Resistance Assay

The drug resistance assay was designed after observing the results obtained from the first cell line, MCF7, to be assayed by the SeaHorse Flux Analyzer, described in chapter 12. A review of the literature revealed the detoxification properties of liquorice in CHM (see section 4.5, and a full discussion in section 20.3). This included some research papers that suggested liquorice may confer resistance to the drugs used in the Seahorse assay. Glabridin was found to protect osteoblast cells from antimycin A induced cytotoxicity (Choi, 2011) and glycyrrhizic acid revealed protective effects against a rotenone induced model of Parkinson's disease in an animal model (Ojha et al., 2016), which was subsequently investigated further with *in vitro* mechanistic studies (Karthikkeyan et al., 2020; Karthikkeyan, G., Prabhu et al., 2021; Karthikkeyan, G., Pervaje et al., 2021; Karthikkeyan et al., 2022). Although there was no literature on liquorice having any effect on oligomycin or FCCP, it prompted an *ad hoc* hypothesis that liquorice was responsible for the reduced effect of the SeaHorse drugs and may have a protective effect on a broad range of toxic substances that modulate mitochondrial activity.

This assay involved repeating the DCFDA assay outlined in section 8.2 but with the introduction of the SeaHorse drugs 10 minutes before the introduction of DCFDA. This timing is recommended in the TMRE instructions where FCCP can be used as a positive control for its effect on collapsing $\Delta\Psi$ m (Abcam, 2022) so was expected to have an effect. DCFDA was chosen over TMRE because changes in ROS measure oxidative stress caused by a number of mitochondrial dysfunctions and so was considered more likely to be measurable in response to the other drugs. The dosage for FCCP was chosen according to the Abcam

protocol with the remaining drug dosages determined based on their proportion to FCCP in the Seahorse assay. This resulted in:

- 1. Oligomycin: 40 µg / mL
- 2. FCCP: 20 µM
- 3. Antimycin A: 40 µM
- 4. Rotenone: 40 µM

9.2. Cisplatin Resistance Assay

The cisplatin resistance assay was designed as a modification of the initial Seahorse drug resistance assay.

Cisplatin (Sigma, UK) was dissolved in dimethylformahide (Sigma, UK) to achieve a stock concentration of 10 mM to be diluted to the appropriate amount on the day of experimentation in fresh media.

Cells were grown and seeded as per section 6.2 and treated as per section 7.3, except divided into groups treated with a high dose of liquorice (2000 μ g / mL) and untreated groups for 24 hours. The cells were then washed with PBS and replaced with cisplatin in low (10 μ M), medium (50 μ M) and high (100 μ M) concentrations for a further 24 hours, after which they were washed once again with PBS and their viability measured with MTT as per section 8.1. A reduction in cell viability of the high dosage cisplatin only treated group was used as a positive control to ensure the cisplatin was killing the cells as intended. Comparisons were drawn between the reduction in cell viability of the cells which were administered cisplatin only and the equivalent group which were administered liquorice for 24 hours prior, followed by the same dose of cisplatin.

9.3. Statistical Analysis

Statistical analysis was performed using the same method of averaging technical repeats, determining changes as percentage of the control and using GraphPad Prism (version 9.5) as outlined in section 8.7. The drug resistance assays differed in that comparisons were made between between a liquorice pretreated group and a group that received no pretreatment, before both were exposed to the same dosage of a specific drug. This made only two groups directly comparable so an unpaired t-test was used for statistical analysis, which is suitable to compare only two groups of unpaired parametric data (Mishra et al., 2019).

Chapter X

10. Adulteration Detection Assays

10.1. Adulteration Detection with Mitochondrial Functional Analysis

Assays using OLTE which scored a significant result (p < 0.05) in any arm were repeated using the adulterated liquorice tea to see if the same result would be obtained. Levels of significance were primarily used to determine if the same result was obtained with mean differences used to determine the degree of difference. This was considered preferable to directly comparing values between the original and adulterated tea samples to better reflect real-world cases where an unknown sample may be tested for its ability to produce a significant known effect and not always compared directly against a reference sample.

10.2. Visual Inspection

A simple visual inspection was undertaken to determine if XLTE could be easily differentiated from OLTE. Eppendorf tubes containing each sample were defrosted and placed in a well lit area of the laboratory and examined for differences in colour, transparency and sediment. A photograph was taken using a Samsung Galaxy A13 50.0 MP camera running software version 13.1.00.58.

10.3. High Performance Thin Layer Chromatography

To determine the likelihood of this form of adulteration being detected through other means, the original and adulterated liquorice samples were subjected to HPTLC analysis. HPTLC instruments from CAMAG (Muttenz, Switzerland) were used, including Automatic TLC Sampler (ATS 4), Automatic Development Chamber (ADC 2) with humidity control, Plate Heater 3, TLC Visualizer 2, and Immersion Device 3.

The reference standards included the original dried herbal material from one of the "3 Licorice" tea bags (Pukka, UK), *G. glabra* from the University of Westminster Polyclinic, and a geno-verified botanical reference sample of *G. uralensis* (Brion Research Institute, Taiwan). The active samples included the OLTE and XLTE.

To prepare the samples, 0.5 g of finely powdered herbal material, or the equivalent of freeze-dried material from the samples, was extracted in 10 mL of ethanol, water 7:3 (v/v), sonicated for 10 minutes, centrifuged at 5000 rpm for 5 minutes, and the supernatant extracted. Since the sample material was limited, the amount of freeze dried material that would have been obtained from 0.125 g of dried material was used in 2.5 mL of solvent. Three sample groups were included:

- 1. OLTE, using 39.3 mg in 2.5 mL of alcohol, water 7:3 (v/v).
- XLTE, using 39.3 mg in 2.5 mL of alcohol, water 7:3 (v/v), to see qualitative differences when used at the same concentration as the original extract, as was used in the biological assays.
- 3. XLTE, using 9.6 mg in 2.5 mL of alcohol, water 7:3 (v/v), to simulate the amount that would be extracted from the same quantity of dried material.

 $2 \ \mu$ L and $5 \ \mu$ L of each sample and reference standard solution were used to aid visual comparison. $2 \ \mu$ L could result in bands from low concentration components being too faded, while 5 μ L could result in bands from high concentration components becoming blurred. Having two concentrations was chosen to increase the likelihood of capturing results in both situations.

For the stationary phase, a 20 x 20 cm aluminium HPTLC silica gel 60 F_{254} plate (Sigma, UK) was used. 2 µL and 5 µL of each of the reference and sample extractions were applied to the plate. The mobile phase used ethyl acetate, formic acid, glacial acetic acid, water 15:1:1:2 (v/v). The plate was developed in a saturated chamber with MgCl₂ at 33% humidity for 20 minutes up to 70 mm from the lower

edge. Plates were then read underivatized under white light and UV 254 nm and UV 366 nm.

For derivatization, a 10% sulphuric acid in methanol solution was used, dipped for 0 seconds at a speed setting of 5 and heated at 100°C for 10 minutes, before being observed under white light and under UV 366 nm.

Besides the appearance or disappearance of particular bands, the overall strength of the columns was compared to determine if HPTLC would be able to conclusively determine if the adulterated tea could be easily recognised, or was within the parameters that might be considered normal variation between crops.

Results

Chapter XI

11. Results from Basic Mitochondrial Assays on Original Liquorice Tea Extract

11.1. Effects of Original Liquorice Tea Extract on Cell Viability

No significant change in cell viability was observed at any of the chosen doses after 24 hours treatment with OLTE (figure 54).



Cell viability measured with 3-(4,5-dimethylthiazol-2-yl)-2,5-diphenyltetrazolium bromide (MTT) represented as % change from untreated control drawn as dotted line at 100%. Data shown as box and whisker plot where the box represents the upper and lower quartiles around the median average shown as a line within the box and the whiskers represent the maximum and minimum values with all data points displayed.

n=8, except MCF7 2000 µg/mL dose added later where n=6.

Figure 54: Results of MTT Assay Showing Changes in Cell Viability After Treatment with Original Liquorice Tea Extract

After finding no cytotoxicity at these doses, three were selected to represent the main dose ranges for the following assays: low (10 μ g / mL), medium (500 μ g / mL) and high (2000 μ g / mL) dosage.

11.2. Effects of Original Liquorice Tea Extract on Reactive Oxygen Species

After 3 hours of treatment with OLTE, ROS increased 16.51% (p < 0.01) in the 500 μ g / mL and 14.88% (p < 0.05) in the 2000 μ g / mL dose groups in the MCF7 cell line (figure 55).



Reactive Oxygen Species (ROS) measured with 2',7'-dichlorodihydrofluorescein diacetate (DCFDA) represented as % change from untreated control drawn as dotted line at 100%. Data shown as box and whisker plot where the box represents the upper and lower quartiles around the median average shown as a line within the box and the whiskers represent the maximum and minimum values with all data points displayed.

n=8 for both groups.

* indicates significance (p < 0.05), ** indicates significance (p < 0.01), compared against control using one way ANOVA with Dunnett's multiple comparisons.

Figure 55: Results of DCFDA Assay Showing Changes in ROS 3 Hours After Treatment with Original Liquorice Tea Extract

After 24 hours treatment with OLTE, no change was observed in either line (figure 56).



Reactive Oxygen Species (ROS) measured with 2',7'-dichlorodihydrofluorescein diacetate (DCFDA) represented as % change from untreated control drawn as dotted line at 100%. Data shown as box and whisker plot where the box represents the upper and lower quartiles around the median average shown as a line within the box and the whiskers represent the maximum and minimum values with all data points displayed.

MCF7 n=10, MCF10A n=8.

Figure 56: Results of DCFDA Assay Showing Changes in ROS 24 Hours After Treatment with Original Liquorice Tea Extract

After 72 hours of treatment with 2000 μ g / mL OLTE, a 14.73% (p < 0.05) drop in ROS was observed in the MCF7 cells (figure 57).



Reactive Oxygen Species (ROS) measured with 2',7'-dichlorodihydrofluorescein diacetate (DCFDA) represented as % change from untreated control drawn as dotted line at 100%. Data shown as box and whisker plot where the box represents the upper and lower quartiles around the median average shown as a line within the box and the whiskers represent the maximum and minimum values with all data points displayed.

MCF7 n=10, MCF10A n=8.

* indicates significance (p < 0.05) compared against control using one way ANOVA with Dunnett's multiple comparisons.

Figure 57: Results of DCFDA Assay Showing Changes in ROS 72 Hours After Treatment with Original Liquorice Tea Extract

11.3. Effects of Original Liquorice Tea Extract on Mitochondrial Membrane Potential

No significant change was seen in $\Delta \Psi m$ after treatment with OLTE at any dose, or any time point, in either cell line (figures 58 - 60).



Mitochondrial Membrane Potential ($\Delta\Psi$ m) measured with tetramethylrhodamine methyl ester (TMRE) represented as % change from untreated control drawn as dotted line at 100%. Data shown as box and whisker plot where the box represents the upper and lower quartiles around the median average shown as a line within the box and the whiskers represent the maximum and minimum values with all data points displayed.

MCF7 n=9, MCF10A n=8.

Figure 58: Results of TMRE Assay Showing Changes in $\Delta \Psi m$ 3 Hours After Treatment with Original Liquorice Tea Extract



Mitochondrial Membrane Potential ($\Delta\Psi$ m) measured with tetramethylrhodamine methyl ester (TMRE) represented as % change from untreated control drawn as dotted line at 100%. Data shown as box and whisker plot where the box represents the upper and lower quartiles around the median average shown as a line within the box and the whiskers represent the maximum and minimum values with all data points displayed.

n=8 for both cell lines.

Figure 59: Results of TMRE Assay Showing Changes in $\Delta \Psi$ m 24 Hours After Treatment with Original Liquorice Tea Extract



Mitochondrial Membrane Potential ($\Delta\Psi$ m) measured with tetramethylrhodamine methyl ester (TMRE) represented as % change from untreated control drawn as dotted line at 100%. Data shown as box and whisker plot where the box represents the upper and lower quartiles around the median average shown as a line within the box and the whiskers represent the maximum and minimum values with all data points displayed.

MCF7 n=10, MCF10A n=8.

Figure 60: Results of TMRE Assay Showing Changes in $\Delta \Psi m$ 72 Hours After Treatment with Original Liquorice Tea Extract

11.4. Effects of Original Liquorice Tea Extract on Calcium Ions

No significant change in Ca²⁺ was observed after 24 hours exposure to OLTE in either cell line (figure 61). MCF10A cells proved to be highly resistant to the assay requiring 4 μ M of Rhod-2 (compared to 1 μ M in MCF7 cells) and modification of the seeding density time to produce any result (see section 8.4) and still had a much wider spread of data when a result was obtained. This may account for the large variation seen in the MCF10A data.



Mitochondrial Ca²⁺ measured with Rhod-2 represented as % change from untreated control drawn as dotted line at 100%. Data shown as box and whisker plot where the box represents the upper and lower quartiles around the median average shown as a line within the box and the whiskers represent the maximum and minimum values with all data points displayed.

MCF7 n = 8; MCF10A n = 6.

Figure 61: Results of Rhod-2 Assay Showing Changes in Ca²⁺ 24 Hours After Treatment with Original Liquorice Tea Extract

11.5. Effects of Original Liquorice Tea Extract on Mitochondrial Morphology

Preliminary confocal microscopy on MCF7 revealed no significant changes in mitochondrial morphology compared to untreated control (figure 62). In all images the mitochondria appear to be fused around the nucleus, apparent by the red stained mitochondria forming a circular network around the nucleus which was stained with NucBlue and so not visible under a red light filter. This is consistent with an absence of raised ROS or cytotoxicity.



Control



24 hr treatment with 10 μg / mL original liquorice tea extract





24 hr treatment with 500 μg / mL original liquorice tea extract

24 hr treatment with 2000 µg / mL original liquorice tea extract

Figure 62: Mitotracker Red Stained Confocal Images of MCF7 Cells After Treatment with Original Liquorice Tea Extract

Since there was no visible evidence of mitochondrial dysfunction in the cell line which showed the greatest perturbations when exposed to OLTE (raised ROS at 3 hours, figure 55, and lowered ROS at 72 hours, figure 57) and no changes in cell viability were seen in either cell line at any dosage (figure 54), it was decided that visual inspection was highly unlikely to produce any useful data so no further microscopy was undertaken.

11.6. Summary of Results from Basic Mitochondrial Assays on Original Liquorice Tea

The results can be summarised as follows:

- 3 hours treatment with 500 µg / mL OLTE saw a 16.51% (p < 0.01) increase in ROS in MCF7 cells
- 3 hours treatment with 2000 μ g / mL OLTE saw a 14.88% (p < 0.05) increase in ROS in MCF7 cells
- 72 hours treatment with 2000 µg / mL OLTE saw a 14.73% (p < 0.05) decrease in ROS in MCF7 cells
- No significant changes were observed on the MCF10A cells

Chapter XII

12. Results from Respiratory Measurements of Original Liquorice Tea Extract in MCF7 Cells

12.1. Effects of Original Liquorice Tea Extract on Mitochondrial Respiration and its Measurement in MCF7 Cells

After 24 hour treatment with OLTE the Seahorse MitoStress assay produced a surprisingly flat curve in response to the drug injections in MCF7 cells, which seemed to become flatter as the dosage increased (figure 63) in contrast to the expected response to the drug injections (figure 53, section 8.6)



MitoStress assay performed with SeaHorse XF_e 24 Flux Analyzer, normalised using protein content measured with Pearce BCA assay to estimate cell count. Points shown as mean average +/- SD. n = 3.

Change at 20 minutes shows effect of administering oligomycin to inhibit ATP synthase. Change at 45 minutes shows effect of administering FCCP disrupt proton gradient and reveal spare capacity. Change at 70 minutes shows effect of administering antimycin A and rotenone to shut down mitochondrial respiration revealing non-mitochondrial respiration.

Figure 63: Results of SeaHorse MitoStress Assay Showing Changes in OCR of MCF7 Cells after Treatment with Original Liquorice Tea Extract

When comparing the basal rates of respiration as a percentage of the control, there was no significant difference in basal OCR (figure 64).



MitoStress assay performed with SeaHorse XF_e^{24} Flux Analyzer, normalised using protein content in mg / mL measured with Bradford and BCA assays to estimate cell count.

Basal OCR calculated by subtracting OCR after administration of antimycin and rotenone from initial OCR. Data shown as mean average +/- SD compared to untreated control shown as dotted line at 100%. n = 3.

Figure 64: Effect of Original Liquorice Tea Extract 24 Hour Treatment on Basal OCR of MCF7 Cells Shown as Percentage in Relation to Control

When plotted as a percentage change from the initial basal rate, the flattening of the expected curve is even more apparent (figure 65).



MitoStress assay performed with SeaHorse XF_e 24 Flux Analyzer, normalised using protein content in mg / mL measured with Pearce BCA assays to estimate cell count shown as % of initial OCR of control. Points shown as mean average +/-SD. n = 3.

Change at 20 minutes shows effect of administering oligomycin to inhibit ATP synthase. Change at 45 minutes shows effect of administering FCCP disrupt proton gradient and reveal spare capacity. Change at 70 minutes shows effect of administering antimycin A and rotenone to shut down mitochondrial respiration revealing non-mitochondrial respiration.

Figure 65: Results of SeaHorse MitoStress Assay Showing Changes in OCR of MCF7 Cells after Treatment with Original Liquorice Tea Extract as percentage of basal OCR

Calculated as a percentage change from the basal rate and compared to the control, 24 hour treatment with 2000 μ g / mL OLTE saw a significant decrease in ATP linked respiration (45.7% compared to 72.24%; p < 0.01; figure 66) and proton leak (54.3% compared to 27.76%; p < 0.01; figure 67), a 26.54% difference.



MitoStress assay performed with SeaHorse XF_e24 Flux Analyzer, normalised using protein content measured with Pearce BCA assays to estimate cell count.

ATP related respiration calculated by subtracting OCR after administration of oligomycin from initial OCR and shown as mean average +/- SD. n = 3.

 ** indicates significance (p < 0.01) compared against control analysed using one way ANOVA with Dunnett's multiple comparisons.

Figure 66: Effect of Original Liquorice Tea Extract 24 Hour Treatment on ATP linked OCR of MCF7 Cells



MitoStress assay performed with SeaHorse XF_e24 Flux Analyzer, normalised using protein content with Pearce BCA assays to estimate cell count.

Proton leak calculated by subtracting OCR after administration of antimycin A & rotenone from OCR after administration of oligomycin and shown as mean average +/- SD. n = 3.

** indicates significance (p < 0.01) compared against control analysed using one-way ANOVA with Dunnett's multiple comparisons.

Figure 67: Effect of 24 hour Original Liquorice Tea Extract Treatment on Mitochondrial Proton Leak in MCF7 Cells

Calculated as a percentage change from the basal rate and compared to the control, 24 hour treatment with 2000 μ g / mL OLTE showed a 78.42% reduction in maximal respiration compared to control, which as below the basal level (90.41% compared to 168.83%; p < 0.05; figure 68), resulting in a spare capacity in the minus figures (-9.59%; p < 0.05) (figure 69).



MitoStress assay performed with SeaHorse XF_e 24 Flux Analyzer, normalised using protein content in measured with Pearce BCA assay to estimate cell count.

Maximal capacity calculated by subtracting OCR after administration of antimycin A and rotenone from OCR after administration of FCCP and shown as mean average +/- SD. n = 3.

 * indicates significance (p < 0.05) compared against the untreated control analysed using one-way ANOVA with Dunnett's multiple comparisons.

Figure 68: Effect of Original Liquorice Tea Extract 24 hour Treatment on Mitochondrial Maximal Capacity in MCF7 Cells



MitoStress assay performed with SeaHorse XF $_{\rm e}$ 24 Flux Analyzer, normalised using protein content in measured with Pearce BCA assay to estimate cell count.

Spare capacity calculated by subtracting initial OCR from OCR administration of FCCP and shown as mean average +/- SD. n = 3.

 * indicates significance (p < 0.05) compared against control analysed using one-way ANOVA with Dunnett's multiple comparions.

Figure 69: Effect of Original Liquorice Tea Extract 24 hour Treatment on Mitochondrial Spare Capacity in MCF7 Cells

12.2. Error Checking the Results of Original Liquorice Tea Extract on Mitochondrial Respiration in MCF7 Cells

These unusual results from the MCF7 Seahorse assay (figures 65-69) prompted an investigation into whether there was a user error or equipment failure. The experiment was repeated using only the high dose and control group at different positions on the plate which confirmed that these findings were genuine and not due to technical error or equipment fault (figure 70).

Although the basal reading starts lower, it is clear that there is almost no effect from the injection of oligomycin and FCCP and only a small change after the injection of antimycin and rotenone. Statistical significance cannot be drawn due to there only being only a single repeat.



MitoStress assay performed with SeaHorse XF_e24 Flux Analyzer, normalised using protein content measured with Pearce BCA assay to estimate cell count as $\mu g / L$. n = 1, with 5 technical repeats for control and 13 technical repeats for 2000 $\mu g / mL$ with positions on plate changed from previous assays.

Change at 20 minutes shows effect of administering oligomycin to inhibit ATP synthase. Change at 45 minutes shows effect of administering FCCP disrupt proton gradient and reveal maximal and spare capacity. Change at 70 minutes shows effect of administering antimycin A and rotenone to shut down mitochondrial respiration revealing non-mitochondrial respiration.

Figure 70: Results of SeaHorse MitoStress Assay Testing For Plate Positioning or Technical Error

12.3. Interpretation of the Results of Original Liquorice Tea Extract on Mitochondrial Respiration in MCF7 Cells as Protection Against Mitochondrial Modulating Drugs

Since there was no previous observations of cytotoxicity or respiratory inhibition at 24 hours with any dosage tested so far (see section 11), and the literature had reports of traditional use for detoxification with some modern mechanistic research supporting this use (see section 4.6), the results were interpreted around an *ad hoc* hypothesis that OLTE may be interfering with the actions of the drugs used in the Seahorse assay.

Reinterpreting the OCR readings as measurements of the effects that the drugs had on respiration, the following patterns were revealed:

Calculated as a percentage change from the basal rate and compared to the control, 24 hour treatment with 2000 μ g / mL OLTE saw a 29.94% reduction in the effect of oligomycin to inhibit OCR (86.22% compared to 56.28%; p < 0.01; figure 71).



MitoStress assay performed with SeaHorse $XF_e 24$ Flux Analyzer, normalised using protein content measured with Pearce BCA assay to estimate cell count.

Data shown as mean average +/- SD. n = 3.

 ** indicates significance (p < 0.01) compared against control analysed using one-way ANOVA with Dunnett's multiple comparisons.

Figure 71: Effect of Original Liquorice Tea Extract 24 Hour Treatment Followed by Oligomycin Administration on OCR of MCF7 Cell

Calculated as a percentage change from the basal rate and compared to the control, 24 hour treatment with 2000 μ g / mL OLTE saw a 43.2% reduction in the effect of FCCP to increase OCR, remaining close to the initial basal reading (99.82% compared to 143.02%; p < 0.01; figure 72).



MitoStress assay performed with SeaHorse XF_e 24 Flux Analyzer, normalised using protein content measured with Pearce BCA assay to estimate cell count.

Data shown as mean average +/- SD. n = 3.

** indicates significance (p < 0.01) compared against control analysed using one-way ANOVA with Dunnett's multiple comparisons.

Figure 72: Effect of Original Liquorice Tea Extract 24 Hour Treatment Followed by FCCP Administration on OCR of MCF7 Cells

Calculated as a percentage change from the basal rate and compared to the control, 24 hour treatment with 2000 μ g / mL OLTE showed the effect of antimycin and rotenone to reduce OCR was reduced by 32.09% (71.95% compared to 39.86%; p < 0.01; figure 73).



MitoStress assay performed with SeaHorse XF_e 24 Flux Analyzer, normalised using protein content measured with Pearce BCA assay to estimate cell count.

Data shown as mean average +/- SD. n = 3.

** indicates significance (p < 0.01) compared against control analysed using one-way ANOVA with Dunnett's multiple comparisons.

Figure 73: Effect of Original Liquorice Tea Extract 24 Hour Treatment Followed by Antimycin A and Rotenone Administration on OCR of MCF7 Cells

12.4. Summary of Results from Original Liquorice Tea on Respiration in MCF7 Cells

The results can be summarised as follows:

- There was a flattening of the predicted SeaHorse curve in response to all drugs including a minus value of spare capacity leading to a reinterpretation of results as the effect of OLTE on mitochondrial modulating drugs.
- 24 hours treatment with 2000 µg / mL OLTE saw a 29.94% (p < 0.01) decrease in the effect of oligomycin in MCF7 cells
- 24 hours treatment with 2000 μ g / mL OLTE saw a 43.2% (p < 0.01) decrease in the effect of FCCP in MCF7 cells
- 24 hours treatment with 2000 μ g / mL OLTE saw a 32.09% (p < 0.01) decrease in the effect of antimycin A and rotenone in MCF7 cells

Chapter XIII

13. Results from Respiratory Measurements of Original Liquorice Tea Extract in MCF10A Cells

13.1. Effects of Original Liquorice Tea Extract on Mitochondrial Respiration and its Measurement in MCF10A Cells

After 24 hours treatment with OLTE, there appeared to be a similar flattening of the expected curve in MCF10A cells (figure 74) to that seen in the MCF7 cells (figure 63, section 12.1), but the most noticeable difference was a decrease in the basal OCR.



MitoStress assay performed with SeaHorse XF_e 24 Flux Analyzer, normalised using protein content in measured with Pearce BCA assay to estimate cell count. Points shown as mean average +/- SD. n = 3.

Change at 20 minutes shows effect of administering oligomycin to inhibit ATP synthase. Change at 45 minutes shows effect of administering FCCP disrupt proton gradient to reveal maximal & spare capacity. Change at 70 minutes shows effect of administering antimycin A and rotenone to shut down mitochondrial respiration by inhibiting complexes I & III revealing non-mitochondrial respiration.

Figure 74: Results of SeaHorse MitoStress Assay Showing Changes in OCR of MCF10A Cells After Treatment with Original Liquorice Tea Extract

When comparing the basal rates of respiration as a percentage of the control, 24 hours treatment with OLTE showed a significant reduction in basal OCR among the 500 μ g / mL (72.6%; p < 0.01) and 2000 μ g / mL (85.16%; p < 0.01) dose groups (figure 75).



MitoStress assay performed with SeaHorse XF $_{\rm e}$ 24 Flux Analyzer, normalised using protein content in measured with Pearce BCA assay to estimate cell count.

Basal OCR calculated by subtracting OCR after administration of antimycin and rotenone from initial OCR. Data shown as mean average +/- SD compared to untreated control shown as dotted line at 100%. n = 3.

** indicates significance (p < 0.01) compared against untreated control using one-way ANOVA with Dunnett's multiple comparions.

Figure 75: Effect of Original Liquorice Tea Extract 24 Hour Treatment on Basal OCR of MCF10A Cells Shown as Percentage in Relation to Control

When plotted as a percentage change from the basal OCR to see relative effects of the drugs, it was shown that the drop in basal OCR is the main factor and the relative effects of the drugs is the same (figure 76). It can also be seen that there is some possible degradation of the oligomycin which makes the initial drop in OCR after its administration a gradual decline in all groups, rather than the sharp decrease usually expected in a Seahorse MitoStress test (see section 8.6, figure 53)



MitoStress assay performed with SeaHorse XF_e24 Flux Analyzer, normalised using protein content in mg / mL measured with Bradford and BCA assays to estimate cell count shown as % of initial OCR of control. Points shown as mean average +/- SD. n = 3.

Change at 20 minutes shows effect of administering oligomycin to inhibit ATP synthase. Change at 45 minutes shows effect of administering FCCP disrupt proton gradient and reveal spare capacity. Change at 70 minutes shows effect of administering Antimycin and Rotenone to shut down mitochondrial respiration revealing non-mitochondrial respiration.

Figure 76: Results of SeaHorse MitoStress Assay Showing Changes in OCR of MCF10A Cells After Treatment with Original Liquorice Tea Extract

This drop in the basal OCR meant that small differences in the modulations of the drugs produced large proportional differences, leading to a high standard deviation in the 2000 μ g / mL group (142.98% in the ATP and proton leak calculations made from the administration of oligomycin and 650.74% in the maximal and spare capacity calculations made from the administration of FCCP) resulting a logarithmic scale being required to display the spare capacity and no significance being able to be drawn (figures 77-80).



MitoStress assay performed with SeaHorse XF_e24 Flux Analyzer, normalised using protein content measured with Pearce BCA assays to estimate cell count.

ATP related respiration calculated by subtracting OCR after administration of oligomycin from initial OCR and shown as mean average +/- SD. n = 3.

Significance assessed using one way ANOVA with Dunnett's multiple comparisons.

Figure 77: Effect of Original Liquorice Tea Extract 24 Hour Treatment on ATP linked OCR of MCF10A Cells


MitoStress assay performed with SeaHorse XF $_{e}$ 24 Flux Analyzer, normalised using protein content with Pearce BCA assays to estimate cell count.

Proton leak calculated by subtracting OCR after administration of antimycin A & rotenone from OCR after administration of oligomycin and shown as mean average +/- SD. n = 3.

Significance assessed using one way ANOVA with Dunnett's multiple comparisons.

Figure 78: Effect of Original Liquorice Tea Extract 24 Hour Treatment on Mitochondrial Proton Leak in MCF10A Cells



MitoStress assay performed with SeaHorse XF e 24 Flux Analyzer, normalised using protein content in measured with Pearce BCA assay to estimate cell count.

Maximal capacity calculated by subtracting OCR after administration of antimycin A and rotenone from OCR after administration of FCCP and shown as mean average +/- SD. n = 3.

Figure 79: Effect of Original Liquorice Tea Extract 24 Hour Treatment on Mitochondrial Maximal Capacity in MCF10A Cells



MitoStress assay performed with SeaHorse XF $_{\rm e}$ 24 Flux Analyzer, normalised using protein content in measured with Pearce BCA assay to estimate cell count.

Spare capacity calculated by subtracting initial OCR from OCR administration of FCCP and shown as mean average +/- SD. n = 3.

Significance assessed using one way ANOVA with Dunnett's multiple comparisons.

Figure 80: Effect of Original Liquorice Tea Extract 24 Hour Treatment on Mitochondrial Spare Capacity in MCF10A Cells

13.2. Interpretation of the Results of Original Liquorice Tea Extract on Mitochondrial Respiration in MCF10A Cells as Protection Against Mitochondrial Modulating Drugs

When analysed as the effects of the Seahorse drugs on OCR, there was no significant difference of any group when treated for 24 hours with OLTE compared to control (figures 81-83).



MitoStress assay performed with SeaHorse XF_e 24 Flux Analyzer, normalised using protein content measured with Pearce BCA assay to estimate cell count.

Data shown as mean average +/- SD. n = 3.

Significance assessed using one way ANOVA with Dunnett's multiple comparisons.

Figure 81: Effect of Original Liquorice Tea Extract 24 Hour Treatment Followed by Oligomycin on OCR of MCF10A Cells



MitoStress assay performed with SeaHorse XF_e 24 Flux Analyzer, normalised using protein content measured with Pearce BCA assay to estimate cell count.

Data shown as mean average +/- SD. n = 3.

Significance assessed using one way ANOVA with Dunnett's multiple comparisons.

Figure 82: Effect of Original Liquorice Tea Extract 24 Hour Treatment Followed by FCCP on OCR of MCF10A Cells



MitoStress assay performed with SeaHorse XF_e 24 Flux Analyzer, normalised using protein content measured with Pearce BCA assay to estimate cell count.

Significance assessed using one way ANOVA with Dunnett's multiple comparisons.

Figure 83: Effect of Original Liquorice Tea Extract 24 Hour Treatment Followed by Antimycin A and Rotenone on OCR of MCF10A Cells

13.3. Summary of Results from Original Liquorice Tea on Respiration in MCF10A Cells

The results can be summarised as follows:

- 24 hours treatment with 2000 μg / mL OLTE saw an 85.16% (p < 0.01) decrease in the effect of basal OCR of MCF10A cells
- 24 hours treatment with 500 µg / mL OLTE saw an 72.6% (p < 0.01) decrease in the effect of basal OCR of MCF10A cells
- There were no other significant changes in respiration among the MCF10A cells.
- Some apparent differences were due to the decrease in basal respiration producing relatively large changes from the basal rate which led to high standard deviation

Data shown as mean average +/- SD. n = 3.

Chapter XIV

14. Results from Original Liquorice Tea Extract on Drug Resistance

14.1. Effects of Original Liquorice Tea Extract on Oligomycin, FCCP, Antimycin A and Rotenone Induced Perturbations in ROS in MCF7 Cells

24 hour pretreatment with OLTE followed by 10 minutes treatment with oligomycin, FCCP, antimycin A and rotenone saw no significant differences between the liquorice pretreated group and the group which received no pretreatment in MCF7 cells with any of the drugs (figure 84). This experiment was reasoned to not be a suitable test for drug resistance (see section 20.1.2) and subsequently abandoned in favour of the cisplatin resistance assays.



ROS production measured with 2',7'-dichlorodihydrofluorescein diacetate (DCFDA) represented as % change untreated control drawn as dotted line at 100%.

Liquorice treated cells were exposed to 2000 μ g/mL of liquorice for 24 hours before being assayed. Mitochondrial drugs were applied 10 minutes before assay at dosages of:oligomycin 40 μ g / mL, FCCP 20 μ M, antimycin 40 μ M, rotenone 40 μ M.

Data shown as mean +/- SD. n=5.

Significance assessed using unpaired t-tests to compare liquorice treated and non-liquorice treated groups for each drug. Additional liquorice only group with no drug included to assess the effect of liquorice alone on ROS.

Figure 84: Effect of Original Liquorice Tea Extract and Mitochondrial Modulating Drugs on ROS in MCF7 cells

14.2. Effects of Original Liquorice Tea Extract on Cisplatin Induced Reduction of Cell Viability in MCF7 Cells

24 hours pretreatment with 2000 μ g / mL OLTE, followed by 24 hours treatment with cisplatin found no significant difference in cell viability between the liquorice pretreated and cisplatin only groups in MCF7 cells, both when compared as change against a group that received neither treatment (figure 85), or when comparing a liquorice pretreated group against the cisplatin only group (figure 86).



Dose (µg/ ml)

Liquorice groups pretreated with 2000 μ g/ml freeze dried liquorice tea extract for 24 hours prior to exposure to cisplatin for 24 hours and presented as the change in cell viability measured with 3-(4,5-dimethylthiazol-2-yl)-2,5-diphenyltetrazolium bromide (MTT) from the untreated control drawn as dotted line at 100%.

Data shown as shown as mean average +/- SD. n=9.

Significance assessed using unpaired t-test to compare each liquorice pretreated group to its untreated group given the same dosage of cisplatin.

Figure 85: Effect of Original Liquorice Tea Extract Pre-treatment on Cytotoxic Effects of Cisplatin in MCF7 Cells



Liquorice groups treated with 2000 μ g/ml liquorice tea extract for 24 hours prior to exposure to cisplatin for 24 hours and presented as difference between the cisplatin only group and the liquorice & cisplatin groups.

Cell viability measured with 3-(4,5-dimethylthiazol-2-yl)-2,5-diphenyltetrazolium bromide (MTT) represented as % change from group not treated with liquorice and given same dose of cisplatin.

Data shown as box and whisker plot where the box represents the upper and lower quartiles around the median average shown as a line within the box and the whiskers represent the maximum and minimum values with all data points displayed. n=9.

Significance assessed using unpaired t-tests to compare liquorice treated and non-liquorice treated groups for each dosage of cisplatin.

Figure 86: Difference Between Original Liquorice Tea Extract Pre-treatment and No Pre-treatment on the Cytotoxic Effect of Cisplatin in MCF7 Cells

14.3. Effects of Original Liquorice Tea Extract on Cisplatin Induced Reduction of Cell Viability in MCF10 Cells

24 hours pretreatment with 2000 μ g / mL OLTE, followed by 24 hours treatment with cisplatin found no significant difference when comparing the groups as changes from a control not treated with either liquorice or cisplatin (figure 87).



Liquorice groups pretreated with 2000 μ g/ml freeze dried liquorice tea extract for 24 hours prior to exposure to cisplatin for 24 hours and presented as the change in cell viability measured with 3-(4,5-dimethylthiazol-2-yl)-2,5-diphenyltetrazolium bromide (MTT) from the untreated control drawn as dotted line at 100%.

Data shown as shown as mean average +/- SD. n=10.

Significance assessed using unpaired t-test to compare each liquorice pretreated group to its untreated group given the same dosage of cisplatin.

Figure 87: Effect of Original Liquorice Tea Extract Pre-treatment on Cytotoxic Effects of Cisplatin in MCF-10 Cells

When the liquorice pretreated group was comprised directly against the group that received no pretreatment with liquorice, there was an 11.79% (p < 0.05) increase in cell viability among the MCF10A cells treated with 50 μ M cisplatin, and a 8.186% (p < 0.001) increase in cell viability among the MCF10A cells treated with 100 μ M cisplatin (figure 88). Comparing the 50 μ M and 100 μ M groups with an unpaired t-test reveals the effect is not significantly different between groups (3.6%; p < 0.5465).



Liquorice groups treated with 2000 μ g/ml liquorice tea extract for 24 hours prior to exposure to cisplatin for 24 hours and presented as difference between the cisplatin only group and the liquorice & cisplatin groups. n=9.

Cell viability measured with 3-(4,5-dimethylthiazol-2-yl)-2,5-diphenyltetrazolium bromide (MTT) represented as % change from group not treated with liquorice and given same dose of cisplatin.

Data shown as box and whisker plot where the box represents the upper and lower quartiles around the median average shown as a line within the box and the whiskers represent the maximum and minimum values with all data points displayed. n=10.

Significance assessed using unpaired t-tests to compare liquorice treated and non-liquorice treated groups for each dosage of cisplatin. * indicates significance (p<0.05) and *** indicates significance (p<0.001) compared to equivalent dose of cisplatin using an unpaired t-test.

Figure 88: Difference Between Original Liquorice Tea Extract Pre-treatment and No Pre-treatment on the Cytotoxic Effect of Cisplatin in MCF10A Cells

14.4. Summary of Results from Original Liquorice Tea on Drug Resistance

The results can be summarised as follows:

- MCF10A cells pretreated for 24 hours with 2000 µg / mL OLTE followed by 24 hour treatment with 50 µM cisplatin had a 11.79% (p < 0.05) increase in cell viability compared to those who did not receive liquorice pretreatment
- MCF10A cells pretreated for 24 hours with 2000 µg / mL OLTE followed by 24 hour treatment with 100 µM cisplatin had a 8.186% (p < 0.001) increase in cell viability compared to those who did not receive liquorice pretreatment

Chapter XV

15. Results from Basic Mitochondrial Assays on Pre-extracted Liquorice Tea Extract

Only the assays which produced significant results were replicated using XLTE. This included the ROS levels of the MCF7 cells after 3 and 72 hours exposure, the Seahorse assay in both cell lines, and the cytoprotective effect of liquorice pre-treatment against cisplatin induced cytotoxicity of MCF10A cells.

15.1. Effects of Pre-extracted Liquorice Tea Extract on ROS in MCF7 Cells

3 hours treatment with XLTE saw no significant difference in ROS in MCF7 cells (figure 89).



Cell viability measured with 3-(4,5-dimethylthiazol-2-yl)-2,5-diphenyltetrazolium bromide (MTT) represented as % change from untreated control drawn as dotted line at 100%.

Data shown as box and whisker plot where the box represents the upper and lower quartiles around the median average shown as a line within the box and the whiskers represent the maximum and minimum values with all data points displayed. n=8.

Significance assessed using one way ANOVA with Dunnett's multiple comparisons.

Figure 89: Results of DCFDA Assay Showing Changes in ROS 3 Hours After Treatment with Pre-extracted Liquorice Tea Extract

72 hours pretreatment with XLTE saw an 18.7% (p < 0.0001) significant drop in ROS in MCF7 cells (figure 90).



Cell viability measured with 3-(4,5-dimethylthiazol-2-yl)-2,5-diphenyltetrazolium bromide (MTT) represented as % change from untreated control drawn as dotted line at 100%.

Data shown as box and whisker plot where the box represents the upper and lower quartiles around the median average shown as a line within the box and the whiskers represent the maximum and minimum values with all data points displayed. n=8.

Significance assessed using one way ANOVA with Dunnett's multiple comparisons. **** indicates significance (p < 0.0001) compared to an untreated control.

Figure 90: Results of DCFDA Assay Showing Changes in ROS 72 Hours After Treatment with Pre-extracted Liquorice Tea Extract

15.2. Summary of Results from Basic Mitochondrial Assays of Pre-extracted Liquorice Tea

The results can be summarised as follows:

 72 hours treatment with 2000 µg / mL XLTE saw an 18.7% (p < 0.0001) decrease in ROS in MCF7 cells

Chapter XVI

16. Results from Respiratory Measurements of Pre-extracted Liquorice Tea Extract in MCF7 Cells

16.1. Effects of Pre-extracted Liquorice Tea Extract on Mitochondrial Respiration and its Measurement in MCF7 Cells

24 hour treatment of MCF7 cells with XLTE showed no flattening of the expected curve of the Seahorse assay (figure 91).



MitoStress assay performed with SeaHorse XF_e 24 Flux Analyzer, normalised using protein content measured with Pearce BCA assay to estimate cell count. Points shown as mean average +/- SD. n = 3.

Change at 20 minutes shows effect of administering oligomycin to inhibit ATP synthase. Change at 45 minutes shows effect of administering FCCP disrupt proton gradient and reveal spare capacity. Change at 70 minutes shows effect of administering antimycin A and rotenone to shut down mitochondrial respiration revealing non-mitochondrial respiration.

Figure 91: Results of SeaHorse MitoStress Assay Showing Changes in OCR of MCF7 Cells after treatment with Pre-extracted Liquorice Tea Extract

When comparing the basal rates of respiration as a percentage of the control, there was no significant difference in basal OCR (figure 92).



MitoStress assay performed with SeaHorse XF $_{\rm e}$ 24 Flux Analyzer, normalised using protein content in mg / mL measured with Bradford and BCA assays to estimate cell count.

Basal OCR calculated by subtracting OCR after administration of antimycin and rotenone from initial OCR. Data shown as mean average +/- SD compared to untreated control shown as dotted line at 100%. n = 3.

Significance assessed using one way ANOVA with Dunnett's multiple comparisons.

Figure 92: Effect of Pre-extracted Liquorice Tea Extract 24 Hour Treatment on Basal OCR of MCF7 Cells Shown as Percentage in Relation to Control

When plotted as a percentage change from the initial basal rate, it is apparent that there is no significant proportional difference between the groups treated for 24 hours with XLTE and the control (figure 93).



MitoStress assay performed with SeaHorse XF_e24 Flux Analyzer, normalised using protein content in mg / mL measured with Pearce BCA assays to estimate cell count shown as % of initial OCR of control. n = 3.

Change at 20 minutes shows effect of administering oligomycin to inhibit ATP synthase. Change at 45 minutes shows effect of administering FCCP disrupt proton gradient and reveal spare capacity. Change at 70 minutes shows effect of administering antimycin A and rotenone to shut down mitochondrial respiration revealing non-mitochondrial respiration.

Figure 93: Results of SeaHorse MitoStress Assay Showing Changes in OCR of MCF7 Cells after treatment with Pre-extracted Liquorice Tea Extract as percentage of basal OCR

There were also no significant differences in any of the reported measures used for the Seahorse MitoStress assay (figures 94-97).



MitoStress assay performed with SeaHorse XF_e 24 Flux Analyzer, normalised using protein content measured with Pearce BCA assays to estimate cell count.

ATP related respiration calculated by subtracting OCR after administration of oligomycin from initial OCR and shown as mean average +/- SD. n = 3.

Significance assessed using one way ANOVA with Dunnett's multiple comparisons.

Figure 94: Effect of Pre-extracted Liquorice Tea Extract 24 Hour Treatment on ATP Linked OCR of MCF7 Cells



MitoStress assay performed with SeaHorse XF_e 24 Flux Analyzer, normalised using protein content with Pearce BCA assays to estimate cell count.

Proton leak calculated by subtracting OCR after administration of antimycin A & rotenone from OCR after administration of oligomycin and shown as mean average +/- SD. n = 3.

Significance assessed using one way ANOVA with Dunnett's multiple comparisons.

Figure 95: Effect of Pre-extracted Liquorice Tea Extract 24 Hour Treatment on Mitochondrial Proton Leak in MCF7 Cells



MitoStress assay performed with SeaHorse XF $_{\rm e}$ 24 Flux Analyzer, normalised using protein content in measured with Pearce BCA assay to estimate cell count.

Maximal capacity calculated by subtracting OCR after administration of antimycin A and rotenone from OCR after administration of FCCP and shown as mean average +/- SD. n = 3.

Significance assessed using one way ANOVA with Dunnett's multiple comparisons.

Figure 96: Effect of Pre-extracted Liquorice Tea Extract 24 hour treatment on Mitochondrial Maximal Capacity in MCF7 Cells



MitoStress assay performed with SeaHorse XF e 24 Flux Analyzer, normalised using protein content in measured with Pearce BCA assay to estimate cell count.

Spare capacity calculated by subtracting initial OCR from OCR administration of FCCP and shown as mean average +/- SD. n = 3.

Significance assessed using one way ANOVA with Dunnett's multiple comparisons.

Figure 97: Effect of Pre-extracted Liquorice Tea Extract 24 Hour Treatment on Mitochondrial Spare Capacity in MCF7 Cells

16.2. Interpretation of the Results of Pre-extracted Liquorice Tea Extract on Mitochondrial Respiration in MCF7 Cells as Protection Against Mitochondrial Modulating Drugs

24 hour treatment of MCF7 cells with XLTE showed no significant difference in terms of the effect the extract was having on the mitochondrial modulating drugs used in the Seahorse MitoStress assay (figures 98-100).



MitoStress assay performed with SeaHorse XF $_{\rm e}$ 24 Flux Analyzer, normalised using protein content measured with Pearce BCA assay to estimate cell count.

Data shown as mean average +/- SD. n = 3.

Significance assessed using one way ANOVA with Dunnett's multiple comparisons.

Figure 98: Effect of Pre-extracted Liquorice Tea Extract 24 Hour Treatment followed by Oligomycin on OCR of MCF7 Cells



MitoStress assay performed with SeaHorse XF $_{\rm e}$ 24 Flux Analyzer, normalised using protein content measured with Pearce BCA assay to estimate cell count.

Data shown as mean average +/- SD. n = 3.

Significance assessed using one way ANOVA with Dunnett's multiple comparisons.

Figure 99: Effect of Pre-extracted Liquorice Tea Extract 24 Hour Treatment Followed by FCCP on OCR of MCF7 Cells



MitoStress assay performed with SeaHorse XF_e 24 Flux Analyzer, normalised using protein content measured with Pearce BCA assay to estimate cell count.

Data shown as mean average +/- SD. n = 3.

Significance assessed using one way ANOVA with Dunnett's multiple comparisons.

Figure 100: Effect of Pre-extracted Liquorice Tea Extract 24 Hour Treatment Followed by Antimycin A and Rotenone on OCR of MCF7 Cells

16.3. Summary of Results from Respiratory Measurements of Pre-extracted Liquorice Tea Extract in MCF7 Cells

The results can be summarised as follows:

 24 hours treatment with XLTE showed no significant effects on respiratory measurements in MCF7 cells.

Chapter XVII

17. Results from Respiratory Measurements of Pre-extracted Liquorice Tea Extract in MCF10A Cells

17.1. Effects of Pre-extracted Liquorice Tea Extract on Mitochondrial Respiration and its Measurement in MCF10A Cells

24 hour treatment of MCF10A cells with XLTE showed a reduction in basal OCR in all dosage groups (figure 101)



MitoStress assay performed with SeaHorse XF_e 24 Flux Analyzer, normalised using protein content in measured with Pearce BCA assay to estimate cell count. Points shown as mean average +/- SD. n = 3.

Change at 20 minutes shows effect of administering oligomycin to inhibit ATP synthase. Change at 45 minutes shows effect of administering FCCP disrupt proton gradient to reveal maximal & spare capacity. Change at 70 minutes shows effect of administering antimycin A and rotenone to shut down mitochondrial respiration by inhibiting complexes I & III revealing non-mitochondrial respiration.

Figure 101: Results of Seahorse MitoStress Assay Showing Changes in OCR of MCF10A cells After Treatment with Pre-extracted Liquorice Tea Extract

24 hour treatment of MCF10A cells with XLTE saw a reduction in basal OCR compared as a percentage against the control (figure 102). At 10 μ g / mL the basal OCR was reduced by 21.12% (p < 0.05); at 500 μ g / mL the basal OCR was reduced by 53.52% (p < 0.0001) and at 2000 μ g / mL the basal OCR was reduced by 56.41% (p < 0.0001).



MitoStress assay performed with SeaHorse XF _e24 Flux Analyzer, normalised using protein content in mg / mL measured with Bradford and BCA assays to estimate cell count.

Basal OCR calculated by subtracting OCR after administration of antimycin and rotenone from initial OCR. Data shown as mean average +/- SD compared to untreated control shown as dotted line at 100%. n = 3.

Significance assessed using one way ANOVA with Dunnett's multiple comparisons.* indicates significance (p < 0.05) and **** indicates significance (p < 0.0001) compared against untreated control.

Figure 102: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment on Basal OCR of MCF10A Cells Shown as Percentage in Relation to Control

No significant effects were seen in any of the other respiratory measurements (figures 103 - 110).



MitoStress assay performed with SeaHorse XF_e24 Flux Analyzer, normalised using protein content in mg / mL measured with Bradford and BCA assays to estimate cell count shown as % of initial OCR. Points shown as mean average +/- SD. n = 3.

Change at 20 minutes shows effect of administering oligomycin to inhibit ATP synthase. Change at 45 minutes shows effect of administering FCCP disrupt proton gradient and reveal spare capacity. Change at 70 minutes shows effect of administering Antimycin and Rotenone to shut down mitochondrial respiration revealing non-mitochondrial respiration.

Figure 103: Results of Seahorse MitoStress Assay Showing Changes in OCR of MCF10A cells After Treatment with Pre-extracted Liquorice Tea Extract as a Percentage of Basal OCR



MitoStress assay performed with SeaHorse XF $_{e}$ 24 Flux Analyzer, normalised using protein content measured with Pearce BCA assays to estimate cell count.

ATP related respiration calculated by subtracting OCR after administration of oligomycin from initial OCR and shown as mean average +/- SD. n = 3.

Significance assessed using one way ANOVA with Dunnett's multiple comparisons.

Figure 104: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment on ATP Linked OCR of MCF10A Cells



MitoStress assay performed with SeaHorse XF $_{e}$ 24 Flux Analyzer, normalised using protein content with Pearce BCA assays to estimate cell count.

Proton leak calculated by subtracting OCR after administration of antimycin A & rotenone from OCR after administration of oligomycin and shown as mean average +/- SD. n = 3.

Significance assessed using one way ANOVA with Dunnett's multiple comparisons.

Figure 105: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment on Mitochondrial Proton Leak in MCF10A Cells



MitoStress assay performed with SeaHorse XF _e 24 Flux Analyzer, normalised using protein content in measured with Pearce BCA assay to estimate cell count.

Maximal capacity calculated by subtracting OCR after administration of antimycin A and rotenone from OCR after administration of FCCP and shown as mean average +/- SD. n = 3.

Significance assessed using one way ANOVA with Dunnett's multiple comparisons.

Figure 106: Effect of Liquorice Tea 24 Hour Treatment on Mitochondrial Maximal Capacity in MCF10A Cells



MitoStress assay performed with SeaHorse XF e 24 Flux Analyzer, normalised using protein content in measured with Pearce BCA assay to estimate cell count.

Spare capacity calculated by subtracting initial OCR from OCR administration of FCCP and shown as mean average +/- SD. n = 3.

Significance assessed using one way ANOVA with Dunnett's multiple comparisons.

Figure 107: Effect of Liquorice Tea 24 Hour Treatment on Mitochondrial Spare Capacity in MCF10A Cells

17.2. Interpretation of the Results of Pre-extracted Liquorice Tea Extract on Mitochondrial Respiration in MCF7 Cells as Protection Against Mitochondrial Modulating Drugs

24 hour treatment of MCF10A cells with XLTE showed no significant difference in the effect of oligomycin and FCCP on OCR (figures 108-109).



MitoStress assay performed with SeaHorse XF_e 24 Flux Analyzer, normalised using protein content measured with Pearce BCA assay to estimate cell count.

Data shown as mean average +/- SD. n = 3.

Significance assessed using one way ANOVA with Dunnett's multiple comparisons.

Figure 108: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Oligomycin on OCR of MCF10A Cells





Data shown as mean average +/- SD. n = 3.

Significance assessed using one way ANOVA with Dunnett's multiple comparisons.

Figure 109: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by FCCP on OCR of MCF7 Cells

24 hour treatment of MCF10A cells with 10 μ g / mL XLTE showed an 8.85% increase in OCR (65.12% compared to 56.27% in control; p < 0.05; figure 110) after administration of antimycin A and rotenone, implying a reduction in the effect of antimycin and rotenone to inhibit OCR.



MitoStress assay performed with SeaHorse XF_e 24 Flux Analyzer, normalised using protein content measured with Pearce BCA assay to estimate cell count.

Data shown as mean average +/- SD. n = 3.

Significance assessed using one way ANOVA with Dunnett's multiple comparisons. * indicates significance (p < 0.05) compared against untreated control.

Figure 110: Effect of Pre-extracted Liquorice Tea 24 Hour Treatment Followed by Antimycin A and Rotenone on OCR of MCF10A Cells

17.3. Summary of Results from Respiratory Measurements of Pre-extracted Liquorice Tea in MCF10A Cells

The results can be summarised as follows:

- 24 hours pretreatment with 2000 μg / mL XLTE showed a 56.41% decrease (p < 0.0001) in basal OCR of MCF10A cells
- 24 hours pretreatment with 500 μg / mL XLTE showed a 53.52% decrease (p < 0.0001) in basal OCR of MCF10A cells
- 24 hours pretreatment with 10 μg / mL XLTE showed a 21.12% decrease (p < 0.05) in basal OCR of MCF10A cells
- 24 hours pretreatment with 10 µg / mL XLTE showed a 8.85% reduction (p < 0.05) in the effect of antimycin A and rotenone to reduce OCR in MCF10A cells.

Chapter XVIII

18. Results from Pre-extracted Liquorice Tea Extract on Drug Resistance in MCF10A Cells

18.1. Effects of Pre-extracted Liquorice Tea Extract on Cisplatin Induced Reduction of Cell Viability in MCF10A Cells

No significant difference in cell viability was seen in MCF10A cells after 24 hours pretreatment with XLTE followed by 24 hours cisplatin treatment (figures 111 - 112).



Liquorice groups pretreated with 2000 μ g/ml freeze dried liquorice tea extract for 24 hours prior to exposure to cisplatin for 24 hours and presented as the change in cell viability measured with 3-(4,5-dimethylthiazol-2-yl)-2,5-diphenyltetrazolium bromide (MTT) from the untreated control drawn as dotted line at 100%.

Data shown as shown as mean average +/- SD. n=8.

Significance assessed using unpaired t-test to compare each liquorice pretreated group to its untreated group given the same dosage of cisplatin.

Figure 111: Effect of Pre-extracted Liquorice Tea Extract Pre-treatment on Cytotoxic Effect of Cisplatin in MCF10A Cells



Liquorice groups treated with 2000 μ g/ml liquorice tea extract for 24 hours prior to exposure to cisplatin for 24 hours and presented as difference between the cisplatin only group and the liquorice & cisplatin groups.

Cell viability measured with 3-(4,5-dimethylthiazol-2-yl)-2,5-diphenyltetrazolium bromide (MTT) represented as % change from group not treated with liquorice and given same dose of cisplatin.

Data shown as box and whisker plot where the box represents the upper and lower quartiles around the median average shown as a line within the box and the whiskers represent the maximum and minimum values with all data points displayed. n=8.

Significance assessed using unpaired t-tests to compare liquorice treated and non-liquorice treated groups for each dosage of cisplatin.

Figure 112: Difference Between Pre-extracted Liquorice Tea Extract Pre-treatment and No Pre-treatment on the Cytotoxic effect of Cisplatin in MCF10A Cells

18.2. Summary of Results from Pre-extracted Liquorice Tea Extract in Drug Resistance

The results can be summarised as follows:

 24 hours pretreatment with 2000 µg / mL XLTE did not produce any significant effect on the cytotoxic effects of cisplatin.

Chapter XIX

19. Results from Other Methods of Adulteration Detection

19.1. Visual Inspection

The rehydrated samples were subjected to visual inspection and revealed that the XLTE was slightly more clear and less yellow in colour than OLTE (figure 113, left). However, this could not be considered a definitive test as a different sample from the same batch under slightly different light made this difference lessen and even disappear entirely (figure 113, right)



Figure 106: Visual Comparison of Original and Pre-extracted Liquorice Tea Extracts.

In each picture OLTE is on the left and the XLTE is on the right.
19.2. High Performance Thin Layer Chromatography

Two sets of each sample were analysed with HPTLC (figs 114 - 118). The individual numbered tracks represent:

- 1. Raw "3 Licorice Tea" extracted in 70:30 v/v ethanol:water as per HPTLC Association guidelines for analysis of liquorice.
- 2. *G. glabra* sample from the University of Westminster Polyclinic extracted in 70:30 v/v ethanol:water
- 3. *G. uralensis* geno-verified botanical reference sample of *G. uralensis* (Brion Research Institute, Taiwan) extracted in 70:30 v/v ethanol:water
- 4. OLTE sample
- 5. XLTE sample at equal strength to OLTE
- 6. XLTE sample at 20% strength of OLTE so simulate the strength of a second extraction from a pre-extracted teabag.

Two sets of tracks are displayed. The left set of tracks show results from using 2 μ L of sample, while right tracks show results from using 5 μ L of sample.

	1 licorice tea	2 G. glabra	3 G. uralensis	4 original samp	5 adulterated s	6 adulterated n	1 licorice tea	2 G. glabra	3 G. uralensis	4 original samp	5 adulterated s	6 adulterated n		
0.9							_	-	-					0.9
0.8														- 0.8
0.7														- 0.7
0.6														- 0.6
0.5														- 0.5
0.4														- 0.4
0.3														- 0.3
0.2														- 0.2
0.1														0.1

Figure 114: HPTLC Results After Development Under Visual Light. No bands are visible prior to viewing under UV light or derivatization.



Figure 115: HPTLC Results After Development Under UV 254 nm



Figure 116: HPTLC Results After Development Under UV 366 nm



Figure 117: HPTLC Results After Derivatization Under Visible Light



Figure 118: HPTLC Results After Derivatization Under UV 366 nm

19.2.1. HPTLC Comparison of Species

Specific marker compounds to differentiate *G. uralensis* and *G. glabra* (Kondo et al., 2007) could be used to identify the species and extraction method. *G. uralensis* (track 3), could be differentiated from *G. glabra* by a band at 0.5 Rf after development and viewing with UV 366 nm, and at 0.65 Rf after derivatization (figure 119) which could be conjectured to be liquiritin that is only present in *G. uralensis* (Rizzato et al. 2017).



Figure 119: HPTLC Comparison of *G. glabra* (left) and *G. uralensis* (right) Left: Developed and viewed under UV 366 nm Middle: Derivatized and viewed under visual light Right: Derivatized and viewed under UV 366 nm

19.2.2. HPTLC Comparison of Extraction Method

7:3 ethanol / water (v/v) extracts of *G. glabra* (tracks 1 & 2) produced a red band at 0.45 Rf after development and viewed under UV 366 nm, which could differentiate them from water extracts (tracks 4, 5 & 6). The most notable are tracks 1 and 4 (figure 120) because these used the exact same starting material, the Pukka "3 Licorice" tea bag, except track 1 was extracted using the ethanol solution, while track 4 was the freeze dried aqueous extraction sample used in the OLTE mitochondrial analysis. This band was speculated to be glabridin, a species-specific secondary metabolite of *G. glabra* and insoluble in water (Tian, Yan & Row, 2008).



Figure 120: HPTLC Comparison of Ethanol (left) and Water (right) Extracts of *G. glabra* After Development and Viewed Under UV 366 nm

19.2.3. HPTLC Comparison of Source

It was even possible to differentiate between the sample taken from the University of Westminster Polyclinic dispensary (track 2), and the Pukka "3 Licorice" tea bags (track 1) due to the presence of many additional brown bands in the polyclinic sample after derivatization (figure 121).



Figure 121: HPTLC Comparison of *G. glabra* from a Commercial Tea Brand (left) and a Herbal Medicine Clinic (right) After Derivatization under UV 366 nm

19.2.4. HPTLC Comparison of Original and Pre-extracted Material

It was possible to differentiate the OLTE (track 4) from the XLTE, when prepared at the dosage that would be expected from the same amount of starting material (i.e. if the tea bag was reused, yielding approximately 20% of the freeze dried material). All bars were visibly faded (track 6) at all points (figure 122).



Figure 122: HPTLC comparison of Original (left) and Pre-extracted (right) Liquorice Tea Extracts at Natural Strength if Extraction was Taken Directly from an Pre-extracted Tea Bag

Left: Developed and viewed under UV 366 nm

Middle: Derivatized and viewed under visual light

Right: Derivatized and viewed under UV 366 nm

When OLTE (track 4) and XLTE prepared at an identical dosage of freeze dried material (track 5), as was used in the mitochondrial functional analysis, the differences were slight and may be within a normal range of variation and were even harder to differentiate after derivatization (figure 123).



Figure 123: HPTLC Comparison of Original (left) and Pre-extracted (right) Liquorice Tea Extracts at the Same Dosage of Freeze Dried Material Left: Developed and viewed under UV 366 nm Middle: Derivatized and viewed under visual light Right: Derivatized and viewed under UV 366 nm Additionally, both OLTE and XLTE at equal strength showed a quenching zone at 0.27 Rf (figure 124) suggesting that both had contained glycyrrhizin, the main constituent commonly searched for in liquorice extracts. OLTE appeared stronger but not to a degree that may be immediately apparent.



Figure 124: HPTLC Comparison of Original (left) and Pre-extracted (right) Liquorice Tea Extracts at the Same Dosage of Freeze Dried Material after Development Under UV 366 Showing Probable Glycyrrhizin Elution Zone

19.3. Summary of Results from Other Methods of Adulteration Detection

The results can be summarised as follows:

- Visual inspection could detect a slight difference but one that was subtle, dependent on lighting and easily overlooked
- HPTLC analysis easily differentiated between species and between an alcohol and water extracts but was much less effective at differentiating between the OLTE and XLTE when administered at the same dosage of freeze dried material
- The difference between the OLTE and XLTE was even less apparent after derivatization and observation under UV 366 nm

Discussion

Chapter XX

20. Discussion of the Results

The results of this study will now be discussed. First, the reasons for the effects observed in this study will be discussed, followed by how these relate to the detection of adulteration and the hypothesis being tested. An examination of the limitations of mitochondrial functional analysis in general and of this study in particular will then be undertaken. Next, the implications these findings may have for other fields of research and practice will be explored. Finally, future work that could be undertaken to expand mitochondrial functional analysis into a tool that can be used to develop an industry standard for functional quality evaluation will be suggested.

20.1. The Effects of Liquorice Tea Extracts on Reactive Oxygen Species

Three significant results were obtained in relation to ROS, the first being a rise in MC7 cells in the 500 μ g / mL and 2000 μ g / mL groups after 3 hours exposure to the OLTE, the second and third being significant reductions in ROS after 72 hours in both the OLTE and XLTE at 2000 μ g / mL doses.

20.1.1. Increases in Reactive Oxygen Species

The first significant result obtained was an increase in ROS among the MCF7 cells treated for 24 hours with 500 μ g / mL and 2000 μ g / mL OLTE for 3 hours. ROS increased by 16.51% (p < 0.01) in the 500 μ g / mL group and 14.88% (p < 0.05) in the 2000 μ g / mL group (figure 55, section 11.2). This is unusual in

relation to the literature as very few studies report any components of liquorice increasing cellular ROS. Two notable exceptions are Fiore et al. (2004) and Singh, Pal and Darokar (2015).

Fiore et al. (2004) observed that glycyrrhetinic acid can cause an increase in H_2O_2 within the mitochondrial matrix in isolated rat liver mitochondria. It achieved this by interfering with the iron-sulphur clusters in Complex I which generates O_2^{\bullet} that is removed by SOD to become H_2O_2 . In the presence of Fe²⁺ this become •OH which are highly reactive and oxidise thiol groups in the presence of Ca²⁺ leading to the opening of the mPTP. This mechanism has been exploited in many attempts to utilise glycyrrhetinic acid to deliver chemotherapeutic drugs directly to mitochondria in liver cancer (Speciale et al., 2022). Glycyrrhetinic acid is the aglycone form of glycyrrhizin and primarily formed by enzymatic hydrolysis of the gut microbiota (Hattori et al., 1983) and present only in small amounts in the plant itself (Graebin, 2018) making a comparison complex, however some sources state that glycyrrhizin and glycyrrhetinic acid as being pharmacologically equivalent (Wahab et al., 2021) and if the dose is equivalent or greater then it may be possible that this effect is achieved.

Fiore et al. (2004) achieved their results based on 10 μ M of glycyrrhetinic acid and, although it is not possible to get a precise molarity on the OLTE due to it being a whole root extract whose precise constituent proportions are unknown, an estimate can be attempted based on the fact that Tian, Yan & Row (2008) managed to extract 2.44 mg / g glycyrrhizin using a water extraction. The total yield of freeze dried material was 377.33 mg / g, making 2000 μ g (2 mg) equivalent to 5.3 mg of raw material:

Dose (mg) \div yield (mg / g) = initial starting material (g) 2 \div 377.33 = 0.0053 g = 5.3 mg

which equates to 12.932 µg of glycyrrhizin:

Extraction rate (mg / g) x starting material (g) = glycyrrhizin content (mg) 2.44 x 0.0053 = 0.012932 mg or 12.932 µg The molecular weight of glycyrrhizin is 822.9 g/mol (NCBI, 2023a), so by using a concentration conversion formula (Abcam, 2024) we can calulate:

Therefore, assuming our yields were the same as Tian, Yan & Row, (2008), we can estimate that approximately 10.6 µM glycyrrhizin was used, close to the concentration that Fiore et al. (2004) used to achieve an increase in ROS in rat liver mitochondria with glycyrrhetinic acid. This does not account for the fact that Tian, Yan & Row (2008) extracted using room temperature water for 240 minutes instead of using boiling water for 15 minutes as was used in this study. This was designed to make their results comparable to other solvents which would evaporate at higher temperatures but ignores the fact that most aqueous herb extractions occur under heated conditions. If we instead take the minimum glycyrrhizin content found in the root as our starting point, assuming that the insoluble material would offset the fact that not all the glycyrrhizin contained in the root would be extracted, then we can expect a minimum of 2% glycyrrhizin (Ministry of Health and Welfare, 2016, p.136), which equates to 7.55 mg / g in our extract, or 3x the amount used by Fiore et al. (2004) with some sources stating that glycyrrhizin content can be as high as 24% (Esmaili & Karami, 2022) increasing this much further. This would place our estimates for a glycyrrhizin molarity at anywhere between 30-360 µM. Either way, we should have had at least as much glycyrrhizin in our extract as Fiore et al. (2004) had glycyrrhetinic acid and maybe more.

Although the dosages used in this study may have been similar, or even higher that that used by Fiore et al. (2004), there are other dissimilarities between the studies. Fiore et al., (2004) used isolated rat mitochondria and so the cellular uptake of OLTE in MCF7 cells has to be accounted for too. Also, the fact that glycyrrhizin and glycyrrhetinic acid are often treated as pharmacologically equivalent (Wahab et al., 2021) refers to the fact that glycyrrhizin, once metabolised, becomes glycyrrhetinic acid which is pharmacologically active and not that they have identical effects *in vitro*. The experiment of Fiore et al. (2004) has not been reproduced using glycyrrhizin in MCF7 cells and an important argument against this mechanism being activated in the current study is that the increase in H_2O_2 was accompanied by a drop in $\Delta\Psi$ m which was not observed at any time point in this study.

Singh, Pal & Darokar (2015) studied the effects of glabridin on multidrug-resistant *Staphylococcus aureus* finding an antibacterial effect at higher concentrations (up to 25 μ g / mL) through the generation of ROS and increased NO leading to lipid peroxidation, detected by malondialdehyde (MDA), and DNA fragmentation. Although glabridin is present in *G. glabra* and considered one of its marker compounds (Rizatto et al., 2017; Avula et al., 2022), it has poor water solubility (Tian, Yan & Row, 2008) so little would be present in the water extracts used in this study. This can be speculatively confirmed by the HPTLC chromatograms under UV 366 nm (figure 116) that revealed a red band present at 0.45Rf in the 70% ethanol extracts (tracks 1 & 2) which is absent in the aqueous extracts used in the mitochondrial studies (tracks 4 & 5). Small amounts of glabridin are extracted in water, equating to 0.18 mg / g compared to 0.93 mg / g in ethanol (Tian, Yan & Row, 2008), which implies that if 0.18 mg of the 377.33 mg extracted from 1 g of liquorice tea is glabridin, then, using the formula above, in 2000 μ g / mL of extract:

mg glabridin in 2 mg of extract = $(0.18 \div 0.0053) \times 1000 = 0.00094$ = 0.954 µg

If we can expect to find 0.95 μ g / mL of glabridin if our 2000 μ g / mL extract, it is unlikely to have exerted a pro-oxidant effect, and may even have exerted an antioxidant effect, as was found by Singh, Pal & Darokar (2015) at lower doses. A complete breakdown of the proportions of all components obtained from an infused aqueous extraction of *G. glabra* could not be found and so these doses can remain only estimates, but a full spectrum quantitative analysis of the OLTE using HPLC, MS, NMR or a hyphenated technique could be undertaken to gain better quantification of the individual constituents.

20.1.2. Decreases in Reactive Oxygen Species

Another possibility emerges from the subsequent results. After 24 hours treatment with 2000 μ g / mL OLTE, ROS measurements with the MCF7 cells had returned to levels equal to control, and after 72 hours treatment with 2000 μ g / mL OLTE, ROS measurements had reduced by 14.73% (p < 0.05) (figure 57, section 11.2). This effect was also seen among the MCF7 cells treated with the adulterated liquorice tea extract, where 2000 μ g / mL treatment for 72 hours saw an even greater reduction in ROS of 18.7% (p < 0.0001) (figure 90, section 15.1). This concurs with the majority of literature on liquorice having an antioxidant effect in cancer cells, primarily due to its phenolic content (Pastorino et al., 2018, Sharma, Katiyar & Agrawal, 2018) but with some activities attributed to glycyrrhizin too (Ageeva et al., 2022).

In the same study that researched pro-oxidant effects of high doses of glabridin, antioxidant effects were found from low doses of 0.78 to 3.12 μ g / mL (Singh, Pal & Darokar, 2015), for which the estimated proportion of glabridin present in the tea samples of 0.95 μ g / mL would correspond. This has been supported by *in vivo* research which showed a 20% reduction in oxidised low density lipoproteins in blood samples taken from human volunteers supplemented with 60 mg glabridin for 12 months (Carmeli & Fogelman, 2009). In the present study, the enhanced antioxidant activity of XLTE could be explained by this effect. The initial water extraction will have removed the most polar compounds from the raw root pieces, meaning that the second extraction may have proportionally higher amounts of non-polar substances like glabridin than the initial extraction, thus exerting a stronger antioxidant effect. If the initial extraction was approximately 0.95 μ g / mL, then the second extraction would be closer to the optimal range (1.56 μ g / mL) that Singh, Pal & Darokar (2015) found for antioxidant activity.

Studies of the whole root extract through different extraction methods have shown that the antioxidant effects are not solely dependent on glabridin. A comparison of different solvent extracts of *G. glabra* showed that aqueous extractions had comparable activity on DPPH, NO radicals, $O_2^{\bullet-}$ and H_2O_2 to less polar solvent extractions, despite achieving a phenolic extraction rate of 2.58 ± 0.014 mg gallic acid equivalents / g, compared to 79.297 ± 0.776 achieved from methanol (Hejazi et al., 2017). This suggests the antioxidant effect in aqueous may be due to components with a greater affinity for water.

Licochalcones have also been implicated in the antioxidant potential of liquorice. Sources differ as to which licochalcones are present in G. glabra with some finding only licochalcone B (Rizatto et al., 2017; Avula et al., 2022), while others find only licochalcone A (Cerulli et al., 2022) and others find both (Fu et al., 2013). Licochalcones A and B have been reported to have antioxidant effects against lipopolysaccharide induced oxidative stress in RAW 264.7 cells with licochalcone B having the superior action in a DCFDA cellular assay, but licochalcone A having a superior radical scavenging potential in an ABTS⁺ assay (Fu et al., 2013). ABTS is a peroxidase substrate that, in the presence of H_2O_2 is oxidised to yield ABTS⁺ whose colour can be measured by spectrophotometry at 600-750 nm (Siddeeg et al., 2021). The reduction in absorbance reflects the ability of the antioxidant to donate a hydrogen atom and neutralise the radical, implying that licochalcone A scavenges ROS through the hydrogen atom transfer method (HAT) as opposed to a single electron transfer (SET) method. This is confirmed by an UPLC-MS/MS evaluation of a licochalcone A DPPH assay which found that licochalcone A favours a HAT mechanism of ROS scavenging in weakly non-polar solutions which approximate organic conditions (Liang et al., 2018). Meanwhile licochalcone B seems to operate through a cellular mechanism, activating the keap1-Nrf2 pathway to raise endogenous antioxidant defences including SOD and GSH (Zhou, B. et al., 2021; Lv et al., 2018) and HO-1 (Huang et al., 2023; Su et al., 2018).

Another flavonoid that could be implicated in the antioxidant capabilities of *G. glabra* is liquiritin apioside. There is very little research on this specific molecule but it has been shown to increase levels of SOD and GSH in a mouse cigarette smoke induced lung disease model (Guan et al., 2012). Some further studies enforce this result by studying liquiritin apioside in relation to cough and finding it

likely to be the primary constituent for the antitussive effect (Kuang et al., 2018; Wei et al., 2020) but these did not investigate antioxidant mechanisms. There is more research on liquiritin, suggesting an inhibitory action on NF-κB through the AMPK/SIRT1 pathway (Tang et al., 2022) and the ERK1/2 pathways along with activation of Nrf2 (Li et al., 2020), restoring deficits in SOD, GSH and CAT (Sun et al., 2010; Zhou et al., 2017). However, liquiritin and liquiritin apioside are not identical molecules and so any comparison of their mechanism should be tentative.

Glycyrrhizin, the main constituent of liquorice water extracts, has also been implicated in antioxidant mechanisms despite being a triterpenoid saponin instead of a phenol. Although there is wide consensus that it has antioxidant potential, there is no consensus on its mechanism. Some argue it is unable to scavenge •OH and O₂•- but is able to scavenge DPPH (Imai et al., 2013; Račková et al., 2007; Polyakov et al., 2006), while others argue that it can scavenge $O_2^{\bullet-}$ and $\bullet OH$ but not DPPH (Kato et al., 2008; Cheel et al., 2010). Others still observed cellular mechanisms such as the increase in GSH and the inhibition of NOX in neutrophils (Beskina et al., 2007). One mechanism that may connect these conflicting results is presented by Ageeva et al. (2022) who, based on earlier findings by Ghandi et al. (2004), used a chemically induced dynamic nuclear polarisation technique to demonstrate that the ROS scavenging effects of glycyrrhizin may be due an ability to capture solvated electrons created by UV irradiated naproxen before they bind to O₂, creating a glycyrrhizin radical anion. They also suggest that glycyrrhizin may act as a trap for harmful short-lived radicals. This implies glycyrrhizin is acting in a different way to most antioxidants that deradicalise ROS through the donation of H or electrons to create stable compounds, or at least less harmful radicals (Leopoldini, Russo & Toscano, 2011) and would explain its inconsistent results in antioxidant assays that depend on HAT or SET from the antioxidant to neutralise the assay reagent (Munteanu & Apetrei, 2021).

20.1.3. A Combined Explanation for Modulations in Reactive Oxygen Species

The concept of hormesis could connect these modulations in ROS. The available literature examining the effects of liquorice extracts to raise ROS remains unconvincing when applied to the quantities of compounds likely to be available in the extracts, but there is much more evidence for an antioxidant effect of liquorice extracts. Cells require a certain amount of oxidative stress in order to maintain cell signalling, proliferation and differentiation functions, with cancer cells tending to favour an environment of slightly raised oxidative stress (Hayes, Dinkova-Kostova & Tew, 2020). A sustained drop in ROS induces a state of reductive stress where these functions can be compromised (Xiao & Loscalzo, 2020). One of the hormetic mechanisms proposed is overcompensation in response to a perturbation in cellular homeostasis (Calabrese et al., 2016). The introduction of a large quantity of antioxidant material which threatens to reduce this optimal environment may initiate an attempt to overcompensate, seen in the rise in ROS in MCF7 cells after 3 hours treatment and returning to homeostasis at 24 hours. Over time this compensation cannot be maintained and after 72 hours treatment with 2000 µg / mL OLTE it succumbs and ROS is reduced, although at 72 hours treatment with 500 µg / mL OLTE, homeostasis appears to be maintained.

The lack of an increase in ROS after 3 hours treatment with XLTE but subsequent drop after 72 hours treatment with 2000 μ g / mL XLTE to below that of the OLTE group may indicate that there is proportionally more of the antioxidant substance in this batch. This could prevent the MCF7 cells from being able to initially overcompensate and only maintain their basal rate of ROS. Further assays at time points between the ones chosen could elucidate when the rise and decline in ROS happens in each extract, and if it happens in the XLTE at all, but at a time point not measured in this study.

The lack of changes in MCF10A cells at any time point suggests that this antioxidant effect applies primarily to sources of ROS generated by cancer. In ordinary cells the majority of ROS is generated as a byproduct of mitochondrial respiration which is initially generated as $O_2^{\bullet-}$ within the mitochondrial matrix before being transformed by SOD to H_2O_2 where it can diffuse across the IMM and OMM into the cytoplasm (see sections 3.1.1 & 3.1.2). Cancer cells have mDNA mutations that cause them to be metabolically reprogrammed and favour aerobic glycolysis as their preferred energy source (Kang, Lee & Lee, 2015). In order to maintain the redox balance optimal for proliferation and growth, additional ROS is generated through an increased expression of NOX which use heme groups to transfer electrons across membranes and transform O₂ into O₂^{•-} (Moloney & Cotter, 2018). This is then rapidly converted into H_2O_2 . Although primarily associated with being located on the plasma membrane and generating extracellular ROS, NOX are also located inside the cell too: on the ER, mitochondrial membranes, the nuclear membrane and specialised domains within the cell (Skonieczna et al., 2017). This makes them them the main source of $O_2^{\bullet-}$ within the cytoplasm and in the external cellular environment. If the theory presented by Ageeva et al. (2022) is correct and glycyrrhizin can act as an electron acceptor and trap for short-lived radicals then we can speculate that the drop in ROS seen only in MCF7 cells is due to glycyrrhizin being able to trap the short-lived O₂^{•-} generated by NOX in this cell line, thus preventing its conversion to H₂O₂ and subsequent modifications. This mechanism would also have little effect on H₂O₂ itself, not being a radical itself and so having no electron to donate and which will be the main form of ROS that reaches the cytoplasm from mitochondrial sources in MCF10A cells.

The lack of changes in $\Delta \Psi m$ would suggest that the mitochondrial function is largely unchanged and that this is primarily due to a ROS scavenging effect. Upregulation of cellular antioxidant defences could be possible too, but these tend to only be observed in studies of liquorice extracts where an initial stressor has depleted them (Zhou et al., 2021; Lv et al., 2018; Huang et al., 2023; Su et al., 2018; Sun et al., 2010; Zhou et al., 2017).

Further experiments could elucidate this mechanism. Assays more specific to particular species of ROS could help determine the relative balance of ROS species. If glycyrrhizin is capable of capturing electrons from $O_2^{\bullet-}$ then an assay specific to this form of ROS such as mitoSOX to measure the mitochondrial $O_2^{\bullet-}$

(Invitrogen, 2022) and a general $O_2^{\bullet-}$ detection kit (Abcam, 2022b) could determine if this ROS species is affected and whether it is from mitochondrial sites or other sources. Since ROS is an important signalling molecule for proliferation, with higher ROS levels associated with hyperproliferation in cancer cells (Hayes, Dinkova-Kostova & Tew, 2020), future tests examining this mechanism could look at the cell viability of MCF7 cells after 72 hours treatment with 2000 µg / mL liquorice tea extracts, as well as longer time frames, to determine if this antioxidant mechanism can have an anti-proliferative effect. Although ROS is a stimulator for cancer cell proliferation, antioxidant therapies have been paradoxically associated with increasing proliferation due to a reduction in ROS, DNA damage and inactivation of the pro-apoptotic function of p53 (Sayin et al., 2014).

20.2. The Effects of Liquorice Tea Extracts on Respiratory Measurements

Some of the most striking results were from the Seahorse Flux Analyzer which produced a reduction in the effect of the mitochondrial modulating drugs used in the assay among the MCF7 cells, and a reduction in basal OCR among the MCF10A cells. After 24 hours treatment with OLTE a significant reduction in the effects of the mitochondrial modulating drugs was seen in the MCF7 cells (see chapter 10) and the basal OCR was observed among the 500 µg / mL (72.6%; p < 0.01) and 2000 µg / mL (85.16%; p < 0.01) groups in MCF10A cells. This was surprising since there was no change in cell viability, $\Delta\Psi$ m or Ca²⁺ in either cell line and no change in ROS in the MCF10A cells (see section 11). Furthermore, this effect was also seen in the MCF10A cells treated for 24 hours with XLTE which appeared to be dose dependent with a 21.12% (p < 0.05) decrease in OCR at 10 µg / mL, 53.52% (p < 0.0001) decrease in OCR at 500 µg / mL and a 56.41% (p < 0.0001) decrease at 2000 µg / mL suggesting this effect was not exhausted by the second extraction.

A search of the literature revealed that despite liquorice having the third most articles written about it in relation to mitochondria (table 8), there was no analysis of the whole root extract using the Seahorse Flux Analyzer.

PubMed Search Term : Mitochondria AND " "	Results
Ginkgo biloba	99
Panax ginseng	84
Glycyrrhiza	66
Astragalus	64
Scutellaria baicalensis	52
Withania somnifera	37
Zingiber officinale	36
Panax notoginseng	36
Camellia sinensis	31
Cinnamomum	29
Sargassum	28
Centella asiatica	26
Garcinia mangostana	20
Coptis	19
Momordica charantia	19
Andrographis	18
Scutellaria barbata	14
Zingiber officinale	13
Piper longum	12

Table 8: Papers Discussing Mitochondria and Specific Herbs.

Searches were conducted on 18th September 2023. List is based on those in Mani, Swargiary & Singh (2020) and modified to include searches for herbs not directly involved in cancer prevention or treatment. Where multiple species or species names are used interchangeably, only the genus was searched for. Where only one specific species has therapeutic uses, or where different species have very different effects, the genus and species name was used.

Both herbs that have more published data than liquorice have had Seahorse evaluations of at least some of their components, with one study on gingko (*G. biloba*; Wang, Z. et al., 2021) and four on asian ginseng (*P. ginseng*; Huang et al., 2019; Liu, W. et al., 2021; Huang et al., 2023; Shin et al., 2021). Other herbs which have attracted far less mitochondrial interest have also been assayed with the Seahorse Flux Analyser including astragalus (*Astragalus mongholicus*, Bunge, *Fabaceae*; Huang et al., 2018), cinnamon (*C. cassia*; Li, X. et al., 2021), gotu kola (*Centella asiatica*, L. Urban, *Apiaceae*; Gray et al., 2018) and bitter gourd (*Momordica charantia*, L., *Cucurbitaceae*; Sur et al., 2019). It was speculated that other researchers may have encountered the same issues and assumed they were erroneous, as was initially thought in this study, and decided not to publish.

Of the data available, some individual compounds found in liquorice have been examined using the Seahorse Flux Analyzer. Jin, X. et al., (2022) found that 12 hours exposure to 100 µM glycyrrhetinic acid down-regulated OXPHOS and fatty acid β-oxidation in in HeLa cervical cancer cells, producing a similar Seahorse curve with inhibited basal, ATP-related and maximal respiration (ibid.: figure 1, graph D). This is similar to the high dose of liquorice tea extract used in this study (figures 63, 74 & 101) with a lower basal rate and flatter curve. They identified the mechanism as inhibition of mitochondrial serine hydroxymethyltransferase 2 (SHMT2) by targeting the folate-binding pocket of SHMT2 and competing with tetrahydrofolic acid, resulting in inhibited proliferation but did not repeat the experiments on normal cell lines. This could explain some of the results in the MCF7 group which had similar reduced ATP-related and maximal OCR but the lack of comparison to normal cells presents a problem because this study found an even greater reduction of basal OCR in the non-cancerous MCF10A cells (figures 74 & 101), yet without any apparent inhibition of function. While SMHT2 is present in both cancerous and non-cancerous cells, its expression supports cancer growth and elevated expression is associated with poor prognosis (Zeng et al., 2021) making this an unsatisfactory model for explaining the drop in basal OCR in MCF10A cells.

Searching beyond liquorice studies, another study looking at Ginsenoside compound K from *P. ginseng* in an oxygen-glucose deprivation / reperfusion (OGD/R) was found with a similar flattened curve and lower basal OCR to that found in this study (Huang et al., 2023). This used rat adrenal medulla pheochromocytoma (PC12) cells subjected to OGD/R to imitate IRI during stroke. The aspect to note is that the group that shows a pattern similar to the high doses of liquorice (figures 63, 74, 91 & 101) is the OGD/R model without any treatment, and to a lesser extent, the low dose group (ibid.; figure 1, graph B). This could also imply that high doses of liquorice are having an inhibitory effect on mitochondrial respiration, similar to OGD/R, but, as a form of IRI, this should induce mitochondrial dysfunction leading to increases in ROS and Ca²⁺ as well as loss of $\Delta\Psi$ m and cell death (He, Z. et al., 2020), none of which were observed in either cell line.

20.2.1. Possible Explanations from Clinical Uses of Liquorice

The literature does include some reports that liquorice has protective effects against IRI. Licochalcone B and D have been reported to have a cardioprotective effect against IRI in rat hearts (Han et al., 2014; Yuan et al. 2015), while glycyrrhizin and magnesium isoglycyrrhizinate been attributed have hepatoprotective roles in IRI models (Kou et al. 2020; Huang, Qin & Lu, 2014). Liquiritin has also been investigated for its potential in preserving the blood brain barrier during stroke (Li, M. et al., 2021). All these studies found reduced infarct size, apoptosis, inflammation and oxidative stress mediated by different pathways depending on the investigations being performed but usually sharing an increase in SOD and GSH, and reduced proinflammatory markers such as MDA and TNF-α.

This relates directly to a clinical usage of liquorice where it is the chief herb in the CHM formula "Prepared Liquorice Decoction" (*Zhi Gan Cao Tang* 炙甘草湯) which has been used since the Han dynasty for irregular pulse and palpitations, and remains one of the most popular formulas for cardiovascular disease in

Taiwan today (Wang, L. et al., 2022), with its reported protective effects against IRI attributed primarily to glycyrrhizin through the inhibition of apoptosis in cardiomyocytes injured by high glucose and hypoxia, and reduction of infarct size in a diabetic myocardial infarction mouse model (Hu, M. et al., 2023). Cardiovascular disease may have been indicated in ancient Europe too with one the few indications given for liquorice in Anglo-saxon England being "chest pain" (van Arsdall, 2002, p.213), or "breast pain" (Pollington, 2008, p.269). This indication survived into the 17th century with Culpeper indicating liquorice for "griefs of the breast" (Culpeper, 1653). Pretreating with mitochondrial targeted antioxidants has been reported to be an effective method of cardioprotection from IRI through reduction in cytochrome c release and subsequent apoptosis (Adlam et al., 2005) but these do not reduce OCR. This finding could suggest that liquorice is able to induce a form of hypoxia that preconditions the cells in order to improve survival in future hypoxic states. Preconditioning the heart against IRI is considered one of the most potent forms of protection from IRI, achieved by inhibition of the mPTP, thus preventing apoptosis (Cadenas, 2018), a mechanism similar to that induced by mitochondrial targeted antioxidants.

20.2.2. Possible Explanations from Mechanistic Research

During hypoxia ROS increases. This is due to metabolic stress which triggers Ca^{2+} and causes NO release from NO synthase which acts as a vasodilator to return normal perfusion to the cells during brief periods of hypoxia but also suppresses Complex IV of the ETC, thereby reducing the use O_2 as a terminal electron acceptor (Burtscher et al., 2022). This causes accumulation of electrons within the ETC which result in increased $O_2^{\bullet-}$ from complex III, succinate dehydrogenase at Complex II being changed to fumarate reductase to enable fumarate to act as an electron acceptor from reverse electron flow (Paddenberg et al., 2003) and bursts of H_2O_2 being generated from Complex I of the ETC which act as an O_2 sensitive signalling molecule (Okoye, Koren & Wojtovich, 2023). This triggers further cellular responses to hypoxia such as Ca^{2+} release from the ER (Desireddi et al., 2010) to initiate more cellular responses, or organism wide

response if the tissues involved form part of the carotid body. HIF-1a, which is stabilised in hypoxic conditions, induces the mitochondria to undergo alterations in protein composition of the complexes in the ETC, a loss of mitochondrial mass through fission and mitophagy and changes in morphology including lower cristae density to compensate for the reduced O_2 as a terminal electron acceptor while maintaining essential ROS signalling functions (Fuhrmann & Brüne, 2017).

The lack of change in ROS or Ca²⁺ at 24 hours and the eventual drop in ROS argue against hypoxia occurring. Normal $\Delta\Psi$ m was maintained at all time points, and ROS and Ca²⁺ was also equal to control at 24 hours (figures 56, 59 & 61). Even if the cells were surviving, the lower metabolic activity induced by these changes should result in less MTT being converted to formazan crystals and therefore the appearance of lower cell viability in this assay (Zhang & Cox, 1996; Liu & Dalgleish, 2009) but this was not observed (figure 54). Instead, the drop in OCR among the MCF10A cells and the reduction of effect in OCR in response to the mitochondrial modulating drugs in MCF7 cells in the Seahorse assay, combined with no significant changes in ROS, $\Delta\Psi$ m, cell viability and Ca²⁺ implies that the mitochondria are respiring normally but using less O₂. This is only possible if there is an alternative terminal electron acceptor, or if the O₂ used as a terminal electron acceptor can be recycled.

The most obvious explanation would be that liquorice increases CAT activity which converts $2H_2O_2$ into $2H_2O + O_2$ which could then diffuse back across the IMM be reused by the cells, thus reducing the amount of O_2 consumed. Whole liquorice root extract has shown increased CAT in *Caenorhabditis elegans* (Maupas, 1900), *Rhabditidae* (Reigada et al., 2019) and Licochalcone A administration has shown increased CAT *in vitro* (Chen et al., 2017) but the majority of assays only show increased CAT in relation to a stressor that has depleted it. Glycyrrhizin has been observed to restore levels of CAT that are depleted in an allergic rhinitis mouse model (Li et al., 2011) and glabridin has been found to raise levels of CAT in monocytes, but only after their depletion through glucose stress (Yehuda et al., 2011). Liquorice may also reduce levels of CAT, as was seen in a rabbit model of vibration stress (Oganesyan, 2002). Since the literature is conflicting, the most direct method of testing this would be to measure

CAT activity in MCF7 and MCF10A cells after exposure to liquorice using a CAT assay kit (Cayman Chemical, 2023).

Another explanation would be that the liquorice is able to provide an alternative electron acceptor to O_2 . This seems unlikely but is not impossible. Alternative electron acceptors exist in anaerobic organisms and even mammalian mitochondria can use alternative substances such as fumarate as final electron acceptors in limited O_2 environments (Paddenberg et al., 2003; Spinelli et al., 2021). This is due to them retaining some of the anaerobic ancestral architecture in their respiratory complexes (Glass, Elbon & Williams, 2023). Some exogenous substances, such as methylene blue, have also been demonstrated to act as electron bridges, bypassing complexes I and III of the ETC by acting as electron receivers and donors, rerouting electrons directly from NADH to cytochrome c (Wen et al., 2011; Tucker, Lu & Zhang, 2018; Sváb et al., 2021) and the antioxidant study of glycyrrhizin by Ageeva et al. (2022) demonstrated the possibility that glycyrrhizin could act as an electron acceptor and prevent the binding of solvated electrons to O_2 .

The main challenge for this model is that the IMM is almost impermeable so it would be necessary for glycyrrhizin to enter the mitochondrial matrix, or the electrons to leave. This could be made possible by the mechanism proposed by Fiore et al. (2004) that glycyrrhetinic acid can trigger the opening of the mPTP. Although generally associated with permanent opening to release cytochrome c and initiate apoptosis, a transient opening of the mPTP for physiological regulatory functions has been suggested (Endlicher et al., 2023), enabling the release of Ca²⁺ and ROS into the cytosol which can then act as signalling molecules, while also accelerating electron flow and lowering mtROS to restore homeostasis. This could enable the transition of electrons, or $O_2^{\bullet-}$ to glycyrrhizin outside the mitochondria whereupon it could act as an electron acceptor, or receive the electron from $O_2^{\bullet-}$, returning it back to O₂. It is even possible that glycyrrhizin could enter the mitochondrial matrix, or at least bind to the IMM through its observed ability to incorporate itself into membranes and enhance their permeability (Selyutina & Polyakov, 2019) enabling the direct capture of electrons leaving complex IV (figure 125). Mitochondrial isolation could be used to detect whether this activity is occurring from liquorice compounds in the cytoplasm or the mitochondria themselves but this process often leads to contamination (Liao et al., 2020). Alternatively, radioactive isotopes have been used for a long time to detect interaction of compounds with mitochondria (Siekevitz, 1961), so a radioactive isotope of glycyrrhizin could be synthesised which can then be traced.





This would explain how the only measure affected was OCR and the response to drugs that should be detectable through modulations in OCR, and why the decrease in OCR is only seen among the MCF10A cells. As non-cancerous cells, MCF10A use OXPHOS more predominantly than the cancerous MCF7 cells which prefer glycolysis even in the presence of oxygen (Hanahan & Weinberg, 2011), enabling them to adapt to the hypoxic environment of tumours (Zheng, 2012). This would also explain why the cells do not appear to be suffering the damaging consequences of reperfusion: if the liquorice were inducing hypoxia, then washing and returning them to normal media would induce IRI and be detected in rising Ca²⁺, ROS and apoptosis from the sudden reinitiation of OXPHOS (figure 31, section 3.1.6.2) but if the cells have been functioning normally on reduced O_2 and the compounds from liquorice responsible for this change were still present within the cell, then only OCR would be affected.

These could explain the unusual results of the SeaHorse experiment due to the measure being OCR and the cells using an alternative substance for their respiration or being able to recycle some of their O₂. One might expect this to produce a drop in ROS as measured by DCFDA but the fluctuation between increasing ROS which causes the opening of the mPTP to restore mitochondrial homeostasis (Endlicher et al., 2023) and the antioxidant function of glycyrrhizin acting on the ROS leaving the mPTP, or directly capturing electrons, could cancel each other out, or at least produce inconsistent changes making significant results difficult to obtain.

This could also explain the drop seen in ROS among the MCF7 cells at the 2000 μ g / mL dose in both extracts due to differences in this cell line. Cancer cells produce additional O₂^{•-} through NOX (see section 20.1.3) and glycyrrhizin may be neutralising this ROS through electron capture after 72 hours but not affecting the glycolytic favouring metabolism of MCF7 cells. Meanwhile in the OXPHOS favouring MCF10As, the electrons captured may be primarily from complex IV of the ETC or recycling O₂^{•-} into O₂ resulting in lowered OCR. This would also explain the improvements seen in IRI models and the clinical use of liquorice in cardiovascular disease: if the cells are able to function closer to normal in a hypoxic environment by consuming less O₂ then they would experience less reperfusion injury when O₂ is returned.

20.2.3. Testing the Theoretical Mechanism

Many experiments can be designed to test this theoretical mechanism. One of the first should test the obvious explanation, that liquorice is interfering with the Seahorse sensors. The cells are washed twice before entering the Seahorse Flux Analyzer so there should only be what is inside the cells and extruded after washing or attached firmly enough to the outside to resist washing and diffuse into the media after soaking. Performing a spectroscopic analysis of the media after assaying could reveal how much of each component of liquorice is in the media after the assay is complete. Following this the assay could be repeated once more but with some wells filled with media spiked with liquorice at the concentration that was found in the Seahorse media during the assay and comparing the O₂ levels with blank controls that receive only media. The Seahorse Flux Analyzer measures OCR, which must be by a comparison of two O₂ readings, so either the device would require recalibration to just take a single O₂ reading, or untreated cells could be used that are given the Seahorse media spiked with the traces of liquorice. A simpler alternative could be to use the media from some treated cells after washing and preparing for the Seahorse assay. To do this, two Seahorse plates would be prepared, one with treated and one with untreated cells. The treated cells would be washed and prepared as in preparation for the Seahorse assay. After a period of incubation equal to the time taken to perform the assay (1-3 hours; usually 45 minutes to 1 hour in a non CO₂ incubator while the MitoStress assay drugs are prepared and 2 hours to perform the actual assay), their media could be removed, filter-sterilised to avoid contamination with any detached treated cells, and applied to the active arm of the washed and untreated cells. These then undergo the MitoStress assay and any unusual fluctuations in the OCR compared to the control group must then be connected to liquorice components that have been effluxed into the media in the time taken to perform the assay.

Once it is certain that the results are not equipment failure, the next test should be an alternative way to measure mitochondrial respiration to compare with the OCR readings, through the direct measurement of the respiratory end product, ATP. Alternative tests to measure ATP concentrations exist (Rajendran et al., 2016) but one of the most sensitive and reliable techniques is the bioluminescent luciferin–luciferase reaction (Morciano et al., 2017; Morciano et al., 2019). Luciferase is an enzyme derived from the North American firefly, *Photinus pyralis* L. *Lampyridae*, which uses ATP to generate its characteristic yellow-green light. When applied to cells, along with the substrate D-luciferin and magnesium ions, it oxidises the substrate into oxyluciferin to produce a luminous flash at a peak emission of 560 nm, whose brightness is proportional to the amount of ATP present and can be detected with luminescence detectors. The limitation of this technique is that it does not differentiate between the ATP of the cells and those from other substances, which may include those from the herbal materials, and which may also have their own luminescence (AIB Staff, 2013). It will also not

differentiate between ATP generated by glycolysis and OXPHOS. This could also be determined through a glycolysis stress test using the Seahorse Flux Analyzer. This assay works in the same way as the MitoStress assay but by injecting different drugs in the sequence and measuring the acidification rate from the generation of lactic acid (Agilent Technologies, 2019b). Initially, antimycin A and rotenone are injected to inhibit OXPHOS and the rate of acidification from glycolysis is measured. This is then followed by an injection of 2-deoxy-D-glucose which inhibits glycolysis giving a set of values that can be used to calculate the glycolytic acidification from other potential sources (figure 126).



Time (minutes)

Figure 126: Seahorse Glycolytic Rate Test Example (created with BioRender.com)

This could suffer from the same problems as the MitoStress assay, since it utilises antimycin and rotenone which could be affected by components in liquorice (see the following section 20.3.1). Alternatively, glucose and lactose levels can be measured with a spectrophotometer or bioanalyzer, or the uptake of a glucose analogue such as 2-[N-(7-nitrobenz-2-oxa-1,3-diaxol-4-yl)amino]-2-deoxy-glucose (2-NBDG) (TeSlaa & Teitell, 2014). These may be useful if interference with the Seahorse sensors or drugs from high levels of liquorice is suspected, although the glycolysis stress test is using different sensors than the OCR assay.

One of the final ways that could be employed to test this mechanism would be actual exposure of cells to hypoxia to see if liquorice treated cells can survive for longer. The research into IRI discussed above can be brought to bear indirectly on this, but the mechanisms involved are more complex than simply O_2 deprivation (see section 3.1.6.2 and figure 31). Attempting to analyse the cells under continuous hypoxic conditions would be challenging and a proxy measurement, such as measuring the activation of HIF-1a could be problematic due to the possibility of components of liquorice inhibiting this transcription factor (Park et al., 2021; Zhou et al., 2022). An in vitro model of IRI cell injury is presented by Li, M. et al. (2021) which involves culturing cells in a basic media, without FBS under hypoxic conditions (1% O₂, 5% CO₂, 94% N₂) for 12 hours, before replacing the media with a complete medium containing FBS and culturing under normal conditions (5% CO₂, 95% air) for a further 8 hours. By exposing cells to various dosages of liquorice for varying times and observing the effect when exposed to hypoxic conditions and subsequent reperfusion, it would be possible to evaluate if liquorice has a genuine effect on IRI and determine its mechanisms more precisely.

20.3. Drug Resistance Assays

The effect of 2000 μ g / mL OLTE to protect MCF10A cells from 24 hours treatment with cisplatin was measured through an improvement in cell viability by 8.186% (p < 0.001) among the 100 μ M cisplatin group and 11.79% (p < 0.05) in the 50 μ M cisplatin group (figure 88, section 14.3). This assay was developed after

receiving the first respiratory results from MCF7 cells (chapter 12) which produced such unusual results, including a spare capacity of -9.59% (p < 0.01) (figure 69, section 12.1), that they demanded an alternative explanation.

20.3.1. Detoxifying Effects of Liquorice

The results obtained from the MCF7 respiratory assays were reminiscent of the traditional CHM use of liquorice as a detoxifying agent (Bensky et al., 2004, p.733; Chen & Chen, 200, p.868). This led to a reinterpretation of the OCR data as modulating the effect of the drugs on the cells (section 12.3). This resulted in discovering that 24 hours treatment with 2000 μ g / mL OLTE reduced the decline in OCR caused by oligomycin by 29.94% (p < 0.01; figure 71), reduced the increase in OCR caused by FCCP by 43.2%% (p < 0.01; figure 72) to remain close to the basal rate (99.82%) and reduced the effect of antimycin A and rotenone by 32.09% (p < 0.01; figure 73). Investigation for the cause of this effect revealed both chemical and biological mechanisms in the literature.

Chemical interactions at the stage of preparation and decoction have been investigated. Liquiritin, found primarily in Chinese liquorice (G. uralensis) was shown to form a complex with aconitine from aconite (A. carmichaelii) reducing the free aconitine in decoctions (Peter et al., 2013). A review of studies (Jiang et al., 2020), including Chinese literature, confirmed this finding that liquorice reduces the toxicity of aconite (A. carmichaelii) decoctions by forming chelate precipitates with the the toxic ester-series alkaloids while also reviewing the interactions of liquorice on several other herbs. Strychnine and brucine concentrations of strychnine nut (Strychnos nux-vomica, L., Loganiaceae) preparations were reduced, also by forming insoluble precipitates. Not all herbal compounds were reduced though, and liquorice increased concentrations of certain therapeutic agents from peony root (Paeonia lactiflora, Pall., Paeoniaceae) including paeoniflorin, oxidized paeoniflorin, benzoyl paeoniflorin and albiflorin. In ginger (Z. officinale) both effects were observed together, with liquorice increasing the extraction of 8-gingerol and 6-shogaol from ginger while reducing 6-gingerol that is considered a slightly toxic component and responsible for the "acrid and drying"

nature of ginger in CHM. Liquorice also increased the ephedrine methylephedrine of ephedra (*Ephedra sinica*, Stapf, *Ephedraceae*) extractions while reducing pseudoephedrine and hydrochloride isomers of ephedrine by forming complex salts. Two other studies found that liquorice interacted with the berberine content of Chinese Goldenthread root (*Coptis chinensis*, Franch., *Ranunculaceae*), forming a sediment that lowers intestinal absorption, both reducing the efficacy of this component but also reducing the digestive side effects associated with it (Zhang, H. et al., 2019; Zhang, J.Q. et al., 2020). While these are of relevance to clinical practice, the drugs used in the Seahorse assay were not prepared with the liquorice extract and so are not able to explain the results found in this study.

Biological mechanisms for a detoxifying effect of liquorice can be divided into four types. Effects were reported on Cytochrome P450 (CYP450) liver enzymes, glucuronidation, drug transporter proteins and direct effects on mitochondria and their associated pathways.

Reviews of the evidence for interaction with CYP450 enzymes suggests that various components of liquorice do modulate several CYP450 enzymes but that the effect is inconsistent and contradictory depending on the administrations used in different studies (Wang et al., 2013). This mechanism is also not relevant to the effects seen in this study since the liquorice does not undergo metabolism by CYP450 enzymes before being introduced to the media.

Glucuronidation remains a possible mechanism but available literature showed that while glycyrrhizin and glycyrrhetinic acid could induce glucuronidation, it could also show inhibitory activity under different circumstances, with glycyrrhetinic acid having more potent inhibitory action (Feng, Ding & Qiu, 2015). This may be due to glycyrrhetinic acid being a substrate for hepatic uridine 5'-diphospho-glucuronosyltransferase that converts it to glycyrrhetinic acid 3-O-glucuronide, responsible for its toxic effects of pseudohyperaldosteronism (Koyama et al., 2017) by competing for aldosterone binding sites. Since the main site of this glucuronidation reaction is also the liver, it is not applicable to the cultures of breast cells that were used in the mitochondrial functional analysis.

The evidence related to drug transporter proteins are the most relevant to this study. They can be expressed on a variety of cells, potentially affecting cell culture work where the substrates are not metabolised by liver enzymes before being introduced to the cells. The review articles had contradictory findings on the effects on drug transporter proteins with Wang et al. (2013) concluding that glycyrrhizin inhibits P-glycoprotein (P-gp) and glycyrrhetinic acid activates it, while Feng, Ding & Qiu (2015) reported the opposite. The most recent study found that several compounds in liquorice (glycyrrhetinic acid, liquiritin, liquiritigenin, isoliquiritin, isoliquiritigenin and licochalcone A) upregulated ATP Binding Cassette (ABC) efflux transporters P-gp, BCRP, and MRP2 and improved the transport of fluorescent Rhodamine 123 from the cells via P-gp, suggesting this to be a generic mechanism by which liquorice could provide protection against a variety of toxins (He et al., 2019). Although these are usually expressed on the cell membrane, they are dependent on ATP to function and on mitochondria for maximum efficiency (Giddings et al, 2021) and have been detected on mitochondria where they engage in biosynthesis, management of oxidative stress and chemoresistance (Schaedler et al., 2015; Zutz et al., 2009), including BCRP (Kobuchi et al., 2012; Zhang, H. et al., 2021). However, of the substances studied by He et al. (2019), glycyrrhetinic acid is the metabolised form of glycyrrhizin so suffers from the uncertainty that glycyrrhizin will have the same effect, while liquiritin and isoliquiritin are found only in *G. uralensis* (Avula et al., 2022), leaving only liquiritigenin, isoliquiritigenin and licochalcone A as potentially directly comparable.

Liquorice appears to have direct effects on mitochondrial pathways that attenuate the toxicity of other substances. Zhang, M. et al. (2019) discovered that liquorice reduced brucine induced nephrotoxicity in rats given *Semen Strychnine* (*Strychnos nux-vomica*, L., *Loganiaceae*) by reducing oxidative stress and activating STAT3 which combined, suppressed the mitochondrial apoptosis pathway. Upadhyay, Mantha & Dhiman (2020) also discovered that liquorice extract was able to protect cardiac cells from doxorubicin induced toxicity. Kim et al. (2004) and Lee et al. (2009) found that liquorice had a protective effect against cadmium poisoning by inhibiting the translocation of pro-apoptotic Bad proteins to the mitochondria. Rashedinia et al. (2019) found a similar effect with aluminium

toxicity in nerve cells, suggesting liquorice could maintain mitochondrial function, reducing ROS and increasing ATP production and induce biogenesis. Yan et al. (2016) also found an indirect effect of glycyrrhizin on mitochondrial ROS induced apoptosis of hepatocytes induced by paracetamol overdose through the inhibition of TNF-a. However, the interaction of liquorice with mitochondria and other drugs can also have the opposite effect. Chen et al. (2020) found the reason for the traditional incompatibility between liquorice and *Radix Euphorbiae Pekinensis* (*Euphorbia pekinensis*, Rupr., *Euphorbiaceae*) was because the pair increased ROS production and activated apoptosis through the intrinsic pathway.

Some studies also revealed a specific protective effect on some of the drugs used in the Seahorse Flux Analyzer. Glabridin was shown to alleviate the toxicity of antimycin A in MC3T3-E1 cells by scavenging ROS and O₂^{•-} generated through the inhibition of Complex III, prevented nitrotyrosine increase and thioredoxin reductase inactivation and restored phosphoinositide 3-kinase, promoting cell survival (Choi, 2011). However, this component was likely not extracted well in the tea samples due to its relative insolubility in water (discussed in section 20.1.1). Glycyrrhizin demonstrated a protective effect in vivo against a rotenone-induced model of Parkinson's disease by restoring SOD, GSH and CAT, thereby reducing MDA, and lowering pro-inflammatory cytokines IL-1^β, IL-6, and TNF- α (Ojha et al., 2016). This was subsequently investigated in vitro (Karthikkeyan et al., 2020; Karthikkeyan, G., Prabhu et al., 2021; Karthikkeyan, G., Pervaje et al., 2021; Karthikkeyan et al., 2022) finding the signalling pathway involved to be primarily inhibition of ERK-1/2 hyperphosphorylation preventing the activation of Bax and Bak and subsequent intrinsic apoptosis (see figure 26, section 3.1.4). Both of these drugs are used in the Seahorse assay, suggesting the possibility that the high doses of liquorice used may be having a protective effect against the drugs being used, confounding the results, and raising the possibility that it could be affecting oligomycin and FCCP too.

20.3.2. Development of the Drug Resistance Assays

The discovery of these potential mechanisms prompted the development of the drug resistance assays. Only the 2000 μ g / mL dose of liquorice was used due to this producing significant modulations of the drug effects in the MCF7 Seahorse assay. Initial investigations utilised the same drugs employed in the Seahorse assay, but measured with DCFDA, to acquire a broad measure of changes in oxidative stress from which to investigate further but did not produce any significant results (figure 84) and gave the appearance that liquorice enhanced the reductive stress induced by the drugs. However, there were several flaws in this assay which could be responsible for these results. The first is the complexity of the assay itself, involving the treatment of 8 different groups (four drugs plus four drugs after liquorice exposure), in addition to controls, which had to all be washed in order to ensure the liquorice was not chemically interacting with the drugs, within 10 minutes. Several repeats failed to generate any results, decided by a lack of any effect from the drug-only groups and were presumably from user error due to this tight timeframe.

The second confounding factor is the antioxidant effects of liquorice (see section 20.1) which could create the illusion of having an additive effect to the drugs when measured using DCFDA. Antimycin A inhibits complex III and rotenone inhibits complex I, which are the main producers of ROS in the ETC (Bleier & Dröse, 2013). Meanwhile, FCCP is known as an uncoupling agent and uncoupling is a physiological mechanism used to reduce the production of ROS (Hass & Barnstable, 2021). Therefore, the antioxidant properties of liquorice could give the appearance of enhancing the effects of these drugs when measured with DCFDA when it is actually functioning as an antioxidant.

Another reason is that a "one size fits all" approach to the timing of the drug administrations may not have been appropriate. Oligomycin has been reported to have no effect on neuronal survival after 30 minutes (Chen et al., 2020) and inhibits OXPHOS after 1 hour (Hao et al., 2010). Therefore, a 10 minute window to observe effects may be sufficient to produce changes in OCR but it may not be
sufficient to see effects from other assays and the same could be true of antimycin A and rotenone. A preferable approach would be to use simpler individual assays optimised for each drug. The first of these could have been to use FCCP as recommended as a positive control for the detection of $\Delta\Psi m$ modulations in a TMRE assay (Abcam, 2022) compared to a liquorice and FCCP group. This could be followed by separate assays for oligomycin, antimycin A and rotenone optimised to detect their specific effects based on examples found in the literature.

Instead, it was decided to use cisplatin, a drug with known cytotoxic effects through induction of mtDNA lesions (Oliveira et al., 1995), causing increased ROS (Choi et al., 2015) and initiating apoptosis primarily through the mitochondrially mediated intrinsic pathway (Ghosh, 2019), whose toxicity could be easily measured through an MTT assay to assess cell viability after 24 hours exposure. Previous and current members of the laboratory team had experience using this drug and its reliability in measuring mitochondrial apoptosis (Mould, 2019; Henley, 2015; Henley et al., 2017) so advice could be sought on its execution. These previous studies had researched "priming," a mechanism whereby increasing oxidative stress in cancer cells without affecting their non-cancerous analogues enhanced the chemotherapeutic potential of cisplatin without increasing overall cytotoxicity. This assay would test for an opposing effect: that liquorice pretreatment would reduce the cytotoxicity of cisplatin treated cells. The dose of OLTE remained at 2000 µg / mL due to this producing a protective role against the drugs in the Seahorse assay.

20.3.3. Cisplatin Resistance Assay Results

Pretreatment with 2000 μ g / mL OLTE followed by 24 hours treatment with cisplatin produced no effect in MCF7 cells (figure 86) but revealed a small but consistently higher cell viability in the MCF10A cells treated with 50 μ M and 100 μ M cisplatin than those not pretreated with 2000 μ g / mL OLTE (figure 88). Those treated with 50 μ M cisplatin saw an 11.79% increase (p < 0.05) and those treated with 100 μ M cisplatin saw a 8.186% increase (p < 0.001). The difference between the protection conferred at 50 μ M and 100 μ M cisplatin was not significant (3.6%;

p < 0.5465), suggesting that this cytoprotective effect is more likely to activate as the dosage of the cytotoxic agent increases but once it is activated, the result is the same.

This result was the opposite of that discovered in the Seahorse respiratory assays where the MCF7 cells displayed a resistance to the drugs while the MCF10A cells showed only a drop in basal OCR. It can be speculated that this is connected to the reasons behind the drop on OCR among the MCF10A cells discussed in 20.2 potentially distorting the respiratory readings and making them an unreliable marker of the true changes in mitochondrial activity.

The cisplatin resistance effects only being seen among the MCF10A cells can be explained through the mechanisms described above. The study by He et al. (2019), revealing an increased expression of ABC efflux transporters induced by several components of liquorice, provides the most likely explanation. Cisplatin resistance has been associated with expression of P-gp (Li et al., 2018), BCRP and MRP2 (Herraez et al., 2012) that were upregulated in response to liquorice constituents. Additionally, the efficiency of ABC transporters is connected to mitochondrial ATP (Giddings et al, 2021) which is the favoured system of ATP generation in non-cancerous cells. In contrast, cancer cells favour glycolysis (Hanahan & Weinberg, 2011), suggesting a less efficient functioning of ABC transporters. The assays conducted by He et al. (2019) to measure expression of ABC proteins including western blot and measurement of efflux using fluorescent dyes effluxed by particular transporters, could elucidate which ones are being expressed.

The antioxidant potential of glycyrrhizin discussed in section 20.1 could also provide another mechanism for the chemoresistant properties of the liquorice treated cells, lowering the elevation in ROS that leads to apoptosis in cisplatin induced cell death (Choi et al., 2015). However, modulations in ROS were only noted in MCF7 cells in the DCFDA assay which were unaffected by pretreatment with liquorice with regards to resistance to cisplatin, although potential cellular mechanisms that restore depleted levels of endogenous antioxidant enzymes but do not initiate reductive stress in ordinary conditions is possible. Further DCFDA analysis after treatment with liquorice and cisplatin at different time points could reveal whether an antioxidant effect is responsible for the enhanced cell viability observed in MCF10A cells after 24 hours pretreatment 2000 μ g / mL OLTE followed by cisplatin.

Chapter XXI

21. Adulteration Detection

21.1. Ability of Mitochondrial Analysis to Detect Functional Adulteration

The alternative hypothesis was that a medicinal herbal tea (liquorice) will induce specific mitochondrial modulations that enable a measurement of functional quality and identification of adulteration (section 5.3). In relation to detecting adulteration and potency, the most important factors are whether these results are unique to one of the liquorice tea samples and reproducible enough to provide a reliable evaluation of quality or adulteration. The areas where there were significant differences induced by the OLTE included the production of ROS in MCF7 cells, the OCR measured by the Seahorse Flux Analyzer, and the cytoprotective effect conferred against cisplatin-induced cytotoxicity in MCF10A cells. Some of these modulations were not replicated by XLTE enabling differentiation between OLTE and XLTE, even when administered at the same dosage of freeze dried material, and which were difficult to differentiate through visual inspection and HPTLC analysis. A discussion of each of these tests and its suitability for adulteration detection will now be undertaken.

21.1.1. Modulations in ROS

The first significant change to be observed that could differentiate the two extracts was the rise in ROS in the MCF7 cells after 3 hours of exposure to 500 μ g / mL OLTE (16.51%; p < 0.01) and 2000 μ g / mL OLTE (14.88%; p < 0.05) which was not reproduced by the XLTE at any dose (figure 127).



Reactive Oxygen Species (ROS) measured with 2',7'-dichlorodihydrofluorescein diacetate (DCFDA) represented as % change from untreated control drawn as dotted line at 100%.

n=8.

* indicates significance (p < 0.05) & ** indicates significance (p < 0.01) compared against control using one way ANOVA with Dunnett's multiple comparisons.

Figure 127: Comparison of Original Liquorice Tea Extract and Pre-extracted Liquorice Tea Extract on ROS Production in MCF7 Cells after 3 Hours

The short time point and ease of this assay would make this rise in ROS among the medium and high dosages appealing for a rapid test. However, many herbs can have pro-oxidant effects on cancer cells while having antioxidant or no effect on normal cells due to their differing metabolism (Mahanta & Challa, 2022; Fernando, Rupasinghe & Hoskin, 2019) making this rise in ROS at 3 hours unlikely to be unique to liquorice. Therefore, this marker used alone, would not suffice for the identification of potency or adulteration in liquorice.

This was followed by returning to levels equal to control after 24 hours and then dropping significantly in the 2000 μ g / mL dosage group in both the OLTE group (14.73%; p < 0.05) and the XLTE group (18.7%; p < 0.0001) (figure 128). The mechanisms for this are discussed in section 20.1.



Reactive Oxygen Species (ROS) measured with 2',7'-dichlorodihydrofluorescein diacetate (DCFDA) represented as % change from untreated control drawn as dotted line at 100%.

n=8.

* indicates significance (p < 0.05) & **** indicates significance (p < 0.0001) compared against control using one way ANOVA with Dunnett's multiple comparisons.

Figure 128: Comparison of Original Liquorice Tea Extract and Pre-extracted Liquorice Tea Extract on ROS Production in MCF7 cells After 72 Hours

The drop in ROS was seen in both groups, to a similar degree (14.73% OLTE; 18.7% XLTE) despite the XLTE being far more significant (p < 0.001 compared to p < 0.05). This would also make differentiation between the two extracts problematic using this test alone. If the combined effect of raising ROS at 3 hours, returning to normal at 24 hours and dropping at 72 hours can be consistently reproduced with other authenticated samples of liquorice and the same pattern is not seen from an adulterated batch, or from other herbs once the data expands, then this combined result could be considered a potential marker using redox status as a fingerprint for *G. glabra*.

21.1.2. Respiratory Measurements

The apparent interference of liquorice with the drugs used in the Seahorse Assay (see section 20.3.1) presents a significant obstacle in the use of the Seahorse MitoStress assay to assess functional potency and adulteration in herbal medicine. The injection of mitochondrial modulating drugs is essential to the Seahorse MitoStress assay (Agilent, 2019a), which relies on calculations between the effects of all four drugs to determine the mitochondrial parameters being measured. If a single drug fails to work, then the results cannot be used, so if liquorice reduces the effectiveness of any of the drugs, then the results of this assay may be compromised. Since liquorice is included in around half of all Chinese formulas (Guo et al., 2014) and many commercially available tea blends (Rolfe, 2023; Yogi Tea, 2023) and supplements (Aura Nutrition 2023a), evaluations using the Seahorse assays may be problematic when being used to evaluate herbal tea blends. This problem may also extend beyond adulteration research which will be discussed fully in section 23.2.

The use of the Seahorse Flux Analyzer may have other issues when used as a method of detecting adulteration too. Firstly, it is a specialised piece of equipment that requires training and may not be available in the laboratories available to many herbal supply companies. In the UK, medicinal teas often come under food regulations that only require the correct ingredients listed and do not require a proof of efficacy, provided they do not make medical claims. When they do, it is possible to be registered under the Traditional Herbal Regulation scheme which only requires evidence of traditional use and not efficacy (Heinrich, 2015). This means that many small companies who supply herbal teas exist and are unlikely to have this specialist piece of equipment, or the expertise in its use, restricting this kind of analysis to larger corporations. Simple assays that only require a sterile hood and incubator, and tests that can be easily performed may be more accessible to these smaller businesses. Ease and affordability will likely appeal to many larger companies too. Second, It also takes approximately 2 hours to take Seahorse readings, during which time the equipment cannot be used by anyone else, with additional preparation before and the protein assay after, placing a bottleneck on the speed of the results compared to assays where many repeats using different populations of cells by multiple members of staff can incubated at the same time. This will be a major obstacle in quality control where industry will want a rapid turnaround in order to confirm the quality of their products for their intended use before sale. However, the Seahorse Flux Analyzer did provide some

valuable insights that inspired simpler assays which obtained results that could be used for functional evaluation making it a useful tool for basic research.

One such assay was the cisplatin resistance assay inspired from the MCF7 respiration results (see section 20.3) and the results from the MCF10A respiratory assays could also inspire other simple tests. The proposed mechanism behind the drop in OCR amongst MCF10A cells without any apparent inhibition in function could also be used to devise some functional mitochondrial assays unique to liquorice. Should liquorice prove to be an effective protective agent against IRI, as proposed in section 20.2, then this model could be used to evaluate adulteration. The *in vitro* model of IRI cell injury presented by Li, M. et al. (2021) described in section 20.2.3, involving FBS and O_2 deprivation and return to normal media to model IRI, could present another test by which to functionally evaluate liquorice.

21.1.3. Cisplatin Resistance

The protective effect of the OLTE against cisplatin induced cytotoxicity occurred only in MCF10A cells and was the best indicator of its functional potency for several reasons. These include its replicability and the fact that it was directly related to a clinical indication.

The effect of OLTE to protect MCF10A cells from cisplatin exposure was highly replicable with only one result out of ten repeats in the 100 μ M cisplatin group being below the non-liquorice treated group, and this only 1.29% below (figure 129). This gave it a significance of p < 0.001, suggesting a strong likelihood of it being repeated in future experiments. Conversely, there was no significant difference in the XLTE treated MCF10A cells, suggesting this result is highly replicable and specific to OLTE, making it the optimal quality indicator found in this study.



Cisplatin dosage

Liquorice groups treated with 2000 μ g/ml freeze dried liquorice tea extract for 24 hours prior to exposure to cisplatin for 24 hours and presented as difference between the cisplatin only group and the liquorice & cisplatin groups. n=10 for original liquorice sample, n=8 for washed out liquorice..

Cell viability measured with 3-(4,5-dimethylthiazol-2-yl)-2,5-diphenyltetrazolium bromide (MTT) represented as % change from group not treated with liquorice and given same dose of cisplatin.

* indicates significance (p<0.05) compared to equivalent dose of cisplatin only and *** indicates significance (p<0.001) compared to equivalent dose of cisplatin using a students t-test.

Figure 129: Comparison of Original and Pre-extracted Liquorice Tea Extract on Cell Viability When Used as Pre-treatment for 24 Hours Before 24 Hour Exposure to Cisplatin in MCF10A cells

This action of protecting MCF10A cells from cisplatin was directly derived from a traditional use (Bensky et al., 2004, p.733; Chen & Chen, 200, p.868) in contemporary clinical practice where it is actively used to counter the side effects of chemotherapy and radiotherapy regimens (Wang, K. et al., 2022; Jain et al., 2022). This makes it representative of a functional effect in relation to a clinical application. It is also reassuring for clinicians that the cytoprotective effect of liquorice tea extract is achievable with a water extract, and hence applicable to commonly used decoctions, and that protection was not conferred to the cancerous MCF7 cell line. The implications of these results for cancer research will be discussed further in section 23.3.

21.1.4. A Question of Dose

One criticism of applying these results to clinical use is the dosage used in the study. The dosage of 2000 µg / mL equates to 10 g of extracted material in the bloodstream of a person with 5 L blood, the average estimated amount in an adult human (Sharma & Sharma, 2023). Since the extraction rate was 377.33 mg / g, this makes it the equivalent of 26.5 g of fresh liquorice, far beyond the safe daily amount of 0.2 mg / kg / day, estimated at 6 g / day by van Gelderen (2000), although doses occasionally reach this level in CHM for short periods (Bensky et al., 2004, p. 732). Furthermore, as no metabolization occurred, this is the equivalent of 26.5 g liquorice being extracted and injected. Although an unrealistic comparison to in vivo dosages, parallels between human and animal in vitro and in vivo trials are still the optimal way to determine mechanisms (Mattes, 2020) and so the parallel between the results of this study and a clinical indication represents a starting point for this kind of research. For quality evaluation purposes, a precise in vivo translation is not necessary. Provided that a measurable and predictable effect is seen in the original sample and not in the adulterated sample, it can be used as a guality marker, and the parallel to a clinical indication implies that the quality indicator may share some of its mechanism with its clinically intended purpose.

21.2. Comparison with Other Methods of Adulteration Detection

To determine whether there is any benefit to mitochondrial testing of herbal quality and potency, it is important to consider whether other methods can achieve the same result, especially if they are faster and more cost effective. In this study two alternative methods were utilised, a simple visual inspection of the two samples, and an HPTLC analysis.

21.2.1. Visual Inspection

Examination by visual inspection is important since there would be no need for further testing if the two extractions could be easily differentiated by visual inspection alone. The XLTE was paler than the original extract, but not to a degree that would cause alarm and this difference disappeared with only slight alterations in positioning and lighting conditions (figure 113).

Further organoleptic testing could have been undertaken. This is still used in the quality evaluation of medicinal herbal materials (Dentali, 2013), especially among herbalists themselves who may not have access to sophisticated laboratory equipment. A comparison of taste could have been undertaken, given that the primary sensory quality attached to liquorice is that of sweetness, but was not performed due to the limited amount of the XLTE material that was obtained after the sample boiled over in the freeze dryer (see section 7.2) and the ease of deceiving this method with added sugar and anise oil, a form of adulterating liquorice so common in confectionary that it accounts for the majority of "liquorice" products sold in the United States (NCCIH, 2020). In addition, the HPTLC analysis suggested that glycyrrhizin content, which is primarily responsible for the sweet taste of liquorice, was not substantially different between the OLTE and XLTE samples. The HPTLC Association (2019) guide shows glycyrrhizin eluting at 0.17 Rf and forming the first quenching zone in the samples of all species. In the HPTLC analysis presented in section 19.2, this zone corresponds to 0.27 Rf and is present in both samples, and only to a slightly faded degree in the XLTE (figure 124), suggesting that they would both likely have tasted sweet.

21.2.2. HPTLC Analysis

The HPTLC analysis (section 19.2) revealed some clear differences between species and extraction solvents but the crucial analysis, to differentiate the functionally adulterated sample from the original when prepared at the same strength, was difficult.

It was easily possible to differentiate between G. glabra and G. uralensis on the presence of liquiritin (figure 119) and the extraction method used for G. glabra by the presence of glabridin (figure 120). It was also possible to differentiate between two samples of G. glabra from different sources, with a number of additional brown bands appearing in the sample sourced from the University of Westminster Polyclinic dispensary (track 2), and the Pukka "3 Licorice" tea bags (track 1) due to the presence of many additional brown bands in the polyclinic sample after derivatization (figure 121). This may be due to one sample being medical grade while the other is food grade liquorice, or due to the different storage conditions. The Pukka "3 Licorice" tea bag had been stored in the laboratory inside its original box and individual envelope, while the dispensary sample was stored inside a large airtight glass jar. The author's own previous experience as a dispensary technician at the University of Westminster suggested that Western herbalists tend to prescribe tinctures over raw herbs, and when they do prescribe raw herbs, it is usually leaves and flowers that infuse quickly, rather than roots and bark which need a longer decocting time for extraction (Jeanroy, 2022), implying that the raw liquorice root could have been stored and exposed to a relatively large volume of air for a long time, changing some of the components.

It was also possible to differentiate the OLTE from the XLTE, when prepared at the dosage that would be expected from the same amount of starting material (if the tea bag was reused, yielding approximately 20% of the freeze dried material). This was observed by a considerable fading of all bars (figure 122).

The most challenging was the comparison between OLTE (track 4) and XLTE when prepared at an identical dosage of freeze dried material (track 5), as was used in the mitochondrial functional analysis. The differences were slight and may be within a normal range of variation and were harder to differentiate after derivatization (figure 123). This may be due to the derivatization process making compounds more visible and masking the quantitative differences between the two extracts. This agrees with the assessment of Gafner et al. (2023), that HPTLC is mainly qualitative, and the detection of pre-extracted material is problematic for this kind of analysis.

Alternative forms of chromatography could have been explored. HPLC is more accurate for quantitative assessments and may have been better suited to detect alterations in the levels of specific compounds (Loescher et al., 2014) which might be able to differentiate the two samples. However, HPTLC is the accepted standard for determination of quality in pharmacopoeias (Cañigueral et al., 2018) and a sample that passes this analysis is likely to pass industry testing too.

21.3. Relationship to the Hypothesis

These results show that some of the mitochondrial tests did produce significant, measurable effects and that some of these effects were specific to the OLTE and not XLTE. Furthermore, comparison by HPTLC, the main method for routine testing of herbal materials today, was challenging whereas the mitochondrial tests produced clear differences. Therefore we can presently reject the null hypothesis and retain the alternative hypothesis that "a medicinal herbal tea will induce specific mitochondrial modulations that enable a measurement of functional potency and identification of adulteration" for further testing.

21.4. Issues Associated with Mitochondrial Functional Analysis to Detect Adulteration

One of the main drawbacks of mitochondrial functional analysis as a means of adulteration detection is the length of time it takes to perform. Each assay requires a day for seeding, the specified treatment time which was between 3 and 72 hours, and the time taken to perform the assay, usually 1-5 hours, and then each assay has to be repeated enough times to perform statistical analysis, a minimum of 3, although more is preferable (Morriswood, 2017). This means that, even if a lab is equipped to be running multiple assays on several distinct populations of cells simultaneously, it will still likely have a turnaround time of weeks instead of hours using chemometric methods.

A second major limitation to mitochondrial functional analysis is that it is impossible to prove adulteration through this method, only that the sample does not meet the standard criteria. While genetic profiling can reveal the presence of DNA that is not inherent to the target plant's DNA and chemical fingerprinting can forensically reveal the addition of foreign substances (see chapter 2), functional analysis can only indicate that a sample does not perform in the assay as expected when compared to a voucher specimen or standard. There can be other reasons for a crop failing to meet the standard such as environmental factors during the development of the plant which can affect the synthesis of secondary metabolites (Li, Y. et al., 2020). These may be deliberately introduced crops from poor growing regions or using substandard cultivation techniques, but they could equally be due to annual variations in weather producing a poor harvest, despite the best efforts of the farmers. Although mitochondrial analysis may not be able to provide a forensic analysis of how the adulteration occurred, it can still serve as a bar above which medicinal herbs are expected to reach or be rejected from use regardless of the cause.

Another potential issue is that mitochondrial functional analysis could be deceived by the addition of substances aimed to affect the mitochondria in similar ways. Quality evaluation and adulteration technologies tend to move together whereby novel technologies developed to assess the quality of herbs are soon matched by fraudulent dealers developing ways to deceive these technologies and make a substandard product appear correct. From the manufacture of wooden pips to make a fake "raspberry jam" appear like the genuine article reported by Sylvia Pankhurst in 1931 (Lawrence, 2003), to attempts to deceive chemometric testing by adding adulterants that mimic chromatography profiles of active or marker compounds (Booker, A., Frommewiler et al., 2016), there is an ongoing competition between those who adulterate supplies and quality assessors trying to detect them. This raises a possibility of adulterants being added that will mimic the mitochondrial activities of plants. For example, as discussed in section 3.2.5, the metabolic effects of cinnamon have been identified as potentially being due to an uncoupling effect which temporarily lowers ATP, followed by a hormetic response to stimulate mitochondrial biogenesis, leading to the browning of adipose tissue (Li, X. et al., 2021; Kwan et al., 2017). This could lead to the addition of other

uncoupling agents, including natural extracts and synthetic drugs which have been discovered to have uncoupling effects (Goedeke & Shulman, 2021). Some of these can cause acute toxic responses (Grundlingh et al., 2011) suggesting that this kind of adulteration could be as dangerous as other functional adulterations of herbs, such as sibutramine in weight loss supplements (Jairoun et al., 2021). This could be especially relevant if the mitochondrial effect being tested for was cytotoxicity against cancer cells that could be mimicked with chemotherapeutic drugs. However, the likelihood of being able to add an adulterant that both mimics the mitochondrial effects and is undetectable through chemometric analysis will be far more challenging than deceiving either method when used alone. This makes mitochondrial functional analysis an additional angle to the orthogonal approach already suggested by Gafner et al., (2023) involving visual, microscopic, organoleptic, chromatographic and spectroscopic methods to keep ahead of fraudulent suppliers. Unfortunately, they also report that fraudulent suppliers nearly always find ways to evade detection, even deliberately sending their adulterated material for quality evaluation to discover what passes the existing tests. Therefore, it is inevitable that methods of deceiving mitochondrial functional analysis will be created. The goal of using mitochondrial functional analysis in quality evaluation, like other methods before it, can only be to create an additional layer of complexity to adulteration, meaning fewer suppliers can engage in the practice unnoticed.

Chapter XXII

22. Limitations of this Study

There were several limitations to this study, including the method of adulteration used, that mixtures of genuine and adulterated supplies were not tested, and that only one herb ended up being tested. This section will only discuss the limitations, while the steps that can be taken to address these will be discussed in the section on future work (section 24).

22.1. Method of Adulteration

One of the main limitations of this study was the method of adulteration that was used. Despite requesting samples of herbs that failed quality evaluation from the supplier of the original herbal tea (Pukka, UK) and contacting another researcher in quality evaluation who was working on liquorice (UCL), no samples were obtained leaving us to devise a method of functional adulteration that we could generate in the laboratory. This means that the method is still only a proof of concept and remains to be tested on real-world samples of adulterated material. Although the form of adulteration chosen is reported in the literature with other herbs (see section 2.6.1), there is no evidence that pre-extracted material is specifically employed for liquorice. In addition, the concerns regarding glycyrrhizin raising blood pressure have created a market for deglycyrrhizinated liquorice (Öztürk et al., 2017), meaning that there is less incentive to sell pre-extracted liquorice as an adulterant when it can be openly marketed and promoted as safer by manufacturers.

Given more time and resources, it could have been possible to locate some real-world samples or create equivalents that more closely imitate how liquorice is known to be functionally adulterated. Liquorice is preferred to be foraged from the wild because cultivated roots lack the same potency (Brinckmann, 2020), which would make cultivated vs. wild foraged roots an excellent comparison. A search for farmed liquorice did not produce any results and this kind of liquorice is probably not marketed as such due to its purported inferiority, but is simply lacking certifications like FairWild, that require products to be traceable to ensure their wild foraged origin (FairWild Foundation, 2010: principle 10.2). Products with this label have a marketing advantage to sellers, suggesting they are ethical and of superior quality, but there is no marketing advantage to advertising an inferior farmed sample as such. For a scientific study, absence of evidence should not be considered evidence of absence (Wright, 1888), so using a sample that simply has no certification should not be taken as an assumption that it is farmed or inferior. A farm could have been contacted directly and liquorice farms have even been reopened in Pontefract, once the home of UK liquorice, for use in sweets (Smithers, 2012). This could also suggest a comparison of liquorice "terroir" since the UK soil and environment should be expected to produce a different product than one from Kazakhstan, Georgia and Egypt, as the sample used in this study is said to come from (Pukka Herbs, 2020a; Rolfe, 2023). An even better comparison would have been if the researcher could have grown samples in different conditions, both ideal and sub-optimal, to reveal the pattern of efficacy and constituents of several different conditions. This would have taken time, resources and agricultural knowledge that were not available to the researcher during this study but could be options in the future, if working with a team.

A similar critique could be made with regard to storage techniques. If planned from the start, it would have been possible to deliberately store the same liquorice tea that was used for the initial tests in an inappropriate condition, e.g. opened and exposed, and when the time came to test the adulterated sample, the badly stored sample could have been used. It was not expected that the characterisation of liquorice would take so long, with the first year seeing laboratory access closed during most of 2020 and restricted for much of the following year due to Covid-19. The sudden nature of the lockdown, which occurred just as the protocol for the original extract was about to be executed, meant that the possibility of taking advantage of this time to naturally age some of the liquorice was not available. Only accelerated shelf-life testing, as used in the food industry, would have been possible but this must be validated against real-life storage conditions to define its accuracy (Betts & Bakhai, 2023), so would not have been appropriate. The liquorice stored in the University of Westminster dispensary could also have been a potential candidate, especially if the additional brown bands seen on the derivatized HPTLC chromatogram (right column in figure 121) were evidence of excessive exposure to air for a long period, but this is only speculation. The dispensary records could be requested to determine how much time had elapsed since the sample had been purchased and opened, and the mitochondrial functional assays repeated on this sample alongside a recently acquired sample for comparison. The alternative explanation was that these differences could be due to food and medical grade liquorice, which could also be the basis of another test: acquiring samples from herbal medicine suppliers and others intended for use in the food or confectionary industries to compare their effects on mitochondrial function.

22.2. Mixing Adulterated and Original Extracts

Another limitation of the study was that it did not look at mixtures of the original and pre-extracted material to discover the proportion at which the effects disappeared. This is a significant limitation because companies that adulterate dried herbs with pre-extracted material, or employ the double extraction method of enhancing their yields for supplements, are likely to combine the two extracts to evade detection. The XLTE yielded approximately 20% of the original extract, so another set of tests, which examined a ratio of 5:1 OLTE / XLTE, would have provided some useful real-world data regarding the potential reduction of effects if the two extracts were combined. Other ratios could also have been tried to find out when the functional effects ceased to be significant. Finally, this study did not try extracting a third and fourth time, until no new material was produced. This could have been useful since any manufacturer that will try to enhance yields by double extraction is likely to try to continue extracting until the yield no longer provides a profitable return.

22.3. Other Herbs and Blends

The original aim of the study was to test several medicinal herbal teas but due to the loss of laboratory access during the coronavirus pandemic, it became only possible to test liquorice. Liquorice was selected because it is present in most herbal blends, both in CHM formulas and commercial infusions (Wang et al., 2013; Guo et al., 2014; Yogi Tea, 2023; Pukka Herbs, 2020a) and due to its presence as a stand-alone tea among the donors' range. Therefore, any effect of liquorice needs to be accounted for when examining blends. Even without the impact of coronavirus on laboratory access, the option of testing multiple herbs and blends may still not have been possible given how many potential avenues for research were revealed during the evaluation of liquorice. Instead, the current study stands as a proof of concept and the full development of a pipeline for many herbs and products may have to be developed over time with greater resources. A plan for this and a priority list of future herbs to profile and blends to be examined first is presented in the section on future work (section 24.4 and 24.5).

Chapter XXIII

23. Implications of the Results for Other Research

One of the great advantages of mitochondrial testing to functionally evaluate medicinal herbal teas is that it can operate reciprocally with other areas of science and medicine. The results gained from mitochondrial quality evaluation can initiate or inform research and practice in other fields, while the current interest in mitochondrial aspects of disease can inform quality evaluators on the parameters that they can look for. As part of the pipeline process to develop standards (see section 24.3), any tests chosen for quality evaluation need to be reproduced to ensure that they are reliable markers and not unique occurrences, since reproducibility in scientific literature can be as low as 10% (Baker, 2016) with estimates of 40% in cancer biology (Errington et al., 2020), a field closely related to mitochondrial functional testing due to the importance of mitochondria in cancer formation and apoptosis (see section 3.1.6.3). This can then feed back to the research community with rebuttals if the results cannot be reliably reproduced and adopted as a standard if they can.

23.1. Implications for Adulteration Research

For the first time, this study shows that the functional adulterations of herbal medicines described in the 5th century by Tao Hongjing may be detectable through mitochondrial profiling. This involves herbs that are the correct species but have been tampered with in ways that reduce their clinical efficacy and can occur through incorrect procedures when growing, harvesting, storing or handling but are still sold as medicines (Liu, 2021). The relative ease of detecting species through genomic or chemometric fingerprinting and functional adulteration through the detection of drugs (see chapter 2) has left a large gap in adulteration research regarding poor quality herbs. The current study suggests that efforts can go further, identifying the functional capability of herbs to achieve their intended result through *in vitro* assays. This marks a break from previous forms of functional

quality evaluation which have not changed since Galen's testing of antivenom on chickens in 2nd century Rome (Karaberopoulos, Karamanou & Androutsos, 2012), or the testing of ginseng (*P. ginseng*) on runners in the 11th century (Shang & Gong, 1994). The costs and ethical considerations of animal and human testing have caused functional evaluation to disappear from routine quality evaluation protocols but *in vitro* protocols could allow them to return.

One important finding from this research is that the assays performed to evaluate functional testing should be relevant to clinical applications. In all the basic tests performed, only a rise in ROS after 3 hours was significant and different between the two samples (figure 127), a finding that returned to normal at 24 hours. The Seahorse results also led to a complex picture among both cell lines (figures 63-69 & 74-83) that remained unexplained in the follow up assays, despite inspiring the cisplatin resistance assay which achieved the most repeatable and clinically applicable result to date. This suggests that when designing functional adulteration tests, the assays employed should also be functional, inducing a cellular model of a pathological state and assaying if the herb can affect this condition. Besides direct applicability to the clinician, the advantage of this approach is that it is equally applicable to single herbs, blends, teas, powders or extracts. It also avoids the issues of substitution discussed in section 2.6.2 because a substitution that is able to achieve a functional effect on a specific cellular model can be considered a functionally correct substitution. Therefore, different substitutions aimed at different ends, like the substitutions for aconite (A. carmichaelii) in Jin Gui Shen Qi Wan made by Aura Herbs and GinSen, described in section 2.6.2, which are focused on replicating different aspects of the original formula, can also be evaluated using different models.

23.2. Implications for Research into Liquorice

These results could have an impact on research into liquorice using drug induced models of disease. The idea that liquorice may have a protective effect against drugs came partly from the research findings of the Seahorse experiment, partly from the traditional use of liquorice in TCM but also from the literature that

suggested liquorice could have a protective effect against Parkinson's disease though the use of a rotenone induced model of disease in rats (Ojha et al., 2016) and cell cultures (Karthikkeyan et al., 2020; Karthikkeyan, G., Prabhu et al., 2021; Karthikkeyan, G., Pervaje et al., 2021; Karthikkeyan et al., 2022). If liquorice can induce a protective effect against the drugs which are being used in these models then there is a risk that they are committing a type I error, rejecting the null hypothesis, when it is in fact true (Banerjee et al., 2009) due to not accounting for the possibility that liquorice could be interfering with the action of the drugs used to induce the model of disease. Rotenone is a compound found in several Leguminosae plants (Wood et al., 2005) and so not a natural part of the development of Parkinson's disease. It is used because its inhibitory action on Complex I of the ETC was found to cause dopaminergic neuron degeneration with Lewy body-like structures forming in the brain and Parkinson-like symptom appearance after intravenous administration to rats (Betarbet et al., 2000). This means that if liquorice can protect cells from the toxic effects of the drug, then it could interfere with the model and not have a protective effect against the natural pathogenesis of Parkinson's disease at all. This model has already come under criticism due to the high mortality associated with intravenous injection and the poor bioavailability by oral administration resulting in more gastrointestinal issues than induction of Parkinson-like effects (Niederberger et al., 2022). The results of the current study add an additional critique that can be applied to any experiments using drug induced models of disease to examine the effects of liquorice.

This detoxifying effect does not have to invalidate all research where a drug is used to initiate a disease state. Sometimes experiments look at endogenous molecules that induce toxicity at high doses. These include studies looking at liquorice having a protective effect on glutamate induced neurotoxicity (Wang, D., Guo et al., 2014; Lee et al., 2018; Yang, E. et al., 2012), or those investigating the effect of liquorice on corticosterone (Zhou et al., 2017; Li, X. et al., 2020) and 7-ketocholesterol (Kim et al., 2009) induced toxicity. These may still be valid since these substances are part of the pathophysiological mechanism under investigation, and a protective effect against them may still confer protection against their natural pathogenesis.

23.3. Implications for Chemotherapy Research

The results found in this study are most relevant to the actions of liquorice tea on chemotherapy. They suggest that if liquorice was to become an adjuvant therapy in cisplatin regimes, then the liquorice used should undertake an *in vitro* cisplatin resistance test to ensure its effectiveness. These findings also add to the current debates regarding liquorice as an adjunct to chemotherapy. Most chemotherapy utilises cytotoxic agents with the intention to raise the already elevated ROS in cancer cells above the threshold to induce apoptosis (Yang, H. et al., 2018). This is a tactic which can be resisted through the expression of ABC efflux transporters that promote chemoresistance (Sharom, 2008). It is therefore reasonable to question the use of a herb with antioxidant activity and which may have the ability to create chemoresistance by promoting the expression of ABC efflux transporters as an adjuvant to chemotherapy.

Some of the findings of this study support a view of caution when taking antioxidants with chemotherapy agents designed to induce apoptosis through ROS production (Lawenda et al., 2008; Harris & DeNicola, 2020). After 72 hours of exposure to 2000 μ g / mL liquorice tea extract, ROS dropped 14.73% in the MCF7 cells compared to control (figure 57). While the natural environment of slightly raised ROS in cancer cells is associated with increased proliferation and metastasis, reductive stress is associated with increased cancer survival, drug resistance and stemness (Chun, Kim & Surh, 2021). As mentioned in section 21.1.3, the doses used in these assays are not realistic representations of the concentrations that would be found in the body. However, in the interests of caution, it does warrant further investigation and implies that regular consumption of liquorice based adjuvant therapies for more than 3 days before chemotherapy could potentially negatively affect the outcome of a regime that depends upon ROS activation.

Other findings in this study suggest that liquorice may actually have potential as an adjuvant in chemotherapy. First, after 3 hours of exposure to liquorice tea extract, ROS increased in the MCF7 cells by 16.51% in the 500 μ g /

mL dosage group, and by 14.88% in the 2000 µg / mL group while leaving MCF10A cells unaffected (figure 55, section 11.2). This would suggest that the timing of liquorice administration could be important, adding nuance to the discussion of whether liquorice is simply good or bad during chemotherapy. Ingestion of liquorice just before, concomitantly, or shortly after chemotherapy could raise ROS in cancer cells enough to enhance the chemotherapeutic effect through "priming" (Mould, 2019; Henley 2015; Henley et al., 2017). The exact optimal time for ingestion will depend on the length of time it takes to metabolise the different compounds of liquorice and for it to reach the cells, as well as how long it takes to leave the body. Further research involving blood analysis for liquorice metabolites at different time points after ingestion could elucidate this. It is also important to determine whether the metabolised forms will have the same effect as those in the tea extract used in this study before hypotheses about clinical efficacy can be developed. However, it suggests that the intervals between consumption of liquorice and chemotherapy, and that duration of consistent consumption may be important factors to consider in future research aimed at reaching a stage of clinical trials to determine whether liquorice is beneficial or detrimental in chemotherapy.

The most important finding in relation to chemotherapy is that liquorice protected MCF10A cells from the cytotoxic effect of cisplatin, but did not offer any protection to MCF7 cells (figure 129). This suggests that the antioxidant effects of liquorice are not interfering with cisplatin and that, if this finding is consistent with further experiments in other cell lines and *in vivo* activity, that liquorice may exert a protective effect on non-cancerous cells while conferring no benefit to cancerous ones. It could even be speculated that the increased survival rate of the liquorice treated MCF10A cells of 8.186% at 100 μ M and 11.79% at 50 μ M cisplatin treatment imply that a greater dose of cisplatin could be tolerated, making the regime even more effective with the same side-effects. Before any conclusions can be drawn, these results would need to be replicated among other cell lines to determine whether this effect is seen only amongst breast cells or in other cells, especially those which limit therapeutic dosing guidelines.

Nephrotoxicity is often the limiting factor in cisplatin therapy, forcing clinicians to balance tumour toxicity against kidney failure (Miller et al., 2010; Volarevic et al., 2019), so determining the cytoprotective effect of liquorice on cisplatin treated kidney cells is necessary to inform clinical trials using liquorice as an adjunct to cisplatin therapy. Some experiments have been performed on this exact question. *In vitro* studies found that liquorice extracts reduced cisplatin induced cytotoxicity in human kidney cells (HK-2 and HEK-293) by reducing ROS, inflammatory and apoptosis markers, including caspase-3 and p53/p21 expression (Ju et al., 2017; Basist et al., 2022). *In vivo* mouse studies support these findings, with glycyrrhizin rescuing cisplatin induced nephrotoxicity and delayed cisplatin induced injury (Arjumand & Sultana, 2011; Wu, Chen & Yen, 2015). This rescue is achieved by preventing oxidative stress, reducing inflammation, restoring antioxidant enzyme levels, decreasing DNA fragmentation and restoring kidney function markers, possibly due to upregulation of Nrf2 and downregulation of NF-κB.

The toxicity of cisplatin also means that it is often used in reduced dosage along with other chemotherapeutic agents (Dasari & Tchounwou, 2014), so it would be prudent to assess the effect of liquorice in conjunction with other chemotherapeutic agents and common combinations. Several studies have found protective effects of various compounds in liquorice against doxorubicin induced cardiotoxicity through different mechanisms with the common effect of reducing ROS induced apoptosis with restoration of antioxidant enzymes (Cheng et al., 2022; Wang, M. et al., 2023; Xu, Z. et al., 2022; Zhang et al., 2011). Only Zhang et al. (2011) also evaluated the tumour inhibitory capacity to confirm that G. uralensis extract did not reduce the efficacy of doxorubicin. One study investigated the interaction of liquorice and paclitaxel (Ha et al., 2020) which found that 14 days pretreatment with G. uralensis decreased systemic exposure and increased clearance in rats, potentially due to upregulation of CYP450 enzyme 3A4 and P-gp, resulting in potential decreased effectiveness. They also reported that a single dose of liquorice did not affect parameters, recalling the findings of this study that the timing of liguorice consumption may be an important factor.

Other herbal formulas have been promoted as adjuvants in chemotherapy routines and could be quality tested for clinical use with similar models. In particular, CHM formulas TJ-41 (Bu Zhong Yi Qi Tang, 補中益氣湯, "Strengthen the Middle and Augment the Qi Decoction") and PHY906 (Huang Qin Tang, 黄芩 湯, "Scutellaria Decoction") have entered phase II and III trials as adjuvants in cancer treatment (Wang, K. et al., 2022). Both of these formulas contain liquorice which may be responsible for some or all of the alleviation of side effects from chemo- and radiotherapy, making the results of this study potentially relevant to these projects. These formulas also claim to have chemosensitizing effects raising the possibility that, if they were to be used in routine clinical practice, it would be advisable to test the quality of each batch. This testing should include assessing for its chemopreventive effects in normal cells and chemosensitizing properties in cancer cells to ensure that batches have the optimal effect. Although the number of potential assays to test this could be immense, a test on the closest available cancer cell line to the one being treated, a healthy analogue for comparison and any cells that present a dose limiting factor with regard to toxicity could be suggested as an initial choice.

Clinical trials relating to the use of liquorice as an adjunct to chemotherapy can also feed information back to the adulteration testing data. It would be useful to know if the liquorice or other herbs being used in a clinical trial also pass the mitochondrial functionality test in order to determine if the two results are linked and whether the data gathered from mitochondrial functional testing relates to real-world applications. As discussed in section 20.1.3, the dosage used in the cisplatin assay is unrealistic for a clinical environment but finding parallels between human and animal in vitro and in vivo trials is still the optimal way to determine mechanisms (Mattes, 2020). Therefore, these two experimental methods can support each other. In designing this approach, it would be unethical to give volunteers a herbal material that had failed quality testing, so the only direction the research could take would be to test the herbs for quality first and then use those which gained a positive result to compare against the clinical trial results to see if this corresponds with a positive clinical trial outcome. This would inevitably bias the results as not all trials will be successful and only herbs that passed mitochondrial functional testing are used, inevitably resulting in failed clinical results appearing to invalidate mitochondrial functional testing. A better option would be to acquire herbs that had already been used in clinical trials, or in trials that are ongoing and subject them to mitochondrial functional analysis to compare the results with the clinical trials. This would depend on the original herbs being stored by the researchers in the case of past trials, and potential degradation from storage that could still compromise the results, especially if the trials were from years ago. Currently ongoing trials would be a more valuable resource but are likely to be few in number.

Chapter XXIV

24. Future Work

Several avenues for potential future work are possible. These include further profiling of liquorice and the profiling of other herbs. The development of a pipeline process is necessary to facilitate the more efficient development of future mitochondrial functional profiles. This will enhance the scientific rigour and efficiency through standardisation, yet it should still address the practical concerns of the herbalists who may use the herbs and the public, especially those buying a medicinal herbal tea for the purported benefits attributed its ingredients.

24.1. Further Profiling of Liquorice

Further profiling of liquorice can be divided into work on other cell lines and models of disease, comparisons between different species of liquorice and the examination of the effects of other forms of adulteration for which mitochondrial functional analysis may be appropriate.

24.1.1. Other Cell Lines

One of the most obvious areas where this work could progress is other cell lines to find out if the same results are obtained or if the results of this study are unique to breast cells. The present findings would indicate that this should be correlated to clinical actions and so the priority cell lines should reflect the tissues liquorice is most likely to affect. Although liquorice is said in CHM to enter all bodily systems (Bensky et al., 2004, p.732), which would suggest it has an effect on all cell lines, there are a few indications that are shared across traditional medical systems which could be used to identify some preferential cell lines.

The main activity of liquorice in WHM is for the treatment of coughs (Bone & Mills, 2013: 719; Hoffman, 2000, p.555), a property it shares with CHM (Bensky et

al., 2004: 732; Chen & Chen, 2000, p.867), Ayurveda (Pole, 2012, p.220) and Unani medicine (Saad & Said, 2011, p.264). This action is also supported by the EMA (2012), suggesting that it is a clinical action that would be valuable to test through mitochondrial functional analysis. An evaluation of the effects of liquorice on several lung cell lines would therefore seem an obvious selection. Since primary cells are harvested directly from living donors, it would be unethical to routinely extract these cells, so only immortalised lines should be used. These can include established epithelial cells (BEAS-2B; American Type Culture Collection, 2023) and the use of human telomerase reverse transcriptase technology which has created immortalised lung fibroblasts (American Type Culture Collection, 2022) and airway smooth muscle (Burgess et al., 2018).

Antitussive drugs primarily act on the nervous system, either peripherally, by suppressing the responsiveness of sensory receptors that initiate the cough reflex, or centrally, at the brainstem, where the neural circuitry for coughing is located (Bolser, Hey & Chapman, 1999), making neuronal cell lines a possibility for research too. PC12 cells are an immortalised cell line often used in the study of nerve physiology and pharmacology due to them sharing common features with neurons such as neurosecretion, ion channels and neurotransmitter receptors, and can be differentiated into sympathetic ganglion neurons by culturing with nerve growth factor (Wiatrak et al., 2020). This could make PC12 cells another potential cell line to observe mitochondrial modulations in response to liquorice. Ayurveda's attribution of liquorice to calm the nervous system (Pole, 2012, p.220) and act as an aphrodisiac (Aronson, 2018) may also suggest that an effect could be observed on neuronal cell lines. Some research has already identified carbenoxolone, a synthetic analogue of glycyrrhetinic acid, as a gap junction blocker in epilepsy and pain research through the inhibition of connexins (Connors, 2012; Bell et al., 2021) which could serve as the basis for mitochondrial functional assays on nerve cells whose activity, when stimulated, should be reduced by exposure to liquorice compared to controls.

Another use for liquorice is to help with digestive ulcers. This indication is found in WHM (Bone & Mills, 2013, pp.721-722), Unani (Saad & Said, 2011, p.264) and Ayurveda (Pole, 2012, p.220). The EMA (2012) monograph does not

concur with this use but does suggest a traditional use for dyspepsia and burning sensation and both glycyrrhizin and carbenoxolone been have used as medicines for ulcers since 1946 with other components also being researched for anti-ulcer activity (Kaczor, 2015). CHM gives liquorice a broader digestive function, attributing its primary action to enhancing digestion (Pi Qi, 脾氣, "Spleen Qi") and treating symptoms of fatigue with decreased appetite and loose stools (Chen & Chen, 2000, p.867) but also includes indications for internal or topical use, singularly or within formulas, for a variety of "sores, swellings and carbuncles" (Bensky et al., 2004, p.733). This is particularly noticeable in the formula Liquorice Decoction to Drain the Heart (Gan Cao Xie Xin Tang, 甘草瀉心湯) where a higher than usual dose of unprocessed liquorice is used in the treatment of ulceration around the mouth, anus and genitals, symptoms which have been interpreted as syphilis or Behçet's Syndrome (Fruehauf 2010), a variable vessel vasculitis that involves ulceration of mucocutaneous tissue around the mouth and genitals (Karadag & Bolek, 2020). This protective activity on gastric mucosa would make immortalised gastric cell lines, such as KMU-CS12 (Applied Biological Materials, n.d.) which are derived from gastric stem cells but have the potential to be tumorigenic (Yang et al., 2009), or an immortalised primary gastric cell line, such as JOK-1 (Okayama et al., 2000), suitable for testing this activity.

Cardiomyocytes would be another useful cell line to study. In relation to this study, the drop in OCR among MCF10A cells after 24 hour treatment with 2000 µg / mL liquorice tea extract (chapters 13 & 17), despite no perturbations being detected in the basic mitochondrial function assays at this time point (chapter 11), make liquorice an interesting herb to study in relation to IRI (see section 20.2). Other research in this field has also suggested glycyrrhizin is an agent responsible for the protective effects in IRI (Hu et al., 2023) and its use in cardiovascular disease in CHM (Wang, L. et al., 2022). Usually, rat myocytes are used in cardiovascular research but human immortalised atrial myocytes are now available which have been transduced with a lentiviral vector to grow in the presence of doxycycline and differentiate into cardiomyocytes when the drug is withdrawn (deVries, 2022). This rapidly replicating human heart cell line could offer a promising target for developing mitochondrial functional analysis protocols for quality evaluation based on the protective effects of liquorice against IRI.

The recent development of induced pluripotent stem cells, which can be differentiated into any cell type (Jimenez-Tellez & Greenway, 2019), offer another possibility for mitochondrial functional adulteration research. These could potentially replace all the limitations of using existing cell lines by creating personalised cellular models. More optimisation is needed before this becomes a realistic opportunity but it suggests that an exciting new era of cellular research may be on the horizon. Organoid models are another recent innovation which more closely approximate the complexity of an organ and its natural physiological processes *in vitro* (Hagen, 2021) and could be used to examine the effects of liquorice on whole organ systems in greater detail.

24.1.2. Actions on Other Disease Models

The findings from this study suggest that cells being exposed to liquorice and measuring their response is not the optimal way to detect adulteration but that the effect of liquorice on clinically relevant models should be used. This implies that as well as testing on a variety of cell lines, the effects of liquorice on cellular models of commonly treated pathologies should be investigated too.

The use of liquorice in the treatment of coughs can be modelled in a number of ways. Infection of immortalised cells with viruses is difficult and primary cell cultures are normally used in researching pathogenesis studies for their similarity to natural disease responses to respiratory tract infections (Rijsbergen et al., 2021). Lung inflammation models are rare in the literature but have been developed using multicellular models composed of macrophages, dendritic cells, and epithelial cells and using lipopolysaccharide to induce the macrophages to shift to their proinflammatory M1 phenotype (Drasler et al., 2021). Although this uses primary macrophages and dendritic cells, they are derived from monocytes isolated from the buffy coats of blood samples from a transfusion centre and so are not as invasive as other primary cells which may need harvesting from biopsy. Pulmonary inflammation involves an overproduction of ROS from both mitochondrial and non-mitochondrial sources. Endogenous sources include

phagocytes which destroy pathogens with a "respiratory burst" of O₂^{•-} generated by NOX, while exogenous sources include medicines, air pollution or cigarette smoke (Zuo & Wijegunawardana, 2021). This increase in ROS initiates pro-inflammatory signalling cascades that attract more phagocytes which release more ROS and result in endothelial damage if not curbed. This would make an assay like DCFDA or MitoSOX a useful tool to measure the levels of ROS in a cellular inflammatory model and compare the protective effects conferred by liquorice, even though it would not be measuring mitochondrial ROS exclusively but also that produced by NOX.

The use of liquorice in the treatment of stomach ulcers can also be modelled in several ways. The protective effect of liquorice against gastric ulcers was observed by Revers in 1946 and led to both the extraction of glycyrrhizin, the subsequent synthesis of carbenoxolone, as well as the observation of its hypertensive side effects (Kaczor, 2015). This led to attempts to treat ulcers using deglycyrrhizinated liquorice with mixed results but is still the main marketing drive for deglycyrrhizinated liquorice, despite the glycyrrhizin derived compounds being the component that was used as prescription medicine for gastric ulcers until its replacement with newer non-steroidal anti-inflammatory drugs that did not affect blood pressure (Aronson, 2018). Once the role of Heliobacter pylori was established as the cause of gastric ulcers by Marshall and Warren in the 1980s, triple therapy with two antibiotics and a proton pump inhibitor became the first line of treatment (Marshall & Adams, 2005). The use of liquorice for gastric ulcers has been discontinued from UK clinical guidelines (National Institute for Health and Care Excellence, 2023) and remains only as a herbal or nutraceutical alternative. Much of the anti-ulcer activity of liquorice is attributed to increased gastric mucosal secretion but antioxidant mechanisms have also been proposed (Bafna & Balaraman, 2005). The pathomechanism of ulcer formation by Heliobacter pylori has been associated with the Duodenal ulcer-promoting gene A (Sharndama & Mba, 2022), whose effector proteins have been shown in vitro to induce the secretion of IL-8 and IL-12 by gastric epithelial cells (Dadashzadeh, 2017). Although its mechanism is not completely understood (de Lima Silva et al., 2021), this could be developed into a similar model to the lung inflammatory model described above, except utilising the induction of inflammation by the effector

proteins encoded by *H. pylori* instead of lipopolysaccharides and measuring the changes in ROS among liquorice treated cells against controls. However, if gastric secretions are the main protective mechanism, then it could be measured by the secretions produced by gastric cells exposed to liquorice compared to controls, which would be simpler than mitochondrial functional assays in this instance.

Although investigating the effects of liquorice from the perspective of traditional indications will reveal the most valuable information to herbalists, many indications have been produced from modern research that would simply require a reproduction of the original experiments, or another assay that measures the same outcome.

One of the most common of these is the induction of apoptosis in cancer cell lines. This has been reported for licochalcones in human bladder cancer (Yuan et al., 2013; Hong et al., 2019), breast cancer (Kang et al., 2017), liver cancer (Wang et al., 2019), oral squamous cell cancer (Oh et al., 2018), oesteosarcoma (Lin et al., 2019) and lung cancer cells (Kim et al., 2015); for glycyrrhizin in breast cancer (Lin et al., 2017) and leukaemia cells (Chueh et al., 2012); glycyrrhetinic acid for lung cancer (Luo et al., 2021), pituitary adenoma (Wang et al., 2014), ovarian cancer (Lee et al., 2010; Yang, J. et al., 2012), leukaemia (Huang et al., 2016) and breast cancer cells (Sharma et al., 2012); and for isoliquiritigenin for cervical cancer (Hirchaud et al., 2013; Hsu et al., 2009), gastric cancer (Ma et al., 2001), prostate cancer (Jung et al., 2006) and melanoma cells (Chen et al., 2019). These would be easy to replicate in mitochondrial functional tests using an MTT assay to measure cell viability. It should be noted that the liquorice extract used in this study had no effect on cell viability, even in the cancerous MCF7 cell line (figure 54), and at doses that are unrealistically high for a clinical comparison (see section 21.1.3), suggesting that these experiments are not representative of a clinically applicable use for liquorice as a whole extract but may be applicable to isolated extracts only.

A number of studies have also reported protection against apoptosis. These are often created *in vitro* using drug models of pathogenic processes and so intersect more directly with the findings of this study that unadulterated liquorice offers a protective role in preventing apoptosis in non-cancerous MCF10A cell lines while offering no protection to the cancerous MCF7 cell line (figures 86 & 88; chapter 14). In a similar finding with clear parallels to this study, Cheng et al. (2022) found that glycyrrhetinic acid conferred a protective effect against cardiotoxicity induced by doxorubicin which would make another simple test based on replicating the same results. Most of the other studies which looked at the cytoprotective effects of liquorice against toxicity induced by other substances looked at neuronal cells and include a glutamate model of excitotoxicity (Lee et al., 2018; Wang, D., Guo et al., 2014; Yang, E. et al., 2012; Teng et al., 2014), corticosterone model of stress induced neurotoxicity (Zhou et al., 2017; Li, X. et al., 2020), 7-ketocholesterol model of oxidative stress induced neurotoxicity (Kim et al., 2009) and the rotenone induced models of Parkinson's disease discussed previously in section 23.2 (Ojha et al., 2016; Karthikkeyan et al., 2020; Karthikkeyan, G., Prabhu et al., 2021; Karthikkeyan, G., Pervaje et al., 2021; Karthikkeyan et al., 2022). Each of these could be tested to find the protective effect which is easiest to replicate and differentiate adulterated products from the original sample.

24.1.3. Comparison Between Species

All of the cultivars used in the present study were from the *G. glabra* species of liquorice so further profiling of other species would potentially reveal different mitochondrial modulating effects, potentially enabling differentiation of species with mitochondrial functional analysis. As noted in section 4.2, different compounds are found in different species with glycyrrhizin common to all, glabridin almost exclusively present in *G. glabra*, liquiritin and glycycoumarin only present in *G. uralensis*, and licochalcones being a dominant characteristic of *G. inflata* (Kondo et al., 2007; Rizzato et al., 2017; Avula et al., 2022). This would suggest that these species may have different therapeutic effects, which could explain the different activities attributed to them in various herbal medical traditions. It is also possible that this may be observable in their ability to modulate mitochondrial activity differently.

The specific findings of this study observed an activity of G. glabra to protect non-cancerous MCF10A cells from cisplatin, whereas the protective effect of liquorice against toxic substances is an action found in Chinese herbal medicine where G. uralensis is predominantly utilised (Bensky et al., 2004: 732; Chen & Chen, 2000: 866). This detoxification property is not commonly recognised in WHM or Ayurveda which primarily use G. glabra (EMA, 2012; Pole 2012: 220), suggesting that a unique component amongst G. uralensis is responsible for this effect. The literature to date reflects this with the study by He et al. (2019), which found that components of liquorice could upregulate ABC efflux transporter expression, mainly looking at liquiritin, its precursors and isomers, predominantly found in G. uralensis. The present study suggests that this activity may be more common across species than previously assumed and possibly due to the actions of common components like glycyrrhizin or liquiritin apioside. Previous research on the role of these compounds as activators or inhibitors of ABC efflux transporters has returned contradictory results (Wang et al., 2013; Feng, Ding & Qiu, 2015), so this study lends strength to the argument that it might increase their expression. Furthermore, it suggests that further study into the detoxifying effects of G. glabra are warranted. When G. glabra is investigated for its protective effects against toxic substances, they are usually investigating drug induced models of disease, discussed in section 23.2, and not testing directly for the effect of G. glabra on administered drugs. This could be a significant oversight in the potential uses for G. glabra but also an invitation for researchers to compare the protective effects of G. glabra against G. uralensis to determine if additional substances in G. uralensis confer additional benefit.

24.2. Other Forms of Adulteration

How liquorice can be adulterated was discussed in section 4.8 and the limitations of the chosen method in section 22.2. Expanding on this, further work could be undertaken to determine if other forms of adulteration could be amenable to mitochondrial analysis.

24.2.1. Storage Conditions

Poor storage conditions can introduce a number of potential factors that affect the quality and potency of medicinal herbs. Storage for up to five years has been reported while farmers wait for market values to adjust to sell their crop for greater profit, at the expense of degradation with loss of metabolites, and fungal and weevil infestation (Booker et al., 2014).

Fungal cultures can affect the quality of liquorice. This is more commonly reported in liquorice intended for food markets than herbal medicine, but likely due to a lack of analyses of herbal medicine products (Altyn & Twarużek, 2020). Fungal contamination could make a suitable candidate for detecting functional adulteration with mitochondrial analysis. Several mycotoxins have mitochondria modulating effects that may be detectable with the analyses described in this study. The most common types of mycotoxins affecting liquorice are aflatoxins and ochratoxin A. Ochratoxin A is produced by several species of Aspergillus and *Penicillium* and has been researched for its mitochondrial dysregulatory activity since the 1970s (Moore & Truelove, 1970; Meisner & Chan, 1974). This research continues today regarding the role of mitochondrial dysfunction in ochratoxin induced neurotoxicity (Pei et al., 2021). Aflatoxins are also produced by Aspergillus species and induce hepatotoxicity through damage to mitochondrial structure and function among other mechanisms (Hua et al., 2021). This would make mitochondrial functional analysis a useful tool for detecting fungal contamination, especially as it could detect whether levels are sufficient to be causing mitochondrial dysfunction, determining safe levels. However, aptasensors can detect mycotoxins at concentrations in the ranges of fg / mL and more rapidly than chromatographic techniques (Shkembi et al., 2021), so they will be able to detect mycotoxins at levels far below the European Food Standards Agency recommended safe dose of 0.4 µg / kg bw / day (Schrenk, 2020). Another issue that could arise if liquorice does protect mitochondria from a range of toxic substances, is that liquorice may mask some of the effects of the mycotoxins, potentially making liquorice a poor candidate for detecting mycotoxin contamination, and perhaps other toxic adulterants.
An area where functional mitochondrial analysis would be useful is in the degradation of herbs stored for extended periods without fungal contamination. Stafford, Jäger & van Griggs (2001) followed up a study by Taylor et al. (1995) to evaluate the bioactive properties of herbs after 6 years of storage and found that, of 19 plants tested for antimicrobial and antifungal activity, 3 lost all activity, 6 retained all activity, and 10 had partial activity. Of the 13 that had been found to have light enhanced activity in the earlier study, all but one had lost this ability. This study is particularly interesting because it used the same herbs as tested 6 years ago and stored in cotton bags at room temperature, the method common in a traditional Nepalese dispensary, so the results were not influenced by potential variations in harvest. Stafford et al. (2005) used a 1 year accelerated ageing model of 9 medicinal herbs, discovering an increase in their antimicrobial capacity but reducing their cyclooxygenase-1 inhibiting capacity, suggesting that some storage may be advisable depending on the intended use of the herb. Laher et al. (2013) compared materials stored for 3 years in airtight containers against those picked fresh from the same site and found changes in their composition but not always to their detriment. Senna petersiana (Bolle) Lock, Fabaceae, had higher levels of all measured content (phenols, flavonoids, tannins and iridoids) than the recently picked dried plant and they concluded that effects were species-specific and that, as well as potential reduced efficacy, there could be an issue with enhanced potency and therefore toxicity among stored materials. However, their use of materials picked from the same site 3 years later could also be influenced by variations in the two harvests and an examination of the stored materials upon collection should also have been undertaken. Ali et al. (2018) looked at the effects of storage on the phenolic content of Piper betle L., Piperaceae and found that dark storage at 5°C affected the content of eugenol and isoeugenol and completely degraded 4-allyl-1,2-diacetoxybenzene with a half-life of 62 days. Only hydroxychavicol was unaffected. Booker et al. (2014) analysed samples of turmeric (C. longa) from a number of storage facilities in India and found significantly lower levels of volatile tumerone in stored samples. In contrast, Rowshan, Bahmanzadegan & Saharkhiz (2013) found that storage of Thymus daenensis Celak., Lamiaceae for 3 months at room temperature caused a decrease in compounds with lower boiling temperatures such as α -pinene,

 α -terpinene and myrcene, but also caused a transformation of γ -terpinene and p-cymene into carvacrol and thymol, which are important quality indicators, resulting in an overall increase of 23-26%. Despite these contradictory findings, the one consistent thread that emerges is that storage does affect medicinal herb constituents and may affect their clinical efficacy in unknown ways.

This would make mitochondrial analysis a suitable method of evaluating the effects of storage on potency because it would bypass the issue of which precise compounds are causing the increase or decrease in efficacy, and instead assess the potency of the herbs on a model of their eventual therapeutic action in living cells. This could also elucidate which components are responsible for their therapeutic effects if combined with chemical analyses to compare which components increase in the more potent herbs. In terms of this study, it would be useful to evaluate the effects of various storage conditions on the mitochondrial modulating effects of liquorice to determine what effect, if any, storage has. This could also be a limiting factor on the existing results, as the freeze dried material was stored at -20°C for the duration of the study, while the adulterated material was prepared from the same tea sample almost 2 years later that had been stored at ambient temperature in the laboratory, inside its original packaging. The samples were also from a commercial product instead of being obtained directly from a plantation, so their previous storage conditions cannot be verified, although Pukka's "Fair for Life" scheme involves paying their farmers a basic minimum wage (Anastov, 2023), reducing the likelihood that herbs are stored for long periods, waiting for market price increases. In order to test this, the liquid from the first extraction of the XLTE could also have been tested for its mitochondrial modulating functions to see if these had been compromised by the storage, or the double extraction.

24.2.2. Wild Harvested vs. Cultivated Liquorice

The superiority of wild liquorice compared to cultivated varieties (Brinckmann, 2020; Wang, H. et al., 2021) introduces the prospect of deliberate or accidental adulteration of the wild harvested crop with a cultivated one. As

demand for liquorice grows and excessive harvesting of wild liquorice reduces its natural habitat (Khaitov et al., 2022), the incidences of suppliers adulterating their wild harvested stock with cultivated specimens are likely to increase. This is precisely the functional adulteration described by Tao Hongjing in the 5th century China (Liu, 2021), where herbs may look good to the untrained eye, but lack efficacy due to their growing conditions not meeting the strict requirements of *Daodi* (section 2.6.1). This would be impossible to detect by genetic analysis and challenging with HPTLC but mitochondrial functional analysis could be an appropriate test to use to determine whether a cultivated crop is capable of producing the same biological effect as genuine wild-harvested liquorice at an equivalent dose.

The first step would be to acquire some cultivated liquorice samples, discussed in section 20.2.1 as potentially weaker functional adulterants, to determine if mitochondrial functional analysis could detect the difference between the two samples and can be adopted as a method of detecting this form of adulteration. If no difference between the wild and cultivated samples can be found, then the superiority of wild harvested liquorice could be called into question and cultivated crops considered as an acceptable substitution to preserve the habitat of the remaining wild plants. Another possibility is that a difference is found in certain tests and not others, in which case the cultivated crop could be adopted for certain purposes where no difference is found, and the wild harvested plants used for the actions where mitochondrial functional analysis found it superior.

24.2.3. Mixing of Adulterated and Genuine Samples

This study shows that mitochondrial testing could be used to detect a pre-extracted liquorice sample from an original sample but in the real world, adulteration is not usually so clearly defined. Most often, an adulterant filler is mixed with the genuine product in order to increase the size and weight of the product being sold, while minimising the risks of detection (Newmaster et al., 2013). This presents an important question for quality evaluation of medicinal herbal teas using mitochondrial functional profiling: what proportion of OLTE:XLTE

is detectable with mitochondrial analysis? This question could be addressed by spiking the OLTE with the XLTE (or samples adulterated in other ways) to see when the effect disappears.

24.2.4. Real-World Sampling

The ultimate purpose of this study was to demonstrate that mitochondrial functional analysis could be used in quality evaluation within the herbal tea industry, and this means that it would eventually have to be tested on some real-world samples. As already mentioned in the limitations (section 22.2), attempts were made to acquire some real-world samples that had failed other forms of adulteration testing, but these may not have been the ideal samples to use. Having already been detected by another, faster method, they would have already been rejected and so mitochondrial analysis would only be confirming that the adulterated herbs also lack potency. The main strength of mitochondrial functional analysis is that it may be able to detect herbs that were genetically and chemically close enough to pass existing inspections, but lacking in potency. Therefore, the aforementioned methods of creating adulterated samples in the laboratory and confirming what forms and levels of adulteration can be detected by this method that other methods cannot, and then applying this to real-world samples would be a superior sequence. However, it could be useful to note if some real-world samples that had been adulterated were capable of passing mitochondrial functional analysis, for example, through the addition of drugs to an incorrect species that could pass mitochondrial functional tests. This could represent an important development in adulteration technology of herbal supply chains should this method become frequently adopted, as already discussed in section 21.3.

24.3. Building a Pipeline

In order to build a fast, efficient method of developing mitochondrial functional quality tests, it is necessary to develop a pipeline. Based on the results found to date, this can be divided into a few necessary stages:

A. Literature Review

The first stage in developing a mitochondrial functional profile of any herb or formula is to review the literature on the specific actions of the herb or herbs. This should draw upon scientific literature on the subject and common uses from herbal medical traditions. There is a precedent for using herbal medicinal traditions to generate hypotheses for scientific testing. Artemisinin, the primary treatment for malaria around the world today, was discovered through the study of a 4th century Chinese text by alchemist Ge Hong (葛洪), that revealed the original method to use sweet wormwood (*Artemesia annua* L., *Asteraceae*) for malaria was to take "a handful of *Qinghao* [sweet wormwood] immersed in two litres of water, wring out the juice and drink it all," which led to a lowering in temperature for the extraction, successful discovery of antiplasmodial activity through laboratory analysis and isolation of the active ingredient (Tu, 2016). A similar process, searching a 10th century Anglo-saxon herbal volume, has prompted scientific investigation into potential new antibiotics for drug resistant *Staphylococcus aureus* and *Pseudomonas aeruginosa* (Fuchs et al., 2018).

Following this approach, certain hypotheses can be generated that can be experimentally tested through mitochondrial functional analysis. These can include complete reproductions of experiments found in the literature, variations of experiments found in the literature, and novel experiments designed to test traditional indications that could reveal their mechanisms and lead to the publication of new research.

B. Reproduction of the Results

Any significant results that could be used for quality evaluation of the herb under analysis should be reproduced using verified samples to ensure they are consistent and reliable. This is important because reproducibility in scientific literature can be low (Baker, 2016; Errington et al., 2020) and adulteration in herbal supplies can be high (Booker A., Frommenwiler et al., 2016; Srirama et al., 2017). This could extend to the materials used in the scientific literature, so it is essential to ensure that any research used to develop a standard is obtained using a validated test on a verified sample to ensure the reproducibility of the result.

C. Investigation into Functional Adulteration

The next area of research would include common forms of adulteration, especially functional adulteration, as described in section 2.6.1, which may be detected through mitochondrial analysis and evade other methods. This information is not always easily available and may require research techniques beyond a literature review, including qualitative techniques such as surveys and interviews with experts involved in quality evaluation of the specific herb or formulas under investigation, farmers, market traders, industry specialists, professional groups and end users such as herbalists that may have experience of batch variability of their supplies. Besides researching the methods of adulteration that have been found or suspected, this fieldwork can also serve as a means to acquire real world samples for the later stages of the investigation.

D. Comparison of Significant Results with Adulterated Samples

To determine whether mitochondrial analysis can be used to differentiate the adulterated samples, it will be necessary to compare the results from the authenticated samples with those of an adulterated sample. If possible, these should be based on real-world concerns, using methods that have been discovered in stage C. These should be artificially created in the laboratory in order to determine the sensitivity of the test and, wherever possible, related to the kinds of functional adulteration that is the strength of this method to detect.

E. Publication of the Results on a Database

All results need to be made available on a database that can be accessed by quality control organisations. This is similar to the HPTLC Atlas (HPTLC Association, 2020), which publishes chromatographic profiles of herbs and the exact protocols to get optimal results, against which test samples can be compared. Herbal quality control organisations could contribute to the data, adding new tests, modifying existing ones for greater optimisation and challenging results which they cannot reproduce. Since the content of this database would be assays that have been replicated with enough frequency to be used as industry standards, it could become a benchmark for researchers to aspire towards since inclusion on the database would show their findings were consistent, similar to the adoption of specific protocols for controls in biological assays, and help mitigate the tendency of journals to only publish novel results (Sharma & Verma, 2019).

24.4. Mitochondrial Profiling of Other Herbs

The future herbs and formulas to be tested will be driven by the needs of industry to have standards for their particular products. Since the donor of the original material was Pukka Herbs, the future herbs that were planned were mainly dictated by the products they manufacture, starting with single herb teas such as liquorice, and gradually increasing in complexity to include two and three herb blends which could be compared against individual herb profiles, and then complex mixtures. These were cross referenced with the likelihood of them having a detectable effect on mitochondria that would enable their profiling based on table 7, section 20.2).

24.4.1. Cinnamon

Cinnamon (*C. verum* seu *cassia, Lauraceae*) has a history as long and rich as liquorice, being especially valued in CHM where the formula *Gui Zhi Tang* (桂枝 湯, "Cinnamon Twig Decoction") is said to be the "crown … of the Chinese herbal medicine" (Scheid et al., 2009, p. 17) and the major leader of all other formulae, around which the entire tradition of formula creation that characterises CHM has evolved (European TCM Association, 2019). Cinnamon usually refers to two main species: *C. verum* J.Presl (syn. *C. zeylanicum* Blume), known as Ceylon cinnamon or true cinnamon; and *C. cassia* (L.) J.Presl (syn. *C. aromaticum* Nees et

Neolitsea cassia (L.) Kosterm.), commonly referred to as Chinese cinnamon or cassia bark. Although Chinese cinnamon retains its traditional uses in acute infectious diseases (Bensky et al., 2004, pp.8-11; Chen & Chen, 2000, pp.40-42), much modern research has focused on the potential therapeutic properties of cinnamon in chronic, non-communicable diseases prevalent today, including the management of diabetes mellitus, glucose intolerance, dyslipidemia, obesity, arteriosclerosis, Alzheimer's and arthritis (Medagama, 2015; Hariri & Ghiasvand, 2016; Jiang, 2019). It also has a potential toxicity profile despite being a commonly consumed spice. The cassia species contains high levels of coumarin which has demonstrated hepatotoxic and carcinogenic properties (Abraham et al., 2010; Wang et al., 2013).

The mitochondrial modulating functions of cinnamon have already been discussed in section 3.2.5 but to briefly summarise, the metabolic nature of these disorders and involvement of oxidative stress suggest a mitochondrial mechanism. The research picture reflects this with 38 papers connecting cinnamon or one of its principal components with mitochondrial function. Only two of these attempted to profile its activity using the SeaHorse Flux Analyzer and only one looked at the herb itself (Li, X. et al., 2021). The other investigated e-cigarette liquids flavoured with cinnamaldehyde (Clapp et al., 2019). Li, X. et al. (2021) found that it induced uncoupling of the protons and electrons in the ETC leading to thermogenesis while Kwan et al. (2017) reported that it induced mitochondrial biogenesis, resulting in the browning of adipose tissue. These suggest a hormetic mechanism whereby uncoupling reduces the ATP being generated in favour of heat, resulting in activation of mitochondrial biogenesis to restore ATP levels and a net increase in mitochondrial ATP generation. Cinnamon also helped to restore insulin sensitivity and reduce apoptosis by reducing the sensitivity of the mPTP in the brains of mice fed a high fructose diet (Couturier et al., 2016). It also helped to stabilise serum glucose, insulin and cholesterol levels, reducing overall body weight while increasing mitochondrial biogenesis in skeletal muscle in another mouse model given a high fat diet (Song et al., 2017), potentially assisting with obesity that is also a causative factor in the development of diabetes mellitus.

These suggest several mitochondrial functional experiments that can be reproduced in order to generate a profile of cinnamon types. The use of antioxidant profiling has already been suggested to differentiate four species of cinnamon (Rana & Sheu, 2023) but their profiling only used simple antioxidant assays (DPPH and ABTS assays) rather looking at the functional effects cinnamon has on ROS in a complex biological system. Different cinnamon species could equally be evaluated using a DCFDA or MitoSOX assay, and the proposed uncoupling effects could be measured using a TMRE assay with its uncoupling potential compared against FCCP as a positive control. This would provide a more complete picture to enable comparison of the species. Uncoupling is a physiological process used to reduce the production of ROS (Hass & Barnstable, 2021), so it could be predicted that both of these assays could be used and the results compared to determine the uncoupling capacity of cinnamon species. Finally, Li, X. et al. (2021) used the Seahorse assay to measure uncoupling through an inhibited respiratory rate and raised proton leak meaning that this assay could also be utilised as a third reference point.

24.4.2. Ginger

Ginger (*Z. officinalis*) has enjoyed enormous popularity as both a culinary spice and medicinal herb throughout the world and over time (Spence, 2023). After liquorice, ginger is perhaps one of the most frequently used herbs in CHM, estimated to be in around half of all formulas (Shahrajabian, Sun & Cheng, 2019). It forms two distinct herbs in the Chinese pharmacopoeia, depending on whether it is fresh or dried (Bensky et al., 2004, p. ???), a division supported by modern chemometric analysis which reveals that heat sensitive gingerols are converted to shogaols as it dehydrates (Jolad et al., 2005).

One of the most common uses for ginger is in the alleviation of gastrointestinal disorders. Clinical trials showing a reduction in vomiting among children with acute gastroenteritis (Nocerino, 2021) have led to speculation that it could ameliorate the burden of child deaths from gastroenteritis related dehydration in parts of the world where pharmaceutical interventions are

unavailable (Matthews-King, 2018). It has also been researched for immunomodulatory activity, having an anti-inflammatory effect through Akt inhibition and NF-KB activation, while reducing oxidative stress via the Nrf2 signalling pathway activation (Ayustaningwarno et al., 2024).

As well as its potential in reducing infant mortality from infectious diseases, ginger has been researched into its role in alleviating the burden of non-communicable diseases such as metabolic disorders and cancer that affect industrialised nations. Seo, Kang & Fang (2021) found that ginger supplementation attenuated the effects of a high fructose diet which is known to induce mitochondrial dysfunction and oxidative stress. Oh et al. (2017) connected this to increased mitochondrial biogenesis in rats fed a high fat diet, with Alhamoud et al. (2023) finding that 6-gingerol may increase uncoupling protein 1 expression and thermogenesis leading to mitochondrial biogenesis and browning of adipose tissue, similar to cinnamon. In regards to cancer, ginger is suspected to act on several mechanisms, including cell cycle arrest, induction of apoptosis, disrupting redox balance, and inhibition of proliferation, angiogenesis, migration and dissemination (Zadorozhna & Mangieri, 2021). Mitochondria are central to apoptosis and redox balance in particular, with many of the other mechanisms having reliance on ATP that mitochondria produce. In one instance, Nedungadi et al. (2021) found that ginger extract induces apoptosis in MDA-MB-231 (triple negative breast cancer) and A549 (non-small cell lung cancer) cell lines by inducing ER stress and drastic reduction in $\Delta \Psi m$ and ATP, along with excessive ROS production. This resulted in translocation of apoptosis inducing factor to the nucleus and DNA fragmentation.

The importance of ginger as a common culinary spice and medicinal herb, used in lay and professional herbalism for communicable and non-communicable diseases, makes it important to understand. Furthermore, its activity on mitochondria may be behind many of these uses making it amenable to mitochondrial analysis but to date no Seahorse analysis has been conducted on ginger.

24.4.3. Tea

Tea (*C. sinensis* (L.) Kuntze, *Theaceae*) is an important herb to understand since the global industry is the largest in the world, worth an estimated US\$22.7 billion market by 2025 (Market Research Future, 2019), more than 5x the entire herbal tea market combined. It also has an ancient history of use in China with traces of tea being found in Western Han dynasty tombs (206 BCE - 9 BC) (Lu et al., 2016) and is still consumed for medicinal purposes as varied as anti-inflammatory, anti-carcinogenic, anti-mutagenic, antioxidative, antimicrobial, hypolipidemic, antidiabetic, angiogenesis prevention, memory enhancement, bone health improvement and skin protective effects when consumed internally (Saeed et al., 2017) and used as an active ingredient in cosmetics (Koch et al., 2019).

Some of these actions are likely to involve mitochondria, especially as some of its constituents are known to have mitochondrial modulatory effects. Caffeine has been demonstrated to induce mitochondrial biogenesis through activation of PGC-10 which may mimic or improve exercise adaptations (Yamada et al., 2022) and increase muscular endurance by acting as a Ca²⁺ ionophore, AMPK activator and triggering myokine release (Takada, Fumoto & Kinugawa, 2022). However, its actions on mitochondria can also induce cardiac toxicity in high enough doses (Sardão, Oliveira & Moreno, 2002; Wink et al., 1999). Epigallocatechin 3-gallate also has both therapeutic and toxic effects if consumed as a concentrated extract (Dekant et al., 2017; Hu et al., 2018), which are mitochondria mediated (Oliveira et al., 2016; James, Kennett & Lambert, 2018).

Despite the popularity of this drink and the mitochondrial modulation activity it possesses, its characterisation is still incomplete with no analysis using the Seahorse Flux Analyser and much disagreement about its exact mechanisms. Furthermore, there are few current reviews of its mitochondrial modulating activity, except some regarding the effect of caffeine of skeletal muscle (Yamada et al., 2022; Takada, Fumoto & Kinugawa, 2022) and the effects of epigallocatechin 3-gallate on the liver (Oliveira et al., 2016) making a thorough review of the current literature and some additional characterisation to fill the current knowledge gaps overdue.

24.4.4. Ginseng

Asian ginseng (*P. ginseng*) is another herb with a history dating back to the earliest classics of CHM and remains popular today. It has been trialled for improvements in psychomotor performance, physical performance, circulation, metabolism. the respiratory erectile dysfunction, alucose system, immunomodulation, mood regulation, antioxidant effects, cancer, menopausal symptoms and dry mouth, with the most promising results in the areas of glucose metabolism and moderating the immune response, suggesting potential implications for several diseases, including type 2 diabetes and chronic respiratory conditions (Shergis et al., 2013). This suggests that mitochondria may be the main target of ginsenosides, the active compounds found in ginseng. Current evidence suggests that ginsenosides may enhance mitochondrial respiration while suppressing glycolysis, encouraging greater reliance on OXPHOS, protect against Ca²⁺ induced apoptosis, and both stimulate mitochondrial biogenesis and fusion, as well as initiate fission and mitophagy of damaged mitochondria (Wang & Roh, 2020). Ginseng also has a favourable toxicology profile (Mancuso & Santangelo, 2017). While this herb has been tested using a full array of mitochondrial assays including the Seahorse Flux Analyzer (Huang et al., 2019), most methods favour alcohol extracts over water extracts making them poor representations of brewing a commercial tea blend, and many CHM formulas, which also favour water decoctions.

24.4.5. Ginkgo

Ginkgo (*G. biloba*) is uncommon to find in any commercial teas, but it is a very popular herbal supplement and received the most results on a search for herbs in relation to mitochondria (table 7, section 20.2). It is also one of the most adulterated herbs in the marketplace with as many as 94% of products being adulterated (Booker, A., Frommenwiler et al., 2016). Industry regulatory standards

demand a flavonoid content of 22-27% quercetin, kaempferol and isorhamnetin and 5-7% terpene lactones (ginkgolides A, B, C at 2.8 - 3.4% and bilobalide at 2.6 - 3.2%) and less than 5 ppm of toxic ginkgolic acids (EMA, 2014). These can all be detected using HPLC-MS techniques (Ding et al., 2006) but supplements have been found containing rutin and quercetin from much cheaper buckwheat sources, which remain difficult to differentiate from the same ginkgo derived flavonoids without using advanced equipment such as NMR spectroscopy (Booker, A., Frommenwiler et al., 2016).

Ginkgo has also attracted attention for its effects on mitochondria long before the current scientific trend with the majority of studies on its mitochondrial modulating activities being published from 1999 to 2012. These were mostly assessing its ability to prevent Alzheimer's disease through selective effects on enzymes that assemble the ETC resulting in a reduction in oxidative stress and an increase in ATP generation which, in turn, reduced β -amyloid plaques and extended lifespan in mouse and *C. elegans* models (Eckert, 2012). The toxic ginkgolic acid component also operates through a mitochondrial mechanism, inhibiting mitochondrial biogenesis and promoting excessive mitophagy (Wang et al., 2019). Indirect effects that may affect mitochondrial function include an inhibitory effect of platelet aggregating factors and vasodilatory action that may improve blood circulation, which will increase oxygenation of ischaemic tissues (Tao et al., 2020). Clinical trials have had mixed results with greater efficacy in relieving symptoms of cognitive impairment than in prevention (Lejri et al., 2019).

This combination of adulteration practices being common, using substances that closely imitate the genuine product making chemometric testing complex, and its proposed mechanism involving modulation of mitochondrial function, make ginkgo supplements a prime candidate for mitochondrial functional testing. One important finding that may emerge from this research is whether the more cheaply derived flavonoids also affect the mitochondrial modulating function of the supplements similarly to authentic ginkgo. If adulterated ginkgo supplements perform comparably in the mitochondrial functional testing, then there may be an argument to evaluate these cheaper flavonoids on their own and start producing products which openly declare a mixture of the two extractions, helping to protect this high demand, endangered plant (Sun, 1998) from overharvesting, while providing a cheaper product for consumers.

24.5. Mitochondrial Profiling of Blends and Formulas

Once some significant individual herbs have been tested, the next stage would be to examine some blends and formulas. An important aspect of formula creation is the principle of synergy between the herbs, creating a combined effect that is greater than the sum of its parts (Zhou et al., 2016). When testing a formula for its mitochondrial modulations in relation to a specific clinical effect, this can be compared against its individual herbs to find out if the effect is equal to the individual herb, greater, or less. For example, in relation to this study, if a formula is being examined for its potential antivenom effects, as Galen's Theriac formula was said to do (Karaberopoulos, Karamanou & Androutsos, 2012), it can be compared against the equivalent dosage of liquorice alone, both in equal dose and compared to the amount of liquorice that is in the formula. This will determine if any effect is solely due to the liquorice alone, or if simply taking an equivalent dose of liquorice may be sufficient. It is also possible that an equivalent dose of liquorice will have the same effect, but would exceed recommended toxicity profiles, suggesting that the formula may be effective at producing the detoxification result without triggering the effects of pseudoaldosteronism, hypokalemia and hypertension that liquorice alone can cause (Nazari, Rameshrad & Hosseinzadeh, 2017). It could also be observed if the detoxifying effect of liquorice enables other toxic herbs to retain their therapeutic effect with fewer side effects when included in a formula, as practised in Chinese herbal medicine (Bensky et al., 2004, p,734).

The aim of starting testing with liquorice was due to the ubiquity of its use in herbal formulas and in commercial medicinal herbal tea blends in the UK such as Pukka and Yogi Tea (Rolfe, 2023; Yogi Tea, 2023). It would be advisable to start examining the synergistic effects of liquorice and other herbs though gradually increasing complexity, starting with herb pairs that use liquorice to modify their actions. Fortunately there are several classical CHM formulas that utilise liquorice with another herb to modify its effect which include:

24.5.1. Cinnamon and Liquorice

Cinnamon and Liquorice decoction is a classic CHM formula from the *Discussion of Cold Damage* (c. 220 C.E.) that was originally described for heart palpitations after a patient suffering from an externally contracted illness had been encourage to sweat excessively (Zhang & Liu, 2016, p. 211-213). The same combination is sold in commercial blends of tea today (Pukka Herbs, 2020b) which would make it a particularly useful combination to understand, being of interest to CHM practitioners, industry and consumers alike.

24.5.2. Liquorice and Ginger

Liquorice and dry ginger is another combination that is derived from the *Discussion of Cold Damage* (c. 220 C.E.), originally described for when a patient suffering from an externally contracted illness experiences cold hands and feet, dry throat, vexation and agitation, and vomiting after being mistakenly given the wrong formula for their condition (Zhang & Liu, 2016, p. 110). This combination has also received some modern research into their interaction, whereby liquorice was observed to increase the extraction of 8-gingerol and 6-shogaol while reducing the "acrid and drying" 6-gingerol component (Jiang et al., 2020). This shift in composition through the addition of liquorice could have important implications for its therapeutic potential when compared against ginger alone.

24.5.3. Peony and Liquorice

Another classic CHM formula consisting of only a pair of herbs featuring liquorice that is of great importance in CHM is peony and liquorice decoction. This consists of peony (*P. lactiflora*) and liquorice, usually in equal measure and was originally used for cramps in the legs but has become widely used for a number of disorders that involve spasms, including abdominal pain (Guo, 2016, pp. 110 & 116-124). While there appears to be little research into peony and mitochondria, if this combination is expected to work on muscle cramps, then it is likely to be

having an effect on the mitochondria rich muscle cells. The precise causes of muscular spasms are unclear, but an ischaemic model has been proposed (Coletti, 2022) which would implicate mitochondrial dysfunction creating an energy crisis due to lack of oxygen inhibiting OXPHOS and therefore the ATP required for muscles to relax as well as contract. If this model is correct, then the findings of this study that liquorice helps maintain normal function on lower oxygen consumption, could be relevant to both the clinical application and theoretical development of this model.

24.5.4. More Complex Blends

One of the main advantages of using an end-point such as the effect on mitochondrial function to assess medicinal herbal teas is that a profile can be generated regardless of the complexity of the formula. Testing these particular herb pairs would have a particular advantage for scientific discovery though: All of the above herbs, cinnamon, peony, ginger and liquorice are combined together, with the addition of jujube berries (Ziziphus jujuba Mill., Rhamnaceae) in the classic CHM formula Cinnamon Twig Decoction (Gui Zhi Tang, 桂枝湯), discussed in section 24.4.1 above, around which the entire formula system of CHM has been devised. Therefore testing these individual herbs and pairs would then enable comparison to a small formula whose individual constituents and main combinations have already been profiled. The combined effect of each herb and pair can then be compared against the formula to determine the proportional influence of each herb, along with new effects arising from the entire formula not seen in any individual herb or combination to establish whether the whole formula has effects that cannot be attributed to any of its parts. Further small variations on this central formula, such as Cinnamon Twig Decoction with Extra Cinnamon (Gui Zhi Jia Gui Tang, 桂枝加桂湯), Cinnamon Twig Decoction minus Peony (Gui Zhi Qu Shao Yao Tang, 桂枝加芍藥湯) and Cinnamon Twig Decoction plus Peony (Gui Zhi Jia Shao Yao Tang, 桂枝加芍藥湯) which, as their name describes, have a higher amounts of cinnamon, or the peony removed, and Cinnamon Twig and Ginseng Decoction (Gui Zhi Ren Shen Tang, 桂枝人參湯) that utilises cinnamon, ginseng, liquorice, dried ginger and one other herb (Atractylodes macrocephala Koidz, *Asteraceae*) (Zhang & Liu, 2016, pp. 362, 73, 687 & 482). These each contain very similar ingredients which will have all been profiled individually and in combination with each other, but have very different indications, so they could become potential future targets against which to assess the effect of combinations and complex formulas compared against each other.

Chapter XXV

25. Conclusions

For the first time mitochondrial functional analysis was used to detect functional adulteration with poor quality herbs of correct species in medicinal herbal teas, an issue that has concerned herbalists for millennia. This has remained largely unaddressed in the literature due to the technical challenges it presents using common existing methods that excel at identifying added drugs or incorrect species, but encounter difficulties when these two criteria are met. To date the main methods of evaluating functional quality were to use animal or human subjects which present ethical and financial challenges today, preventing routine use. Instead, only post-hoc evaluation of the efficacy of medicinal herbal teas is usually performed by herbalists or the consumers of medicinal herbal teas who evaluate their effects based on experience. Mitochondrial functional analysis presents an alternative that can provide objective measurements based on the mitochondrial modulations that are produced by herbal tea extractions in relation to their intended use. This can be used alongside existing methods of quality evaluation to ensure that herbs are both correct species, do not have added drugs and produce effects on cell populations that relate to their clinical use.

The strength of mitochondrial functional analysis is that mitochondria have been identified as central to many biological functions and are the likely target for many herbs, especially those in the class of the adaptogens. This means that the complex effects brought about by herbs can be evaluated through their effects on a complex biological system, avoiding the issues of having to use marker compounds as a proxy. This is especially problematic when active compounds are unknown, or in the case of blends and formulas, where the potential ingredients responsible for the effects may complex and naturally vary between harvests and batches, or may be substituted for legitimate reasons. Despite this variability in chemotype and genotype, the intended effect on the biological system should be equivalent and can be evaluated directly through mitochondrial functional analysis. In this study several modulations in mitochondrial function were identified from exposure to a liquorice tea extract including modulations in ROS and respiration. In particular, the reduction in effects brought about by mitochondrial modulating drugs used in the Seahorse assay was correlated with the clinical use of liquorice in CHM and further assays identified the liquorice tea extract to have a cytoprotective effect in non-cancerous MCF10A cells treated with cisplatin that was not reproduced in either the cancerous MCF7 cell line, or when using the adulterated tea. This indicates a unique property to the unadulterated liquorice tea extract that could be used as a functional quality indicator. Furthermore, while HPTLC analysis was easily able to detect other common types of adulteration, such as incorrect species and different extraction methods, it was very challenging to detect any difference between the two extracts that produced significantly different mitochondrial functional profiles.

Other potential mitochondrial functional quality markers were also discovered including a drop in basal OCR in MCF10A cells despite no apparent decline in other mitochondrial functional markers. A potential link between this and the contemporary use of liquorice in CHM for cardiovascular disease was proposed and this was presented as another possible mitochondrial functional test that could be developed. As knowledge progresses, superior mitochondrial functional markers of the clinical effects of liquorice could be found and made into better standards. Other herbs and blends were presented for priority profiling and a pipeline process to streamline the procedure was presented. This pipeline process should look at traditional indications for the herbs as well as scientific literature in order to devise hypotheses which can be tested, adding to the knowledge base of these herbs and repeatable standards for comparison.

Over the course of this work, much has been learned that could improve the development of new quality assessment measures in future. A review of historical and contemporary literature illustrated the extent of the problem of adulteration of medicinal herbs. The developments in technology leading to modern methods were explored and the principle behind each was evaluated, finding their specific strengths and limitations. Recent discoveries regarding the role of mitochondria in

cellular adaptation and cell fate were reviewed, and the centrality of mitochondrial modulation to traditional medical techniques with non-specific targets was speculated. Techniques for evaluating mitochondrial function were examined and their mechanisms understood in order to develop a system for evaluating the effect of herbs on mitochondrial function. Practical skills were also developed in cell culture and experimental assays to achieve the results presented in this study. In discussing these results, the effect of liquorice on the drugs used in the Seahorse assay led to researching how cells detoxify harmful substances and reviewing studies suggesting that liquorice may encourage the expression of efflux transporters and maintain the mitochondrial ATP production necessary for them to function at maximal efficiency. Another unexpected result that implied that liquorice treated cells were consuming less oxygen than untreated cells led to further investigation on the antioxidant mechanisms of liquorice that proposed glycyrrhizin can capture electrons and act as a trap for short-lived radicals, simultaneously explaining the reduced oxidative stress in cancerous cell lines which deliberately produce radicals in order to maintain an optimal redox state, and the ability to maintain normal function while consuming less oxygen in liquorice treated non-cancerous cells. Further potential mitochondrial functional tests for liquorice and other popular herbs with mitochondrial modulating functions were also perused and discussed. These attest to the fact that by incorporating mitochondrial functional testing into existing methods of adulteration detection, a new dimension can be added to current approaches, addressing a deficiency in modern techniques that has been a concern to those using herbs for their medicinal benefits for thousands of years.

26. References

- Abcam (2022). *ab113852- TMRE Mitochondrial Membrane Potential Assay Kit.* Retrieved from https://www.abcam.com/products/assay-kits/tmre-mitochondrial-membran e-potential-assay-kit-ab113852.html (accessed 20 November 2022).
- Abcam (2024). *How to convert between different concentrations*. Retrieved from

https://www.abcam.com/help/how-to-convert-between-different-concentrat ions (accessed 23 January 2024).

- Abcam (2022). ab139476 ROS/Superoxide Detection Assay Kit (Cell-based). Retrieved from https://www.abcam.com/ps/products/139/ab139476/documents/ROS-Sup eroxide-Detection-Assay-protocol-book-v12c-ab139476%20(website).pdf (accessed 25 January 2024)
- Accum, F. C. (1822). A Treatise on Adulterations of Food, and Culinary Poisons. London: Longman, Hurst, Rees, Orme, and Brown. Retrieved from https://wellcomecollection.org/works/t3gc9j9p/items?canvas=17 (23 November 2022).
- Adlam, V. J., Harrison, J. C., Porteous, C. M., James, A. M., Smith, R. A. J., Murphy, M. P., & Sammut, I. A. (2005). Targeting an antioxidant to

mitochondria decreases cardiac ischemia-reperfusion injury. *The FASEB Journal*, *19*, 1088–1095. doi:10.1096/fj.05-3718com

- Advertising Standards Agency (2021). Advertising complementary and alternative medicine and therapies. *CAP News*. Retrieved from https://www.asa.org.uk/news/advertising-complementary-and-alternative-medicine-and-therapies.html (accessed 9 August 2023).
- Ageeva, A. A., Kruppa, A. I., Magin, I. M., Babenko, S. V., Leshina, T. V., & Polyakov, N. E. (2022). New Aspects of the Antioxidant Activity of Glycyrrhizin Revealed by the CIDNP Technique. *Antioxidants*, *11*, 1591. doi:10.3390/antiox11081591
- Agilent Technologies (2019). *Agilent Seahorse XF Glycolytic Rate Assay Kit: User Guide Kit 103344-100*. Retrieved from https://www.agilent.com/cs/library/usermanuals/public/103344-400.pdf (accessed 30 September 2023).
- AIB Staff. (2013, October 11). A Q&A on ATP Bioluminescence Assay. Quality Assurance and *Food Safety Magazine*. Retrieved from https://www.qualityassurancemag.com/article/aib1013-atp-bioluminescen ce-assay/ (accessed 19 November 2020).
- Al-Khafaji, M. (2008). The Avicenna Centre: New Developments in the Practice and Dispensing of Chinese Medicine. *RCHM Journal*, 5. Retrieved from

https://www.avicenna.co.uk/news-lifestyle/the-avicenna-centre-new-devel

opments-in-the-practice-and-dispensing-of-chinese-medicine-268/ (accessed 11 March 2023).

- Al-Khafaji, M. (2016). *Avicenna Herb Supplier Field Trip*. Brighton & Hove. 18th March 2018.
- Alam, J., Subhan, F., Ullah, I., Shahid, M., Ali, G., & Sewell, R. D. E. (2017).
 Synthetic and natural antioxidants attenuate cisplatin-induced vomiting. *BMC Pharmacology and Toxicology*, 18, 4.
 doi:10.1186/s40360-016-0110-9
- Alberts, B., Johnson, A., Lewis, J., Morgan, D., Raff, M., Roberts, K., & Walter, P. (2002). *The Mitochondrion*. Retrieved from https://www.ncbi.nlm.nih.gov/books/NBK26894/ (accessed 11 October 2023).
- Alhamoud, Y., Wu, J., Ahmad, M. I., Chen, F., Feng, F., & Wang, J. (2023).
 6-Gingerol induces browning of white adipose tissue to improve obesity through PI3K/AKT-RPS6-mediated thermogenesis pathway. *Food Frontiers*, *4*(3), 1285–1297. doi:10.1002/fft2.251
- Ałtyn, I., & Twarużek, M. (2020). Mycotoxin Contamination Concerns of
 Herbs and Medicinal Plants. *Toxins*, 12(3), 182.
 doi:10.3390/toxins12030182
- American Botanical Council. (2022, February 7). *Licorice Root. Herbal Medicine: Expanded Commission E.* Retrieved from

https://www.herbalgram.org/resources/expanded-commission-e-monogra phs/licorice-root/ (accessed 17 August 2023).

- American Type Culture Collection. (2022). hTERT Lung FibroblastCRL-4058TM.Retrievedhttps://www.atcc.org/api/pdf/product-sheet?id=CRL-4058(accessed 1October 2023).
- American Type Culture Collection. (2023). *BEAS-2B CRL-3588TM*. Retrieved from https://www.atcc.org/api/pdf/product-sheet?id=CRL-3588 (accessed 1 October 2023).
- Anastov, M. (2023). *Why we are certified Fair for Life*. Retrieved from https://www.pukkaherbs.com/uk/en/wellbeing-articles/why-we-are-certified -fair-for-life (accessed 28 September 2023).
- Apekey, T. A., & Khokhar, S. (2009). Survey on the Consumption of *Cinnamon-Containing Foods and Drinks by the UK Population*. Retrieved from

https://www.food.gov.uk/sites/default/files/media/document/final-report-cin namon-containing%20foods.pdf (accessed 17 July 2023).

APG IV. (2016). An update of the Angiosperm Phylogeny Group classification for the orders and families of flowering plants: APG IV.
Botanical Journal of the Linnean Society, 181(1), 1–20. doi:10.1111/boj.12385

- Archchinese.com. (2023). 氣. Retrieved from https://www.archchinese.com/chinese_english_dictionary.html?find=%E6 %B0%A3 (accessed 9 November 2023).
- Archibald, A. (2020). The Differences Between Wine and Whiskey Barrels,
 Explained. *Wine Enthusiast*. Retrieved from https://news.wine.co.za/news.aspx?NEWSID=36962 (accessed 31 December 2023).
- Arjumand, W., & Sultana, S. (2011). Glycyrrhizic acid: A phytochemical with a protective role against cisplatin-induced genotoxicity and nephrotoxicity. *Life Sciences*, *89*, 422–429. doi:10.1016/j.lfs.2011.06.016
- Aronson, J. (2018). Jeffrey Aronson: When I use a word . . . Liquorice and glycyrrhetinic acid. *The BMJ Opinion*. Retrieved from https://blogs.bmj.com/bmj/2018/09/14/jeffrey-aronson-when-i-use-a-word-liquorice-and-glycyrrhetinic-acid/ (accessed 15 December 2023).
- Arthur, H., Joubert, E., Beer, D. D., Malherbe, C. J., & Witthuhn, R. C. (2011). Phenylethanoid glycosides as major antioxidants in Lippia multiflora herbal infusion and their stability during steam pasteurisation of plant material. *Food Chemistry*, *127*, 581–588. doi:10.1016/j.foodchem.2011.01.044
- Aryal, S. (2018). *Nuclear Magnetic Resonance (NMR) Spectroscopy*. Retrieved from

https://microbenotes.com/nuclear-magnetic-resonance-nmr-spectroscopy/ (accessed 31 November 2020).

- Ashenhurst, J. (2022). *Introduction To UV-Vis Spectroscopy*. Retrieved from https://www.masterorganicchemistry.com/2016/09/16/introduction-to-uv-vi s-spectroscopy/ (accessed 19 April 2024).
- Associated Press. (2020). Man dies after eating bag of licorice every day for a few weeks. The Guardian, September 24, 2020. Retrieved from https://www.theguardian.com/us-news/2020/sep/24/man-licorice-dies-mas sachusetts (accessed 25 September 2020).
- Athreya, K., & Xavier, M. F. (2017). Antioxidants in the Treatment of Cancer.
 Nutrition and Cancer, 69, 1099–1104.
 doi:10.1080/01635581.2017.1362445
- Aura Nutrition. (2023a). *Jin Kui Shen Qi Wan*. Retrieved from https://auranutrition.com/collections/water/products/jin-kui-shen-qi-wan (accessed 10 June 2023).
- Aura Nutrition. (2023b). *Search results: 53 results for "liquorice."* Retrieved from https://auranutrition.com/search?q=liquorice (accessed 30 November 2023).
- Avula, B., Bae, J.-Y., Chittiboyina, A. G., Wang, Y.-H., Wang, M., Zhao, J., Ali, Z., Brinckmann, J. A., Li, J., Wu, C., & Khan, I. A. (2022). Chemometric analysis and chemical characterization for the botanical

identification of Glycyrrhiza species (G. glabra, G. uralensis, G. inflata, G. echinata and G. lepidota) using liquid chromatography-quadrupole time of flight mass spectrometry (LC-QToF). *Journal of Food Composition and Analysis*, *112*, 104679. doi:10.1016/j.jfca.2022.104679

- Ayustaningwarno, F., Anjani, G., Ayu, A. M., & Fogliano, V. (2024). A critical review of Ginger's (Zingiber officinale) antioxidant, anti-inflammatory, and immunomodulatory activities. *Frontiers in Nutrition*, *11*. doi:10.3389/fnut.2024.1364836
- Azadbakht, M., Monadi, T., Esmaeili, Z., Chabra, A., & Tavakoli, N. (2018). Formulation and evaluation of licorice shampoo in comparison with commercial shampoo. *Journal of Pharmacy And Bioallied Sciences*, *10*, 208. doi:10.4103/JPBS.JPBS_243_17
- Bailly, C., & Vergoten, G. (2020). Glycyrrhizin: An alternative drug for the treatment of COVID-19 infection and the associated respiratory syndrome? *Pharmacology & Therapeutics*, *214*, 107618. doi:10.1016/j.pharmthera.2020.107618
- Bailly, C. (2022). Forsythosides as essential components of *Forsythia* -based traditional chinese medicines used to treat inflammatory diseases and COVID-19. *World Journal of Traditional Chinese Medicine*, 8, 1. doi:10.4103/wjtcm.WJTCM_36_21
- Baker, M. (2016). 1,500 scientists lift the lid on reproducibility. *Nature*, 533, 452–454. doi:10.1038/533452a

- Ballard, J. W. O., & Towarnicki, S. G. (2020). Mitochondria, the gut microbiome and ROS. *Cellular Signalling*, 75, 109737.
 doi:10.1016/j.cellsig.2020.109737
- Banerjee, A., Chitnis, U. B., Jadhav, S. L., Bhawalkar, J. S., & Chaudhury, S.
 (2009). Hypothesis testing, type I and type II errors. *Industrial Psychiatry Journal*, *18*, 127. doi:10.4103/0972-6748.62274
- Bánki, O., Roskov, Y., Döring, M., Ower, G., Vandepitte, L., Hobern, D.,
 Remsen, D., Schalk, P., DeWalt, R. E., Keping, M., Miller, J., Orrell, T.,
 Aalbu, R., Adlard, R., Adriaenssens, E. M., Aedo, C., Aescht, E., Akkari,
 N., Alexander, S., ... The Royal Botanic Gardens Kew. (2022). Catalogue
 of Life Checklist. *Catalogue of Life*. doi:10.48580/dfq8
- Banoth, B., & Cassel, S. L. (2018). Mitochondria in innate immune signaling. *Translational Research*, *202*, 52–68. doi:10.1016/j.trsl.2018.07.014
- Bardell, D. (2004). The Invention of the Microscope. *Bios*, 75, 78–84. Retrieved from https://www.jstor.org/stable/4608700 (accessed 11 March 2023).
- Basist, P., Zahiruddin, S., Khan, M. U., Gautam, G., Jan, B., Khan, M. A., R., & Ahmad, S. (2022). Metabolite profiling Parveen, and nephroprotective potential of Glycyrrhiza glabra L. roots against cisplatin-induced nephrotoxicity in vitro and in vivo. Iranian Journal of Basic Medical Sciences. 25. 1286-1298. doi:10.22038/IJBMS.2022.65478.14404

- Bell, R. F., Moreira, V. M., Kalso, E. A., & Yli-Kauhaluoma, J. (2021). Liquorice for pain? *Therapeutic Advances in Psychopharmacology*, *11*, 204512532110248. doi:10.1177/20451253211024873
- Bensky, D., Clavey, S., Stöger, E., & Gamble, A. (2004). *Chinese Herbal Medicine: Materia Medica* (3rd edition). Seattle, WA: Eastland Press.
- Benzie, I. F. F., & Wachtel-Galor, S. (eds.) (2011). *Herbal Medicine: Biomolecular and Clinical Aspects* (2nd edition). Retrieved from https://www.ncbi.nlm.nih.gov/books/NBK92771/ (accessed 9 August 2023).
- Bernardi, N. F., Bordino, M., Bianchi, L., & Bernardi, L. (2017). Acute fall and long-term rise in oxygen saturation in response to meditation. *Psychophysiology*, *54*, 1951–1966. doi:10.1111/psyp.12972
- Beskina, O. A., Abramov, A. Y., Gabdulkhakova, A. G., Miller, A. V., Safronova, V. G., & Zamaraeva, M. V. (2007). Possible mechanisms for antioxidant activity of glycyrrhizinic acid. *Biochemistry (Moscow) Supplement Series B: Biomedical Chemistry*, 1, 29–34. doi:10.1134/S1990750807010040
- Betarbet, R., Sherer, T. B., MacKenzie, G., Garcia-Osuna, M., Panov, A. V.,
 & Greenamyre, J. T. (2000). Chronic systemic pesticide exposure reproduces features of Parkinson's disease. *Nature Neuroscience*, *3*, 1301–1306. doi:10.1038/81834

- Betts, G., & Bakhai, B. (2023). Accelerated shelf-life testing of food and beverage products. *Campden BRI*. Retrieved from https://www.campdenbri.co.uk/blogs/accelerated-shelf-life-testing.php (accessed 18 September 2023).
- Bicchi, C., Cordero, C., Iori, C., Rubiolo, P., Sandra, P., Yariwake, J. H., &
 Zuin, V. G. (2003). SBSE-GC-ECD/FPD in the Analysis of Pesticide
 Residues in *Passiflora alata* Dryander Herbal Teas. *Journal of Agricultural and Food Chemistry*, *51*(1), 27–33. doi:10.1021/jf020630f
- Bleier, L., & Dröse, S. (2013). Superoxide generation by complex III: From mechanistic rationales to functional consequences. *Biochimica et Biophysica Acta (BBA) Bioenergetics*, 1827, 1320–1331. doi:10.1016/j.bbabio.2012.12.002
- Bock, F. J., & Tait, S. W. G. (2020). Mitochondria as multifaceted regulators of cell death. *Nature Reviews Molecular Cell Biology*, *21*, 85–100. doi:10.1038/s41580-019-0173-8
- Boengler, K., Lochnit, G., & Schulz, R. (2018). Mitochondria "THE" target of myocardial conditioning. *American Journal of Physiology-Heart and Circulatory Physiology*, 315, H1215–H1231. doi:10.1152/ajpheart.00124.2018
- Bokil, A., & Sancho, P. (2019). Mitochondrial determinants of chemoresistance. *Cancer Drug Resistance*. doi:10.20517/cdr.2019.46

- Bolser, D. C., Hey, J. A., & Chapman, R. W. (1999). Influence of central antitussive drugs on the cough motor pattern. *Journal of Applied Physiology*, *86*, 1017–1024. doi:10.1152/jappl.1999.86.3.1017
- Bone, K., & Mills, S. (2012). *Principles and Practice of Phytotherapy: Modern Herbal Medicine* (2nd edition). Edinburgh: Churchill Livingstone.
- Booker, A., Agapouda, A., Frommenwiler, D. A., Scotti, F., Reich, E., & Heinrich, M. (2018). St John's wort (Hypericum perforatum) products an assessment of their authenticity and quality. *Phytomedicine*, *40*, 158–164. doi:10.1016/j.phymed.2017.12.012
- Booker, A., Frommenwiler, D., Reich, E., Horsfield, S., & Heinrich, M. (2016). Adulteration and poor quality of Ginkgo biloba supplements. *Journal of Herbal Medicine*, *6*, 79–87. doi:10.1016/j.hermed.2016.04.003
- Booker, A., Frommenwiler, D., Johnston, D., Umealajekwu, C., Reich, E., & Heinrich, M. (2014). Chemical variability along the value chains of turmeric (Curcuma longa): A comparison of nuclear magnetic resonance spectroscopy and high performance thin layer chromatography. *Journal of Ethnopharmacology*, *152*, 292–301. doi:10.1016/j.jep.2013.12.042
- Bracken, M. B. (2009). Why animal studies are often poor predictors of human reactions to exposure. *Journal of the Royal Society of Medicine*, *102*, 120–122. doi:10.1258/jrsm.2008.08k033

- Bravo-Sagua, R., Parra, V., López-Crisosto, C., Díaz, P., Quest, A. F. G., &
 Lavandero, S. (2017). Calcium Transport and Signaling in Mitochondria. *Comprehensive Physiology*, 7, 623–634. doi:10.1002/cphy.c160013
- Bremer, B. (2007). Linnaeus' sexual system and flowering plant phylogeny. *Nordic Journal of Botany*, 25, 5–6. doi:10.1111/j.0107-055X.2007.00098 12.x
- Brinckmann, J. A. (2020). The Long Road to Sustainable Licorice. Journal of Medicinal Plant Conservation: A United Plant Savers Publication, 19–21. Retrieved from https://www.researchgate.net/publication/342481218_The_long_road_to_ sustainable licorice (accessed 24 November 2022).
- Britti, E., Delaspre, F., Tamarit, J., & Ros, J. (2018). Mitochondrial calcium signalling and neurodegenerative diseases. *Neuronal Signaling*, 2. doi:10.1042/NS20180061
- Brown, D. (2014). *RHS Encyclopedia Of Herbs* (2nd ed.). London: Dorling Kindersley.
- Buckman, J. F., Hernández, H., Kress, G. J., Votyakova, T. V., Pal, S., & Reynolds, I. J. (2001). MitoTracker labeling in primary neuronal and astrocytic cultures: influence of mitochondrial membrane potential and oxidants. *Journal of Neuroscience Methods*, *104*, 165–176. doi:10.1016/S0165-0270(00)00340-X

- Burgess, J. K., Ketheson, A., Faiz, A., Rempel, K. A. L., Oliver, B. G., Ward,
 J. P. T., & Halayko, A. J. (2018). Phenotype and Functional Features of
 Human Telomerase Reverse Transcriptase Immortalized Human Airway
 Smooth Muscle Cells from Asthmatic and Non-Asthmatic Donors. *Scientific Reports*, *8*, 805. doi:10.1038/s41598-017-18429-0
- Burnstock, G. (2009). Acupuncture: A novel hypothesis for the involvement of purinergic signalling. *Medical Hypotheses*, 73, 470–472. doi:10.1016/j.mehy.2009.05.031
- Burtscher, J., Mallet, R. T., Pialoux, V., Millet, G. P., & Burtscher, M. (2022). Adaptive Responses to Hypoxia and/or Hyperoxia in Humans. *Antioxidants & Redox Signaling*, *37*, 887–912. doi:10.1089/ars.2021.0280
- Bush, J. F. (2002). "By Hercules! The More Common the Wine, the More Wholesome!" Science and the Adulteration of Food and Other Natural Products in Ancient Rome. *Food and Drug Law Journal*, *57*, 573–602.
 Retrieved from https://www.jstor.org/stable/26660496 (accessed 26 September 2021).
- Butler, A. (2022). Liquorice and Chinese herbal medicine: An epistemological challenge (V. Lo, D. Yang, & M. Stanley-Baker, eds.). doi:10.4324/9780203740262-58
- Cadenas, S. (2018). ROS and redox signaling in myocardial ischemia-reperfusion injury and cardioprotection. *Free Radical Biology and Medicine*, *117*, 76–89. doi:10.1016/j.freeradbiomed.2018.01.024

- Calabrese, E. J. (2013). Hormetic mechanisms. Critical Reviews in Toxicology, 43(7), 580–606. doi:10.3109/10408444.2013.808172
- Calabrese, E. J., & Baldwin, L. A. (1998). Hormesis as a biological hypothesis. *Environmental Health Perspectives*, *106*, 357–362. doi:10.1289/ehp.98106s1357
- Calabrese, Edward J., Dhawan, G., Kapoor, R., Iavicoli, I., & Calabrese, V. (2016). HORMESIS: A Fundamental Concept with Widespread Biological and Biomedical Applications. *Gerontology*, 62, 530–535. doi:10.1159/000441520
- Calabrese, E. J., Iavicoli, I., & Calabrese, V. (2013). Hormesis: Its impact on medicine and health. *Human & Experimental Toxicology*, *32*, 120–152. doi:10.1177/0960327112455069
- CAMAG. (n.d.). *Process Steps: Derivitization*. Retrieved from https://www.camag.com/process-steps/derivatization (accessed 16 December 2020).
- Cañigueral, S., Frommenwiler, D. A., Reich, E., & Vila, R. (2018). 'High performance thin-layer chromatography (HPTLC) in the quality control of herbal products' in D. Muñoz-Torrero, Y. Cajal, & J. M. Llobet (eds.), *Recent Advances in Pharmaceutical Sciences: Vol. VIII*. Delhi: AkiNik Publications. pp.119-136. Retrieved from https://diposit.ub.edu/dspace/bitstream/2445/128132/1/T_1548757317mu noz%207.pdf (accessed 16 September 2023).

- Cantó, C., & Auwerx, J. (2009). PGC-1α, SIRT1 and AMPK, an energy sensing network that controls energy expenditure. *Current Opinion in Lipidology*, 20(2), 98–105. doi:10.1097/MOL.0b013e328328d0a4
- Cardanho-Ramos, C., Faria-Pereira, A., & Morais, V. A. (2020). Orchestrating mitochondria in neurons: Cytoskeleton as the conductor. *Cytoskeleton*, 77(3–4), 65–75. doi:10.1002/cm.21585
- Carrillo-Garmendia, A., Madrigal-Perez, L. A., & Regalado-Gonzalez, C. (2023). The multifaceted role of quercetin derived from its mitochondrial mechanism. *Molecular and Cellular Biochemistry*. doi:10.1007/s11010-023-04833-w
- Carmeli, E., & Fogelman, Y. (2009). Antioxidant effect of polyphenolic glabridin on LDL oxidation. *Toxicology and Industrial Health*, *25*, 321–324. doi:10.1177/0748233709103034
- Carmines, E. L., Lemus, R., & Gaworski, C. L. (2005). Toxicologic evaluation of licorice extract as a cigarette ingredient. *Food and Chemical Toxicology*, *43*, 1303–1322. doi:10.1016/j.fct.2005.01.012
- Cayman Chemical. (2021). *Rhod-2 AM*. Retrieved from https://www.caymanchem.com/product/19355/rhod-2-amulates,557%20a nd%20581%20nm%2C%20respectively (accessed 29 January 2021).
- Cayman Chemical. (2022). Product Information: 18β-Glycyrrhetinic Acid,ItemNo.11845.Retrievedfrom

https://cdn.caymanchem.com/cdn/insert/11845.pdf (accessed 25 August 2024).

- Cayman Chemical. (2023). Catalase Assay Kit. Retrieved from https://cdn.caymanchem.com/cdn/insert/707002.pdf (accessed 8 November 2023).
- Cebi, N., Yilmaz, M. T., & Sagdic, O. (2017). A rapid ATR-FTIR spectroscopic method for detection of sibutramine adulteration in tea and coffee based on hierarchical cluster and principal component analyses. *Food Chemistry*, 229, 517–526. doi:10.1016/j.foodchem.2017.02.072
- Centre for the Promotion of Imports from Developing Countries. (2018). *Exporting cloves to Europe*. Retrieved from https://www.cbi.eu/market-information/spices-herbs/cloves (accessed 21 September 2023).
- Cerulli, A., Masullo, M., Montoro, P., & Piacente, S. (2022). Licorice (Glycyrrhiza glabra, G. uralensis, and G. inflata) and Their Constituents as Active Cosmeceutical Ingredients. *Cosmetics*, 9, 7. doi:10.3390/cosmetics9010007
- Chan, K.-T., Wu, H.-Y., Tin, W.-Y., But, P. P.-H., Cheung, S. C.-H., & Shaw,
 P.-C. (2024). Ethnopharmacology of five flowers herbal tea, a popular traditional beverage in Hong Kong and South China. *Journal of Ethnobiology and Ethnomedicine*, 20(1), 36. doi:10.1186/s13002-024-00674-z
- Charaka (2000). *Caraka Samhita* (P. V. Sharma, ed.) Uttar Pradesh: Chaukhambha Orientalia.
- Checa, J., & Aran, J. M. (2020). Reactive Oxygen Species: Drivers of Physiological and Pathological Processes. *Journal of Inflammation Research*, *13*, 1057–1073. doi:10.2147/JIR.S275595
- Cheel, J., Antwerpen, P. van, Tůmová, L., Onofre, G., Vokurková, D., Zouaoui-Boudjeltia, K., Vanhaeverbeek, M., & Nève, J. (2010). Free radical-scavenging, antioxidant and immunostimulating effects of a licorice infusion (Glycyrrhiza glabra L.). *Food Chemistry*, *122*, 508–517. doi:10.1016/j.foodchem.2010.02.060
- Chen, J. K., & Chen, T. T. (2000). *Chinese Medical Herbology and Pharmacology* (L. Crampton, ed.). City of Industry, CA: Art of Medicine.
- Chen, J.-L., Chang, C.-J., Wang, J.-Y., Wen, C.-S., Tseng, L.-M., Chang, W.-C., Noomhorm, N., Liu, H.-J., Chen, W.-S., Chiu, J.-H., & Shyr, Y.-M. (2014). In Vitro and In Vivo Effects of Jia-Wei-Xiao-Yao-San in Human Breast Cancer MCF-7 Cells Treated With Tamoxifen. *Integrative Cancer Therapies*, *13*, 226–239. doi:10.1177/1534735414520970
- Chen, Y., Feng, Q., Li, M., Cai, Z., Chen, Y., Wang, L., Teng, J., Chen, Y., Wang, W., Rein, G., Tang, B. Q., & Bai, X. (2020). The Effect of TCM Herbs on Mitochondrial Functions: The Linkage between Qi and Mitochondria. *Chinese Medicine*, 11(04), 127–141. doi:10.4236/cm.2020.114008

- Chen, M., Geng, D., Yang, X., Liu, X., Liu, S., Ding, P., Pang, Y., Du, M., Hu,
 X., & Wang, R. (2020). In Vitro Nephrotoxicity Induced by Herb-Herb
 Interaction between Radix Glycyrrhizae and Radix Euphorbiae
 Pekinensis. *Oxidative Medicine and Cellular Longevity*, 2020, 1–16.
 doi:10.1155/2020/6894751
- Chen, Y., Feng, Q., Li, M., Cai, Z., Chen, Y., Wang, L., Teng, J., Chen, Y., Wang, W., Rein, G., Tang, B. Q., & Bai, X. (2020). The Effect of TCM Herbs on Mitochondrial Functions: The Linkage between Qi and Mitochondria. *Chinese Medicine*, 11(04), 127–141. doi:10.4236/cm.2020.114008
- Chen, X., Liu, Z., Meng, R., Shi, C., & Guo, N. (2017). Antioxidative and anticancer properties of Licochalcone A from licorice. *Journal of Ethnopharmacology*, *198*, 331–337. doi:10.1016/j.jep.2017.01.028
- Chen, X.-Y., Ren, H.-H., Wang, D., Chen, Y., Qu, C.-J., Pan, Z.-H., Liu, X.-N., Hao, W.-J., Xu, W.-J., Wang, K.-J., Li, D.-F., & Zheng, Q.-S. (2019).
 Isoliquiritigenin Induces Mitochondrial Dysfunction and Apoptosis by Inhibiting mitoNEET in a Reactive Oxygen Species-Dependent Manner in A375 Human Melanoma Cells. *Oxidative Medicine and Cellular Longevity*, 2019, 1–12. doi:10.1155/2019/9817576
- Chen, X., Sun, X., Hua, H., Yi, Y., Li, H., & Chen, C. (2019). Quality evaluation of decoction pieces of Rhizoma Atractylodis Macrocephalae by near infrared spectroscopy coupled with chemometrics. *Spectrochimica*

Acta Part A: Molecular and Biomolecular Spectroscopy, 221, 117169. doi:10.1016/j.saa.2019.117169

- Chen, H., Tian, J., Guo, L., & Du, H. (2020). Caspase inhibition rescuesF1Fo ATP synthase dysfunction-mediated dendritic spine elimination.Scientific Reports, 10(1), 17589. doi:10.1038/s41598-020-74613-9
- Cheng, X., Qiu, L., & Wang, F. (2020). 18α-Glycyrrhetinic acid (GA) ameliorates fructose-induced nephropathy in mice by suppressing oxidative stress, dyslipidemia and inflammation. *Biomedicine & Pharmacotherapy*, *125*, 109702. doi:10.1016/j.biopha.2019.109702
- Cheng, X., Li, B., Han, Z., Zhang, F., Jiang, Z., Yang, J., Luo, Q., & Tang, L. (2022). Qualitative and quantitative analysis of the major components in Qinghao Biejia decoction by UPLC-Orbitrap Fusion-MS/MS and UPLC-QQQ-MS/MS and evaluation of their antibacterial activities. *Phytochemical Analysis*, *33*(5), 809–825. doi:10.1002/pca.3131
- Cheng, Y., Wu, X., Nie, X., Wu, Y., Zhang, C., Lee, S. M.-Y., Lv, K., Leung, G. P.-H., Fu, C., Zhang, J., & Li, J. (2022). Natural compound glycyrrhetinic acid protects against doxorubicin-induced cardiotoxicity by activating the Nrf2/HO-1 signaling pathway. *Phytomedicine*, *106*, 154407. doi:10.1016/j.phymed.2022.154407
- Cheung, E. (2023). Licorice compound suppresses 'silent killer' pancreatic cancer, Hong Kong study finds, paving way for improved treatment. *South China Morning Post.* Retrieved from

https://www.scmp.com/news/hong-kong/health-environment/article/32302 43/licorice-compound-found-be-effective-suppressing-silent-killer-pancrea tic-cancer-hong-kong-study (accessed 16 August 2023).

- Chistiakov, D. A., Shkurat, T. P., Melnichenko, A. A., Grechko, A. V., & Orekhov, A. N. (2018). The role of mitochondrial dysfunction in cardiovascular disease: a brief review. *Annals of Medicine*, *50*, 121–127. doi:10.1080/07853890.2017.1417631
- Choi, E. M. (2011). Glabridin protects osteoblastic MC3T3-E1 cells against antimycin A induced cytotoxicity. *Chemico-Biological Interactions*, 193, 71–78. doi:10.1016/j.cbi.2011.05.007
- Choi, Y.-M., Kim, H.-K., Shim, W., Anwar, M. A., Kwon, J.-W., Kwon, H.-K.,
 Kim, H. J., Jeong, H., Kim, H. M., Hwang, D., Kim, H. S., & Choi, S.
 (2015). Mechanism of Cisplatin-Induced Cytotoxicity Is Correlated to
 Impaired Metabolism Due to Mitochondrial ROS Generation. *PLOS ONE*,
 10, e0135083. doi:10.1371/journal.pone.0135083
- Choules, M. P., Bisson, J., Simmler, C., McAlpine, J. B., Giancaspro, G., Bzhelyansky, A., Niemitz, M., & Pauli, G. F. (2020). NMR reveals an undeclared constituent in custom synthetic peptides. *Journal of Pharmaceutical and Biomedical Analysis*, *178*, 112915. doi:10.1016/j.jpba.2019.112915

- Chrzanowski, J., Chrzanowska, A., & Graboń, W. (2021). Glycyrrhizin: An old weapon against a novel coronavirus. *Phytotherapy Research*, 35, 629–636. doi:10.1002/ptr.6852
- Chu, X. T., la Cruz, J. de, Hwang, S. G., & Hong, H. (2014). Tumorigenic Effects of Endocrine-disrupting Chemicals are Alleviated by Licorice (Glycyrrhiza glabra) Root Extract through Suppression of AhR Expression in Mammalian Cells. *Asian Pacific Journal of Cancer Prevention*, *15*, 4809–4813. doi:10.7314/APJCP.2014.15.12.4809
- Chua, L. S., Lau, C. H., Chew, C. Y., Ismail, N. I. M., & Soontorngun, N. (2018). Phytochemical profile of Orthosiphon aristatus extracts after storage: Rosmarinic acid and other caffeic acid derivatives. *Phytomedicine*, *39*, 49–55. doi:10.1016/j.phymed.2017.12.015
- Chueh, F.-S., Hsiao, Y.-T., Chang, S.-J., Wu, P.-P., Yang, J.-S., Lin, J.-J.,
 Chung, J.-G., & Lai, T.-Y. (2012). Glycyrrhizic acid induces apoptosis in
 WEHI-3 mouse leukemia cells through the caspase- and
 mitochondria-dependent pathways. *Oncology Reports*, *28*, 2069–2076.
 doi:10.3892/or.2012.2029
- Chun, K.-S., Kim, D.-H., & Surh, Y.-J. (2021). Role of Reductive versus Oxidative Stress in Tumor Progression and Anticancer Drug Resistance. *Cells*, *10*, 758. doi:10.3390/cells10040758
- Ciarlo, L., Marzoli, F., Minosi, P., Matarrese, P., & Pieretti, S. (2021). Ammonium Glycyrrhizinate Prevents Apoptosis and Mitochondrial

Dysfunction Induced by High Glucose in SH-SY5Y Cell Line and Counteracts Neuropathic Pain in Streptozotocin-Induced Diabetic Mice. *Biomedicines*, *9*, 608. doi:10.3390/biomedicines9060608

- CiteQ Biologics. (n.d.). BCA and Bradford protein assays. Retrieved from https://citeqbiologics.com/support/bca-and-bradford-protein-assays/ (accessed 31 August 2023).
- Clapp, P. W., Lavrich, K. S., van Heusden, C. A., Lazarowski, E. R., Carson, J. L., & Jaspers, I. (2019). Cinnamaldehyde in flavored e-cigarette liquids temporarily suppresses bronchial epithelial cell ciliary motility by dysregulation of mitochondrial function. *American Journal of Physiology-Lung Cellular and Molecular Physiology*, 316, L470–L486. doi:10.1152/ajplung.00304.2018
- Clark, J. (2019). *The Mass Spectrometer*. Retrieved from https://www.chemguide.co.uk/analysis/masspec/howitworks.html (accessed 13 November 2020).
- Clark, A., & Mach, N. (2017). The Crosstalk between the Gut Microbiota and
 Mitochondria during Exercise. *Frontiers in Physiology*, 8.
 doi:10.3389/fphys.2017.00319
- Cobley, J. N., Fiorello, M. L., & Bailey, D. M. (2018). 13 reasons why the brain is susceptible to oxidative stress. *Redox Biology*, *15*, 490–503. doi:10.1016/j.redox.2018.01.008

- Cocetta, V., Ragazzi, E., & Montopoli, M. (2020). 'Links between cancer metabolism and cisplatin resistance' in L. Galluzzi (ed.), *International Review of Cell and Molecular Biology*, vol 354. Elsevier. pp.107–164. doi:10.1016/bs.ircmb.2020.01.005
- Çoku, J., Booth, D. M., Skoda, J., Pedrotty, M. C., Vogel, J., Liu, K., Vu, A., Carpenter, E. L., Ye, J. C., Chen, M. A., Dunbar, P., Scadden, E., Yun, T. D., Nakamaru-Ogiso, E., Area-Gomez, E., Li, Y., Goldsmith, K. C., Reynolds, C. P., Hajnoczky, G., & Hogarty, M. D. (2022). Reduced ER–mitochondria connectivity promotes neuroblastoma multidrug resistance. *The EMBO Journal*, 41(8). doi:10.15252/embj.2021108272
- Coletti, R. H. (2022). The ischemic model of chronic muscle spasm and pain. *European Journal of Translational Myology*, *32*(1). doi:10.4081/ejtm.2022.10323
- Connors, B. W. (2012). Tales of a Dirty Drug: Carbenoxolone, Gap Junctions, and Seizures. *Epilepsy Currents*, *12*, 66–68. doi:10.5698/1535-7511-12.2.66
- Cooper, G. M. (2000). 'The Mechanism of Oxidative Phosphorylation' in *The Cell: A Molecular Approach* (2nd edition). Sunderland MA: Sinauer Associates. Retrieved from https://www.ncbi.nlm.nih.gov/books/NBK9885/ (accessed 11 October 2023).
- Corsi, F. M., Galgani, S., Gasparini, C., Giacanelli, M., & Piazza, G. (1983). Acute hypokalemic myopathy due to chronic licorice ingestion: report of a

case. The Italian Journal of Neurological Sciences, 4, 493–497. doi:10.1007/BF02125633

- Coulter, I., Snider, P., & Neil, A. (2019). Vitalism-A Worldview Revisited: A Critique Of Vitalism And Its Implications For Integrative Medicine. *Integrative Medicine (Encinitas, Calif.)*, 18, 60–73. Available from https://www.ncbi.nlm.nih.gov/pubmed/32549817 (accessed 28 January 2024).
- Cousins, J. N., & Fernández, G. (2019). *The impact of sleep deprivation on declarative memory*. doi:10.1016/bs.pbr.2019.01.007
- Couturier, K., Hininger, I., Poulet, L., Anderson, R. A., Roussel, A.-M., Canini, F., & Batandier, C. (2016). Cinnamon intake alleviates the combined effects of dietary-induced insulin resistance and acute stress on brain mitochondria. *The Journal of Nutritional Biochemistry*, 28, 183–190. doi:10.1016/j.jnutbio.2015.10.016
- Cowell, W., Ireland, T., Vorhees, D., & Heiger-Bernays, W. (2017). Ground Turmeric as a Source of Lead Exposure in the United States. *Public Health Reports*, *132*, 289–293. doi:10.1177/0033354917700109
- Cox, D. (2022). Can our mitochondria help to beat long Covid? *The Guardian*. Retrieved from https://www.theguardian.com/science/2022/jun/26/can-our-mitochondria-h elp-to-beat-long-covid (accessed 23 October 2022).

- Cross, T. (2016). What Happens When UV Detection Is Just Not Enough? *Thermo Scientific Analyte Guru*. Retrieved from https://analyteguru.com/what-happens-when-uv-detection-is-just-not-enou gh/ (accessed 15 November 2020).
- CS Analytical. (2020). *Comparison of Atomic Spectroscopy Techniques and the Advantages of ICP-MS vs AA & ICP-OES*. Retrieved from https://csanalytical.com/comparison-atomic-spectroscopy-techniques-adv antages-icp-ms-vs-aa-icp-oes/ (accessed 18 June 2024).
- Culpeper, N. (1653). *The Complete Herbal*. Retrieved from https://www.gutenberg.org/files/49513/49513-h/49513-h.htm (accessed 18 August 2023).
- Czechowski, T., Weathers, P. J., Brodelius, P. E., Brown, G. D., & Graham, I.
 A. (2020). Editorial: Artemisinin—From Traditional Chinese Medicine to Artemisinin Combination Therapies; Four Decades of Research on the Biochemistry, Physiology, and Breeding of Artemisia annua. *Frontiers in Plant Science*, *11*. doi:10.3389/fpls.2020.594565
- Dadashzadeh, K. (2017). Relevance of Helicobacter pylori dupA and OipA genotypes and development of gastric disease. *Biomedical Research*, *28*, 8179–8183. Retrieved from https://www.alliedacademies.org/articles/relevance-of-helicobacter-pyloridupa-and-oipa-genotypes-and-development-of-gastric-disease.pdf (accessed 2 October 2023).

- Danaboina, G., Mishra, S., Sharma, D., Kumar, A., Mehra, V., Kumar, R., Dahiya, M., Dhobi, M., Kalaiselvan, V., Raghuvanshi, R. S., & Twiari, R. (2023). Targeted enrichment of Glycyrrhiza glabra root extract and bioactive compound(s) quantification—A potential scaffold for phytopharmaceutical ingredient development. *Chinese Journal of Analytical Chemistry*, *51*(5), 100247. doi:10.1016/j.cjac.2023.100247
- Dartier, J., Lemaitre, E., Chourpa, I., Goupille, C., Servais, S., Chevalier, S.,
 Mahéo, K., & Dumas, J.-F. (2017). ATP-dependent activity and
 mitochondrial localization of drug efflux pumps in doxorubicin-resistant
 breast cancer cells. *Biochimica et Biophysica Acta (BBA) General Subjects*, *1861*, 1075–1084. doi:10.1016/j.bbagen.2017.02.019
- Dasari, S., & Tchounwou, P. B. (2014). Cisplatin in cancer therapy: Molecular mechanisms of action. *European Journal of Pharmacology*, 740, 364–378. doi:10.1016/j.ejphar.2014.07.025
- Datta, S., & Jaiswal, M. (2021). Mitochondrial calcium at the synapse. *Mitochondrion*, *59*, 135–153. doi:10.1016/j.mito.2021.04.006
- de Castro, O., Comparone, M., di Maio, A., del Guacchio, E., Menale, B., Troisi, J., Aliberti, F., Trifuoggi, M., & Guida, M. (2017). What is in your cup of tea? DNA Verity Test to characterize black and green commercial teas. *PLOS ONE*, 12(5), e0178262. doi:10.1371/journal.pone.0178262
- de Ligt, M., Timmers, S., & Schrauwen, P. (2015). Resveratrol and obesity: Can resveratrol relieve metabolic disturbances? *Biochimica et Biophysica*

Acta (BBA) - Molecular Basis of Disease, 1852(6), 1137–1144. doi:10.1016/j.bbadis.2014.11.012

- de Vries, A. A. F. (2022, January 7). (h)iAM what I am. Nature Biomedical Engineering. Retrieved from https://go.nature.com/31xAQCF (accessed 1 October 2023).
- Dekant, W., Fujii, K., Shibata, E., Morita, O., & Shimotoyodome, A. (2017).
 Safety assessment of green tea based beverages and dried green tea extracts as nutritional supplements. *Toxicology Letters*, 277, 104–108. doi:10.1016/j.toxlet.2017.06.008
- Demine, S., Renard, P., & Arnould, T. (2019). Mitochondrial Uncoupling: A
 Key Controller of Biological Processes in Physiology and Diseases. *Cells*, *8*, 795. doi:10.3390/cells8080795
- Deng, H., & Shen, X. (2013). The Mechanism of Moxibustion: Ancient Theory and Modern Research. *Evidence-Based Complementary and Alternative Medicine*, *2013*, 1–7. doi:10.1155/2013/379291
- Dentali, S. (2013). Organoleptic Analysis of Herbal Ingredients. Retrieved from

https://ahpa.gomembers.com/AHPAResources/TechnicalGuidance/AHPA GuidanceDocuments/TabId/401/ArtMID/1244/ArticleID/1019/Organoleptic -Analysis-of-Herbal-Ingredients.aspx (accessed 16 September 2023).

- Deutch, M. R., Grimm, D., Wehland, M., Infanger, M., & Krüger, M. (2019).Bioactive Candy: Effects of Licorice on the Cardiovascular System.*Foods*, *8*, 495. doi:10.3390/foods8100495
- DeVito, N. J., & Goldacre, B. (2019). Catalogue of bias: publication bias.
 BMJ Evidence-Based Medicine, 24, 53–54.
 doi:10.1136/bmjebm-2018-111107
- Dhariwala, M. Y., & Ravikumar, P. (2019). An overview of herbal alternatives in androgenetic alopecia. *Journal of Cosmetic Dermatology*, jocd.12930. doi:10.1111/jocd.12930
- Ding, Y., Brand, E., Wang, W., & Zhao, Z. (2022). Licorice: Resources, applications in ancient and modern times. *Journal of Ethnopharmacology*, *298*, 115594. doi:10.1016/j.jep.2022.115594
- Ding, S., Dudley, E., Plummer, S., Tang, J., Newton, R. P., & Brenton, A. G. (2006). Quantitative determination of major active components in *Ginkgo biloba* dietary supplements by liquid chromatography/mass spectrometry. *Rapid Communications in Mass Spectrometry*, 20, 2753–2760. doi:10.1002/rcm.2646
- Ding, K., Xu, Y., Li, N., & Xu, H. (2022). Research on the Reuse of Licorice Residue. *Highlights in Science, Engineering and Technology*, 14, 378–385.
 Retrieved from https://drpress.org/ojs/index.php/HSET/article/view/1850/1769 (accessed 13 September 2023).

- Diomede, L., Beeg, M., Gamba, A., Fumagalli, O., Gobbi, M., & Salmona, M.(2021). Can Antiviral Activity of Licorice Help Fight COVID-19 Infection?*Biomolecules*, *11*, 855. doi:10.3390/biom11060855
- Dong, F., Xie, M., Xu, M., Lu, L., Miao, Y., Zhang, P., Li, X., Gui, X., & Liu, R. (2023). Quality consistency evaluation of chemical composition and pharmacology of Shaoyao-Gancao decoction dispensing granules and traditional decoction. *Biomedical Chromatography*, 37(12). doi:10.1002/bmc.5745
- Dong, T. T. X., Zhao, K. J., Gao, Q. T., Ji, Z. N., Zhu, T. T., Li, J., Duan, R., Cheung, A. W. H., & Tsim, K. W. K. (2006). Chemical and Biological Assessment of a Chinese Herbal Decoction Containing Radix Astragali and Radix Angelicae Sinensis: Determination of Drug Ratio in Having Optimized Properties. *Journal of Agricultural and Food Chemistry*, *54*, 2767–2774. doi:10.1021/jf0531631
- Drasler, B., Karakocak, B. B., Tankus, E. B., Barosova, H., Abe, J., Sousa de Almeida, M., Petri-Fink, A., & Rothen-Rutishauser, B. (2020). An Inflamed Human Alveolar Model for Testing the Efficiency of Anti-inflammatory Drugs in vitro. *Frontiers in Bioengineering and Biotechnology*, *8*. doi:10.3389/fbioe.2020.00987
- Drawell. (2023). *What Type of Gas Can Be Used in Gas Chromatography*. Drawell: Artist of Science. Retrieved from

https://www.drawellanalytical.com/what-type-of-gas-can-be-used-in-gas-c hromatography/ (accessed 10 July 2024).

- du Preez, B. V. P., de Beer, D., Moelich, E. I., Muller, M., & Joubert, E. (2020). Development of chemical-based reference standards for rooibos and honeybush aroma lexicons. *Food Research International*, 127, 108734. doi:10.1016/j.foodres.2019.108734
- Duan, B.-Z., Fang, H.-L., Li, X.-W., Huang, L.-F., Wang, P., & Chen, S.-L. (2017). Survey of traditional Dai medicine reveals species confusion and potential safety concerns: a case study on Radix Clerodendri Japonicum. *Chinese Journal of Natural Medicines*, *15*, 417–426. doi:10.1016/S1875-5364(17)30063-8
- Dyad Labs. (2018). *HPLC vs. UPLC*. Retrieved from https://dyadlabs.com/wp-content/uploads/2018/10/hplc_uplc_onesheet.pd f (accessed 11 November 2020).
- Eckert, A. (2012). Mitochondrial effects of *Ginkgo biloba extract*.
 International Psychogeriatrics, 24, S18–S20.
 doi:10.1017/S1041610212000531
- Einstein, A. (1997). *On the Quantum Theory of Radiation* (A. Engel, ed.). Retrieved from https://einsteinpapers.press.princeton.edu/vol6-trans/232 (accessed 15 March 2023).

- EI-Khoury, J. M. (2018). *Financial considerations for purchasing a mass spectrometer*. Retrieved from https://www.msacl.org/presenter_slides/2018_US_deborah.french_82527 _slides.pdf (accessed 14 November 2020).
- Elmore, S. (2007). Apoptosis: A Review of Programmed Cell Death. *Toxicologic Pathology*, *35*, 495–516. doi:10.1080/01926230701320337
- Emery Pharma. (2018). *A Step-By-Step Guide to 1D and 2D NMR Interpretation*. Retrieved from https://emerypharma.com/blog/a-step-by-step-guide-to-1d-and-2d-nmr-int erpretation/ (accessed 16 November 2020).
- Emwas, A.-H., Roy, R., McKay, R. T., Tenori, L., Saccenti, E., Gowda, G. A.
 N., Raftery, D., Alahmari, F., Jaremko, L., Jaremko, M., & Wishart, D. S.
 (2019). NMR Spectroscopy for Metabolomics Research. *Metabolites*, *9*, 123. doi:10.3390/metabo9070123
- Endlicher, R., Drahota, Z., Štefková, K., Červinková, Z., & Kučera, O.
 (2023). The Mitochondrial Permeability Transition Pore—Current Knowledge of Its Structure, Function, and Regulation, and Optimized Methods for Evaluating Its Functional State. *Cells*, *12*, 1273. doi:10.3390/cells12091273
- Ernst, E. (2011). How Much of CAM Is Based on Research Evidence? *Evidence-Based Complementary and Alternative Medicine*, *2011*, 1–3. doi:10.1093/ecam/nep044

- Errington, T. M., Mathur, M., Soderberg, C. K., Denis, A., Perfito, N., Iorns,
 E., & Nosek, B. A. (2021). Investigating the replicability of preclinical cancer biology. *eLife*, *10*. doi:10.7554/eLife.71601
- Eruslanov, E., & Kusmartsev, S. (2010). 'Identification of ROS Using Oxidized DCFDA and Flow-Cytometry' in D. Armstrong (ed.) Advanced Protocols in Oxidative Stress II. Methods in Molecular Biology, vol 594.
 Totowa, NJ: Humana Press. pp. 57–72. doi:10.1007/978-1-60761-411-1_4
- Esmaeili, H., & Karami, A. (2022). 'Phytochemical, Pharmacological, and Health Benefits of Glycyrrhiza glabra' in P. Balakrishnan & S. Gopi (eds.). *Handbook of Nutraceuticals and Natural Products*. Hoboken, NJ: John Wiley & Sons. pp.361–381. doi:10.1002/9781119746843.ch28
- Etheridge, C. (2023). Panel Discussion. *BHMA Stakeholders Meeting*. Chelsea Physic Garden, London. 21st September 2023.
- European Commission. (2023). EU coordinated action "From the Hives" (Honey 2021-2022). *Food Safety*. Retrieved from https://food.ec.europa.eu/safety/eu-agri-food-fraud-network/eu-coordinate d-actions/honey-2021-2022_en (accessed 30 March 2023).
- European Medicines Agency. (2018). *Cimicifugae rhizoma herbal medicinal product*. Retrieved from https://www.ema.europa.eu/en/medicines/herbal/cimicifugae-rhizoma (accessed 31 July 2024).

European Medicines Agency. (2012). *Community herbal monograph on Glycyrrhiza glabra L. and/or Glycyrrhiza inflata Bat. and/or Glycyrrhiza uralensis Fisch., radix.* Retrieved from https://www.ema.europa.eu/en/documents/herbal-monograph/final-comm unity-herbal-monograph-glycyrrhiza-glabra-l/glycyrrhiza-inflata-bat/glycyrr hiza-uralensis-fisch-radix en.pdf (accessed 21 November 2020).

European Medicines Agency. (2012). *Community herbal monograph on Glycyrrhiza glabra L. and/or Glycyrrhiza inflata Bat. and/or Glycyrrhiza uralensis Fisch., radix.* Retrieved from https://www.ema.europa.eu/en/documents/herbal-monograph/final-comm unity-herbal-monograph-glycyrrhiza-glabra-l/glycyrrhiza-inflata-bat/glycyrr hiza-uralensis-fisch-radix_en.pdf (accessed 23 November 2022).

European Medicines Agency. (2014). *Assessment report on Ginkgo biloba L., folium.* Retrieved from https://www.ema.europa.eu/en/documents/herbal-report/draft-assessment -report-ginkgo-biloba-l-folium_en.pdf (accessed 13 Novemerb 2023).

- European TCM Association. (2019). *Perspectives: On Gui Zhi Tang*. Retrieved from https://etcma.org/latest-news/perspectives-on-gui-zhi-tang/ (accessed 28 January 2024).
- FairWild Foundation. (2010). FairWild Standard Version 2.0. Retrieved from https://static1.squarespace.com/static/5bec424b297114f64cb908d8/t/5cc

9724ee4966be23ada7273/1556705876096/FairWild-Standard-V2.pdf (accessed 18 September 2023).

- Fan, S., Li, B., Tian, Y., Feng, W., & Niu, L. (2022). Comprehensive characterization and identification of chemical constituents in Yangwei decoction using ultra high performance liquid chromatography coupled with electrospray ionization quadrupole time-of-flight tandem mass spectrometry. *Journal of Separation Science*, 45(5), 1006–1019. doi:10.1002/jssc.202100723
- Fang, Y., Wang, Z., Quan, Q., Li, Z., Pan, K., Lei, Y., Yao, X., Li, X., Shen,
 X., Koidis, A., & Lei, H. (2023). Developing an ultrasensitive immunochromatographic assay for authentication of an emergent fraud aminopyrine in herbal tea. *Food Chemistry*, 406, 135065. doi:10.1016/j.foodchem.2022.135065
- Farhan, S., Shamsi, S., Alam, T., & Perveen, A. (2020). Saffron (Crocus sativus L.): A Review of its Ethnopharmacological value. *American Journal of PharmTech Research*, 10, 41–58. doi:10.46624/ajptr.2020.v10.i4.005
- Félix-Silva, J., Gomes, J. A. S., Fernandes, J. M., Moura, A. K. C., Menezes, Y. A. S., Santos, E. C. G., Tambourgi, D. v., Silva-Junior, A. A., Zucolotto, S. M., & Fernandes-Pedrosa, M. F. (2018). Comparison of two Jatropha species (Euphorbiaceae) used popularly to treat snakebites in Northeastern Brazil: Chemical profile, inhibitory activity against Bothrops

erythromelas venom and antibacterial activity. *Journal of Ethnopharmacology*, 213, 12–20. doi:10.1016/j.jep.2017.11.002

- Feng, Y., Hu, J., Wang, F., Li, B., Qian, Q., Wang, X., & Niu, L. (2022).
 Quality evaluation of the classical prescription, Danggui Jianzhong decoction, using ultra-high-performance liquid chromatography fingerprint, chemical pattern recognition, and network pharmacology. *Journal of Separation Science*, *45*(20), 3838–3851. doi:10.1002/jssc.202200327
- Feng, X., Ding, L., & Qiu, F. (2015). Potential drug interactions associated with glycyrrhizin and glycyrrhetinic acid. *Drug Metabolism Reviews*, 47, 229–238. doi:10.3109/03602532.2015.1029634
- Ferdowsian, H. R., & Gluck, J. P. (2015). The Ethical Challenges of Animal Research. *Cambridge Quarterly of Healthcare Ethics*, 24, 391–406. doi:10.1017/S0963180115000067
- Fernando, W., Rupasinghe, H. P. V., & Hoskin, D. W. (2019). Dietary phytochemicals with anti-oxidant and pro-oxidant activities: A double-edged sword in relation to adjuvant chemotherapy and radiotherapy? *Cancer Letters*, 452, 168–177. doi:10.1016/j.canlet.2019.03.022
- Ferreres, F., Grosso, C., Gil-Izquierdo, A., Valentão, P., Azevedo, C., & Andrade, P. B. (2014). HPLC-DAD-ESI/MSn analysis of phenolic compounds for quality control of Grindelia robusta Nutt. and bioactivities.

Journal of Pharmaceutical and Biomedical Analysis, 94, 163–172. doi:10.1016/j.jpba.2014.01.046

- Fetissov, S. O., & Crook, M. A. (2011). What increased consumption of licorice may reveal in anorexia nervosa. *Nutrition*, 27, 853–854. doi:10.1016/j.nut.2011.05.004
- Finkel, T., Menazza, S., Holmström, K. M., Parks, R. J., Liu, J., Sun, J., Liu, J., Pan, X., & Murphy, E. (2015). The Ins and Outs of Mitochondrial Calcium. *Circulation Research*, *116*, 1810–1819. doi:10.1161/CIRCRESAHA.116.305484
- Fiore, C., Eisenhut, M., Ragazzi, E., Zanchin, G., & Armanini, D. (2005). A history of the therapeutic use of liquorice in Europe. *Journal of Ethnopharmacology*, 99, 317–324. doi:10.1016/j.jep.2005.04.015
- Fiore, C., Salvi, M., Palermo, M., Sinigaglia, G., Armanini, D., & Toninello, A. (2004). On the mechanism of mitochondrial permeability transition induction by glycyrrhetinic acid. *Biochimica et Biophysica Acta (BBA) Bioenergetics*, *1658*, 195–201. doi:10.1016/j.bbabio.2004.05.012
- Flaws, B. (1994). *Chinese Medicinal Wines and Elixirs*. Boulder, CO: Blue Poppy Press.
- Flockhart, M., Nilsson, L. C., Tais, S., Ekblom, B., Apró, W., & Larsen, F. J. (2021). Excessive exercise training causes mitochondrial functional

impairment and decreases glucose tolerance in healthy volunteers. *Cell Metabolism*, 33, 957-970.e6. doi:10.1016/j.cmet.2021.02.017

- Forsyth, J. E., Weaver, K. L., Maher, K., Islam, M. S., Raqib, R., Rahman,
 M., Fendorf, S., & Luby, S. P. (2019). Sources of Blood Lead Exposure in
 Rural Bangladesh. *Environmental Science & Technology*, *53*(19),
 11429–11436. doi:10.1021/acs.est.9b00744
- Fortune Business Insights. (2022). *Herbal Medicine Market*. Retrieved from https://www.fortunebusinessinsights.com/herbal-medicine-market-106320 (accessed 9 March 2023).
- Foster, S. (2013). Exploring the Peripatetic Maze of Black Cohosh Adulteration: A Review of the Nomenclature, Distribution, Chemistry, Market Status, Analytical Methods, and Safety. *HerbalGram*, *98*, 32–51. Retrieved from http://cms.herbalgram.org/herbalgram/issue98/hg98feat-blackcohosh.html (accessed 29 July 2024).
- Fraunhofer, J. (1815). Denkschriften der Königlichen Akademie der Wissenschaften zu München (Vol. 5). Munchen: Bayerische Akademie der Wissenschaften.
 Retrieved from https://ia802602.us.archive.org/14/items/denkschriftender05munc/denksc hriftender05munc.pdf (accessed 28 January 2024).
- Frommenwiler, D. A., Booker, A., Vila, R., Heinrich, M., Reich, E., & Cañigueral, S. (2019). Comprehensive HPTLC fingerprinting as a tool for

a simplified analysis of purity of ginkgo products. *Journal of Ethnopharmacology*, 243. doi:10.1016/j.jep.2019.112084

- Frost, J. (2019). Using Post Hoc Tests with ANOVA. *Statistics by Jim*. Retrieved from https://statisticsbyjim.com/anova/post-hoc-tests-anova/ (accessed 31 August 2023).
- Frost, R., Bhamra, S. K., Pendry, B., & Heinrich, M. (2021). COVID-19 and herbal practice: A United Kingdom practitioner survey. Advances in Integrative Medicine, 8, 256–260. doi:10.1016/j.aimed.2021.09.003
- Fruehauf, H. (1995). Commonly Used Chinese Herb Formulas for the Treatment of Mental Disorders. *Journal of Chinese Medicine*, 48. Retrieved from https://classicalchinesemedicine.org/chinese-herb-formulas-treatment-me ntal-disorders/ (accessed 21 August 2023).
- Fruehauf, H. (2010). Gancao Xiexin Tang: A Forgotten Key Remedy for the Treatment of Toxic Skin Conditions. *Classical Chinese Medicine*. Retrieved from https://classicalchinesemedicine.org/wp-content/uploads/2016/03/fruehau f_gcxxtENG.pdf (accessed 1 October 2023).
- Fu, Y., Chen, J., Li, Y.-J., Zheng, Y.-F., & Li, P. (2013). Antioxidant and anti-inflammatory activities of six flavonoids separated from licorice. *Food Chemistry*, *141*, 1063–1071. doi:10.1016/j.foodchem.2013.03.089

- Fuchs, A. L., Weaver, A. J., Tripet, B. P., Ammons, M. C. B., Teintze, M., & Copié, V. (2018). Characterization of the antibacterial activity of Bald's drug resistant Staphylococcus evesalve against aureus and Pseudomonas aeruginosa. PLOS ONE, 13, e0208108. doi:10.1371/journal.pone.0208108
- Fuhrmann, D. C., & Brüne, B. (2017). Mitochondrial composition and function under the control of hypoxia. *Redox Biology*, *12*, 208–215. doi:10.1016/j.redox.2017.02.012
- Gafner, S., Blumenthal, M., Foster, S., Cardellina, J. H., Khan, I. A., & Upton, R. (2023). Botanical Ingredient Forensics: Detection of Attempts to Deceive Commonly Used Analytical Methods for Authenticating Herbal Dietary and Food Ingredients and Supplements. *Journal of Natural Products*, *86*, 460–472. doi:10.1021/acs.jnatprod.2c00929
- Gandhi, N. M., Maurya, D. K., Salvi, V., Kapoor, S., Mukherjee, T., & Nair, C.
 K. K. (2004). Radioprotection of DNA by Glycyrrhizic Acid Through Scavenging free Radicals. *Journal of Radiation Research*, 45(3), 461–468. doi:10.1269/jrr.45.461
- Gangele, P., & Paliwal, M. (2020). Upavasa (fasting) for health & longevity: an overview. *CellMed*, *10*, 10.1-10.5. doi:10.5667/CellMed.2020.0010
- Gannon, N. P., Schnuck, J. K., Mermier, C. M., Conn, C. A., & Vaughan, R.A. (2015). trans-Cinnamaldehyde stimulates mitochondrial biogenesis

through PGC-1α and PPARβ/δ leading to enhanced GLUT4 expression. *Biochimie*, *119*, 45–51. doi:10.1016/j.biochi.2015.10.001

- George, S., & Abrahamse, H. (2020). Redox Potential of Antioxidants in
 Cancer Progression and Prevention. *Antioxidants*, 9, 1156.
 doi:10.3390/antiox9111156
- Gerontakos, S. E., Casteleijn, D., Shikov, A. N., & Wardle, J. (2020). A Critical Review to Identify the Domains Used to Measure the Effect and Outcome of Adaptogenic Herbal Medicines. *The Yale Journal of Biology and Medicine*, 93, 327–346. Available from https://www.ncbi.nlm.nih.gov/pubmed/32607092 (accessed 28 January 2024).
- Ghosh, S. (2019). Cisplatin: The first metal based anticancer drug. *Bioorganic Chemistry*, *88*, 102925. doi:10.1016/j.bioorg.2019.102925
- Giddings, E. L., Champagne, D. P., Wu, M.-H., Laffin, J. M., Thornton, T. M., Valenca-Pereira, F., Culp-Hill, R., Fortner, K. A., Romero, N., East, J., Cao, P., Arias-Pulido, H., Sidhu, K. S., Silverstrim, B., Kam, Y., Kelley, S., Pereira, M., Bates, S. E., Bunn, J. Y., Fiering, S., Matthews, D., Robey, R., Stich, D., D'Alessandro, A., Rincon, M. (2021). Mitochondrial ATP fuels ABC transporter-mediated drug efflux in cancer chemoresistance. *Nature Communications*, *12*, 2804. doi:10.1038/s41467-021-23071-6

- GinSen. (n.d.). *Jin Gui Shen Qi Wan*. Retrieved from https://www.ginsen-london.com/infertility-treatments/jin-gui-shen-qi-tang/ (accessed 10 June 2023).
- Glass, J. B., Elbon, C. E., & Williams, L. D. (2023). Something old, something new, something borrowed, something blue: the anaerobic microbial ancestry of aerobic respiration. *Trends in Microbiology*, *31*, 135–141. doi:10.1016/j.tim.2022.08.006
- Goedeke, L., & Shulman, G. I. (2021). Therapeutic potential of mitochondrial uncouplers for the treatment of metabolic associated fatty liver disease and NASH. *Molecular Metabolism*, 46, 101178. doi:10.1016/j.molmet.2021.101178
- Gordon, J., Brown, M., & Reynolds, M. (2018). Cell-Based Methods for Determination of Efficacy for Candidate Therapeutics in the Clinical Management of Cancer. *Diseases*, *6*, 85. doi:10.3390/diseases6040085
- Graebin, C. S. (2018). The Pharmacological Activities of Glycyrrhizinic Acid
 ("Glycyrrhizin") and Glycyrrhetinic Acid. *Sweetners*, 245–261.
 doi:10.1007/978-3-319-27027-2_15
- Gray, N. E., Zweig, J. A., Caruso, M., Zhu, J. Y., Wright, K. M., Quinn, J. F.,
 & Soumyanath, A. (2018). Centella asiatica attenuates hippocampal mitochondrial dysfunction and improves memory and executive function in β-amyloid overexpressing mice. *Molecular and Cellular Neuroscience*, 93, 1–9. doi:10.1016/j.mcn.2018.09.002

- Grew, N. (1682). The Anatomy of Plants: with an idea of a philosophical history of plants: and several other lectures read before the Royal Society. Retrieved from https://archive.org/details/mobot31753000008869 (accessed 25 September 2021).
- Grieve, M. (1931). Licorice. Retrieved from https://www.botanical.com/botanical/mgmh/l/liquor32.html (accessed 18 August 2023).
- Griggs, J. K., Manandhar, N. P., Towers, G. H. N., & Taylor, R. S. L. (2001).
 The effects of storage on the biological activity of medicinal plants from
 Nepal. *Journal of Ethnopharmacology*, 77, 247–252.
 doi:10.1016/S0378-8741(01)00291-4
- Grundlingh, J., Dargan, P. I., El-Zanfaly, M., & Wood, D. M. (2011).
 2,4-Dinitrophenol (DNP): A Weight Loss Agent with Significant Acute Toxicity and Risk of Death. *Journal of Medical Toxicology*, *7*, 205–212. doi:10.1007/s13181-011-0162-6
- Guan, H., Wang, Q., Mei, Y., Ran, J., Zeng, F., Cai, H., Wang, D., Yang, S.,
 Zhang, M., Shi, Y., Liao, S., & Li, P. (2024). A multistep approach for
 exploring quality markers of Shengjiang Xiexin decoction by integrating
 plasma pharmacochemistry-pharmacokinetics-pharmacology. *Journal of Pharmaceutical and Biomedical Analysis*, 241, 115999.
 doi:10.1016/j.jpba.2024.115999

- Guan, Y., Li, F., Hong, L., Yan, X., Tan, G., He, J., Dong, X., Bao, M., & Xie,
 Q. (2012). Protective effects of liquiritin apioside on cigarette smoke-induced lung epithelial cell injury. *Fundamental & Clinical Pharmacology*, *26*, 473–483. doi:10.1111/j.1472-8206.2011.00956.x
- Guilhaus, M. (2005). *Mass Spectrometry* | *Time-of-Flight*. doi:10.1016/B0-12-369397-7/00347-2
- Guilhaus, M. (2005). Mass Spectrometry | Time-of-Flight. In *Encyclopedia of Analytical* Science (pp. 412–423). Elsevier.
 doi:10.1016/B0-12-369397-7/00347-2
- Guo, L., Cho, S. Y., Kang, S. S., Lee, S.-H., Baek, H.-Y., & Kim, Y. S. (2007).
 Orthogonal array design for optimizing extraction efficiency of active constituents from Jakyak-Gamcho Decoction, the complex formula of herbal medicines, Paeoniae Radix and Glycyrrhizae Radix. *Journal of Ethnopharmacology*, *113*, 306–311. doi:10.1016/j.jep.2007.06.013
- Guo, J., Shang, E., Zhao, J., Fan, X., Duan, J., Qian, D., Tao, W., & Tang, Y. (2014). Data mining and frequency analysis for licorice as a "Two-Face" herb in Chinese Formulae based on Chinese Formulae Database. *Phytomedicine*, *21*, 1281–1286. doi:10.1016/j.phymed.2014.07.006
- Gustafson, C. (2015). Mark Blumenthal: Quality and Efficacy of Herbal Medicines. *Integrative Medicine (Encinitas, Calif.)*, 14, 54–59. Available from https://www.ncbi.nlm.nih.gov/pubmed/26770158 (accessed 28 January 2024).

- Guzelmeric, E., Ristivojević, P., Vovk, I., Milojković-Opsenica, D., & Yesilada, E. (2017). Quality assessment of marketed chamomile tea products by a validated HPTLC method combined with multivariate analysis. *Journal of Pharmaceutical and Biomedical Analysis*, *132*, 35–45. doi:10.1016/j.jpba.2016.09.030
- Ha, Y., Wang, T., Li, J., Li, J., Lu, R., Li, J., Chen, L., & Gan, P. (2020).
 Herb-Drug Interaction Potential of Licorice Extract and Paclitaxel: A
 Pharmacokinetic Study in Rats. *European journal of drug metabolism and pharmacokinetics*, 45(2), 257–264. doi:10.1007/s13318-019-00593-5
- Hagen, S. J. (2021). Mucosal defense: gastroduodenal injury and repair mechanisms. *Current Opinion in Gastroenterology*, *37*, 609–614. doi:10.1097/MOG.000000000000775
- Hahm, Y. H., Cho, K., & Ahn, Y. H. (2021). Compositional Characteristics of Glucuronide Conjugates in Regional Glechoma hederacea var. longituba Herbal Extracts Using a Set of Polyphenolic Marker Compounds. *Plants*, *10*(11), 2353. doi:10.3390/plants10112353
- Halt, M. (1998). Moulds and mycotoxins in herb tea and medicinal plants. *European Journal of Epidemiology*, 14, 269–274.
 doi:10.1023/a:1007498613538
- Han, X., Song, Z., Liu, J., Zhang, Y., Wu, M., & Liu, H. (2023). Effects of Different Drying Methods on the Quality of Bletilla striata Scented Tea. *Molecules*, 28(6), 2438. doi:10.3390/molecules28062438

- Han, J., Wang, D., Yu, B., Wang, Y., Ren, H., Zhang, B., Wang, Y., & Zheng,
 Q. (2014). Cardioprotection against Ischemia/Reperfusion by
 Licochalcone B in Isolated Rat Hearts. *Oxidative Medicine and Cellular Longevity*, 2014, 1–11. doi:10.1155/2014/134862
- Hanahan, D. (2022). Hallmarks of Cancer: New Dimensions. *Cancer Discovery*, *12*, 31–46. doi:10.1158/2159-8290.CD-21-1059
- Hanahan, D., & Weinberg, R. A. (2000). The Hallmarks of Cancer. *Cell*, *100*(1), 57–70. doi:10.1016/S0092-8674(00)81683-9
- Hanahan, D., & Weinberg, R. A. (2011). Hallmarks of Cancer: The Next Generation. *Cell*, *144*, 646–674. doi:10.1016/j.cell.2011.02.013
- Haneef, J., Shaharyar, M., Husain, A., Rashid, M., Mishra, R., Siddique, N.
 A., & Pal, M. (2013). Analytical methods for the detection of undeclared synthetic drugs in traditional herbal medicines as adulterants. *Drug Testing and Analysis*, *5*, 607–613. doi:10.1002/dta.1482
- Hao, W., Chang, C.-P. B., Tsao, C.-C., & Xu, J. (2010). Oligomycin-induced
 Bioenergetic Adaptation in Cancer Cells with Heterogeneous Bioenergetic
 Organization. *Journal of Biological Chemistry*, 285, 12647–12654.
 doi:10.1074/jbc.M109.084194
- Haq, I. ul, Taj, R., Nafees, M., & Hussain, A. (2024). Mycotoxin detection in selected medicinal plants using chromatographic techniques. *Biomedical Chromatography*, *38*(4). doi:10.1002/bmc.5831

- Hara, T., Matsui, H., & Shimizu, H. (2014). Suppression of Microbial Metabolic Pathways Inhibits the Generation of the Human Body Odor Component Diacetyl by Staphylococcus spp. *PLoS ONE*, *9*, e111833. doi:10.1371/journal.pone.0111833
- Hariri, M., & Ghiasvand, R. (2016). 'Cinnamon and Chronic Diseases' in S.
 Gupta, S. Prasad & B. Aggarwal (eds.), *Drug Discovery from Mother Nature. Advances in Experimental Medicine and Biology*, vol. 929. Cham: Springer. pp. 1–24. doi:10.1007/978-3-319-41342-6_1
- Harman, D. (1956). Aging: A Theory Based on Free Radical and Radiation
 Chemistry. *Journal of Gerontology*, *11*, 298–300.
 doi:10.1093/geronj/11.3.298
- Harper, D. (1998). Early Chinese Medical Literature. London: Routledge.
- Harris, I. S., & DeNicola, G. M. (2020). The Complex Interplay between Antioxidants and ROS in Cancer. *Trends in Cell Biology*, *30*, 440–451. doi:10.1016/j.tcb.2020.03.002
- Hasan, A. A., Kalinina, E., Nuzhina, J., Volodina, Y., Shtil, A., & Tatarskiy, V. (2023). Potentiation of Cisplatin Cytotoxicity in Resistant Ovarian Cancer SKOV3/Cisplatin Cells by Quercetin Pre-Treatment. *International Journal of Molecular Sciences*, *24*, 10960. doi:10.3390/ijms241310960

- Hass, D. T., & Barnstable, C. J. (2021). Uncoupling proteins in the mitochondrial defense against oxidative stress. *Progress in Retinal and Eye Research*, 83, 100941. doi:10.1016/j.preteyeres.2021.100941
- Hassall, A. H. (1876). Food: its adulterations, and the methods for their detection. London: Long Mans, Green & Co. Retrieved from https://wellcomecollection.org/works/z3kguj4b/items (accessed 23 November 2022).
- Hattori, M., Sakamoto, T., Kobashi, K., & Namba, T. (1983). Metabolism of Glycyrrhizin by Human Intestinal Flora. *Planta Medica*, *48*, 38–42. doi:10.1055/s-2007-969875
- Hayashi, H., & Sudo, H. (2009). Economic importance of licorice. *Plant Biotechnology*, 26, 101–104. Retrieved from https://www.jstage.jst.go.jp/article/plantbiotechnology/26/1/26_1_101/_pdf (accessed 25 August 2023).
- Hayashi, H., Hattori, S., Inoue, K., Khodzhimatov, O., Ashurmetov, O., Ito, M., & Honda, G. (2003). Field Survey of Glycyrrhiza Plants in Central Asia
 (3). Chemical Characterization of G. glabra Collected in Uzbekistan. *Chemical and Pharmaceutical Bulletin*, *51*(11), 1338–1340.
 doi:10.1248/cpb.51.1338
- Hayes, J. D., Dinkova-Kostova, A. T., & Tew, K. D. (2020). Oxidative Stress in Cancer. *Cancer Cell*, 38, 167–197. doi:10.1016/j.ccell.2020.06.001

- He, S. (2013). Chinese herbal dose in ancient and modern times: a comparative study. *Journal of Traditional Chinese Medicine*, *33*, 268–271. doi:10.1016/S0254-6272(13)60138-7
- He, F., Wan, G., & Chen, J. (2022). Pancreatic lipase and alpha-glucosidase inhibitors screening from *Schisandra chinensis* based on spectrum-effect relationship and ultra-high-performance liquid chromatography-tandem mass spectrometry. *Journal of Separation Science*, *45*(23), 4198–4208. doi:10.1002/jssc.202200541
- He, J.-R., Yu, S.-G., Tang, Y., & Illes, P. (2020). Purinergic signaling as a basis of acupuncture-induced analgesia. *Purinergic Signalling*, 16, 297–304. doi:10.1007/s11302-020-09708-z
- He, Z., Ning, N., Zhou, Q., Khoshnam, S. E., & Farzaneh, M. (2020).
 Mitochondria as a therapeutic target for ischemic stroke. *Free Radical Biology and Medicine*, 146, 45–58.
 doi:10.1016/j.freeradbiomed.2019.11.005
- He, Y., Ci, X., Xie, Y., Yi, X., Zeng, Y., Li, Y., & Liu, C. (2019). Potential detoxification effect of active ingredients in liquorice by upregulating efflux transporter. *Phytomedicine*, 56, 175–182. doi:10.1016/j.phymed.2018.10.033
- Heinrich, M. (2015). Quality and safety of herbal medical products: regulation and the need for quality assurance along the value chains.

British Journal of Clinical Pharmacology, 80, 62–66. doi:10.1111/bcp.12586

- Hejazi, I. I., Khanam, R., Mehdi, S. H., Bhat, A. R., Moshahid Alam Rizvi,
 M., Islam, A., Thakur, S. C., & Athar, F. (2017). New insights into the antioxidant and apoptotic potential of Glycyrrhiza glabra L. during hydrogen peroxide mediated oxidative stress: An in vitro and in silico evaluation. *Biomedicine & Pharmacotherapy*, *94*, 265–279. doi:10.1016/j.biopha.2017.06.108
- Henley, A. (2015). Pretreating with cannabidiol (CBD) enhances chemotherapy : evidence for mitochondrial 'priming'. PhD Thesis. Imperial College London. doi:10.25560/47970
- Henley, A. B., Yang, L., Chuang, K.-L., Sahuri-Arisoylu, M., Wu, L.-H., Bligh,
 S. W. A., & Bell, J. D. (2017). Withania somnifera Root Extract Enhances
 Chemotherapy through 'Priming'. *PLOS ONE*, *12*, e0170917.
 doi:10.1371/journal.pone.0170917
- Herb, M., & Schramm, M. (2021). Functions of ROS in Macrophages and
 Antimicrobial Immunity. *Antioxidants*, 10, 313.
 doi:10.3390/antiox10020313
- Hernández-Abreu, O., Durán-Gómez, L., Best-Brown, R., Villalobos-Molina,
 R., Rivera-Leyva, J., & Estrada-Soto, S. (2011). Validated liquid chromatographic method and analysis of content of tilianin on several extracts obtained from Agastache mexicana and its correlation with

vasorelaxant effect. Journal of Ethnopharmacology, 138(2), 487–491. doi:10.1016/j.jep.2011.09.041

- Herraez, E., Gonzalez-Sanchez, E., Vaquero, J., Romero, M. R., Serrano,
 M. A., Marin, J. J. G., & Briz, O. (2012). Cisplatin-Induced
 Chemoresistance in Colon Cancer Cells Involves FXR-Dependent and
 FXR-Independent Up-Regulation of ABC Proteins. *Molecular Pharmaceutics*, 9, 2565–2576. doi:10.1021/mp300178a
- Herst, P. M., & Berridge, M. V. (2007). Cell surface oxygen consumption: A major contributor to cellular oxygen consumption in glycolytic cancer cell lines. *Biochimica et Biophysica Acta (BBA) Bioenergetics*, 1767, 170–177. doi:10.1016/j.bbabio.2006.11.018
- Higgins, M., Izadi, A., & Kaviani, M. (2020). Antioxidants and Exercise
 Performance: With a Focus on Vitamin E and C Supplementation. *International Journal of Environmental Research and Public Health*, *17*, 8452. doi:10.3390/ijerph17228452
- Hirchaud, F., Hermetet, F., Ablise, M., Fauconnet, S., Vuitton, D., Prétet, J.-L., & Mougin, C. (2013). Isoliquiritigenin Induces Caspase-Dependent
 Apoptosis via Downregulation of HPV16 E6 Expression in Cervical
 Cancer Ca Ski Cells. *Planta Medica*, 79, 1628–1635.
 doi:10.1055/s-0033-1350956

- Hoek, J. B., Cahill, A., & Pastorino, J. G. (2002). Alcohol and mitochondria:
 A dysfunctional relationship. *Gastroenterology*, *122*, 2049–2063.
 doi:10.1053/gast.2002.33613
- Hoffman, D. (2003). *Medical Herbalism: The Science and Practice of Herbal Medicine*. Rochester, VM: Healing Arts Press.
- Hong, S. H., Cha, H.-J., Hwang-Bo, H., Kim, M. Y., Kim, S. Y., Ji, S. Y., Cheong, J., Park, C., Lee, H., Kim, G.-Y., Moon, S.-K., Yun, S. J., Chang, Y.-C., Kim, W.-J., & Choi, Y. H. (2019). Anti-Proliferative and Pro-Apoptotic Effects of Licochalcone A through ROS-Mediated Cell Cycle Arrest and Apoptosis in Human Bladder Cancer Cells. *International Journal of Molecular Sciences*, *20*, 3820. doi:10.3390/ijms20153820
- HPTLC Association. (2020). *HPTLC Atlas*. Retrieved from https://www.hptlc-association.org/atlas/hptlc-atlas.cfm (accessed 16 March 2020).
- HPTLC Association. (2019). *Glycyrrhiza glabra and Glycyrrhiza uralensis root, rhizome, and stolon*. Personal communication by email from Eike Reich, 5th October 2023.
- Hsu, Y.-L., Chia, C.-C., Chen, P.-J., Huang, S.-E., Huang, S.-C., & Kuo, P.-L. (2009). Shallot and licorice constituent isoliquiritigenin arrests cell cycle progression and induces apoptosis through the induction of ATM/p53 and initiation of the mitochondrial system in human cervical carcinoma HeLa

cells. *Molecular Nutrition & Food Research*, *53*, 826–835. doi:10.1002/mnfr.200800288

- Hsueh, J.-Y., Shih, C.-L., Huang, Y.-T., Su, Y.-C., Wang, C.-C., Ho, Y.-L., Yu, C.-C., Lee, S.-S., Li, W.-C., Lin, J.-G., Lin, L.-C., Ma, I.-T., Chang, Y.-S., Chang, K.-H., Chen, C.-F., Chen, C.-T., Chen, I.-S., Don, M.-J., Liu, W.-L., ... Lo, C.-F. (eds.). (2022). *Taiwan Herbal Pharmacopeia 4th Edition English Version*. Taipei: Ministry Health and Welfare. Retrieved from https://www.mohw.gov.tw/dl-81227-a4342f00-f6fe-4347-8a17-058b7d6d0 d89.html (accessed 28 January 2024).
- Hu, W., Hou, J., Liu, W., Gu, X., Yang, Y., Shang, H., & Zhang, M. (2023).
 Online pharmaceutical process analysis of Chinese medicine using a miniature mass spectrometer: Extraction of active ingredients as an example. *Journal of Pharmaceutical Analysis*, *13*(5), 535–543. doi:10.1016/j.jpha.2023.03.005
- Hu, M., Li, H., Ni, S., & Wang, S. (2023). The protective effects of Zhi-Gan-Cao-Tang against diabetic myocardial infarction injury and identification of its effective constituents. *Journal of Ethnopharmacology*, 309, 116320. doi:10.1016/j.jep.2023.116320
- Hu, J., Webster, D., Cao, J., & Shao, A. (2018). The safety of green tea and green tea extract consumption in adults Results of a systematic review. *Regulatory Toxicology and Pharmacology*, 95, 412–433. doi:10.1016/j.yrtph.2018.03.019
- Hu, C., Yang, J., Qi, Z., Wu, H., Wang, B., Zou, F., Mei, H., Liu, J., Wang,
 W., & Liu, Q. (2022). Heat shock proteins: Biological functions,
 pathological roles, and therapeutic opportunities. *MedComm*, *3*.
 doi:10.1002/mco2.161
- Hua, Z., Liu, R., Chen, Y., Liu, G., Li, C., Song, Y., Cao, Z., Li, W., Li, W., Lu,
 C., & Liu, Y. (2021). Contamination of Aflatoxins Induces Severe
 Hepatotoxicity Through Multiple Mechanisms. *Frontiers in Pharmacology*,
 11. doi:10.3389/fphar.2020.605823
- Huang, Q., Li, J., Chen, J., Zhang, Z., Xu, P., Qi, H., Chen, Z., Liu, J., Lu, J.,
 Shi, M., Zhang, Y., Ma, Y., Zhao, D., & Li, X. (2023). Ginsenoside compound K protects against cerebral ischemia/reperfusion injury via Mul1/Mfn2-mediated mitochondrial dynamics and bioenergy. *Journal of Ginseng Research*, 47, 408–419. doi:10.1016/j.jgr.2022.10.004
- Huang, Y., Kwan, K. K. L., Leung, K. W., Yao, P., Wang, H., Dong, T. T., & Tsim, K. W. K. (2019). Ginseng extracts modulate mitochondrial bioenergetics of live cardiomyoblasts: a functional comparison of different extraction solvents. *Journal of Ginseng Research*, *43*, 517–526. doi:10.1016/j.jgr.2018.02.002
- Huang, Y.-C., Kuo, C.-L., Lu, K.-W., Lin, J.-J., Yang, J.-L., Wu, R., Wu, P.-P.,
 & Chung, J.-G. (2016). 18α-Glycyrrhetinic Acid Induces Apoptosis of
 HL-60 Human Leukemia Cells through Caspases- and

Mitochondria-Dependent Signaling Pathways. *Molecules*, *21*, 872. doi:10.3390/molecules21070872

- Huang, X., Qin, J., & Lu, S. (2014). Magnesium isoglycyrrhizinate protects hepatic L02 cells from ischemia/reperfusion induced injury. *International Journal of Clinical and Experimental Pathology*, 7, 4755–4764. Available from https://www.ncbi.nlm.nih.gov/pubmed/25197346 (accessed 28 January 2024).
- Huang, J., Zhu, Y., Li, S., Jiang, H., Chen, N., Xiao, H., Liu, J., Liang, D.,
 Zheng, Q., Tang, J., & Meng, X. (2023). Licochalcone B confers protective
 effects against LPS-Induced acute lung injury in cells and mice through
 the Keap1/Nrf2 pathway. Redox Report, 28(1).
 doi:10.1080/13510002.2023.2243423
- Husni, E., Nahari, F., Wirasti, Y., Wahyuni, F. S., & Dachriyanus. (2015).
 Cytotoxicity study of ethanol extract of the stem bark of asam kandis (Garcinia cowa Roxb.) on T47D breast cancer cell line. *Asian Pacific Journal of Tropical Biomedicine*, 5, 249–252.
 doi:10.1016/S2221-1691(15)30013-7
- Hussain, H., Ali, I., Wang, D., Hakkim, F. L., Westermann, B., Ahmed, I.,
 Ashour, A. M., Khan, A., Hussain, A., Green, I. R., & Shah, S. T. A.
 (2021). Glycyrrhetinic acid: a promising scaffold for the discovery of anticancer agents. *Expert Opinion on Drug Discovery*, *16*, 1497–1516. doi:10.1080/17460441.2021.1956901

- iBOL (2020). DNA Barcoding: A tool for specimen identification and species discovery. Retrieved from https://ibol.org/about/dna-barcoding/ (accessed 16 November 2020).
- ICH (2019). Impurities: Guideline for Residual Solvents Q3C(R6). Retrieved from https://database.ich.org/sites/default/files/Q3C-R6_Guideline_ErrorCorrec tion_2019_0410_0.pdf (accessed 11 March 2023).
- Ilyas, M. (1980). Spices in India: III. *Economic Botany*, *34*, 236–259. doi:10.1007/BF02858643
- Imai, K., Takagi, Y., Iwazaki, A., & Nakanishi, K. (2013). Radical scavenging ability of glycyrrhizin. *Free Radicals and Antioxidants*, 3, 40–42. doi:10.1016/j.fra.2013.04.003
- Ingraham, P. (2019). Do You Believe in Qi? How to embrace a central concept of Eastern mysticism without being a flake. *Pain Science*. Retrieved from https://www.painscience.com/articles/do-you-believe-in-qi.php (accessed 30 November 2021).

Invitrogen. (2022). *MitoSOXTM Green and MitoSOXTM Red Mitochondrial Superoxide Indicators*. Retrieved from https://assets.thermofisher.com/TFS-Assets%2FLSG%2Fmanuals%2FM AN0028459-MitoSOXGreenMitoSOXRedMitochondrialSuperoxideIndicat ors-UG.pdf (accessed 25 January 2024).

- Isbrucker, R. A., & Burdock, G. A. (2006). Risk and safety assessment on the consumption of Licorice root (Glycyrrhiza sp.), its extract and powder as a food ingredient, with emphasis on the pharmacology and toxicology of glycyrrhizin. *Regulatory Toxicology and Pharmacology*, *4*6, 167–192. doi:10.1016/j.yrtph.2006.06.002
- Jain, R., Hussein, M. A., Pierce, S., Martens, C., Shahagadkar, P., & Munirathinam, G. (2022). Oncopreventive and oncotherapeutic potential of licorice triterpenoid compound glycyrrhizin and its derivatives: Molecular insights. *Pharmacological Research*, *178*, 106138. doi:10.1016/j.phrs.2022.106138
- James, L. (2015). A.G. Schneiderman Asks Major Retailers To Halt Sales Of Certain Herbal Supplements As DNA Tests Fail To Detect Plant Materials Listed On Majority Of Products Tested. Retrieved from https://ag.ny.gov/press-release/2015/ag-schneiderman-asks-major-retaile rs-halt-sales-certain-herbal-supplements-dna (accessed 25 September 2023).
- James, K. D., Kennett, M. J., & Lambert, J. D. (2018). Potential role of the mitochondria as a target for the hepatotoxic effects of (-)-epigallocatechin-3-gallate in mice. *Food and Chemical Toxicology*, *111*, 302–309. doi:10.1016/j.fct.2017.11.029
- Jansen, O., Tchinda, A. T., Loua, J., Esters, V., Cieckiewicz, E., Ledoux, A., Toukam, P. D., Angenot, L., Tits, M., Balde, A. M., & Frédérich, M. (2017).

Antiplasmodial activity of Mezoneuron benthamianum leaves and identification of its active constituents. *Journal of Ethnopharmacology*, *203*, 20–26. doi:10.1016/j.jep.2017.03.021

- Jeanroy, A. (2022). How to Make Herbal Infusions. *The Spruce Eats*. Retrieved from https://www.thespruceeats.com/how-to-make-an-herbal-infusion-1762142 (accessed 24 December 2023).
- Jeney, V., Eaton, J. W., Balla, G., & Balla, J. (2013). Natural History of the Bruise: Formation, Elimination, and Biological Effects of Oxidized Hemoglobin. *Oxidative Medicine and Cellular Longevity*, *2013*, 1–9. doi:10.1155/2013/703571
- Jiang, T. A. (2019). Health Benefits of Culinary Herbs and Spices. *Journal of AOAC INTERNATIONAL*, *102*, 395–411. doi:10.5740/jaoacint.18-0418
- Jiang, M., Yang, L., Zou, L., Zhang, L., Wang, S., Zhong, Z., Wang, Y., & Li,
 P. (2024). A comprehensive quality evaluation for Huangqi Guizhi Wuwu
 decoction by integrating UPLC-DAD/MS chemical profile and
 pharmacodynamics combined with chemometric analysis. *Journal of Ethnopharmacology*, 319, 117325. doi:10.1016/j.jep.2023.117325
- Jiang, M., Zhao, S., Yang, S., Lin, X., He, X., Wei, X., Song, Q., Li, R., Fu,
 C., Zhang, J., & Zhang, Z. (2020). An "essential herbal medicine"—licorice: A review of phytochemicals and its effects in

combination preparations. *Journal of Ethnopharmacology*, 249, 112439. doi:10.1016/j.jep.2019.112439

- Jimenez-Tellez, N., & Greenway, S. C. (2019). Cellular models for human cardiomyopathy: What is the best option? World Journal of Cardiology, 11(10), 221–235. doi:10.4330/wjc.v11.i10.221
- Jin, P., Jiang, J., Zhou, L., Huang, Z., Nice, E. C., Huang, C., & Fu, L. (2022). Mitochondrial adaptation in cancer drug resistance: prevalence, mechanisms, and management. *Journal of Hematology & Oncology*, *15*, 97. doi:10.1186/s13045-022-01313-4
- Jin, X., Li, L., Peng, Q., Gan, C., Gao, L., He, S., Tan, S., Pu, W., Liu, Y., Gong, Y., Yao, Y., Wang, G., Liu, X., Gong, M., Lei, P., Zhang, H., Qi, S., Xu, H., Hu, H., Dong, B., Peng, Y., Su, D., & Dai, L. (2022). Glycyrrhetinic acid restricts mitochondrial energy metabolism by targeting SHMT2. *IScience*, 25(5), 104349. doi:10.1016/j.isci.2022.104349
- Jiwaji University. (n.d.). Adulteration of Drugs of Natural Origin. Retrieved from

http://www.jiwaji.edu/pdf/ecourse/pharmaceutical/Adulteration%20of%20d rugs%20of%20natural%20origin.pdf (accessed 24 November 2022).

Jolad, S. D., Lantz, R. C., Chen, G. J., Bates, R. B., & Timmermann, B. N. (2005). Commercially processed dry ginger (Zingiber officinale): Composition and effects on LPS-stimulated PGE2 production.

 Phytochemistry,
 66(13),
 1614–1635.

 doi:10.1016/j.phytochem.2005.05.007

Joshu, E. (2023). Chemical found in popular candy suppresses 'silent killer' pancreatic cancer, study indicates. *Daily Mail* Aug 11, 2023. Retrieved from https://www.dailymail.co.uk/health/article-12397561/Chemical-popular-ca ndy-suppresses-silent-killer-pancreatic-cancer-study-indicates.html

(accessed 11 August 2023).

- Ju, S.-M., Kim, M.-S., Jo, Y.-S., Jeon, Y.-M., Bae, J.-S., Pae, H.-O., & Jeon, B.-H. (2017). Licorice and its active compound glycyrrhizic acid ameliorates cisplatin-induced nephrotoxicity through inactivation of p53 by scavenging ROS and overexpression of p21 in human renal proximal tubular cells. European epithelial Review for Medical and Pharmacological 21. Sciences. 890-899. Available from https://www.ncbi.nlm.nih.gov/pubmed/28272690 (accessed 28 January 2024).
- Jung, J. I., Lim, S. S., Choi, H. J., Cho, H. J., Shin, H.-K., Kim, E. J., Chung, W.-Y., Park, K.-K., & Park, J. H. Y. (2006). Isoliquiritigenin induces apoptosis by depolarizing mitochondrial membranes in prostate cancer cells. *The Journal of Nutritional Biochemistry*, *17*, 689–696. doi:10.1016/j.jnutbio.2005.11.006

- Jungbauer, A., & Medjakovic, S. (2014). Phytoestrogens and the metabolic syndrome. *The Journal of Steroid Biochemistry and Molecular Biology*, *139*, 277–289. doi:10.1016/j.jsbmb.2012.12.009
- Kaczor, T. (2015). Deglycyrrhizinated Licorice for Gastrointestinal Ulcers.
 Natural Medicine Journal. Retrieved from https://www.naturalmedicinejournal.com/journal/deglycyrrhizinated-licorice -gastrointestinal-ulcers (accessed 2 October 2023).
- Kalkavan, H., Chen, M. J., Crawford, J. C., Quarato, G., Fitzgerald, P., Tait,
 S. W. G., Goding, C. R., & Green, D. R. (2022). Sublethal cytochrome c
 release generates drug-tolerant persister cells. *Cell*, *185*, 3356-3374.e22.
 doi:10.1016/j.cell.2022.07.025
- Kan, H., Zhang, D., Chen, W., Wang, S., He, Z., Pang, S., Qu, S., & Wang,
 Y. (2023). Identification of anti-inflammatory components in Panax ginseng of Sijunzi Decoction based on spectrum-effect relationship. *Chinese Herbal Medicines*, 15(1), 123–131.
 doi:10.1016/j.chmed.2022.04.003
- Kang, D., Shin, W.-C., Kim, T., Kim, S., Kim, H., Cho, J.-H., Song, M.-Y., & Chung, W.-S. (2023). Systematic review and meta-analysis of the anti-obesity effect of cupping therapy. *Medicine*, *102*, e34039. doi:10.1097/MD.00000000034039
- Kang, T., Seo, J., Oh, H., Yoon, G., Chae, J., & Shim, J. (2017). Licochalcone A Suppresses Specificity Protein 1 as a Novel Target in

Human Breast Cancer Cells. *Journal of Cellular Biochemistry*, *118*, 4652–4663. doi:10.1002/jcb.26131

- Kang, S. W., Lee, S., & Lee, E. K. (2015). ROS and energy metabolism in cancer cells: alliance for fast growth. *Archives of Pharmacal Research*, 38, 338–345. doi:10.1007/s12272-015-0550-6
- Karaberopoulos, D., Karamanou, M., & Androutsos, G. (2012). The theriac
 in antiquity. *The Lancet*, 379, 1942–1943.
 doi:10.1016/S0140-6736(12)60846-0
- Karadag, O., & Bolek, E. C. (2020). Management of Behcet's syndrome. *Rheumatology*, *59*, iii108–iii117. doi:10.1093/rheumatology/keaa086
- Karimi, A., Majlesi, M., & Rafieian-Kopaei, M. (2015). Herbal versus synthetic drugs; beliefs and facts. *Journal of Nephropharmacology*, *4*, 27–30. Available from https://www.ncbi.nlm.nih.gov/pubmed/28197471 (accessed 28 January 2024).
- Karioti, A., Chiarabini, L., Alachkar, A., Chehna, M. F., Vincieri, F. F., & Bilia,
 A. R. (2014). HPLC–DAD and HPLC–ESI-MS analyses of Tiliae flos and
 its preparations. *Journal of Pharmaceutical and Biomedical Analysis*, *100*, 205–214. doi:10.1016/j.jpba.2014.08.010
- Karthikkeyan, G., Behera, S. K., Upadhyay, S. S., Pervaje, R., Prasad, T. S.
 K., & Modi, P. K. (2022). Metabolomics analysis highlights *Yashtimadhu* (*Glycyrrhiza glabra* L.) - mediated neuroprotection in a rotenone-induced

cellular model of Parkinson's disease by restoring the mTORC1-AMPK1 axis in autophagic regulation. *Phytotherapy Research*, *36*, 2207–2222. doi:10.1002/ptr.7449

- Karthikkeyan, G., Najar, M. A., Pervaje, R., Pervaje, S. K., Modi, P. K., & Prasad, T. S. K. (2020). Identification of Molecular Network Associated with Neuroprotective Effects of Yashtimadhu (*Glycyrrhiza glabra* L.) by Quantitative Proteomics of Rotenone-Induced Parkinson's Disease Model. *ACS Omega*, *5*, 26611–26625. doi:10.1021/acsomega.0c03420
- Karthikkeyan, G., Pervaje, R., Pervaje, S. K., Prasad, T. S. K., & Modi, P. K. (2021). Prevention of MEK-ERK-1/2 hyper-activation underlines the neuroprotective effect of Glycyrrhiza glabra L. (Yashtimadhu) against rotenone-induced cellular and molecular aberrations. *Journal of Ethnopharmacology*, 274, 114025. doi:10.1016/j.jep.2021.114025
- Karthikkeyan, G., Prabhu, A., Pervaje, R., Pervaje, S. K., Modi, P. K., & Prasad, T. S. K. (2021). Data on dose-dependent cytotoxicity of rotenone and neuroprotection conferred by Yashtimadhu (Glycyrrhiza glabra L.) in an in vitro Parkinson's disease model. *Data in Brief*, *39*, 107535. doi:10.1016/j.dib.2021.107535
- Kato, T., Horie, N., Hashimoto, K., Satoh, K., Shimoyama, T., Kaneko, T.,
 Kusama, K., & Sakagami, H. (2008). Bimodal effect of glycyrrhizin on
 macrophage nitric oxide and prostaglandin E2 production. *In Vivo*(Athens, Greece), 22, 583–586. Available from

https://www.ncbi.nlm.nih.gov/pubmed/18853750 (accessed 28 January 2024).

- Kattoor, A. J., Pothineni, N. V. K., Palagiri, D., & Mehta, J. L. (2017).
 Oxidative Stress in Atherosclerosis. *Current Atherosclerosis Reports*, *19*, 42. doi:10.1007/s11883-017-0678-6
- Kauffman, M., Kauffman, M., Traore, K., Zhu, H., Trush, M., Jia, Z., & Li, Y.
 (2016). MitoSOX-Based Flow Cytometry for Detecting Mitochondrial ROS. *Reactive Oxygen Species*. doi:10.20455/ros.2016.865
- Kaul, P., Passafiume, J., Sargent, R. C., & O'Hara, B. F. (2010). Meditation acutely improves psychomotor vigilance, and may decrease sleep need. *Behavioral and Brain Functions*, 6, 47. doi:10.1186/1744-9081-6-47
- Kendall, D. E. (2008). Energy Meridian Misconceptions of Chinese
 Medicine. Schweizerische Zeitschrift Für Ganzheitsmedizin / Swiss
 Journal of Integrative Medicine, 20, 112–117. doi:10.1159/000286171
- Kennedy, E. P., & Lehninger, A. L. (1949). Oxidation of Fatty Acids and Tricarboxylic Acid Cycle Intermediates by Isolated Rat Liver Mitochondria. *Journal of Biological Chemistry*, 179, 957–972. doi:10.1016/S0021-9258(19)51289-3
- Keshari, P. (2022). Controversy, Adulteration and Substitution: Burning Problems in Ayurveda Practices. doi:10.5772/intechopen.98220

- Khaitov, B., Karimov, A., Khaitbaeva, J., Sindarov, O., Karimov, A., & Li, Y. (2022). Perspectives of Licorice Production in Harsh Environments of the Aral Sea Regions. *International Journal of Environmental Research and Public Health*, *19*, 11770. doi:10.3390/ijerph191811770
- Kim, S. E. (2023). What's in traditional Chinese herbal medicine, and do its prescriptions work? *Chemical & Engineering News*, *101*. Retrieved from https://cen.acs.org/biological-chemistry/natural-products/Whats-in-traditio nal-Chinese-herbal-medicine/101/i12 (accessed 2 October 2023).
- Kim, S. C., Byun, S. H., Yang, C. H., Kim, C. Y., Kim, J. W., & Kim, S. G. (2004). Cytoprotective effects of Glycyrrhizae radix extract and its active component liquiritigenin against cadmium-induced toxicity (effects on bad translocation and cytochrome c-mediated PARP cleavage). *Toxicology*, 197, 239–251. doi:10.1016/j.tox.2004.01.010
- Kim, K. H., Yoon, G., Cho, J. J., Cho, J. H., Cho, Y. S., Chae, J.-I., & Shim, J.-H. (2015). Licochalcone A induces apoptosis in malignant pleural mesothelioma through downregulation of Sp1 and subsequent activation of mitochondria-related apoptotic pathway. *International Journal of Oncology*, 46, 1385–1392. doi:10.3892/ijo.2015.2839
- Kim, D. E., Youn, Y. C., Kim, Y. K. I., Hong, K. M., & Lee, C. S. (2009).Glycyrrhizin Prevents 7-Ketocholesterol Toxicity Against DifferentiatedPC12 Cells by Suppressing Mitochondrial Membrane Permeability

Change. *Neurochemical Research*, *34*, 1433–1442. doi:10.1007/s11064-009-9930-y

- Kobuchi, H., Moriya, K., Ogino, T., Fujita, H., Inoue, K., Shuin, T., Yasuda,
 T., Utsumi, K., & Utsumi, T. (2012). Mitochondrial Localization of ABC
 Transporter ABCG2 and Its Function in 5-Aminolevulinic Acid-Mediated
 Protoporphyrin IX Accumulation. *PLoS ONE*, *7*, e50082.
 doi:10.1371/journal.pone.0050082
- Koch, Zagórska, Marzec, & Kukula-Koch. (2019). Applications of Tea (Camellia sinensis) and its Active Constituents in Cosmetics. *Molecules*, 24, 4277. doi:10.3390/molecules24234277
- Kokalj, M., Štih, K., & Kreft, S. (2014). Herbal Tea Identification Using
 Mid-Infrared Spectroscopy. *Planta Medica*, *80*, 1023–1028.
 doi:10.1055/s-0034-1382904
- Kondo, K., Shiba, M., Nakamura, R., Morota, T., & Shoyama, Y. (2007).
 Constituent Properties of Licorices Derived from Glycyrrhiza uralensis, G.
 glabra, or G. inflata Identified by Genetic Information. *Biological and Pharmaceutical Bulletin*, *30*, 1271–1277. doi:10.1248/bpb.30.1271
- Kong, J., Xiang, Q., Shi, G., Xu, Z., Ma, X., Wang, Y., Xuan, Z., & Xu, F. (2023). Licorice protects against ulcerative colitis via the Nrf2/PINK1-mediated mitochondrial autophagy. *Immunity, Inflammation and Disease*, *11*. doi:10.1002/iid3.757

- Kong, R., Zhu, X., Meteleva, E. S., Polyakov, N. E., Khvostov, M. v., Baev,
 D. S., Tolstikova, T. G., Dushkin, A. v., & Su, W. (2018). Atorvastatin calcium inclusion complexation with polysaccharide arabinogalactan and saponin disodium glycyrrhizate for increasing of solubility and bioavailability. *Drug Delivery and Translational Research*, *8*, 1200–1213. doi:10.1007/s13346-018-0565-x
- Kou, X., Zhu, J., Xie, X., Hao, M., & Zhao, Y. (2020). The protective effect of glycyrrhizin on hepatic ischemia-reperfusion injury in rats and possible related signal pathway. *Iranian Journal of Basic Medical Sciences*, 23, 1232–1238. doi:10.22038/ijbms.2020.44101.10334
- Kowalska, A., & Kalinowska-Lis, U. (2019). 18β-Glycyrrhetinic acid: its core biological properties and dermatological applications. *International Journal of Cosmetic Science*, ics.12548. doi:10.1111/ics.12548
- Koyama, M., Shirahata, T., Hirashima, R., Kobayashi, Y., Itoh, T., & Fujiwara, R. (2017). Inhibition of UDP-glucuronosyltransferase (UGT)-mediated glycyrrhetinic acid 3- O -glucuronidation by polyphenols and triterpenoids. *Drug Metabolism and Pharmacokinetics*, *32*, 218–223. doi:10.1016/j.dmpk.2017.04.003
- Kuang, Y., Li, B., Fan, J., Qiao, X., & Ye, M. (2018). Antitussive and expectorant activities of licorice and its major compounds. *Bioorganic & Medicinal Chemistry*, *26*, 278–284. doi:10.1016/j.bmc.2017.11.046

- Kuhn, R., & Lederer, E. (1931). Fraktionierung und Isomerisierung des Carotins. *Die Naturwissenschaften*, *19*, 306–306. doi:10.1007/BF01520417
- Kurop, M. K., Huyen, C. M., Kelly, J. H., & Blagg, B. S. J. (2021). The heat shock response and small molecule regulators. *European Journal of Medicinal Chemistry*, 226, 113846. doi:10.1016/j.ejmech.2021.113846
- Kwan, H. Y., Wu, J., Su, T., Chao, X.-J., Liu, B., Fu, X., Chan, C. L., Lau, R.
 H. Y., Tse, A. K. W., Han, Q. bin, Fong, W. F., & Yu, Z. (2017). Cinnamon induces browning in subcutaneous adipocytes. *Scientific Reports*, *7*, 2447. doi:10.1038/s41598-017-02263-5
- Laher, F., Aremu, A. O., Van Staden, J., & Finnie, J. F. (2013). Evaluating the effect of storage on the biological activity and chemical composition of three South African medicinal plants. *South African Journal of Botany*, *88*, 414–418. doi:10.1016/j.sajb.2013.09.003
- Lai, J.-N., Wu, C.-T., & Wang, J.-D. (2012). Prescription Pattern of Chinese Herbal Products for Breast Cancer in Taiwan: A Population-Based Study. *Evidence-Based Complementary and Alternative Medicine*, 2012, 1–7. doi:10.1155/2012/891893
- Lajeunesse, D. (1994). Effect of 17β-estradiol on the human osteosarcoma cell line MG-63. *Bone and Mineral*, 24(1), 1–16. doi:10.1016/S0169-6009(08)80126-2

- Lam, S.-C., Lam, S.-F., Zhao, J., & Li, S.-P. (2016). Rapid Identification and Comparison of Compounds with Antioxidant Activity in Coreopsis tinctoria Herbal Tea by High-Performance Thin-Layer Chromatography Coupled with DPPH Bioautography and Densitometry. *Journal of Food Science*, *81*, C2218–C2223. doi:10.1111/1750-3841.13402
- Lamming, D. W., Wood, J. G., & Sinclair, D. A. (2004). Small molecules that regulate lifespan: evidence for xenohormesis. *Molecular Microbiology*, *53*, 1003–1009. doi:10.1111/j.1365-2958.2004.04209.x
- Languon, S., Tuffour, I., Quayson, E. E., Appiah-Opong, R., & Quaye, O. (2018). In Vitro Evaluation of Cytotoxic Activities of Marketed Herbal Products in Ghana. *Journal of Evidence-Based Integrative Medicine*, 23, 2515690X1879072. doi:10.1177/2515690X18790723
- Lathe, W. C., Williams, J. M., Mangan, M. E., & Karolchik, D. (2008).
 Genomic Data Resources: Challenges and Promises. *Nature Education*,
 1, 2. Retrieved from
 https://www.nature.com/scitable/topicpage/genomic-data-resources-challe
 nges-and-promises-743721/ (accessed 16 November 2020).
- Lawenda, B. D., Kelly, K. M., Ladas, E. J., Sagar, S. M., Vickers, A., & Blumberg, J. B. (2008). Should Supplemental Antioxidant Administration
 Be Avoided During Chemotherapy and Radiation Therapy? *JNCI Journal of the National Cancer Institute*, *100*, 773–783. doi:10.1093/jnci/djn148

- Lawrence, F. (2003). Chalk and cheese. *The Guardian*, May 17, 2003. Retrieved from https://www.theguardian.com/food/focus/story/0,,956600,00.html (accessed 22 November 2022).
- Lee, M. R. (2018). Liquorice (*Glycyrrhiza Glabra*): The Journey of the Sweet Root from Mesopotamia to England. *Journal of the Royal College of Physicians of Edinburgh*, *48*, 378–382. doi:10.4997/jrcpe.2018.419
- Lee, J., Durst, R. W., & Wrolstad, R. E. (2005). Determination of total monomeric anthocyanin pigment content of fruit juices, beverages, natural colorants, and wines by the pH differential method: collaborative study. *Journal of AOAC International*, *88*, 1269–1278. Retrieved from https://www.ars.usda.gov/ARSUserFiles/37108/pdf/leejaoac.pdf (accessed 20 May 2023).
- Lee, M.-S., Hwang, Y.-S., Lee, J., & Choung, M.-G. (2014). The characterization of caffeine and nine individual catechins in the leaves of green tea (Camellia sinensis L.) by near-infrared reflectance spectroscopy. *Food Chemistry*, 158, 351–357. doi:10.1016/j.foodchem.2014.02.127
- Lee, D. G., Min, J.-S., Lee, H.-S., & Lee, D.-S. (2018). Isoliquiritigenin attenuates glutamate-induced mitochondrial fission via calcineurin-mediated Drp1 dephosphorylation in HT22 hippocampal

neuron cells. *NeuroToxicology*, 68, 133–141. doi:10.1016/j.neuro.2018.07.011

- Lee, J. R., Park, S. J., Lee, H.-S., Jee, S. Y., Seo, J., Kwon, Y. K., Kwon, T.
 K., & Kim, S. C. (2009). Hepatoprotective Activity of Licorice Water
 Extract against Cadmium-Induced Toxicity in Rats. *Evidence-Based Complementary and Alternative Medicine*, 6, 195–201.
 doi:10.1093/ecam/nem078
- Lee, C. S., Yang, J. C., Kim, Y. J., Jang, E.-R., Kim, W., & Myung, S. C. (2010). 18β-Glycyrrhetinic acid potentiates apoptotic effect of trichostatin A on human epithelial ovarian carcinoma cell lines. *European Journal of Pharmacology*, *649*, 354–361. doi:10.1016/j.ejphar.2010.09.047
- Lejri, I., Agapouda, A., Grimm, A., & Eckert, A. (2019). Mitochondria- and Oxidative Stress-Targeting Substances in Cognitive Decline-Related Disorders: From Molecular Mechanisms to Clinical Evidence. *Oxidative Medicine and Cellular Longevity*, *2019*, 1–26. doi:10.1155/2019/9695412
- Leon, C., & Lin, Y.-L. (2017). *Chinese Medicinal Plants, Herbal Drugs and Substitutes: An Identification Guide*. London: Kew Publishing.
- Leong, P. K., Leung, H. Y., Chan, W. M., & Ko, K. M. (2018). Differences in the Mechanisms by Which Yang-Invigorating and Qi-Invigorating Chinese Tonifying Herbs Stimulate Mitochondrial ATP Generation Capacity. *Chinese Medicine*, 09, 63–74. doi:10.4236/cm.2018.92005

- Leopoldini, M., Russo, N., & Toscano, M. (2011). The molecular basis of working mechanism of natural polyphenolic antioxidants. *Food Chemistry*, *125*, 288–306. doi:10.1016/j.foodchem.2010.08.012
- Li, S. (2022). *Ben Cao Gang Mu, Volume III: Mountain Herbs, Fragrant Herbs* (trans. P. Unschuld). Oakland, CA: University of California Press.
- Li, H., Wang, P.-P., Lin, Z.-Z., Wang, Y.-L., Gui, X.-J., Fan, X.-H., Dong, F.-Y., Zhang, P.-P., Li, X.-L., & Liu, R.-X. (2024). Identification of Bletilla striata and related decoction pieces: a data fusion method combining electronic nose, electronic tongue, electronic eye, and high-performance liquid chromatography data. *Frontiers in Chemistry*, *11*. doi:10.3389/fchem.2023.1342311
- Li, X., Zhu, L.-J., Lv, J., & Cao, X. (2023). Purinoceptor: a novel target for hypertension. *Purinergic Signalling*, *19*, 185–197. doi:10.1007/s11302-022-09852-8
- Li, J., Luo, Y., Zhan, L., Gu, Y., Zhang, W., Wen, Q., Feng, Y., & Tan, T. (2022). Comprehensive chemical profiling of the flowers of Citrus aurantium L. var. amara Engl. and uncovering the active ingredients of lipid lowering. *Journal of Pharmaceutical and Biomedical Analysis*, *211*, 114621. doi:10.1016/j.jpba.2022.114621
- Li, T., Wu, F., Zhang, A., Dong, H., Ullah, I., Lin, H., Miao, J., Sun, H., Han, Y., He, Y., & Wang, X. (2022). High-Throughput Chinmedomics Strategy Discovers the Quality Markers and Mechanisms of Wutou Decoction

Therapeutic for Rheumatoid Arthritis. *Frontiers in Pharmacology*, 13. doi:10.3389/fphar.2022.854087

- Li, M.-T., Xie, L., Jiang, H.-M., Huang, Q., Tong, R.-S., Li, X., Xie, X., & Liu,
 H.-M. (2022). Role of Licochalcone A in Potential Pharmacological
 Therapy: A Review. *Frontiers in Pharmacology*, 13.
 doi:10.3389/fphar.2022.878776
- Li, M., Ke, J., Deng, Y., Chen, C., Huang, Y., Bian, Y., Guo, S., Wu, Y., Zhang, H., Liu, M., & Han, Y. (2021). The Protective Effect of Liquiritin in Hypoxia/Reoxygenation-Induced Disruption on Blood Brain Barrier. *Frontiers in Pharmacology*, 12. doi:10.3389/fphar.2021.671783
- Li, Y., Liu, S.-S., Guo, Z.-Y., Yi, H., Li, C., Chen, L.-M., Gao, H.-M., Yan, L.-H., Zhang, W.-W., Feng, X.-X., Zhao, J.-Y., Liu, X.-Q., & Wang, Z.-M. (2021). Discovery of potential pharmacodynamic ingredients of Dang-Gui-Si-Ni decoction based on absorbed ingredients and molecular docking. *Journal of Ethnopharmacology*, 275, 114045. doi:10.1016/j.jep.2021.114045
- Li, X., Lu, H.-Y., Jiang, X.-W., Yang, Y., Xing, B., Yao, D., Wu, Q., Xu, Z.-H.,
 & Zhao, Q.-C. (2021). Cinnamomum cassia extract promotes thermogenesis during exposure to cold via activation of brown adipose tissue. *Journal of Ethnopharmacology*, 266, 113413. doi:10.1016/j.jep.2020.113413

- Li, J., Wang, X., Bu, D., Zou, J., Xun, S., Wang, Y., Jia, Y., Yu, S., Wang, W., Zheng, J., Hou, J., Zhang, X., & Wang, C. (2021). Exploration of the active components and pharmacological mechanism of Compound Longmaining for the treatment of myocardial infarction. *Frontiers in Bioscience-Landmark*, 26(10), 813. doi:10.52586/4990
- Li, Y., Kong, D., Fu, Y., Sussman, M. R., & Wu, H. (2020). The effect of developmental and environmental factors on secondary metabolites in medicinal plants. *Plant Physiology and Biochemistry*, 148, 80–89. doi:10.1016/j.plaphy.2020.01.006
- Li, X., Qin, X., Tian, J., Gao, X., Wu, X., Du, G., & Zhou, Y. (2020). Liquiritin protects PC12 cells from corticosterone-induced neurotoxicity via regulation of metabolic disorders, attenuation ERK1/2-NF-κB pathway, activation Nrf2-Keap1 pathway, and inhibition mitochondrial apoptosis pathway. *Food and Chemical Toxicology*, *146*, 111801. doi:10.1016/j.fct.2020.111801
- Li, S., Gao, M., Li, Z., Song, L. S., Gao, X., Han, J., Wang, F., Chen, Y., Li,
 W., & Yang, J. (2018). p53 and P-glycoprotein influence chemoresistance
 in hepatocellular carcinoma. *Frontiers in Bioscience*, 10, 833.
 doi:10.2741/e833
- Li, X.-L., Zhou, A.-G., Zhang, L., & Chen, W.-J. (2011). Antioxidant Status and Immune Activity of Glycyrrhizin in Allergic Rhinitis Mice. *International Journal of Molecular Sciences*, *12*, 905–916. doi:10.3390/ijms12020905

- Liang, W., Weimei, Z., Chen, Y., Sun, J., Guo, F., Hu, J., Gao, W., & Li, X. (2022). Quality evaluation of different varieties of rhubarb based on multicomponents and bioactivity: Application to quality control in the production of rhubarb decoction pieces. *Biomedical Chromatography*, 36(6). doi:10.1002/bmc.5368
- Liang, G., Yang, J., Liu, T., Wang, S., Wen, Y., Han, C., Huang, Y., Wang, R., Wang, Y., Hu, L., Wang, G., Li, F., Tyndall, J. D. A., Deng, L., Du, D., & Xia, Q. (2021). A multi-strategy platform for quality control and Q-markers screen of Chaiqin chengqi decoction. *Phytomedicine*, *85*, 153525. doi:10.1016/j.phymed.2021.153525
- Liang, M., Li, X., Ouyang, X., Xie, H., & Chen, D. (2018). Antioxidant Mechanisms of Echinatin and Licochalcone A. *Molecules*, *24*, 3. doi:10.3390/molecules24010003
- Liang, X.-M., Jin, Y., Feng, J.-T., & Ke, Y.-X. (2011). 'Identification and Structure Elucidation of Compounds from Herbal Medicines' in W. J. H.
 Liu, (ed.). *Traditional Herbal Medicine Research Methods: Identification, Analysis, Bioassay, and Pharmaceutical and Clinical Studies*. Hoboken, NJ: John Wiley & Sons, Ltd. pp. 139-224.
- Liao, P.-C., Bergamini, C., Fato, R., Pon, L. A., & Pallotti, F. (2020). *Isolation* of mitochondria from cells and tissues (pp. 3–31). doi:10.1016/bs.mcb.2019.10.002

Lim, T. K. (2016). *Glycyrrhiza glabra*. doi:10.1007/978-94-017-7276-1_18

- Lin, R.-C., Yang, S.-F., Chiou, H.-L., Hsieh, S.-C., Wen, S.-H., Lu, K.-H., & Hsieh, Y.-H. (2019). Licochalcone A-Induced Apoptosis Through the Activation of p38MAPK Pathway Mediated Mitochondrial Pathways of Apoptosis in Human Osteosarcoma Cells In Vitro and In Vivo. *Cells*, *8*, 1441. doi:10.3390/cells8111441
- Lin, S.-C., Chu, P.-Y., Liao, W.-T., Wu, M.-Y., Tsui, K.-H., Lin, L.-T., Huang, C.-H., Chen, L.-L., & Li, C.-J. (2017). Glycyrrhizic acid induces human MDA-MB-231 breast cancer cell death and autophagy via the ROS-mitochondrial pathway. *Oncology Reports*. doi:10.3892/or.2017.6123
- Lin, J.-H., Yang, K.-T., Ting, P.-C., Lee, W.-S., Lin, D.-J., & Chang, J.-C. (2023).
 Licochalcone a improves cardiac functions after ischemia-reperfusion via reduction of ferroptosis in rats. *European Journal of Pharmacology*, 957, 176031. doi:10.1016/j.ejphar.2023.176031
- Ling, B., Michel, D., Sakharkar, M., & Yang, J. (2016). Evaluating the cytotoxic effects of the water extracts of four anticancer herbs against human malignant melanoma cells. *Drug Design, Development and Therapy*, *10*, 3563–3572. doi:10.2147/DDDT.S119214
- Linnaeus, C. (1735). Systema naturae : in quo proponuntur naturae regna tria secundum classes, ordines, genera et species (Reprint 1744). Retrieved from https://archive.org/details/CaroliLinnaeiSy00LinnA (accessed 25 November 2021).

- Linnaeus, C. (1753). *Species Plantarum*. Retrieved from https://www.biodiversitylibrary.org/page/26068030 (accessed 20 March 2023).
- Liu, H.-W. (2011). 'Extraction and Isolation of Compounds from Herbal Medicine' in W. J. H. Liu (ed.) *Traditional Herbal Medicine Research Methods: Identification, Analysis, Bioassay, and Pharmaceutical and Clinical Studies*. Hoboken, NJ: John Wiley & Sons, Ltd. pp. 81-138.
- Liu, Y. (2021). *Transforming Poisons*. Retrieved from https://uw.manifoldapp.org/read/healing-with-poisons-d74f2492-5f8c-4898 -8cd1-546023f82ab8/section/69b5c6b9-093c-4946-8c17-3fd351944c06 (accessed 13 September 2021).
- Liu, W. M., & Dalgleish, A. G. (2009). MTT assays can underestimate cell numbers. *Cancer Chemotherapy and Pharmacology*, *64*, 861–862. doi:10.1007/s00280-009-1047-0
- Liu, G., Li, S., Zhang, N., Wei, N., Wang, M., Liu, J., Xu, Y., Li, Y., Sun, Q.,
 Li, Y., Li, F., Yu, P., Liu, M., Wang, Y., Zhai, H., & Wang, Y. (2022).
 Sequential grade evaluation method exploration of Exocarpium Citri
 Grandis (Huajuhong) decoction pieces based on "network prediction →
 grading quantization → efficacy validation". *Journal of Ethnopharmacology*, 291, 115149. doi:10.1016/j.jep.2022.115149
- Liu, J., Deng, X., Liang, X., & Li, L. (2021). The phytoestrogen glabrene prevents osteoporosis in ovariectomized rats through upregulation of the

canonical Wnt/β-catenin signaling pathway. Journal of Biochemical and Molecular Toxicology, 35(2). doi:10.1002/jbt.22653

- Liu, P.-S., & Ho, P.-C. (2018). Mitochondria: A master regulator in macrophage and T cell immunity. *Mitochondrion*, *41*, 45–50. doi:10.1016/j.mito.2017.11.002
- Liu, W., Zhang, S.-X., Ai, B., Pan, H.-F., Zhang, D., Jiang, Y., Hu, L.-H., Sun,
 L.-L., Chen, Z.-S., & Lin, L.-Z. (2021). Ginsenoside Rg3 Promotes Cell
 Growth Through Activation of mTORC1. Frontiers in Cell and
 Developmental Biology, 9. doi:10.3389/fcell.2021.730309
- Liu, J., Zhang, Q., Li, R.-L., Wei, S.-J., Huang, C.-Y., Gao, Y.-X., & Pu, X.-F. (2020). The traditional uses, phytochemistry, pharmacology and toxicology of Cinnamomi ramulus : a review. Journal of Pharmacy and Pharmacology, 72(3), 319–342. doi:10.1111/jphp.13189
- Liu, X., Zhang, Y., Wu, M., Ma, Z., Huang, Z., Tian, F., Dong, S., Luo, S., Zhou, Y., Zhang, J., Li, N., He, X., & Cao, H. (2020). The scientific elucidation of daodi medicinal materials. *Chinese Medicine*, *15*, 86. doi:10.1186/s13020-020-00367-1
- Loescher, C. M., Morton, D. W., Razic, S., & Agatonovic-Kustrin, S. (2014). High performance thin layer chromatography (HPTLC) and high performance liquid chromatography (HPLC) for the qualitative and quantitative analysis of Calendula officinalis—Advantages and limitations.

Journal of Pharmaceutical and Biomedical Analysis, 98, 52–59. doi:10.1016/j.jpba.2014.04.023

- Lombardi, V. R. M., Carrera, I., & Cacabelos, R. (2017). In Vitro Screening for Cytotoxic Activity of Herbal Extracts. *Evidence-Based Complementary and Alternative Medicine*, *2017*, 1–8. doi:10.1155/2017/2675631
- Lombardo, M., Feraco, A., Camajani, E., Caprio, M., & Armani, A. (2023). Health Effects of Red Wine Consumption: A Narrative Review of an Issue That Still Deserves Debate. *Nutrients*, *15*, 1921. doi:10.3390/nu15081921
- Lopaschuk, G. D., Karwi, Q. G., Tian, R., Wende, A. R., & Abel, E. D. (2021). Cardiac Energy Metabolism in Heart Failure. *Circulation Research*, *128*, 1487–1513. doi:10.1161/CIRCRESAHA.121.318241
- Lončar, M., Jakovljević, M., Šubarić, D., Pavlić, M., Buzjak Služek, V., Cindrić, I., & Molnar, M. (2020). Coumarins in Food and Methods of Their Determination. *Foods*, 9(5), 645. doi:10.3390/foods9050645
- Lopaschuk, G. D., Karwi, Q. G., Tian, R., Wende, A. R., & Abel, E. D. (2021). Cardiac Energy Metabolism in Heart Failure. *Circulation Research*, 128(10), 1487–1513. doi:10.1161/CIRCRESAHA.121.318241
- Lou, S.-K., Wong, K.-L., Li, M., But, P., Tsui, S., & Shaw, P.-C. (2010). An integrated web medicinal materials DNA database: MMDBD (Medicinal Materials DNA Barcode Database). *BMC Genomics*, *11*, 402. doi:10.1186/1471-2164-11-402

- Lowe, D. T. (2017). Cupping therapy: An analysis of the effects of suction on skin and the possible influence on human health. *Complementary Therapies in Clinical Practice*, 29, 162–168. doi:10.1016/j.ctcp.2017.09.008
- Lu, G., Qiao, J., Wang, L., Liu, H., Wu, G., Zhu, Y., Zhao, Y., Xie, G., & Qin,
 M. (2022). An integrated study of Violae Herba (Viola philippica) and five adulterants by morphology, chemical compositions and chloroplast genomes: insights into its certified plant origin. *Chinese Medicine*, *17*(1), 32. doi:10.1186/s13020-022-00585-9
- Lu, H., Zhang, J., Yang, Y., Yang, X., Xu, B., Yang, W., Tong, T., Jin, S., Shen, C., Rao, H., Li, X., Lu, H., Fuller, D. Q., Wang, L., Wang, C., Xu, D., & Wu, N. (2016). Earliest tea as evidence for one branch of the Silk Road across the Tibetan Plateau. *Scientific Reports*, 6, 18955. doi:10.1038/srep18955
- Lu, J., Wang, J.-S., & Kong, L.-Y. (2011). Anti-inflammatory effects of Huang-Lian-Jie-Du decoction, its two fractions and four typical compounds. *Journal of Ethnopharmacology*, *134*, 911–918. doi:10.1016/j.jep.2011.01.049
- Luo, Y.-H., Wang, C., Xu, W.-T., Zhang, Y., Zhang, T., Xue, H., Li, Y.-N., Fu, Z.-R., Wang, Y., & Jin, C.-H. (2021). 18β-Glycyrrhetinic Acid Has Anti-Cancer Effects via Inducing Apoptosis and G2/M Cell Cycle Arrest,

and Inhibiting Migration of A549 Lung Cancer Cells. *OncoTargets and Therapy*, *14*, 5131–5144. doi:10.2147/OTT.S322852

- Lv, H., Xiao, Q., Zhou, J., Feng, H., Liu, G., & Ci, X. (2018). Licochalcone A
 Upregulates Nrf2 Antioxidant Pathway and Thereby Alleviates
 Acetaminophen-Induced Hepatotoxicity. *Frontiers in Pharmacology*, *9*.
 doi:10.3389/fphar.2018.00147
- Ma, H.-L., Wang, J.-W., Chen, Y.-J., Cheng, J.-L., & Lai, Z.-T. (2017). Rapid authentication of starch adulterations in ultrafine granular powder of Shanyao by near-infrared spectroscopy coupled with chemometric methods. *Food Chemistry*, 215, 108–115. doi:10.1016/j.foodchem.2016.07.156
- Ma, J., Fu, N.-Y., Pang, D.-B., Wu, W.-Y., & Xu, A.-L. (2001). Apoptosis
 Induced by Isoliquiritigenin in Human Gastric Cancer MGC-803 Cells. *Planta Medica*, 67, 754–757. doi:10.1055/s-2001-18361
- Mahanta, S. K., & Challa, S. R. (2022). *Phytochemicals as Pro-oxidants in Cancer*. doi:10.1007/978-981-16-1247-3_209-1
- Mahamoud, R., Bowman, D. T., Ward, W. E., & Mangal, V. (2024).
 Assessing the stability of polyphenol content in red rooibos herbal tea using traditional methods and high-resolution mass spectrometry: Implications for studying dietary interventions in preclinical rodent studies. *Food Chemistry*, 448, 139068. doi:10.1016/j.foodchem.2024.139068

- Malik, J., Frankova, A., Drabek, O., Szakova, J., Ash, C., & Kokoska, L. (2013). Aluminium and other elements in selected herbal tea plant species and their infusions. *Food Chemistry*, 139, 728–734. doi:10.1016/j.foodchem.2013.02.013
- Mancuso, C., & Santangelo, R. (2017). Panax ginseng and Panax quinquefolius: From pharmacology to toxicology. *Food and Chemical Toxicology*, *107*, 362–372. doi:10.1016/j.fct.2017.07.019
- Mander, B. A., Winer, J. R., & Walker, M. P. (2017). Sleep and Human Aging. *Neuron*, *94*, 19–36. doi:10.1016/j.neuron.2017.02.004
- Mani, S., Swargiary, G., & Singh, K. K. (2020). Natural Agents Targeting
 Mitochondria in Cancer. *International Journal of Molecular Sciences*, *21*,
 6992. doi:10.3390/ijms21196992
- Marchetti, L., Rossi, M. C., Pellati, F., Benvenuti, S., & Bertelli, D. (2020).
 HR- 1 H NMR spectroscopy and multivariate statistical analysis to determine the composition of herbal mixtures for infusions. *Phytochemical Analysis*, ca.3002. doi:10.1002/pca.3002
- Marchi, S., Patergnani, S., Missiroli, S., Morciano, G., Rimessi, A.,
 Wieckowski, M. R., Giorgi, C., & Pinton, P. (2018). Mitochondrial and
 endoplasmic reticulum calcium homeostasis and cell death. *Cell Calcium*,
 69, 62–72. doi:10.1016/j.ceca.2017.05.003

- Maria Pia, G. D., Sara, F., Mario, F., & Lorenza, S. (2018). Biological Effects of Licochalcones. Mini-Reviews in Medicinal Chemistry, 19(8), 647–656. doi:10.2174/1389557518666180601095420
- Marshall, A. C. (2020). *Traditional Chinese Medicine and Clinical Pharmacology*. doi:10.1007/978-3-319-68864-0_60
- Marshall, B., & Adams, P. C. (2008). *Helicobacter Pylori*: A Nobel Pursuit? *Canadian Journal of Gastroenterology*, 22, 895–896.
 doi:10.1155/2008/459810
- Martin, A. J. P., & Synge, R. L. M. (1941). A new form of chromatogram employing two liquid phases. *Biochemical Journal*, 35, 1358–1368. doi:10.1042/bj0351358
- Massey, D., Saydah, S., Adamson, B., Lincoln, A., Aukerman, D. F., Berke,
 E. M., Sikka, R., & Krumholz, H. M. (2022). Prevalence of COVID-19 and
 Long COVID in Collegiate Student Athletes from Spring 2020 to Fall
 2021: A Retrospective Survey. *MedRxiv*.
 doi:10.1101/2022.06.12.22276048
- Matthews-King, A. (2018). Ginger reduces serious vomiting in gastroenteritis and "could save lives", finds clinical trial. *Independent*, 10 May 2018. Retrieved from https://www.independent.co.uk/news/health/ginger-vomiting-bug-gastroen teritis-child-deaths-diarrhoea-norovirus-a8345361.html (accessed 14 August 2024).

- Mattes, W. B. (2020). In vitro to in vivo translation. *Current Opinion in Toxicology*, 23–24, 114–118. doi:10.1016/j.cotox.2020.09.001
- Maxwell, J. T., Tsai, C.-H., Mohiuddin, T. A., & Kwong, J. Q. (2018). Analyses of Mitochondrial Calcium Influx in Isolated Mitochondria and Cultured Cells. *Journal of Visualized Experiments*. doi:10.3791/57225
- Measurlabs. (2024). *Atomic absorption spectroscopy*. Retrieved from https://measurlabs.com/methods/atomic-absorption-spectroscopy/ (accessed 18 June 2024).
- Medagama, A. B. (2015). The glycaemic outcomes of Cinnamon, a review of the experimental evidence and clinical trials. *Nutrition Journal*, *14*, 108. doi:10.1186/s12937-015-0098-9
- Meisner, H., & Chan, S. (1974). Ochratoxin A, an inhibitor of mitochondrial transport systems. *Biochemistry*, *13*, 2795–2800. doi:10.1021/bi00711a002
- Melhuish Beaupre, L. M., Brown, G. M., Braganza, N. A., Kennedy, J. L., & Gonçalves, V. F. (2022). Mitochondria's role in sleep: Novel insights from sleep deprivation and restriction studies. *The World Journal of Biological Psychiatry*, 23(1), 1–13. doi:10.1080/15622975.2021.1907723
- Merkytė, V., Longo, E., Windisch, G., & Boselli, E. (2020). Phenolic
 Compounds as Markers of Wine Quality and Authenticity. *Foods*, 9(12), 1785. doi:10.3390/foods9121785

- Merry, T. L., & Ristow, M. (2016). Mitohormesis in exercise training. *Free Radical Biology and Medicine*, 98, 123–130.
 doi:10.1016/j.freeradbiomed.2015.11.032
- Medicines & Healthcare products Regulatory Agency (2014). *Banned and restricted herbal ingredients*. Available from https://www.gov.uk/government/publications/list-of-banned-or-restricted-h erbal-ingredients-for-medicinal-use/banned-and-restricted-herbal-ingredie nts (accessed 10 June 2023).
- Miao, L.-Y., Chu, T. T. H., Li, P., Jiang, Y., & Li, H.-J. (2019). Cimicifuga heracleifolia is therapeutically similar to black cohosh in relieving menopausal symptoms: evidence from pharmacological and metabolomics studies. *Chinese Journal of Natural Medicines*, *17*(6), 435–445. doi:10.1016/S1875-5364(19)30051-2
- Micheli, L., Nencini, C., Galluzzi, P., Pippi, F., & Menchiari, A. (2014).
 Hepatotoxicity of Teucrium chamaedrys L. decoction: role of difference in the harvesting area and preparation method. *Indian Journal of Pharmacology*, *46*, 181. doi:10.4103/0253-7613.129313
- Miller, G. M., & Stripp, R. (2007). A study of western pharmaceuticals contained within samples of Chinese herbal/patent medicines collected from New York City's Chinatown. *Legal Medicine*, 9, 258–264. doi:10.1016/j.legalmed.2007.04.001

- Miller, V. J., Villamena, F. A., & Volek, J. S. (2018). Nutritional Ketosis and
 Mitohormesis: Potential Implications for Mitochondrial Function and
 Human Health. *Journal of Nutrition and Metabolism*, 2018, 1–27.
 doi:10.1155/2018/5157645
- Miller, R. P., Tadagavadi, R. K., Ramesh, G., & Reeves, W. B. (2010).
 Mechanisms of Cisplatin Nephrotoxicity. *Toxins*, *2*, 2490–2518.
 doi:10.3390/toxins2112490
- Mills, E. L., Kelly, B., & O'Neill, L. A. J. (2017). Mitochondria are the powerhouses of immunity. *Nature Immunology*, *18*, 488–498. doi:10.1038/ni.3704
- Misgeld, T., & Schwarz, T. L. (2017). Mitostasis in Neurons: Maintaining
 Mitochondria in an Extended Cellular Architecture. *Neuron*, *96*, 651–666.
 doi:10.1016/j.neuron.2017.09.055
- Mishra, P., Singh, U., Pandey, C., Mishra, P., & Pandey, G. (2019). Application of student's t-test, analysis of variance, and covariance. *Annals of Cardiac Anaesthesia*, *22*, 407. doi:10.4103/aca.ACA_94_19
- Mitchell, P. (1961). Coupling of Phosphorylation to Electron and Hydrogen Transfer by a Chemi-Osmotic type of Mechanism. *Nature*, *191*, 144–148. doi:10.1038/191144a0

- Mittal, R. D. (2015). Tandem Mass Spectroscopy in Diagnosis and Clinical Research. *Indian Journal of Clinical Biochemistry*, 30, 121–123. doi:10.1007/s12291-015-0498-9
- Mohamed, A. A., Zhang, X., & Jan, Y.-K. (2023). Evidence-based and adverse-effects analyses of cupping therapy in musculoskeletal and sports rehabilitation: A systematic and evidence-based review. *Journal of Back and Musculoskeletal Rehabilitation*, 36, 3–19. doi:10.3233/BMR-210242
- Mohapatra, S., Iqubal, A., Ansari, M. J., Jan, B., Zahiruddin, S., Mirza, M. A.,
 Ahmad, S., & Iqbal, Z. (2022). Benefits of Black Cohosh (Cimicifuga racemosa) for Women Health: An Up-Close and In-Depth Review. *Pharmaceuticals*, *15*(3), 278. doi:10.3390/ph15030278
- Moher, D., Liberati, A., Tetzlaff, J., & Altman, D. G. (2009). Preferred Reporting Items for Systematic Reviews and Meta-Analyses: The PRISMA Statement. *PLoS Medicine*, 6, e1000097. doi:10.1371/journal.pmed.1000097
- Moloney, J. N., & Cotter, T. G. (2018). ROS signalling in the biology of cancer. Seminars in Cell & Developmental Biology, 80, 50–64. doi:10.1016/J.SEMCDB.2017.05.023
- Monbaliu, S., Wu, A., Zhang, D., Van Peteghem, C., & Saeger, S. D. (2010). Multimycotoxin UPLC-MS/MS for Tea, Herbal Infusions and the Derived

Drinkable Products. *Journal of Agricultural and Food Chemistry*, 58, 12664–12671. doi:10.1021/jf1033043

- Moore, J. H., & Truelove, B. (1970). Ochratoxin A: Inhibition of Mitochondrial
 Respiration. *Science*, *168*, 1102–1103.
 doi:10.1126/science.168.3935.1102
- Moorhouse, T. P., Zhou, Z.-M., Shao, M.-L., Zhou, Y., Elwin, A., D'Cruze, N.
 C., & Macdonald, D. W. (2022). Substitutes for wildlife-origin materials as described in China's "TCM" research literature. *Global Ecology and Conservation*, *34*, e02042. doi:10.1016/j.gecco.2022.e02042
- Morciano, G., Imamura, H., Patergnani, S., Pedriali, G., Giorgi, C., & Pinton,
 P. (2020). *Measurement of ATP concentrations in mitochondria of living cells using luminescence and fluorescence approaches*.
 doi:10.1016/bs.mcb.2019.10.007
- Morciano, G., Sarti, A. C., Marchi, S., Missiroli, S., Falzoni, S., Raffaghello,
 L., Pistoia, V., Giorgi, C., di Virgilio, F., & Pinton, P. (2017). Use of
 luciferase probes to measure ATP in living cells and animals. *Nature Protocols*, *12*, 1542–1562. doi:10.1038/nprot.2017.052
- Morriswood, B. (2017). Seeing triple (a short guide to experiment reproducibility). *Total Internal Reflection*. Retrieved from https://totalinternalreflectionblog.com/2017/09/06/seeing-triple-a-short-gui de-to-experiment-reproducibility/ (accessed 24 December 2023).

- Mould, R. (2019). *"Priming" as a novel strategy to enhance chemotherapeutic efficacy*. PhD Thesis. University of Westminster. doi:10.34737/qx75q
- Mousavi, S. S., Karami, A., Saharkhiz, M. J., Etemadi, M., & Zarshenas, M.
 M. (2022). Evaluation of metabolites in Iranian Licorice accessions under salinity stress and Azotobacter sp. inoculation. *Scientific Reports*, *12*(1), 15837. doi:10.1038/s41598-022-20366-6
- Müller-Maatsch, J., Schweiggert, R. M., & Carle, R. (2016). Adulteration of anthocyanin- and betalain-based coloring foodstuffs with the textile dye 'Reactive Red 195' and its detection by spectrophotometric, chromatic and HPLC-PDA-MS/MS analyses. *Food Control*, 70, 333–338. doi:10.1016/j.foodcont.2016.06.012
- Munteanu, I. G., & Apetrei, C. (2021). Analytical Methods Used in Determining Antioxidant Activity: A Review. International Journal of Molecular Sciences, 22, 3380. doi:10.3390/ijms22073380
- Nagar, P. S., Rane, S., & Dwivedi, M. (2022). LC-MS/MS standardization and validation of glycyrrhizin from the roots of Taverniera cuneifolia: A potential alternative source of Glycyrrhiza glabra. *Heliyon*, *8*, e10234. doi:10.1016/j.heliyon.2022.e10234
- Nagendra, R. P., Maruthai, N., & Kutty, B. M. (2012). Meditation and Its Regulatory Role on Sleep. *Frontiers in Neurology*, 3. doi:10.3389/fneur.2012.00054
- Naiker, M., Anderson, S., Johnson, J. B., Mani, J. S., Wakeling, L., & Bowry,
 V. (2020). Loss of *trans* -resveratrol during storage and ageing of red
 wines. *Australian Journal of Grape and Wine Research*, *26*, 385–387.
 doi:10.1111/ajgw.12449
- Naithani, V., & Kakkar, P. (2004). An Evaluation of Residual Organochlorine
 Pesticides in Popular Indian Herbal Teas. *Archives of Environmental Health:* An International Journal, 59(8), 426–430.
 doi:10.3200/AEOH.59.8.426-430
- National Center for Biotechnology Information. (2023a). PubChem Compound Summary for CID 14982, Glycyrrhizin. *PubChem*. Available from https://pubchem.ncbi.nlm.nih.gov/compound/Glycyrrhizin (accessed 25 August 2023).
- National Center for Biotechnology Information. (2023b). PubChem Compound Summary for CID 10114, Enoxolone. *PubChem*. Available from https://pubchem.ncbi.nlm.nih.gov/compound/enoxolone (accessed 25 August 2023).
- National Center for Biotechnology Information (2023c). PubChem Compound Summary for CID 5754, Hydrocortisone. *Pubchem*. Available from https://pubchem.ncbi.nlm.nih.gov/compound/Hydrocortisone (accessed 25 August 2023).
- National Center for Biotechnology Information. (2023d). PubChem Compound Summary for CID 503737, Liquiritin. *PubChem*. Available

https://pubchem.ncbi.nlm.nih.gov/compound/Liquiritin (accessed 25 August 2023).

- National Center for Biotechnology Information. (2023e). PubChem Compound Summary for CID 114829, Liquiritigenin. *PubChem*. Available from https://pubchem.ncbi.nlm.nih.gov/compound/Liquiritigenin (accessed 25 August 2023).
- National Center for Biotechnology Information. (2023f). PubChem Compound Summary for CID 638278, Trihydroxychalcone. *PubChem*. Available from https://pubchem.ncbi.nlm.nih.gov/compound/Trihydroxychalcone (accessed 25 August 2023).
- National Center for Biotechnology Information. (2023g). PubChem Compound Summary for CID 10076238, Liquiritin apioside. *PubChem*. Available from https://pubchem.ncbi.nlm.nih.gov/compound/Liquiritin-apioside (accessed 25 August 2023).
- National Center for Biotechnology Information. (2023h). PubChem Compound Summary for CID 124052, Glabridin. *PubChem*. Available from https://pubchem.ncbi.nlm.nih.gov/compound/Glabridin (accessed 25 August 2023).
- National Center for Biotechnology Information. (2023i). PubChem Compound Summary for CID 480774, Glabrene. *PubChem*. Available

from https://pubchem.ncbi.nlm.nih.gov/compound/Glabrene (accessed 25 August 2023).

- National Center for Biotechnology Information. (2023j). PubChem Compound Summary for CID 5317756, Glycycoumarin. *PubChem*. Available from https://pubchem.ncbi.nlm.nih.gov/compound/94805-82-0 (accessed 9 October 2023).
- National Center for Biotechnology Information. (2023k). PubChem Compound Summary for CID 5318998, Licochalcone a. *PubChem*. Available from https://pubchem.ncbi.nlm.nih.gov/compound/Licochalcone-a (accessed 25 August 2023).
- National Institute for Health. (2020). *Black Cohosh*. Retrieved from https://ods.od.nih.gov/factsheets/BlackCohosh-HealthProfessional/ (accessed 31 July 2024).
- National Institute for Health and Care Excellence. (2023). Dyspepsia proven peptic ulcer. *Clinical Knowledge Summaries*. Retrieved from https://cks.nice.org.uk/topics/dyspepsia-proven-peptic-ulcer/ (accessed 4 January 2024).
- Naviaux, R. K. (2023). Mitochondrial and metabolic features of salugenesis and the healing cycle. *Mitochondrion*, 70, 131–163. doi:10.1016/j.mito.2023.04.003

- Nazarewicz, R. R., Bikineyeva, A., & Dikalov, S. I. (2013). Rapid and Specific Measurements of Superoxide Using Fluorescence Spectroscopy. *SLAS Discovery*, *18*, 498–503. doi:10.1177/1087057112468765
- Nazari, S., Rameshrad, M., & Hosseinzadeh, H. (2017). Toxicological Effects of Glycyrrhiza glabra (Licorice): A Review. *Phytotherapy Research*, *31*, 1635–1650. doi:10.1002/ptr.5893
- National Center for Complementary and Integrative Health (2020). *Licorice Root.* Retrieved from https://www.nccih.nih.gov/health/licorice-root (accessed 25 August 2023).
- Nedungadi, D., Binoy, A., Vinod, V., Vanuopadath, M., Nair, S. S., Nair, B.
 G., & Mishra, N. (2021). Ginger extract activates caspase independent paraptosis in cancer cells via ER stress, mitochondrial dysfunction, AIF translocation and DNA damage. *Nutrition and Cancer*, *73*(1), 147–159. doi:10.1080/01635581.2019.1685113
- Newby, K. (2023). The Spice Seller's Secret. *Stanford Medical Magazine*. Retrieved from https://stanmed.stanford.edu/turmeric-lead-risk-detect/ (accessed 9 December 2023).
- Newmaster, S. G., Grguric, M., Shanmughanandhan, D., Ramalingam, S., & Ragupathy, S. (2013). DNA barcoding detects contamination and substitution in North American herbal products. *BMC Medicine*, *11*, 222. doi:10.1186/1741-7015-11-222

- Newton, I. (1730). *Opticks: or a treatise of the Reflections, Refractions, Inflections and Colours of Light* (4th edition). Retrieved from https://www.gutenberg.org/cache/epub/33504/pg33504-images.html (accessed 18 November 2022).
- Ng, G. (2017). *What Are the Advantages of HPLC Over GC?* Retrieved from https://sciencing.com/advantages-hplc-over-gc-6822919.html (accessed 6 January 2021).
- Niederberger, E., Wilken-Schmitz, A., Manderscheid, C., Schreiber, Y., Gurke, R., & Tegeder, I. (2022). Non-Reproducibility of Oral Rotenone as a Model for Parkinson's Disease in Mice. *International Journal of Molecular Sciences*, 23, 12658. doi:10.3390/ijms232012658
- Nybom, H., Weising, K., & Rotter, B. (2014). DNA fingerprinting in botany:
 past, present, future. *Investigative Genetics*, 5, 1.
 doi:10.1186/2041-2223-5-1
- Oganesyan, K. R. (2002). Antioxidant Effect of Licorice Root on Blood Catalase Activity in Vibration Stress. *Bulletin of Experimental Biology and Medicine*, *134*, 135–136. doi:10.1023/A:1021123928740
- O'Kelly, F. (2021). Liquorice: how different sugars impact flavour and production. *Ragus*. May 13, 2021. Retrieved from https://www.ragus.co.uk/liquorice/ (accessed 27 August 2023).

- Oganesyan, K. R. (2002). Antioxidant Effect of Licorice Root on Blood Catalase Activity in Vibration Stress. *Bulletin of Experimental Biology and Medicine*, 134(2), 135–136. doi:10.1023/A:1021123928740
- Oh, H., Seo, J., Lee, M., Kim, C., Kim, E., Yoon, G., Cho, S., Cho, Y. S., Choi, H. W., Shim, J., & Chae, J. (2018). Licochalcone C induced apoptosis in human oral squamous cell carcinoma cells by regulation of the JAK2/STAT3 signaling pathway. *Journal of Cellular Biochemistry*, *119*, 10118–10130. doi:10.1002/jcb.27349
- Ojha, S., Javed, H., Azimullah, S., Khair, S. B. A., & Haque, M. E. (2016).
 Glycyrrhizic acid Attenuates Neuroinflammation and Oxidative Stress in Rotenone Model of Parkinson's Disease. *Neurotoxicity Research*, *29*, 275–287. doi:10.1007/s12640-015-9579-z
- Okayama, N., Fowler, M. R., Jennings, S. R., Patel, P., Specian, R., Alexander, B., Jackson, T. H., Oshima, T., Shannon, K., & Alexander, J. S. (2000). Characterization of JOK-1, a human gastric epithelial cell line. *In Vitro Cellular & Developmental Biology Animal*, 36, 228. doi:10.1290/1071-2690(2000)036<0228:COJAHG>2.0.CO;2
- Oliveira, M. R., Nabavi, S. F., Daglia, M., Rastrelli, L., & Nabavi, S. M. (2016). Epigallocatechin gallate and mitochondria—A story of life and death. *Pharmacological Research*, *104*, 70–85. doi:10.1016/j.phrs.2015.12.027

- Olivero, O. A., Semino, C., Kassim, A., Lopez-Larraza, D. M., & Poirier, M.
 C. (1995). Preferential binding of cisplatin to mitochondrial DNA of Chinese hamster ovary cells. *Mutation Research Letters*, *346*, 221–230. doi:10.1016/0165-7992(95)90039-X
- Omar, H. R., Komarova, I., El-Ghonemi, M., Fathy, A., Rashad, R.,
 Abdelmalak, H. D., Yerramadha, M. R., Ali, Y., Helal, E., & Camporesi, E.
 M. (2012). Licorice abuse: time to send a warning message. *Therapeutic Advances in Endocrinology and Metabolism*, *3*, 125–138.
 doi:10.1177/2042018812454322
- Omicron UK. (2018). *Herbal Medicines and Botanical Supplements*. Retrieved from https://www.omicron-uk.com/herbal-medicines-and-botanicals (accessed 16 December 2020).
- Otto, A. M. (2016). Warburg effect(s)—a biographical sketch of Otto
 Warburg and his impacts on tumor metabolism. *Cancer & Metabolism*, *4*,
 5. doi:10.1186/s40170-016-0145-9
- Özzeybek, G., Erarpat, S., & Bakırdere, S. (2020). Simple, Sensitive, and Selective High Performance Liquid Chromatographic (HPLC) Method for the Determination of Buturon in Herbal Tea, Dried Blueberry, and Cranberry Samples and Evaluation of Its Stability in Gastric Conditions. *Analytical Letters*, 53(10), 1525–1535. doi:10.1080/00032719.2019.1710746

- Painter, T. (2018). The Differences Between HPLC & GC. Retrieved from https://sciencing.com/differences-between-hplc-gc-7456786.html (accessed 2 January 2020).
- Pandey, A. K., Samota, M. K., & Sanches Silva, A. (2023). Mycotoxins along the tea supply chain: A dark side of an ancient and high valued aromatic beverage. *Critical Reviews in Food Science and Nutrition*, 63(27), 8672–8697. doi:10.1080/10408398.2022.2061908
- Pannu, N., & Bhatnagar, A. (2019). Resveratrol: from enhanced biosynthesis and bioavailability to multitargeting chronic diseases. *Biomedicine* & *Pharmacotherapy*, 109, 2237–2251.
 doi:10.1016/j.biopha.2018.11.075
- Panossian, A. (2023). Challenges in phytotherapy research. *Frontiers in Pharmacology*, *14*. doi:10.3389/fphar.2023.1199516
- Panossian, A., & Wikman, G. (2010). Effects of Adaptogens on the Central Nervous System and the Molecular Mechanisms Associated with Their Stress—Protective Activity. *Pharmaceuticals*, *3*, 188–224. doi:10.3390/ph3010188
- Panossian, A., Wikman, G., & Wagner, H. (1999). Plant adaptogens III. Earlier and more recent aspects and concepts on their mode of action. *Phytomedicine*, *6*, 287–300. doi:10.1016/S0944-7113(99)80023-3

- Park, M. K., Ji, J., Haam, K., Han, T.-H., Lim, S., Kang, M.-J., Lim, S. S., & Ban, H. S. (2021). Licochalcone A inhibits hypoxia-inducible factor-1α accumulation by suppressing mitochondrial respiration in hypoxic cancer cells. *Biomedicine & Pharmacotherapy*, *133*, 111082. doi:10.1016/j.biopha.2020.111082
- Parveen, I., Gafner, S., Techen, N., Murch, S., & Khan, I. (2016). DNA
 Barcoding for the Identification of Botanicals in Herbal Medicine and
 Dietary Supplements: Strengths and Limitations. *Planta Medica*, *82*, 1225–1235. doi:10.1055/s-0042-111208
- Pastorino, G., Cornara, L., Soares, S., Rodrigues, F., & Oliveira, M. B. P. P. (2018). Liquorice (*Glycyrrhiza glabra*): A phytochemical and pharmacological review. *Phytotherapy Research*, *32*, 2323–2339. doi:10.1002/ptr.6178
- Patel, D. N., Li, L., Kee, C.-L., Ge, X., Low, M.-Y., & Koh, H.-L. (2014). Screening of synthetic PDE-5 inhibitors and their analogues as adulterants: Analytical techniques and challenges. Journal of Pharmaceutical 87. and Biomedical Analysis. 176–190. doi:10.1016/j.jpba.2013.04.037
- Pei, X., Zhang, W., Jiang, H., Liu, D., Liu, X., Li, L., Li, C., Xiao, X., Tang, S., & Li, D. (2021). Food-Origin Mycotoxin-Induced Neurotoxicity: Intend to Break the Rules of Neuroglia Cells. *Oxidative Medicine and Cellular Longevity*, 2021, 1–14. doi:10.1155/2021/9967334

- Peng, W., Qin, R., Li, X., & Zhou, H. (2013). Botany, phytochemistry, pharmacology, and potential application of Polygonum cuspidatum Sieb.et Zucc.: A review. *Journal of Ethnopharmacology*, *148*, 729–745. doi:10.1016/j.jep.2013.05.007
- Perry, S. W., Norman, J. P., Barbieri, J., Brown, E. B., & Gelbard, H. A. (2011). Mitochondrial membrane potential probes and the proton gradient:
 a practical usage guide. *BioTechniques*, *50*, 98–115. doi:10.2144/000113610
- Persistence Market Research. (2023). *Traditional Chinese Medicine Market*. Retrieved from https://www.persistencemarketresearch.com/market-research/traditional-c hinese-medicine-market.asp (accessed 6 January 2024).
- Phaniendra, A., Jestadi, D. B., & Periyasamy, L. (2015). Free Radicals:
 Properties, Sources, Targets, and Their Implication in Various Diseases. *Indian Journal of Clinical Biochemistry*, 30, 11–26.
 doi:10.1007/s12291-014-0446-0
- Pharmacognosy. (n.d.). Liquorice Uses, Botanical Source, Characters,AndChemicalConstituents.Retrievedfromhttps://thepharmacognosy.com/liquorice/ (accessed 24 November 2022).
- Picard, M., McEwen, B. S., Epel, E. S., & Sandi, C. (2018). An energetic view of stress: Focus on mitochondria. *Frontiers in Neuroendocrinology*, 49, 72–85. doi:10.1016/j.yfrne.2018.01.001

- Pinches IV, J. L., Pinches, Y. L., Johnson, J. O., Haddad, N. C., Boueri, M. G., Oke, L. M., & Haddad, G. E. (2022). Could "cellular exercise" be the missing ingredient in a healthy life? Diets, caloric restriction, and exercise-induced hormesis. *Nutrition*, 99–100, 111629. doi:10.1016/j.nut.2022.111629
- Pinela, J., Barros, L., Antonio, A., Carvalho, A., Oliveira, M., & Ferreira, I. (2016). Quality Control of Gamma Irradiated Dwarf Mallow (Malva neglecta Wallr.) Based on Color, Organic Acids, Total Phenolics and Antioxidant Parameters. *Molecules*, *21*, 467. doi:10.3390/molecules21040467
- Pirsalami, S., Bagherpour, S., Ebrahim Bahrololoom, M., & Riazi, M. (2021).
 Adsorption efficiency of glycyrrhiza glabra root toward heavy metal ions:
 Experimental and molecular dynamics simulation study on removing copper ions from wastewater. *Separation and Purification Technology*, 275, 119215. doi:10.1016/j.seppur.2021.119215
- Pliny the Elder (1857). The Surrentine Wines: Three Remedies. The Alban Wines: Two Remedies. The Falernian Wines: Six Remedies. In *Natural History of Pliny: Vol. IV, Book XXIII*. (trans. J. Bostock, & H. T. Riley).
 London: Taylor and Francis. Retrieved from http://www.perseus.tufts.edu/hopper/text?doc=Perseus%3Atext%3A1999.
 02.0137%3Abook%3D23%3Achapter%3D20 (accessed 29 May 2023).

- Pole, S. (2012). *Ayurvedic Medicine: The Principles of Traditional Practice*. London: Singing Dragon.
- Poljsak, B., & Milisav, I. (2019). The Role of Antioxidants in Cancer, Friends
 or Foes? *Current Pharmaceutical Design*, 24, 5234–5244.
 doi:10.2174/1381612825666190123112647
- Pollington, S. (2008). Leechcraft: Early English Charms, Plantlore, and Healing. Little Downham, Ely: Anglo-Saxon Books.
- Polyakov, N. E., Leshina, T. V., Salakhutdinov, N. F., Konovalova, T. A., & Kispert, L. D. (2006). Antioxidant and redox properties of supramolecular complexes of carotenoids with β-glycyrrhizic acid. *Free Radical Biology and Medicine*, *40*, 1804–1809. doi:10.1016/j.freeradbiomed.2006.01.015
- Popović, K., Nikolić, M. A., Dražić, B., Jovanović, D., & Tanasković, S. (2022). Assessment of the Quality of Herbal Teas from Šabac, Serbia in Terms of the Content of Heavy Metals. *Pharmaceutical Chemistry Journal*, *56*(6), 827–837. doi:10.1007/s11094-022-02716-w
- Prasad, S., Gupta, S. C., & Tyagi, A. K. (2017). Reactive oxygen species (ROS) and cancer: Role of antioxidative nutraceuticals. *Cancer Letters*, 387, 95–105. doi:10.1016/j.canlet.2016.03.042
- Process NMR Associates (2007). *Maltodextrin Adulteration of Licorice Observed by NMR*. Retrieved from

https://www.process-nmr.com/licorice-adulteration-with-maltodextrin/ (accessed 23 November 2022).

- Professional Register of TCM. (2022). *Medical Conditions Treated by TCM*. Retrieved from https://prtcm.org/medical-conditions-treated-by-tcm/ (accessed 9 August 2023).
- Pukka Herbs (2020). *Three Licorice*. Retrieved from https://www.pukkaherbs.com/uk/en/products/organic-teas/three-licorice.ht ml (accessed 21 November 2020).
- Pukka Herbs (2023). Cinnamon Tea. Retrieved from https://www.pukkaherbs.com/uk/en/hero-herbs/cinnamon (accessed 27 September 2023).
- Qin, Y., Zhao, Q., Zhou, D., Shi, Y., Shou, H., Li, M., Zhang, W., & Jiang, C. (2024). Application of flash GC e-nose and FT-NIR combined with deep learning algorithm in preventing age fraud and quality evaluation of pericarpium citri reticulatae. *Food Chemistry: X*, 21, 101220. doi:10.1016/j.fochx.2024.101220
- Qin, J., Chen, J., Peng, F., Sun, C., Lei, Y., Chen, G., Li, G., Yin, Y., Lin, Z.,
 Wu, L., Li, J., Liu, W., Peng, C., & Xie, X. (2022). Pharmacological activities and pharmacokinetics of liquiritin: A review. *Journal of Ethnopharmacology*, 293, 115257. doi:10.1016/j.jep.2022.115257

- Qin, Y., Zhou, R., Huang, J., Jin, J., Zhou, Q., Liu, H., Xiao, J., Zhao, Y., Shu, J., Zhang, S., & Huang, L. (2019). Comprehensive comparison of the anti-inflammatory activity and chemical consistency of traditional Chinese medicine formula granules with Ge-Gen decoction as a representative sample. *Biomedical Chromatography*, 33. doi:10.1002/bmc.4689
- Qu, Z., Bing, Y., Zhang, T., Zheng, Y., Wu, S., Ji, C., Li, W., & Zou, X. (2023).
 Screening of Q-markers for the wine-steamed Schisandra chinensis decoction pieces in improving allergic asthma. *Chinese Medicine*, *18*(1), 10. doi:10.1186/s13020-023-00712-0
- Quirós, P. M., Mottis, A., & Auwerx, J. (2016). Mitonuclear communication in homeostasis and stress. Nature Reviews Molecular Cell Biology, 17(4), 213–226. doi:10.1038/nrm.2016.23
- Rabinowitz, J. D., & Enerbäck, S. (2020). Lactate: the ugly duckling of energy metabolism. *Nature Metabolism*, 2, 566–571. doi:10.1038/s42255-020-0243-4
- Račková, L., Jančinová, V., Petríková, M., Drábiková, K., Nosáľ, R., Štefek,
 M., Košťálová, D., Prónayová, N., & Kováčová, M. (2007). Mechanism of
 anti-inflammatory action of liquorice extract and glycyrrhizin. *Natural Product Research*, 21(14), 1234–1241. doi:10.1080/14786410701371280
- Rahimifard, M., Navaei-Nigjeh, M., Mahroui, N., Mirzaei, S., Siahpoosh, Z., D, P., Nili-Ahmadabadi, A., Mohammadirad, A., Baeeri, M., Hajiaghaie,

R., & Abdollahi, M. (2014). Improvement in The Function of Isolated Rat Pancreatic Islets through Reduction of Oxidative Stress Using Traditional Iranian Medicine. *Cell Journal*, *16*, 147–163. Retrieved from https://www.ncbi.nlm.nih.gov/pubmed/24567945 (accessed 28 January 2024).

- Raja, P. M. V., & Barron, A. R. (2020). UV-Visible Spectroscopy. Retrieved from https://chem.libretexts.org/@go/page/55881 (accessed 15 November 2020).
- Rajendran, M., Dane, E., Conley, J., & Tantama, M. (2016). Imaging
 Adenosine Triphosphate (ATP). *The Biological Bulletin*, 231, 73–84.
 doi:10.1086/689592
- Ralla, T., Salminen, H., Braun, K., Edelmann, M., Dawid, C., Hofmann, T., & Weiss, J. (2020). Investigations into the Structure-Function Relationship of the Naturally-Derived Surfactant Glycyrrhizin: Emulsion Stability. *Food Biophysics*, *15*, 288–296. doi:10.1007/s11483-020-09624-0
- Ramachandra, C. J. A., Hernandez-Resendiz, S., Crespo-Avilan, G. E., Lin,
 Y.-H., & Hausenloy, D. J. (2020). Mitochondria in acute myocardial infarction and cardioprotection. *EBioMedicine*, 57, 102884. doi:10.1016/j.ebiom.2020.102884
- Ramachandran, A. P., Samarakoon, S. M. S., Harisha, C. R., Prasad, M. S., Chandola, H. M., & Shukla, V. J. (2012). Pharmacognostical and phytochemical evaluation of Vara Asanadi Kwatha. *AYU (An International*

Quarterly Journal of Research in Ayurveda), *33*(1), 130. doi:10.4103/0974-8520.100330

- Ramalingam, M., Kim, H., Lee, Y., & Lee, Y.-I. (2018). Phytochemical and Pharmacological Role of Liquiritigenin and Isoliquiritigenin From Radix Glycyrrhizae in Human Health and Disease Models. *Frontiers in Aging Neuroscience*, *10*. doi:10.3389/fnagi.2018.00348
- Rana, P., & Sheu, S.-C. (2023). Discrimination of four Cinnamomum species
 by proximate, antioxidant, and chemical profiling: towards quality
 assessment and authenticity. *Journal of Food Science and Technology*,
 60, 2639–2648. doi:10.1007/s13197-023-05788-y
- Rashedinia, M., Saberzadeh, J., Bakhtiari, T. K., Hozhabri, S., & Arabsolghar, R. (2019). Glycyrrhizic Acid Ameliorates Mitochondrial Function and Biogenesis Against Aluminum Toxicity in PC12 Cells. *Neurotoxicity Research*, *35*, 584–593. doi:10.1007/s12640-018-9967-2
- Razmovski-Naumovski, V., Zhou, X., Wong, H. Y., Kam, A., Pearson, J., &
 Chan, K. (2020). Chromatographic, Chemometric and Antioxidant
 Assessment of the Equivalence of Granules and Herbal Materials of
 Angelicae Sinensis Radix. *Medicines*, 7(6), 35.
 doi:10.3390/medicines7060035
- Reigada, I., Moliner, C., Valero, M. S., Weinkove, D., Langa, E., & Rincón,C. G. (2020). Antioxidant and Antiaging Effects of Licorice on the

Caenorhabditis elegans Model. Journal of Medicinal Food, 23, 72–78. doi:10.1089/jmf.2019.0081

- Reinholds, I., Bogdanova, E., Pugajeva, I., Alksne, L., Stalberga, D.,
 Valcina, O., & Bartkevics, V. (2020). Determination of Fungi and
 Multi-Class Mycotoxins in Camelia sinensis and Herbal Teas and Dietary
 Exposure Assessment. *Toxins*, *12*, 555. doi:10.3390/toxins12090555
- Reisch, M. S. (2015). A Renaissance For NMRs, Big And Small. *Chemical and Engineering News*, 93. Retrieved from https://cen.acs.org/articles/93/i37/Renaissance-NMRs-Big-Small.html (accessed 14 November 2020).
- Reiter, R., Tan, D., Rosales-Corral, S., Galano, A., Zhou, X., & Xu, B. (2018). Mitochondria: Central Organelles for Melatonin's Antioxidant and Anti-Aging Actions. *Molecules*, *23*, 509. doi:10.3390/molecules23020509
- Reiter, R. J., Mayo, J. C., Tan, D.-X., Sainz, R. M., Alatorre-Jimenez, M., &
 Qin, L. (2016). Melatonin as an antioxidant: under promises but over delivers. *Journal of Pineal Research*, *61*, 253–278. doi:10.1111/jpi.12360
- Reusch, W. (2013). Infrared Spectroscopy. Retrieved from https://www2.chemistry.msu.edu/faculty/reusch/VirtTxtJml/Spectrpy/InfraR ed/infrared.htm (accessed 15 November 2020).
- Riddle, J. M. (1985). *Dioscorides on Pharmacy and Medicine*. Austin, TX: University of Texas Press.

- Rijsbergen, L. C., van Dijk, L. L. A., Engel, M. F. M., de Vries, R. D., & de Swart, R. L. (2021). In Vitro Modelling of Respiratory Virus Infections in Human Airway Epithelial Cells A Systematic Review. *Frontiers in Immunology*, *12*. doi:10.3389/fimmu.2021.683002
- Riss, T. L., Moravec, R. A., Niles, A. L., Duellman, S., Benink, H. A., Worzella, T. J., & Minor, RizzL. (2004). *Cell Viability Assays*. Available from https://www.ncbi.nlm.nih.gov/pubmed/23805433 (accessed 28 January 2024).
- Rizzato, G., Scalabrin, E., Radaelli, M., Capodaglio, G., & Piccolo, O.
 (2017). A new exploration of licorice metabolome. *Food Chemistry*, 221, 959–968. doi:10.1016/j.foodchem.2016.11.068
- Roger, A. J., Muñoz-Gómez, S. A., & Kamikawa, R. (2017). The Origin and
 Diversification of Mitochondria. *Current Biology*, 27, R1177–R1192.
 doi:10.1016/j.cub.2017.09.015
- Rolfe, V. (2023). *The amazing benefits of licorice*. Retrieved from https://www.pukkaherbs.com/uk/en/wellbeing-articles/why-were-wild-abou t-licorice (accessed 3 September 2023).
- Rossi, A., Pizzo, P., & Filadi, R. (2019). Calcium, mitochondria and cell metabolism: A functional triangle in bioenergetics. *Biochimica et Biophysica Acta (BBA) Molecular Cell Research*, 1866, 1068–1078. doi:10.1016/j.bbamcr.2018.10.016

- Rowshan, V., Bahmanzadegan, A., & Saharkhiz, M. J. (2013). Influence of storage conditions on the essential oil composition of Thymus daenensis
 Celak. *Industrial Crops and Products*, 49, 97–101. doi:10.1016/j.indcrop.2013.04.029
- Rubio, C., Lucas, J. R. D., Gutiérrez, A. J., Glez-Weller, D., Pérez Marrero,
 B., Caballero, J. M., Revert, C., & Hardisson, A. (2012). Evaluation of
 metal concentrations in mentha herbal teas (Mentha piperita, Mentha
 pulegium and Mentha species) by inductively coupled plasma
 spectrometry. *Journal of Pharmaceutical and Biomedical Analysis*, *71*,
 11–17. doi:10.1016/j.jpba.2012.07.015
- Saad, B., & Said, O. (2011). Greco-Arab and Islamic herbal medicine: traditional system, ethics, safety, efficacy, and regulatory issues.
 Hoboken, NJ: John Wiley & Sons.
- Sack, H. (2018). Joseph von Fraunhofer and the Solar Spectrum. *SciHi*. Retrieved from http://scihi.org/joseph-fraunhofer-solar-spectrum/ (accessed 14 March 2023).
- Saeed, M., Naveed, M., Arif, M., Kakar, M. U., Manzoor, R., Abd El-Hack, M.
 E., Alagawany, M., Tiwari, R., Khandia, R., Munjal, A., Karthik, K.,
 Dhama, K., Iqbal, H. M. N., Dadar, M., & Sun, C. (2017). Green tea
 (Camellia sinensis) and I-theanine: Medicinal values and beneficial applications in humans—A comprehensive review. *Biomedicine & Pharmacotherapy*, *95*, 1260–1275. doi:10.1016/j.biopha.2017.09.024

- Sahar Baghal-Sadriforoush, Morteza Bagheri, Isa Abdi Rad, & Fattah Sotoodehnejadnematalahi. (2022). Melatonin Sensitizes OVCAR-3 Cells to Cisplatin through Suppression of PI3K/Akt Pathway. *Cellular and Molecular Biology*, 68(4), 158–169. doi:10.14715/cmb/2022.68.4.19
- Saiki, R. K., Gelfand, D. H., Stoffel, S., Scharf, S. J., Higuchi, R., Horn, G. T., Mullis, K. B., & Erlich, H. A. (1988). Primer-Directed Enzymatic Amplification of DNA with a Thermostable DNA Polymerase. *Science*, 239, 487–491. doi:10.1126/science.2448875
- Santhosh Kumar, J. U., Krishna, V., Seethapathy, G. S., Ganesan, R., Ravikanth, G., & Shaanker, R. U. (2018). Assessment of adulteration in raw herbal trade of important medicinal plants of India using DNA barcoding. *3 Biotech*, 8(3), 135. doi:10.1007/s13205-018-1169-3
- Santulli, C., Rallini, M., Puglia, D., Gabrielli, S., Torre, L., & Marcantoni, E. (2020). Characterization of Licorice Root Waste for Prospective Use as
 Filler in more Eco-Friendly Composite Materials. *Processes*, *8*, 733. doi:10.3390/pr8060733
- Sardão, V. A., Oliveira, P. J., & Moreno, A. J. M. (2002). Caffeine Enhances the Calcium-Dependent Cardiac Mitochondrial Permeability Transition: Relevance for Caffeine Toxicity. *Toxicology and Applied Pharmacology*, 179(1), 50–56. doi:10.1006/taap.2001.9334
- Sawada, Y., Matsuda, F., & Hirai, MY. (2008). *MassBank Record: MSBNK-RIKEN_ReSpect-PS044001: Glycyrrhizin, glycyrrhizic Acid,*

3-O-(2-O-beta-D-Glucopyranuronosyl-alpha-D-glucopyranuronosyl)-18bet a-glycyrrhetinic acid; LC-ESI-QQ; MS2. Plant Science Center, RIKEN. Received from https://massbank.eu/MassBank/RecordDisplay?id=MSBNK-RIKEN_ReSp ect-PS044009&dsn=RIKEN_ReSpect (accessed 9 July 2024).

- Sax, J. K. (2015). Dietary Supplements are Not all Safe and Not all Food:
 How the Low Cost of Dietary Supplements Preys on the Consumer. *American Journal of Law & Medicine*, 41, 374–394.
 doi:10.1177/0098858815591523
- Sayin, V. I., Ibrahim, M. X., Larsson, E., Nilsson, J. A., Lindahl, P., & Bergo,
 M. O. (2014). Antioxidants Accelerate Lung Cancer Progression in Mice. *Science Translational Medicine*, 6. doi:10.1126/scitranslmed.3007653
- Schaedler, T. A., Faust, B., Shintre, C. A., Carpenter, E. P., Srinivasan, V., van Veen, H. W., & Balk, J. (2015). Structures and functions of mitochondrial ABC transporters. *Biochemical Society Transactions*, *43*, 943–951. doi:10.1042/BST20150118
- Scheid, V., Bensky, D., Ellis, A., & Barolet, R. (2009). Chinese Herbal Medicine: Formulas & Strategies (2nd edition). Seattle, WA: Eastland Press.
- Schmeda-Hirschmann, G., Rodriguez, J. A., Theoduloz, C., Astudillo, S. L., Feresin, G. E., & Tapia, A. (2003). Free-radical Scavengers and

Antioxidants from Peumus boldus Mol. ("Boldo"). *Free Radical Research*, 37(4), 447–452. doi:10.1080/1071576031000090000

- Schreiber, A. (2017). Advantages of Using Triple Quadrupole over Single Quadrupole Mass Spectrometry to Quantify and Identify the Presence of Pesticides in Water and Soil Samples. p. 1. Retrieved from https://collateral.sciex.com/manuals/msTripleQuad-Pesticides-Testing/ms TripleQuad-Pesticides-Testing rebrand.pdf (accessed 26 April 2021).
- Schulz, T. J., Zarse, K., Voigt, A., Urban, N., Birringer, M., & Ristow, M. (2007). Glucose Restriction Extends Caenorhabditis elegans Life Span by Inducing Mitochondrial Respiration and Increasing Oxidative Stress. *Cell Metabolism*, *6*, 280–293. doi:10.1016/j.cmet.2007.08.011
- Schunk, P. F. T., Kalil, I. C., Pimentel-Schmitt, E. F., Lenz, D., de Andrade, T. U., Ribeiro, J. S., & Endringer, D. C. (2016). ICP-OES and Micronucleus Test to Evaluate Heavy Metal Contamination in Commercially Available Brazilian Herbal Teas. *Biological Trace Element Research*, *172*, 258–265. doi:10.1007/s12011-015-0566-2
- Sedova, I., Kiseleva, M., & Tutelyan, V. (2018). Mycotoxins in Tea:
 Occurrence, Methods of Determination and Risk Evaluation. *Toxins*, *10*, 444. doi:10.3390/toxins10110444
- Sella, A. (2013). Kirchhoff's spectroscope. Chemistry World. Retrieved from https://www.chemistryworld.com/opinion/kirchhoffs-spectroscope/6547.art icle (accessed 14 March 2023).

- Selye, H. (1936). A Syndrome produced by Diverse Nocuous Agents. *Nature*, *138*, 32–32. doi:10.1038/138032a0
- Selyutina, O. Y., & Polyakov, N. E. (2019). Glycyrrhizic acid as a multifunctional drug carrier – From physicochemical properties to biomedical applications: A modern insight on the ancient drug. *International Journal of Pharmaceutics*, 559, 271–279. doi:10.1016/j.ijpharm.2019.01.047
- Selyutina, O. Y., Polyakov, N. E., Korneev, D. V., & Zaitsev, B. N. (2016).
 Influence of glycyrrhizin on permeability and elasticity of cell membrane:
 perspectives for drugs delivery. *Drug Delivery*, 23, 848–855.
 doi:10.3109/10717544.2014.919544
- Seo, S. H., Fang, F., & Kang, I. (2021). Ginger (Zingiber officinale)
 Attenuates Obesity and Adipose Tissue Remodeling in High-Fat Diet-Fed
 C57BL/6 Mice. International Journal of Environmental Research and
 Public Health, 18(2), 631. doi:10.3390/ijerph18020631
- Seo, C.-S., Yoo, S.-R., Jeong, S.-J., & Ha, H. (2019). Quantification of the constituents of the traditional Korea medicine, Samryeongbaekchul-san, and assessment of its antiadipogenic effect. *Saudi Pharmaceutical Journal*, 27, 145–153. doi:10.1016/j.jsps.2018.09.008
- Seyfried, T. N., & Chinopoulos, C. (2021). Can the Mitochondrial Metabolic Theory Explain Better the Origin and Management of Cancer than Can

the Somatic Mutation Theory? *Metabolites*, *11*, 572. doi:10.3390/metabo11090572

- Shah, K. (2020). *Glycyrrhiza, Biological Sources, Morphology, Chemical Constituents, Adulterants, Medicinal Uses and MCQ for GPAT*. Retrieved from https://gpatindia.com/liquorice/ (accessed 23 November 2022).
- Shahrajabian, M. H., Sun, W., & Cheng, Q. (2019). Clinical aspects and health benefits of ginger (*Zingiber officinale*) in both traditional Chinese medicine and modern industry. *Acta Agriculturae Scandinavica, Section B Soil* & *Plant Science*, *69*(6), 546–556. doi:10.1080/09064710.2019.1606930
- Shang, Z., & Gong, Y. (1994). Ben Cao Tu Jing (Atlas of Materia Medica). Retrieved from https://www.jameslindlibrary.org/ben-cao-tu-jing-11th-century/ (accessed 8 October 2023).
- Sharma, V., Katiyar, A., & Agrawal, R. C. (2018). Glycyrrhiza glabra: Chemistry and Pharmacological Activity. doi:10.1007/978-3-319-27027-2_21
- Sharma, R., & Sharma, S. (2023). *Physiology, Blood Volume*. Available from https://www.ncbi.nlm.nih.gov/pubmed/30252333 (accessed 28 January 2024).

- Sharma, H., & Verma, S. (2019). Is positive publication bias really a bias, or an intentionally created discrimination toward negative results? *Saudi Journal of Anaesthesia*, *13*, 352. doi:10.4103/sja.SJA_124_19
- Sharma, G., Kar, S., Palit, S., & Das, P. K. (2012). 18β-glycyrrhetinic acid induces apoptosis through modulation of Akt/FOXO3a/Bim pathway in human breast cancer MCF-7 cells. *Journal of Cellular Physiology*, 227, 1923–1931. doi:10.1002/jcp.22920
- Sharom, F. J. (2008). ABC multidrug transporters: structure, function and role in chemoresistance. *Pharmacogenomics*, 9, 105–127. doi:10.2217/14622416.9.1.105
- Shen, Y., van Beek, T. A., Claassen, F. W., Zuilhof, H., Chen, B., & Nielen,
 M. W. F. (2012). Rapid control of Chinese star anise fruits and teas for neurotoxic anisatin by Direct Analysis in Real Time high resolution mass spectrometry. *Journal of Chromatography A*, 1259, 179–186. doi:10.1016/j.chroma.2012.03.058
- Shergis, J. L., Zhang, A. L., Zhou, W., & Xue, C. C. (2013). Panax ginseng in Randomised Controlled Trials: A Systematic Review. Phytotherapy Research, 27, 949–965. doi:10.1002/ptr.4832
- Shim, W.-B., Kim, K., Ofori, J. A., Chung, Y.-C., & Chung, D.-H. (2012).
 Occurrence of Aflatoxins in Herbal Medicine Distributed in South Korea. *Journal of Food Protection*, 75, 1991–1999.
 doi:10.4315/0362-028X.JFP-12-190

Shimadzu. (n.d.). Principle of Optical Emission Spectrometry. Retrieved from

https://www.shimadzu.com/an/products/elemental-analysis/oes-app/princi ple-of-optical-emission-spectrometry/index.html (accessed 3 December 2020).

- Shin, J., Song, M.-H., Oh, J.-W., Keum, Y.-S., & Saini, R. K. (2020).
 Pro-oxidant Actions of Carotenoids in Triggering Apoptosis of Cancer
 Cells: A Review of Emerging Evidence. *Antioxidants*, 9, 532.
 doi:10.3390/antiox9060532
- Shin, S. J., Nam, Y., Park, Y. H., Kim, M.-J., Lee, E., Jeon, S. G., Bae, B.-S.,
 Seo, J., Shim, S.-L., Kim, J.-S., Han, C.-K., Kim, S., Lee, Y. Y., & Moon,
 M. (2021). Therapeutic effects of non-saponin fraction with rich
 polysaccharide from Korean red ginseng on aging and Alzheimer's
 disease. *Free Radical Biology and Medicine*, *164*, 233–248.
 doi:10.1016/j.freeradbiomed.2020.12.454
- Shkembi, X., Svobodova, M., Skouridou, V., Bashammakh, A. S., Alyoubi,
 A. O., & O'Sullivan, C. K. (2022). Aptasensors for mycotoxin detection: A review. *Analytical Biochemistry*, 644, 114156. doi:10.1016/j.ab.2021.114156
- Siddeeg, A., AlKehayez, N. M., Abu-Hiamed, H. A., Al-Sanea, E. A., & AL-Farga, A. M. (2021). Mode of action and determination of antioxidant

activity in the dietary sources: An overview. Saudi Journal of Biological Sciences, 28, 1633–1644. doi:10.1016/j.sjbs.2020.11.064

- Siekevitz, P. (1961). Use of Radioactive Isotopes in Studying the Metabolism of Mitochondria. In Radioactive Isotopes in Physiology Diagnostics and Therapy / Künstliche Radioaktive Isotope in Physiologie Diagnostik und Therapie (pp. 259–267). Springer Berlin Heidelberg. doi:10.1007/978-3-642-49762-9 24
- Sil, R., Ray, D., & Chakraborti, A. S. (2015). Glycyrrhizin ameliorates metabolic syndrome-induced liver damage in experimental rat model.
 Molecular and Cellular Biochemistry, 409, 177–189. doi:10.1007/s11010-015-2523-y
- Sil, R., & Chakraborti, A. S. (2016). Oxidative Inactivation of Liver Mitochondria in High Fructose Diet-Induced Metabolic Syndrome in Rats: Effect of Glycyrrhizin Treatment. *Phytotherapy Research*, *30*, 1503–1512. doi:10.1002/ptr.5654
- Silveira, D., Prieto-Garcia, J. M., Boylan, F., Estrada, O., Fonseca-Bazzo, Y.
 M., Jamal, C. M., Magalhães, P. O., Pereira, E. O., Tomczyk, M., &
 Heinrich, M. (2020). COVID-19: Is There Evidence for the Use of Herbal
 Medicines as Adjuvant Symptomatic Therapy? *Frontiers in Pharmacology*, *11*. doi:10.3389/fphar.2020.581840

- Simmler, C., Pauli, G. F., & Chen, S.-N. (2013). Phytochemistry and biological properties of glabridin. *Fitoterapia*, 90, 160–184. doi:10.1016/j.fitote.2013.07.003
- Simoes-Pires, C., Hostettmann, K., Haouala, A., Cuendet, M., Falquet, J., Graz, B., & Christen, P. (2014). Reverse pharmacology for developing an anti-malarial phytomedicine. The example of Argemone mexicana. *International Journal for Parasitology: Drugs and Drug Resistance*, 4(3), 338–346. doi:10.1016/j.ijpddr.2014.07.001
- Singh, A., Kukreti, R., Saso, L., & Kukreti, S. (2019). Oxidative Stress: A Key Modulator in Neurodegenerative Diseases. *Molecules*, *24*, 1583. doi:10.3390/molecules24081583
- Singhuber, J., Zhu, M., Prinz, S., & Kopp, B. (2009). Aconitum in Traditional Chinese Medicine—A valuable drug or an unpredictable risk? *Journal of Ethnopharmacology*, *126*, 18–30. doi:10.1016/j.jep.2009.07.031
- Sionneau, P. (1995). Introduction to the Use of Processed Chinese Medicinals to Enhance Therapeutic Effects (trans. B. Flaws). Boulder, CO: Blue Poppy Press.
- Skonieczna, M., Hejmo, T., Poterala-Hejmo, A., Cieslar-Pobuda, A., & Buldak, R. J. (2017). NADPH Oxidases: Insights into Selected Functions and Mechanisms of Action in Cancer and Stem Cells. Oxidative Medicine and Cellular Longevity, 2017, 1–15. doi:10.1155/2017/9420539

- Slimen, I. B., Najar, T., Ghram, A., Dabbebi, H., Mrad, M. B., & Abdrabbah,
 M. (2014). Reactive oxygen species, heat stress and oxidative-induced mitochondrial damage. A review. *International Journal of Hyperthermia*, *30*, 513–523. doi:10.3109/02656736.2014.971446
- Smithers, R. (2012). Liquorice to grow again in Pontefract. *The Guardian*,
 July 30, 2012. Retrieved from
 https://www.theguardian.com/uk/the-northerner/2012/jul/30/liquorice-pont
 efract (accessed 18 September 2023).
- Soderberg, T. (2023). *Infrared spectroscopy*. Retrieved from https://chem.libretexts.org/Courses/SUNY_Oneonta/Chem_221%3A_Org anic_Chemistry_I_%28Bennett%29/1%3ALecture_Textbook/04%3A_Stru cture_Determination_I-_UV-Vis_and_Infrared_Spectroscopy_Mass_Spect rometry/4.04%3A__Infrared_spectroscopy (accessed 19 May 2023).
- Somjen, D., Katzburg, S., Vaya, J., Kaye, A. M., Hendel, D., Posner, G. H.,
 & Tamir, S. (2004). Estrogenic activity of glabridin and glabrene from licorice roots on human osteoblasts and prepubertal rat skeletal tissues. *The Journal of Steroid Biochemistry and Molecular Biology*, *91*, 241–246. doi:10.1016/j.jsbmb.2004.04.008
- Somjen, D., Knoll, E., Vaya, J., Stern, N., & Tamir, S. (2004). Estrogen-like activity of licorice root constituents: glabridin and glabrene, in vascular tissues in vitro and in vivo. *The Journal of Steroid Biochemistry and Molecular Biology*, 91(3), 147–155. doi:10.1016/j.jsbmb.2004.04.003

- Somssich, M. (2022). A Short History of Plant Light Microscopy. *Current Protocols*, 2. doi:10.1002/cpz1.577
- Song, Z. H., Ji, Z. N., Lo, C. K., Dong, T. T., Zhao, K. J., LI, O. T., Haines, C. J., Kung, S. D., & Tsim, K. W. (2004). Chemical and Biological Assessment of a Traditional Chinese Herbal Decoction Prepared from Radix Astragali and Radix Angelicae Sinensis: Orthogonal Array Design to Optimize the Extraction of Chemical Constituents. *Planta Medica*, *70*, 1222–1227. doi:10.1055/s-2004-835855
- Song, M. Y., Kang, S. Y., Kang, A., Hwang, J. H., Park, Y.-K., & Jung, H. W.
 (2017). *Cinnamomum cassia* Prevents High-Fat Diet-Induced Obesity in Mice through the Increase of Muscle Energy. *The American Journal of Chinese Medicine*, *45*, 1017–1031. doi:10.1142/S0192415X17500549
- Speciale, A., Muscarà, C., Molonia, M. S., Cristani, M., Cimino, F., & Saija,
 A. (2022). Recent Advances in Glycyrrhetinic Acid-Functionalized
 Biomaterials for Liver Cancer-Targeting Therapy. *Molecules*, *27*, 1775.
 doi:10.3390/molecules27061775
- Spence, C. (2023). Ginger: The pungent spice. International Journal of Gastronomy and Food Science, 33, 100793.
 doi:10.1016/j.ijgfs.2023.100793
- Spinelli, J. B., Rosen, P. C., Sprenger, H.-G., Puszynska, A. M., Mann, J. L.,Roessler, J. M., Cangelosi, A. L., Henne, A., Condon, K. J., Zhang, T.,Kunchok, T., Lewis, C. A., Chandel, N. S., & Sabatini, D. M. (2021).

Fumarate is a terminal electron acceptor in the mammalian electron transport chain. *Science*, *374*, 1227–1237. doi:10.1126/science.abi7495

- Srinivasan, K., Pandey, A. K., Livingston, A., & Venkatesh, S. (2021). Roles of host mitochondria in the development of COVID-19 pathology: Could mitochondria be a potential therapeutic target? *Molecular Biomedicine*, *2*, 38. doi:10.1186/s43556-021-00060-1
- Srirama, R., Santhosh Kumar, J. U., Seethapathy, G. S., Newmaster, S. G., Ragupathy, S., Ganeshaiah, K. N., Uma Shaanker, R., & Ravikanth, G. (2017). Species Adulteration in the Herbal Trade: Causes, Consequences and Mitigation. *Drug Safety*, 40, 651–661. doi:10.1007/s40264-017-0527-0
- Stafford, G. I., Jäger, A. K., & van Staden, J. (2005). Effect of storage on the chemical composition and biological activity of several popular South African medicinal plants. *Journal of Ethnopharmacology*, *97*, 107–115. doi:10.1016/j.jep.2004.10.021
- Stanley-Baker, M. (2022). 'Qi 氣: A means for cohering natural knowledge'
 in V. Lo, D. Yang, & M. Stanley-Baker (eds.), *Routledge Handbook of Chinese Medicine*. London: Routledge. pp.23-50.
 doi:10.4324/9780203740262-4
- Stoeckle, M. Y., Gamble, C. C., Kirpekar, R., Young, G., Ahmed, S., & Little, D. P. (2011). Commercial Teas Highlight Plant DNA Barcode Identification

Successes and Obstacles. *Scientific Reports*, *1*, 42. doi:10.1038/srep00042

- Stringer, W. W., & Rossiter, H. B. (2017). Hormesis, mithridatism and Paracelsus: A little oxidative stress goes a long way. *Hypertension Research*, 40, 29–30. doi:10.1038/hr.2016.104
- Su, X., Wu, L., Hu, M., Dong, W., Xu, M., & Zhang, P. (2017). Glycyrrhizic acid: A promising carrier material for anticancer therapy. *Biomedicine & Pharmacotherapy*, *95*, 670–678. doi:10.1016/j.biopha.2017.08.123
- Su, X., Li, T., Liu, Z., Huang, Q., Liao, K., Ren, R., Lu, L., Qi, X., Wang, M., Chen, J., Zhou, H., Leung, E. L.-H., Pan, H., Liu, J., Wang, H., Huang, L., & Liu, L. (2018). Licochalcone A activates Keap1-Nrf2 signaling to suppress arthritis via phosphorylation of p62 at serine 349. *Free Radical Biology and Medicine*, *115*, 471–483. doi:10.1016/j.freeradbiomed.2017.12.004
- Sun, Y.-X., Tang, Y., Wu, A.-L., Liu, T., Dai, X.-L., Zheng, Q.-S., & Wang, Z.-B. (2010). Neuroprotective effect of liquiritin against focal cerebral ischemia/reperfusion in mice via its antioxidant and antiapoptosis properties. *Journal of Asian Natural Products Research*, *12*, 1051–1060. doi:10.1080/10286020.2010.535520
- Sun, H., Calabrese, E. J., Lin, Z., Lian, B., & Zhang, X. (2020). Similarities between the Yin/Yang Doctrine and Hormesis in Toxicology and

Pharmacology. *Trends in Pharmacological Sciences*, *41*, 544–556. doi:10.1016/j.tips.2020.05.004

- Sun, Q., Li, Y., Shi, L., Hussain, R., Mehmood, K., Tang, Z., & Zhang, H. (2022). Heavy metals induced mitochondrial dysfunction in animals:
 Molecular mechanism of toxicity. *Toxicology*, *469*, 153136. doi:10.1016/j.tox.2022.153136
- Sun Ten (2015). *Manufacturing*. Retrieved from https://www.suntenglobal.com/_en/Quality-Manufacturing.php (accessed 10 March 2023).
- Sur, S., Nakanishi, H., Flaveny, C., Ippolito, J. E., McHowat, J., Ford, D. A.,
 & Ray, R. B. (2019). Inhibition of the key metabolic pathways, glycolysis and lipogenesis, of oral cancer by bitter melon extract. *Cell Communication and Signaling*, *17*, 131. doi:10.1186/s12964-019-0447-y
- Susruta. (2010). Susruta-Samhita; Text with English translation, Dalhana's commentary alongwith critical notes with 5 appendices and index (2nd edition; P. V. Sharma, ed.). Varanasi: Chaukhambha Visvabharati.
- Sváb, G., Kokas, M., Sipos, I., Ambrus, A., & Tretter, L. (2021). Methylene Blue Bridges the Inhibition and Produces Unusual Respiratory Changes in Complex III-Inhibited Mitochondria. Studies on Rats, Mice and Guinea Pigs. *Antioxidants*, 10(2), 305. doi:10.3390/antiox10020305

- Takada, S., Fumoto, Y., & Kinugawa, S. (2022). Ergogenic effects of caffeine are mediated by myokines. *Frontiers in Sports and Active Living*, *4*. doi:10.3389/fspor.2022.969623
- Tamir, S., Eizenberg, M., Somjen, D., Izrael, S., & Vaya, J. (2001).
 Estrogen-like activity of glabrene and other constituents isolated from licorice root. *The Journal of Steroid Biochemistry and Molecular Biology*, 78, 291–298. doi:10.1016/S0960-0760(01)00093-0
- Tang, Y., Ou, S., Ye, L., & Wang, S. (2022). Pharmacological Activities and Pharmacokinetics of Glycycoumarin. *Revista Brasileira de Farmacognosia*, 33, 471–483. doi:10.1007/s43450-022-00342-x
- Tao, Y., Zhu, F., Pan, M., Liu, Q., & Wang, P. (2022). Pharmacokinetic, Metabolism, and Metabolomic Strategies Provide Deep Insight Into the Underlying Mechanism of Ginkgo biloba Flavonoids in the Treatment of Cardiovascular Disease. *Frontiers in Nutrition*, 9. doi:10.3389/fnut.2022.857370
- Taylor, T. (2015). *Important Aspects of UV Detection for HPLC*. Retrieved from:

https://www.chromatographyonline.com/view/important-aspects-uv-detecti on-hplc (accessed 19th April 2024)

Taylor, R. S., Manandhar, N. P., & Towers, G. H. N. (1995). Screening of selected medicinal plants of Nepal for antimicrobial activities. *Journal of Ethnopharmacology*, 46, 153–159. doi:10.1016/0378-8741(95)01242-6

- Teng, L., Meng, Q., Lu, J., Xie, J., Wang, Z., Liu, Y., & Di, W. (2014). Liquiritin modulates ERK- and AKT/GSK-3β-dependent pathways to protect against glutamate-induced cell damage in differentiated PC12 cells. *Molecular Medicine Reports*, 10, 818–824. doi:10.3892/mmr.2014.2289
- TeSlaa, T., & Teitell, M. A. (2014). *Techniques to Monitor Glycolysis*. doi:10.1016/B978-0-12-416618-9.00005-4
- Tetzlaff-Deas, B. (2023). Chemical found in popular sweet suppresses "silent killer" pancreatic cancer. *Mirror*, August 18th, 2023. Retrieved from https://www.mirror.co.uk/news/health/chemical-found-popular-sweet-foun d-30676265 (accessed 10 August 2023).
- Tewari, S. N., & Trembalowicz, F. C. (1968). Some experience with deglycyrrhizinated liquorice in the treatment of gastric and duodenal ulcers with special reference to its spasmolytic effect. *Gut*, *9*, 48–51. doi:10.1136/gut.9.1.48
- ThermoFisher Scientific. (n.d.). Protein Assays. Retrieved from https://www.thermofisher.com/uk/en/home/life-science/protein-biology/prot ein-assays-analysis/protein-assays.html (accessed 11 January 2022).
- Thomford, N., Senthebane, D., Rowe, A., Munro, D., Seele, P., Maroyi, A., & Dzobo, K. (2018). Natural Products for Drug Discovery in the 21st
 Century: Innovations for Novel Drug Discovery. *International Journal of Molecular Sciences*, *19*, 1578. doi:10.3390/ijms19061578

- Thorndike, L. (1922). Galen: The Man and His Times. *The Scientific Monthly*, *14*, 83–93. Available from http://www.jstor.org/stable/6569 (accessed 24 September 2021).
- Tian, J. J., Levy, M., Zhang, X., Sinnott, R., & Maddela, R. (2022).
 Counteracting health risks by Modulating Homeostatic Signaling. *Pharmacological Research*, *182*, 106281. doi:10.1016/j.phrs.2022.106281
- Tian, Y., Tian, W., Li, T., & Xu, J. (2021). Commentary: The Effects of Different Fluorescent Indicators in Observing the Changes of The Mitochondrial Membrane Potential During Oxidative Stress-Induced Mitochondrial Injury of Cardiac H9c2 Cells. *Journal of Cardiology and Cardiovascular Sciences*, *5*, 1–3. Retrieved from https://www.cardiologyresearchjournal.com/articles/commentary-the-effec ts-of-different-fluorescent-indicators-in-observing-the-changes-of-the-mito chondrial-membrane-potential-during-oxidative-stress-induced-mitochond rial-injury-of-cardiac-h9c2-cells.html (accessed 30 August 2023).
- Tian, M., Yan, H., & Row, K. (2008). Extraction of Glycyrrhizic Acid and Glabridin from Licorice. *International Journal of Molecular Sciences*, *9*, 571–577. doi:10.3390/ijms9040571
- Ting, A., Chow, Y., & Tan, W. (2013). Microbial and heavy metal contamination in commonly consumed traditional Chinese herbal medicines. *Journal of Traditional Chinese Medicine*, 33, 119–124. doi:10.1016/S0254-6272(13)60112-0
- Tohma, H. S., & Gulçin, I. (2010). Antioxidant and Radical Scavenging Activity of Aerial Parts and Roots of Turkish Liquorice (*Glycyrrhiza Glabra* L.). *International Journal of Food Properties*, *13*, 657–671. doi:10.1080/10942911003773916
- Trabelsi, K., Ammar, A., Boujelbane, M. A., Puce, L., Garbarino, S., Scoditti, E., Boukhris, O., Khanfir, S., Clark, C. C. T., Glenn, J. M., Alhaj, O. A., Jahrami, H., Chtourou, H., & Bragazzi, N. L. (2022). Religious fasting and its impacts on individual, public, and planetary health: Fasting as a "religious health asset" for a healthier, more equitable, and sustainable society. *Frontiers in Nutrition*, 9. doi:10.3389/fnut.2022.1036496
- Trebak, M., & Kinet, J.-P. (2019). Calcium signalling in T cells. *Nature Reviews Immunology*, *19*, 154–169. doi:10.1038/s41577-018-0110-7
- Tschanz, D. W. (2003). A Short History of Islamic Pharmacy. Journal for the International Society for the History of Islamic Medicine, 1, 11–17. Retrieved from https://www.ishim.net/ishimj/3/03.pdf (accessed 21 September 2021).
- Tso, L.-D. (2009). Modernizing a Traditional Path to Healing. *Taiwan Today*. Retrieved from https://taiwantoday.tw/news.php?post=22233&unit=12,29,33,45 (accessed 30 August 2023).

- Tu, Y. (2016). Artemisinin—A Gift from Traditional Chinese Medicine to the
 World (Nobel Lecture). Angewandte Chemie International Edition, 55,
 10210–10226. doi:10.1002/anie.201601967
- Turner, D. (2020). Gas Chromatography How a Gas Chromatography Machine Works, How To Read a Chromatograph and GCxGC. Retrieved from

https://www.technologynetworks.com/analysis/articles/gas-chromatograph y-how-a-gas-chromatography-machine-works-how-to-read-a-chromatogra ph-and-gcxgc-335168 (accessed 20 November 2020).

- U.S. Food and Drug Administration. (2024). Investigation of Elevated Lead
 & Chromium Levels: Cinnamon Applesauce Pouches. Retrieved from https://www.fda.gov/food/outbreaks-foodborne-illness/investigation-elevat
 ed-lead-chromium-levels-cinnamon-applesauce-pouches-november-2023
 (accessed 14 June 2024).
- Ugoeze, K. C., Amadi, N., Okoronkwo, N. A., Abali, S. O., Oluigbo, K. E., & Chinko, B. C. (2023). GC-FID guided Identification and Quantification of detectable Phytochemicals in selected Commercial Chamomile Herbal Tea. *International Journal of Applied Biology and Pharmaceutical Technology*, *14*(01). doi:10.26502/ijabpt.202120
- Upadhyay, S., Mantha, A. K., & Dhiman, M. (2020). Glycyrrhiza glabra (Licorice) root extract attenuates doxorubicin-induced cardiotoxicity via alleviating oxidative stress and stabilising the cardiac health in H9c2

cardiomyocytes. *Journal of Ethnopharmacology*, 258, 112690. doi:10.1016/j.jep.2020.112690

- van Arsdall, A. (2002). *Medieval Herbal Remedies [The Old English Herbarium and Anglo-Saxon Medicine]*. New York, NY: Routledge.
- van der Hoff, G. R., & van Zoonen, P. (1999). Trace analysis of pesticides by gas chromatography. *Journal of Chromatography A*, 843(1–2), 301–322. doi:10.1016/S0021-9673(99)00511-7
- van Dinteren, S., Meijerink, J., Witkamp, R., van Ieperen, B., Vincken, J.-P.,
 & Araya-Cloutier, C. (2022). Valorisation of liquorice (*Glycyrrhiza*) roots:
 antimicrobial activity and cytotoxicity of prenylated (iso)flavonoids and
 chalcones from liquorice spent (*G. glabra*, *G. inflata*, and *G. uralensis*). *Food & Function*, *13*(23), 12105–12120. doi:10.1039/D2FO02197H
- van Gelderen, C. E. M., Bijlsma, J. A., van Dokkum, W., & Savelkoull, T. J.
 F. (2000). Glycyrrhizic acid: the assessment of a no effect level. *Human & Experimental Toxicology*, 19(8), 434–439.
 doi:10.1191/096032700682694251
- van Steenwijk, H. P., Bast, A., & de Boer, A. (2021). Immunomodulating Effects of Fungal Beta-Glucans: From Traditional Use to Medicine. *Nutrients*, 13(4), 1333. doi:0.3390/nu13041333

- Vasconcelos, N. G., Croda, J., & Simionatto, S. (2018). Antibacterial mechanisms of cinnamon and its constituents: A review. *Microbial Pathogenesis*, *120*, 198–203. doi:10.1016/j.micpath.2018.04.036
- Venton, B. J. (2020). Ultraviolet-Visible (UV-Vis) Spectroscopy. Retrieved from https://www.jove.com/v/10204/ultraviolet-visible-uv-vis-spectroscopy (accessed 15 November 2020).
- Vergara, R. C., Jaramillo-Riveri, S., Luarte, A., Moënne-Loccoz, C., Fuentes, R., Couve, A., & Maldonado, P. E. (2019). The Energy Homeostasis Principle: Neuronal Energy Regulation Drives Local Network Dynamics Generating Behavior. *Frontiers in Computational Neuroscience*, *13*. doi:10.3389/fncom.2019.00049
- Verichek Tech Services. (2016). Spectroscopy vs. Spectrometry What's the Difference in OES? Retrieved from https://verichek.net/spectroscopy-vs-spectrometry.html (accessed 18 May 2023).
- Viegas, C., Simões, A. B., Faria, M., Gomes, B., Cervantes, R., Dias, M., Carolino, E., Twaruzek, M., Kosicki, R., Viegas, S., & Caetano, L. A. (2023). Tea contamination by mycotoxins and azole-resistant mycobiota The need of a One Health approach to tackle exposures. *International Journal of Food Microbiology*, 385, 110015. doi:10.1016/j.ijfoodmicro.2022.110015

- Viegas, C., Simões, A. B., Faria, M., Gomes, B., Cervantes, R., Dias, M., Carolino, E., Twaruzek, M., Kosicki, R., Viegas, S., & Caetano, L. A. (2023). Tea contamination by mycotoxins and azole-resistant mycobiota The need of a One Health approach to tackle exposures. *International Journal of Food Microbiology*, 385, 110015. doi:10.1016/j.ijfoodmicro.2022.110015
- Visioli, F., Mucignat-Caretta, C., Anile, F., & Panaite, S.-A. (2022). Traditional and Medical Applications of Fasting. *Nutrients*, *14*, 433. doi:10.3390/nu14030433
- Viraragavan, A., Hlengwa, N., de Beer, D., Riedel, S., Miller, N., Bowles, S., Walczak, B., Muller, C., & Joubert, E. (2020). Model development for predicting *in vitro* bio-capacity of green rooibos extract based on composition for application as screening tool in quality control. *Food & Function*, *11*(4), 3084–3094. doi:10.1039/C9FO02480H
- Volarevic, V., Djokovic, B., Jankovic, M. G., Harrell, C. R., Fellabaum, C., Djonov, V., & Arsenijevic, N. (2019). Molecular mechanisms of cisplatin-induced nephrotoxicity: a balance on the knife edge between renoprotection and tumor toxicity. *Journal of Biomedical Science*, *26*, 25. doi:10.1186/s12929-019-0518-9
- Wahab, S., Annadurai, S., Abullais, S. S., Das, G., Ahmad, W., Ahmad, M.F., Kandasamy, G., Vasudevan, R., Ali, M. S., & Amir, M. (2021).Glycyrrhiza glabra (Licorice): A Comprehensive Review on Its

Phytochemistry, Biological Activities, Clinical Evidence and Toxicology. *Plants*, *10*, 2751. doi:10.3390/plants10122751

- Walkowiak-Bródka, A., Piekuś-Słomka, N., Wnuk, K., & Kupcewicz, B. (2022). Analysis of White Mulberry Leaves and Dietary Supplements, ATR-FTIR Combined with Chemometrics for the Rapid Determination of 1-Deoxynojirimycin. *Nutrients*, *14*(24), 5276. doi:10.3390/nu14245276
- Wallace, D. C. (2008). Mitochondria as Chi. *Genetics*, *179*, 727–735. doi:10.1534/genetics.104.91769
- Wan, S., Xie, X., Yang, G., & Feng, F. (2024). Discovery of the toxicity-related quality markers and mechanisms of Zhi-Zi-Hou-Po decoction based on Chinmedomics combined with differentially absorbed components and network pharmacology. *Journal of Ethnopharmacology*, *320*, 117408. doi:10.1016/j.jep.2023.117408
- Wan, X., Wu, W., Zang, Z., Li, K., Naeem, A., Zhu, Y., Chen, L., Zhong, L., Zhu, W., & Guan, Y. (2022). Investigation of the potential curative effects of Gui-Zhi-Jia-Ge-Gen decoction on wind-cold type of common cold using multidimensional analysis. *Journal of Ethnopharmacology*, 298, 115662. doi:10.1016/j.jep.2022.115662
- Wang, X., Dong, Y.-T., Hu, X.-M., Zhang, J.-Z., Shi, N.-R., Zuo, Y.-Q., &
 Wang, X. (2023). The circadian regulation of extracellular ATP. *Purinergic Signalling*, 19(1), 283–295. doi:10.1007/s11302-022-09881-3

- Wang, L., Li, Y.-J., Yang, X., Yang, B., Zhang, X., Zhang, J., Zhang, Q.,
 Cheng, X.-D., Wang, J.-H., & Yu, N.-W. (2023). Purinergic signaling: a potential therapeutic target for ischemic stroke. *Purinergic Signalling*, 19(1), 173–183. doi:10.1007/s11302-022-09905-y
- Wang, M., Liu, M., Tang, L., Shen, L., Xiao, J., & Li, R. (2023). Liquiritin reduces ferroptosis in doxorubicin-induced cardiotoxicity through targeting SLC7A11/GPX4 pathway. *Naunyn-Schmiedeberg's Archives of Pharmacology*. doi:10.1007/s00210-023-02515-4
- Wang, Y., Xi, X., Wang, L., & Chen, Y. (2023). HPTLC-Bioluminescent Bioautography Screening of Herbal Teas for Adulteration with Hypolipidemic Drugs. *Biosensors*, *13*(3), 392. doi:10.3390/bios13030392
- Wang, M.-J., Yang, B.-R., Jing, X.-Y., Wang, Y.-Z., Kang, L., Ren, K., & Kang, L. (2023). P2Y1R and P2Y2R: potential molecular triggers in muscle regeneration. *Purinergic Signalling*, 19(1), 305–313. doi:10.1007/s11302-022-09885-z
- Wang, H., Yang, Z., Zhang, X., Xie, J., Xie, Y., Gokulnath, P., Vulugundam,
 G., & Xiao, J. (2023). Antioxidants Supplementation During Exercise:
 Friends or Enemies for Cardiovascular Homeostasis? *Journal of Cardiovascular Translational Research*, 16(1), 51–62.
 doi:10.1007/s12265-022-10297-y
- Wang, L.-S., Yen, P.-T., Weng, S.-F., Hsu, J.-H., & Yeh, J.-L. (2022). Clinical Patterns of Traditional Chinese Medicine for Ischemic Heart Disease

Treatment: A Population-Based Cohort Study. *Medicina*, *58*, 879. doi:10.3390/medicina58070879

- Wang, K.-L., Yu, Y.-C., Chen, H.-Y., Chiang, Y.-F., Ali, M., Shieh, T.-M., & S.-M. (2022). Recent Advances Hsia. in Glycyrrhiza glabra (Licorice)-Containing Herbs Alleviating Radiotherapyand Chemotherapy-Induced Adverse Reactions in Cancer Treatment. Metabolites, 12(6), 535. doi:10.3390/metabo12060535
- Wang, H., Song, W., Tao, W., Zhang, J., Zhang, X., Zhao, J., Yong, J., Gao,
 X., & Guo, L. (2021). Identification wild and cultivated licorice by
 multidimensional analysis. *Food Chemistry*, 339, 128111.
 doi:10.1016/j.foodchem.2020.128111
- Wang, Z., Zhang, F., Liu, W., Sheng, N., Sun, H., & Zhang, J. (2021).
 Impaired tricarboxylic acid cycle flux and mitochondrial aerobic respiration during isoproterenol induced myocardial ischemia is rescued by bilobalide. *Journal of Pharmaceutical Analysis*, *11*, 764–775. doi:10.1016/j.jpha.2020.08.008
- Wang, W., Long, H., Huang, W., Zhang, T., Xie, L., Chen, C., Liu, J., Xiong,
 D., & Hu, W. (2020). Bu-Shen-Huo-Xue Decoction Ameliorates Diabetic
 Nephropathy by Inhibiting Rac1/PAK1/p38MAPK Signaling Pathway in
 High-Fat Diet/Streptozotocin-Induced Diabetic Mice. *Frontiers in Pharmacology*, *11*. doi:10.3389/fphar.2020.587663

- Wang, J., Liao, A.-M., Thakur, K., Zhang, J.-G., Huang, J.-H., & Wei, Z.-J. (2019). Licochalcone B Extracted from Glycyrrhiza uralensis Fisch Induces Apoptotic Effects in Human Hepatoma Cell HepG2. *Journal of Agricultural and Food Chemistry*, 67(12), 3341–3353. doi:10.1021/acs.jafc.9b00324
- Wang, W., Wang, M., Ruan, Y., Tan, J., Wang, H., Yang, T., Li, J., & Zhou,
 Q. (2019). Ginkgolic Acids Impair Mitochondrial Function by Decreasing
 Mitochondrial Biogenesis and Promoting FUNDC1-Dependent Mitophagy. *Journal of Agricultural and Food Chemistry*, 67(36), 10097–10106.
 doi:10.1021/acs.jafc.9b04178
- Wang, D., Calabrese, E. J., Lian, B., Lin, Z., & Calabrese, V. (2018).
 Hormesis as a mechanistic approach to understanding herbal treatments in traditional Chinese medicine. *Pharmacology & Therapeutics*, *184*, 42–50. doi:10.1016/j.pharmthera.2017.10.013
- Wang, L., Yan, T., Yang, L., Zhang, K., & Jia, J. (2017). Monitoring the Consistency Quality and Antioxidant Activity of Da Hong Pao Teas by HPLC Fingerprinting. *Journal of Chromatographic Science*, *55*, 528–535. doi:10.1093/chromsci/bmx005
- Wang, X., Liu, Y., Wang, L., Han, J., & Chen, S. (2016). A Nucleotide
 Signature for the Identification of Angelicae Sinensis Radix (Danggui) and
 Its Products. *Scientific Reports*, *6*, 34940. doi:10.1038/srep34940

- Wang, D., Guo, T. Q., Wang, Z. Y., Lu, J. H., Liu, D. P., Meng, Q. F., Xie, J.,
 Zhang, X. L., Liu, Y., & Teng, L. S. (2014). ERKs and mitochondria-related
 pathways are essential for glycyrrhizic acid-mediated neuroprotection
 against glutamate-induced toxicity in differentiated PC12 cells. *Brazilian Journal of Medical and Biological Research*, 47(9), 773–779.
 doi:10.1590/1414-431X20143760
- Wang, D., Wong, H.-K., Feng, Y.-B., & Zhang, Z.-J. (2014).
 18beta-Glycyrrhetinic acid induces apoptosis in pituitary adenoma cells via ROS/MAPKs-mediated pathway. *Journal of Neuro-Oncology*, *116*, 221–230. doi:10.1007/s11060-013-1292-2
- Wang, X., Zhang, H., Chen, L., Shan, L., Fan, G., & Gao, X. (2013). Liquorice, a unique "guide drug" of traditional Chinese medicine: A review of its role in drug interactions. *Journal of Ethnopharmacology*, *150*, 781–790. doi:10.1016/j.jep.2013.09.055
- Wang, F., & Roh, Y. S. (2020). Mitochondrial connection to ginsenosides.
 Archives of Pharmacal Research, 43, 1031–1045.
 doi:10.1007/s12272-020-01279-2
- Wang, J., Sasse, A., & Sheridan, H. (2019). *Traditional Chinese Medicine: From Aqueous Extracts to Therapeutic Formulae*.
 doi:10.5772/intechopen.85733

- Wang, D., & Zhou, J. (2023). Purinergic receptor: a crucial regulator of adipose tissue functions. *Purinergic Signalling*, 19, 273–281. doi:10.1007/s11302-022-09907-w
- Warburg, O. (1956). On the Origin of Cancer Cells. *Science*, *123*, 309–314. doi:10.1126/science.123.3191.309
- Webb, R., Hughes, M., Thomas, A., & Morris, K. (2017). The Ability of
 Exercise-Associated Oxidative Stress to Trigger Redox-Sensitive
 Signalling Responses. *Antioxidants*, *6*, 63. doi:10.3390/antiox6030063
- Wegmüller, K. (2015). The Problem with Qi: Vitalism, Science and the Soul of Traditional Chinese Medicine. The Journal of Chinese Medicine, 108(June), 49–55. Retrieved from https://www.academia.edu/19708797/The_Problem_with_Qi_Vitalism_Sci ence_and_the_Soul_of_Traditional_Chinese_Medicine (accessed 10 November 2023).
- Wei, W., Gao, X., Zhao, L., Zhuang, J., Jiao, Y., & Xu, F. (2020). Liquiritin apioside attenuates laryngeal chemoreflex but not mechanoreflex in rat pups. *American Journal of Physiology-Lung Cellular and Molecular Physiology*, 318, L89–L97. doi:10.1152/ajplung.00306.2019
- Wen, Y., Li, W., Poteet, E. C., Xie, L., Tan, C., Yan, L.-J., Ju, X., Liu, R.,Qian, H., Marvin, M. A., Goldberg, M. S., She, H., Mao, Z., Simpkins, J.W., & Yang, S.-H. (2011). Alternative Mitochondrial Electron Transfer as a

Novel Strategy for Neuroprotection. *Journal of Biological Chemistry*, 286, 16504–16515. doi:10.1074/jbc.M110.208447

- Wiatrak, B., Kubis-Kubiak, A., Piwowar, A., & Barg, E. (2020). PC12 Cell
 Line: Cell Types, Coating of Culture Vessels, Differentiation and Other
 Culture Conditions. *Cells*, *9*, 958. doi:10.3390/cells9040958
- Wilde, M. (2020). How to Make Medicine from Mushrooms: Part 1. Wild Medicine.
 Medicine.
 Retrieved from https://wildmedicine.substack.com/p/how-to-make-medicine-from-mushro oms (accessed 6 January 2024).
- Wilms, S. (2017). *Shen Nong Ben Cao Jing: The Divine Farmer's Classic of Materia Medica* (3rd Edition). Corbett, OR: Happy Goat Productions.
- Wink, C. S., Rossowska, M. J., Jr., F. J., Yazdani, M., & Nakamoto, T. (1999). Effects of Caffeine on Heart Mitochondria in Newborn Rats. *Neonatology*, 76, 114–119. doi:10.1159/000014149
- Wong, H. S., Cheung, W. F., Tang, W. L., & Ko, K. M. (2012).
 "Qi-Invigorating" Chinese Tonic Herbs (<i>Shens</i>) Stimulate
 Mitochondrial ATP Generation Capacity in H9c2 Cardiomyocytes
 <i>in Situ</i> and Rat Hearts <i>ex Vivo</i> Chinese
 Medicine, 03, 101–105. doi:10.4236/cm.2012.32016
- Wood, S., Fryer, G., Tan, L. L. F., & Cleary, C. (2020). Dry cupping for musculoskeletal pain and range of motion: A systematic review and

meta-analysis. *Journal of Bodywork and Movement Therapies*, 24, 503–518. doi:10.1016/j.jbmt.2020.06.024

- Wood, D., Alsahaf, H., Streete, P., Dargan, P., & Jones, A. (2005). Fatality after deliberate ingestion of the pesticide rotenone: a case report. *Critical Care*, 9, R280. doi:10.1186/cc3528
- Woodland, D. W. (2003). *History and Development of Classification*. Retrieved from http://sytsma.botany.wisc.edu/fieldbotany/pdf/WoodlandChpt10.pdf (accessed 19 March 2023).
- Woodley, S. B. (2021). A brief history of herbal quality assessment: From Ancient Rome to 21st century mitochondria. *Herbal Reality*. Retrieved from

https://www.herbalreality.com/herbalism/safety/brief-history-herbal-qualityassessment-ancient-rome-21st-century-mitochondria/ (accessed 21 January 2021).

- Woodley, S. B., Mould, R. R., Sahuri-Arisoylu, M., Kalampouka, I., Booker,
 A., & Bell, J. D. (2021). Mitochondrial Function as a Potential Tool for
 Assessing Function, Quality and Adulteration in Medicinal Herbal Teas. *Frontiers in Pharmacology*, *12*. doi:10.3389/fphar.2021.660938
- Wright, W. (1888). The Empire of the Hittites. Journal of the Transactions of
 The Victoria Institute, or Philosophical Society of Great Britain, 21, 55–72.
 Retrieved from

https://babel.hathitrust.org/cgi/pt?id=hvd.32044090845264&seq=80 (accessed 29 December 2023).

- Wu, F., Li, X., Jiang, B., Yan, J., Zhang, Z., Qin, J., Yu, W., & Gao, Z. (2018).
 Glycyrrhetinic Acid Functionalized Nanoparticles for Drug Delivery to
 Liver Cancer. *Journal of Biomedical Nanotechnology*, 14(11), 1837–1852.
 doi:10.1166/jbn.2018.2638
- Wu, Y., Zhang, Z., Chen, T., Cheng, C., Zhang, Z., Zhou, H., & Luo, P. (2020). Comparison of two Polygonum chinense varieties used in Chinese cool tea in terms of chemical profiles and antioxidant/anti-inflammatory activities. *Food Chemistry*, *310*, 125840. doi:10.1016/j.foodchem.2019.125840
- Wu, X.-M., Zhang, N., Li, J.-S., Yang, Z.-H., Huang, X.-L., & Yang, X.-F.
 (2023). Purinergic receptors mediate endothelial dysfunction and participate in atherosclerosis. *Purinergic Signalling*, *19*, 265–272. doi:10.1007/s11302-021-09839-x
- Wu, C.-H., Chen, A.-Z., & Yen, G.-C. (2015). Protective Effects of Glycyrrhizic Acid and 18β-Glycyrrhetinic Acid against Cisplatin-Induced Nephrotoxicity in BALB/c Mice. *Journal of Agricultural and Food Chemistry*, 63, 1200–1209. doi:10.1021/jf505471a
- Wu, H.-Y., Yang, K.-M., & Chiang, P.-Y. (2018). Roselle Anthocyanins:
 Antioxidant Properties and Stability to Heat and pH. *Molecules*, 23, 1357.
 doi:10.3390/molecules23061357

- Xiao, L.-Y., Wang, X.-R., Ye, Y., Yang, J.-W., Cao, Y., Ma, S.-M., Li, T.-R., & Liu, C.-Z. (2018). Applications of Acupuncture Therapy in Modulating Plasticity of Central Nervous System. *Neuromodulation: Technology at the Neural Interface*, *21*, 762–776. doi:10.1111/ner.12724
- Xiao, W., & Loscalzo, J. (2020). Metabolic Responses to Reductive Stress.
 Antioxidants & Redox Signaling, 32, 1330–1347.
 doi:10.1089/ars.2019.7803
- Xie, H., Li, Y., Wang, J., Lei, Y., Koidis, A., Li, X., Shen, X., Xu, Z., & Lei, H.
 (2022). Broad-specific immunochromatography for simultaneous detection of various sulfonylureas in adulterated multi-herbal tea. *Food Chemistry*, 370, 131055. doi:10.1016/j.foodchem.2021.131055
- Xin, T., Li, R., Lou, Q., Lin, Y., Liao, H., Sun, W., Guan, M., Zhou, J., & Song, J. (2022). Application of DNA barcoding to the entire traditional Chinese medicine industrial chain: A case study of Rhei Radix et Rhizoma. *Phytomedicine*, *105*, 154375. doi:10.1016/j.phymed.2022.154375
- Xin, T., Li, X., Yao, H., Lin, Y., Ma, X., Cheng, R., Song, J., Ni, L., Fan, C., & Chen, S. (2015). Survey of commercial Rhodiola products revealed species diversity and potential safety issues. *Scientific Reports*, *5*, 8337. doi:10.1038/srep08337
- Xiong, C., Huang, C.-H., Wu, L., Xu, R., Xue, J.-P., Liu, Z.-G., & Sun, W. (2022). Identification of Andrographis Herba and its common products

using mini-barcode. *Chinese Journal of Natural Medicines*, 20(5), 393–400. doi:10.1016/S1875-5364(22)60157-2

- Xu, Z., Hu, Z., Xu, H., Zhang, L., Li, L., Wang, Y., Zhu, Y., Yang, L., & Hu, D. (2022). Liquiritigenin alleviates doxorubicin-induced chronic heart failure via promoting ARHGAP18 and suppressing RhoA/ROCK1 pathway. *Experimental Cell Research*, *411*, 113008. doi:10.1016/j.yexcr.2022.113008
- Xu, Y., Li, C., Chen, T., Li, X., Wu, X., Zhang, Q., & Zhao, L. (2022).
 Quantitative Analysis of the Multicomponent and Spectrum–Effect
 Correlation of the Antispasmodic Activity of Shaoyao-Gancao Decoction. *Journal of Analytical Methods in Chemistry*, 2022, 1–13.
 doi:10.1155/2022/2279404
- Xu, N., Li, M., Wang, P., Wang, S., & Shi, H. (2022). Spectrum-Effect Relationship Between Antioxidant and Anti-inflammatory Effects of Banxia Baizhu Tianma Decoction: An Identification Method of Active Substances With Endothelial Cell Protective Effect. *Frontiers in Pharmacology*, *13*. doi:10.3389/fphar.2022.823341
- Yadav, K., Gupta, T., & Aeri, V. (2022). High performance thin layer chromatography based chemo profiling of Ashvagandharishta and its antidepressant activity. *Journal of Chromatography B*, *1204*, 123334. doi:10.1016/j.jchromb.2022.123334

- Yamada, A. K., Pimentel, G. D., Pickering, C., Cordeiro, A. V., & Silva, V. R.
 R. (2022). Effect of caffeine on mitochondrial biogenesis in the skeletal muscle A narrative review. *Clinical Nutrition ESPEN*, *51*, 1–6. doi:10.1016/j.clnesp.2022.09.001
- Yan, T., Wang, H., Zhao, M., Yagai, T., Chai, Y., Krausz, K. W., Xie, C., Cheng, X., Zhang, J., Che, Y., Li, F., Wu, Y., Brocker, C. N., Gonzalez, F. J., Wang, G., & Hao, H. (2016). Glycyrrhizin Protects against Acetaminophen-Induced Acute Liver Injury via Alleviating Tumor Necrosis Factor -Mediated Apoptosis. *Drug Metabolism and Disposition*, *44*, 720–731. doi:10.1124/dmd.116.069419
- Yang, F., Zou, Y., Li, C., Li, J., Zang, Y., Peng, X., Wang, J., Liu, E.-H., Tong, S., & Chu, C. (2022). Discovery of potential hypoglycemic metabolites in Cassiae Semen by coupling UHPLC-QTOF-MS/MS combined plant metabolomics and spectrum-effect relationship analyses. *Food & Function*, *13*(19), 10291–10304. doi:10.1039/D2FO00562J
- Yang, L., Wang, D., Zhang, Z., Jiang, Y., & Liu, Y. (2022). Isoliquiritigenin alleviates diabetic symptoms via activating AMPK and inhibiting mTORC1 signaling in diet-induced diabetic mice. *Phytomedicine*, *98*, 153950. doi:10.1016/j.phymed.2022.153950
- Yang, X., Tian, X., Zhou, Y., Liu, Y., Li, X., Lu, T., Yu, C., & He, L. (2018). Evidence-Based Study to Compare Daodi Traditional Chinese Medicinal Material and Non-Daodi Traditional Chinese Medicinal Material.

Evidence-Based Complementary and Alternative Medicine, *2018*, 1–12. doi:10.1155/2018/6763130

- Yang, H., Villani, R. M., Wang, H., Simpson, M. J., Roberts, M. S., Tang, M., & Liang, X. (2018). The role of cellular reactive oxygen species in cancer chemotherapy. *Journal of Experimental & Clinical Cancer Research*, 37(1), 266. doi:10.1186/s13046-018-0909-x
- Yang, E.-J., Min, J. S., Ku, H.-Y., Choi, H.-S., Park, M., Kim, M. K., Song, K.-S., & Lee, D.-S. (2012). Isoliquiritigenin isolated from Glycyrrhiza uralensis protects neuronal cells against glutamate-induced mitochondrial dysfunction. *Biochemical and Biophysical Research Communications*, 421(4), 658–664. doi:10.1016/j.bbrc.2012.04.053
- Yang, J. C., Myung, S. C., Kim, W., & Lee, C. S. (2012). 18β-Glycyrrhetinic acid potentiates Hsp90 inhibition-induced apoptosis in human epithelial ovarian carcinoma cells via activation of death receptor and mitochondrial pathway. *Molecular and Cellular Biochemistry*, 370, 209–219. doi:10.1007/s11010-012-1412-x
- Yang, Y.-C., Wang, S.-W., Wu, I.-C., Chang, C.-C., Huang, Y.-L., Lee, O. K., Chang, J.-G., Chen, A., Kuo, F.-C., Wang, W.-M., & Wu, D.-C. (2009). A tumorigenic homeobox (HOX) gene expressing human gastric cell line derived from putative gastric stem cell. *European Journal of Gastroenterology* & *Hepatology*, 21, 1016–1023. doi:10.1097/MEG.0b013e32831d7b9c

- Yap, K. Y.-L., Chan, S. Y., & Lim, C. S. (2007). Authentication of traditional Chinese medicine using infrared spectroscopy: distinguishing between ginseng and its morphological fakes. *Journal of Biomedical Science*, *14*, 265–273. doi:10.1007/s11373-006-9133-3
- Yehuda, I., Madar, Z., Szuchman-Sapir, A., & Tamir, S. (2011). Glabridin, a phytoestrogen from licorice root, up-regulates manganese superoxide dismutase, catalase and paraoxonase 2 under glucose stress. *Phytotherapy Research*, 25, 659–667. doi:10.1002/ptr.3318
- Yi, L., Qi, L.-W., Li, P., Ma, Y.-H., Luo, Y.-J., & Li, H.-Y. (2007). Simultaneous determination of bioactive constituents in Danggui Buxue Tang for quality control by HPLC coupled with a diode array detector, an evaporative light scattering detector and mass spectrometry. *Analytical and Bioanalytical Chemistry*, 389, 571–580. doi:10.1007/s00216-007-1431-8
- Yogi Tea. (2023). Search for "liquorice." Retrieved from https://www.yogitea.com/en/search/?q=liquorice (accessed 30 November 2023).
- Yokozawa, T., Cho, E. J., Rhyu, D. Y., Shibahara, N., & Aoyagi, K. (2005).
 Glycyrrhizae Radix attenuates peroxynitrite-induced renal oxidative damage through inhibition of protein nitration. *Free Radical Research*, *39*, 203–211. doi:10.1080/10715760400027888
- You, H., Gershon, H., Goren, F., Xue, F., Kantowski, T., & Monheit, L. (2022). Analytical strategies to determine the labelling accuracy and

economically-motivated adulteration of "natural" dietary supplements in the marketplace: Turmeric case study. *Food Chemistry*, *370*, 131007. doi:10.1016/j.foodchem.2021.131007

- Yu, L., Wang, Y., Zhang, H., Li, M., Chen, G., Hao, J., & Xie, M. (2023).
 Involvement of purinergic P2Y1R in antidepressant-like effects of electroacupuncture treatment on social isolation stress mice. *Purinergic Signalling*, *19*, 55–68. doi:10.1007/s11302-021-09827-1
- Yang, F., Zou, Y., Li, C., Li, J., Zang, Y., Peng, X., Wang, J., Liu, E.-H., Tong, S., & Chu, C. (2022). Discovery of potential hypoglycemic metabolites in Cassiae Semen by coupling UHPLC-QTOF-MS/MS combined plant metabolomics and spectrum-effect relationship analyses. *Food & Function*, *13*(19), 10291–10304. doi:10.1039/D2FO00562J
- Yuan, X., Niu, H., Wang, P., Lu, J., Zhao, H., Liu, S., Zheng, Q., & Li, C. (2015). Cardioprotective Effect of Licochalcone D against Myocardial Ischemia/Reperfusion Injury in Langendorff-Perfused Rat Hearts. *PLOS ONE*, *10*, e0128375. doi:10.1371/journal.pone.0128375
- Yuan, X., Li, D., Zhao, H., Jiang, J., Wang, P., Ma, X., Sun, X., & Zheng, Q. (2013). Licochalcone A-Induced Human Bladder Cancer T24 Cells Apoptosis Triggered by Mitochondria Dysfunction and Endoplasmic Reticulum Stress. *BioMed Research International*, 2013, 1–9. doi:10.1155/2013/474272

- Yun, J., & Finkel, T. (2014). Mitohormesis. *Cell Metabolism*, *19*, 757–766.doi:10.1016/j.cmet.2014.01.011
- Zadorozhna, M., & Mangieri, D. (2021). Mechanisms of Chemopreventive and Therapeutic Proprieties of Ginger Extracts in Cancer. *International Journal of Molecular Sciences*, *22*(12), 6599. doi:10.3390/ijms22126599
- Zampieri, L. X., Grasso, D., Bouzin, C., Brusa, D., Rossignol, R., & Sonveaux, P. (2020). Mitochondria Participate in Chemoresistance to Cisplatin in Human Ovarian Cancer Cells. *Molecular Cancer Research*, *18*, 1379–1391. doi:10.1158/1541-7786.MCR-19-1145
- Zeng, J.-G., Tan, M.-L., Peng, X., & Luo, Q. (2011). 'Standardization and Quality Control of Herbal Extracts and Products' in W. J. H. Liu, (ed) *Traditional Herbal Medicine Research Methods: Identification, Analysis, Bioassay, and Pharmaceutical and Clinical Studies*. Hoboken, NJ: John Wiley & Sons, Inc. pp. 377-428.
- Zeng, Y., Zhang, J., Xu, M., Chen, F., Zi, R., Yue, J., Zhang, Y., Chen, N., &
 Chin, Y. E. (2021). Roles of Mitochondrial Serine
 Hydroxymethyltransferase 2 (SHMT2) in Human Carcinogenesis. *Journal of Cancer*, *12*, 5888–5894. doi:10.7150/jca.60170
- Zhang, Z. D., & Cox, G. (1996). MTT assay overestimates human airway smooth muscle cell number in culture. *Biochemistry and Molecular Biology International*, 38, 431–436. Available from

https://www.ncbi.nlm.nih.gov/pubmed/8829601 (accessed 28 January 2024).

- Zhang, Z., & Liu, G. (2016). *Discussion of Cold Damage (Shang Han Lun)*. London: Singing Dragon.
- Zhang, Y., & Wong, H. S. (2021). Are mitochondria the main contributor of reactive oxygen species in cells? *Journal of Experimental Biology*, 224(5). doi:/10.1242/jeb.221606
- Zhang, W., Huai, Y., Miao, Z., Qian, A., & Wang, Y. (2019). Systems
 Pharmacology for Investigation of the Mechanisms of Action of Traditional
 Chinese Medicine in Drug Discovery. *Frontiers in pharmacology*, 10, 743.
 doi:10.3389/fphar.2019.00743
- Zhang, X., Liao, W., Ding, X., Zhang, Y., Long, C., Zhou, Q., Wang, Y., Wu,
 H., & Tan, G. (2024). Multiple-matrices metabolomics combined with serum pharmacochemistry for discovering the potential targets and active constituents of Qifu decoction against heart failure. *Journal of Pharmaceutical and Biomedical Analysis*, 244, 116114. doi:10.1016/j.jpba.2024.116114
- Zhang, Z., Chen, W., Zhang, S., Bai, J., Liu, B., Yung, K. K.-L., & Ko, J. K.-S. (2022). Isoliquiritigenin inhibits pancreatic cancer progression through blockade of p38 MAPK-regulated autophagy. *Phytomedicine*, 106, 154406. doi:10.1016/j.phymed.2022.154406

- Zhang, M., Geng, T., Jing, S.-H., Li, L., Wang, K., Zhang, Q., Zhang, Y., Zhou, S.-K., Gao, P., Ding, A.-W., & Zhang, L. (2022). Chemical profile and miscarriage prevention evaluation of Jiao-Ai Decoction, a classical traditional Chinese formula. *Journal of Pharmaceutical and Biomedical Analysis*, 217, 114832. doi:10.1016/j.jpba.2022.114832
- Zhang, B., Pan, C., Feng, C., Yan, C., Yu, Y., Chen, Z., Guo, C., & Wang, X.
 (2022). Role of mitochondrial reactive oxygen species in homeostasis regulation. *Redox Report*, 27, 45–52. doi:10.1080/13510002.2022.2046423
- Zhang, X., Wang, J., Liang, J., Liu, Z., Shen, X., Liu, Y., Li, X., Xu, Z., Lei, Y.,
 & Lei, H. (2022). A novel self-aggregated gold nanoparticles based on sensitive immunochromatographic assays for highly detection of opium poppy in herbal teas. *Food Chemistry*, 390, 133188. doi:10.1016/j.foodchem.2022.133188
- Zhang, C., Li, X., Li, P., Jiang, Y., & Li, H. (2021). Consistency evaluation between dispensing granule and traditional decoction from Coptidis Rhizoma by using an integrated quality-based strategy. *Phytochemical Analysis*, 32, 153–164. doi:10.1002/pca.2905
- Zhang, H., Han, X., Wang, Z., Wang, Z., Cui, Y., Tian, R., Zhu, Y., Han, B.,
 Liu, H., Zuo, X., Ren, S., Tian, J., Niu, R., & Zhang, F. (2021).
 Mitochondrial Breast Cancer Resistant Protein Sustains the Proliferation and Survival of Drug-Resistant Breast Cancer Cells by Regulating

Intracellular Reactive Oxygen Species. *Frontiers in Cell and Developmental Biology*, 9. doi:10.3389/fcell.2021.719209

- Zhang, H., Chi, X., Pan, W., Wang, S., Zhang, Z., Zhao, H., Wang, Y., Wu, Z., Zhou, M., Ma, S., Zhao, Q., & Ma, K. (2020). Antidepressant mechanism of classical herbal formula lily bulb and Rehmannia decoction: insights from gene expression profile of medial prefrontal cortex of mice with stress-induced depression-like behavior. *Genes, Brain and Behavior, 19.* doi:10.1111/gbb.12649
- Zhang, R., Tan, Z., Zhao, J., Wen, Y., Fan, S., & Liu, C. (2020). Determination of pyrethroid residues in herbal tea using temperature-controlled ionic liquid dispersive liquid-liquid microextraction by high performance liquid chromatography. *Scientific Reports*, *10*(1), 4709. doi:10.1038/s41598-020-61755-z
- Zhang, J.-Q., Wang, R., Zhou, T., Zhao, Q., Zhao, C.-C., & Ma, B.-L. (2020).
 Pharmacokinetic incompatibility of the Huanglian-Gancao herb pair. *BMC Complementary Medicine and Therapies*, 20(1), 61.
 doi:10.1186/s12906-020-2845-5
- Zhang, H., Wang, J., Chen, Y., Shen, X., Jiang, H., Gong, X., & Yan, J. (2019). Establishing the chromatographic fingerprint of traditional Chinese medicine standard decoction based on quality by design approach: A case study of *Licorice*. *Journal of Separation Science*, *42*, 1144–1154. doi:10.1002/jssc.201800989

- Zhang, M., Wang, C., Cai, H., Wen, J., & Fang, P. (2019). Licorice Extracts Attenuate Nephrotoxicity Induced by Brucine Through Suppression of Mitochondria Apoptotic Pathway and STAT3 Activation. *Current Medical Science*, 39(6), 890–898. doi:10.1007/s11596-019-2126-z
- Zhang, C., Liu, Z., Zheng, Y., Geng, Y., Han, C., Shi, Y., Sun, H., Zhang, C.,
 Chen, Y., Zhang, L., Guo, Q., Yang, L., Zhou, X., & Kong, L. (2018).
 Glycyrrhetinic Acid Functionalized Graphene Oxide for Mitochondria
 Targeting and Cancer Treatment In Vivo. *Small*, *14*, 1703306.
 doi:10.1002/smll.201703306
- Zhang, J., Wang, X., Vikash, V., Ye, Q., Wu, D., Liu, Y., & Dong, W. (2016). ROS and ROS-Mediated *Cellular Signaling. Oxidative Medicine and Cellular Longevity*, 2016, 1–18. doi:10.1155/2016/4350965
- Zhang, C.-E., Xiong, Y., Dong, Q., Gao, D., Zhang, L.-L., Ma, L.-N., Peng, C., Dong, X.-P., & Yan, D. (2014). Comparison of reversed-phase liquid chromatography and hydrophilic interaction chromatography for the fingerprint analysis of *Radix isatidis*. *Journal of Separation Science*, *37*, 1141–1147. doi:10.1002/jssc.201301159
- Zhang, L., Yang, Y., Yu, L., Wang, Y., Liu, L., & Fan, X. (2011). Cardioprotective Effects of *Glycyrrhiza uralensis* Extract Against Doxorubicin- Induced Toxicity. *International Journal of Toxicology*, 30, 181–189. doi:10.1177/1091581810393033

- Zhang, Z., Yung, K. K.-L., & Ko, J. K.-S. (2022). Therapeutic Intervention in Cancer by Isoliquiritigenin from Licorice: A Natural Antioxidant and Redox Regulator. Antioxidants, 11(7), 1349. doi:10.3390/antiox11071349
- Zhang, H.-M., & Zhang, Y. (2014). Melatonin: a well-documented antioxidant with conditional pro-oxidant actions. *Journal of Pineal Research*, 57(2), 131–146. doi:10.1111/jpi.12162
- Zhao, Z. (2017). The Original Source of Modern Research on Chinese Medicinal Materials: Bencao Texts. *Alternative, Complementary* & *Integrative Medicine*, 3, 1–9. doi:10.24966/ACIM-7562/100045
- Zhao, M., Xia, W., Zhang, P., Xie, Q., Mu, W., Tang, L., Liu, Z., Han, L., & Peng, D. (2024). Ultra-performance liquid chromatography-quadrupole time-of-flight mass combined with UNIFI to study the mechanism of Tao Hong Si Wu Decoction in the treatment of postpartum blood stasis. *Journal of Separation Science*, *47*(5). doi:10.1002/jssc.202300871
- Zhao, M., Linghu, K.-G., Xiao, L., Hua, T., Zhao, G., Chen, Q., Xiong, S.,
 Shen, L., Yu, J., Hou, X., Hao, E., Du, Z., Deng, J., Bai, G., Chen, X., Li,
 L., Li, P., & Yu, H. (2022). Anti-inflammatory/anti-oxidant properties and
 the UPLC-QTOF/MS-based metabolomics discrimination of three yellow
 camellia species. *Food Research International*, *160*, 111628.
 doi:10.1016/j.foodres.2022.111628

- Zhao, R., Jiang, S., Zhang, L., & Yu, Z. (2019). Mitochondrial electron transport chain, ROS generation and uncoupling (Review). *International Journal of Molecular Medicine*. doi:10.3892/ijmm.2019.4188
- Zhao, J., Wu, H.-L., Niu, J.-F., Yu, Y.-J., Yu, L.-L., Kang, C., Li, Q., Zhang, X.-H., & Yu, R.-Q. (2012). Chemometric resolution of coeluting peaks of eleven antihypertensives from multiple classes in high performance liquid chromatography: A comprehensive research in human serum, health product and Chinese patent medicine samples. *Journal of Chromatography B*, 902, 96–107. doi:10.1016/j.jchromb.2012.06.032
- Zhao, Z., Guo, P., & Brand, E. (2012). The formation of daodi medicinal materials. *Journal of Ethnopharmacology*, 140(3), 476–481.
 doi:10.1016/j.jep.2012.01.048
- Zhao, Z., Chen, H. (2014). *Chinese Medicinal Identification: An Illustrated Approach* (P. Guo & E. Brand, eds.). Taos, NM: Paradigm Publications.
- Zheng, J. (2012). Energy metabolism of cancer: Glycolysis versus oxidative phosphorylation (Review). Oncology Letters, 4, 1151–1157.
 doi:10.3892/ol.2012.928
- Zheng, Y., Liu, S., Fan, C., Zeng, H., Huang, H., Tian, C., Lu, Z., Cao, H.,
 Liu, J., & Yu, L. (2020). Holistic quality evaluation of Qingwen Baidu
 Decoction and its anti-inflammatory effects. *Journal of Ethnopharmacology*, 263, 113145. doi:10.1016/j.jep.2020.113145

- Zhou, M., Wu, J., Chang, H., Fang, Y., Zhang, D., & Guo, Y. (2023).
 Adenosine signaling mediate pain transmission in the central nervous system. *Purinergic Signalling*, 19, 245–254. doi:10.1007/s11302-021-09826-2
- Zhou, J., Nie, R.-C., Yin, Y.-X., Cai, X.-X., Xie, D., & Cai, M.-Y. (2022).
 Protective Effect of Natural Antioxidants on Reducing Cisplatin-Induced
 Nephrotoxicity. *Disease Markers*, 2022, 1–17. doi:10.1155/2022/1612348
- Zhou, H., Liu, N., Yan, Z., Yu, D., Wang, L., Wang, K., Wei, X., & Wu, A. (2021). Development and validation of the one-step purification method coupled to LC-MS/MS for simultaneous determination of four aflatoxins in fermented tea. *Food Chemistry*, 354, 129497. doi:10.1016/j.foodchem.2021.129497
- Zhou, B., Wang, H., Zhang, B., & Zhang, L. (2021). Licochalcone B attenuates neuronal injury through anti-oxidant effect and enhancement of Nrf2 pathway in MCAO rat model of stroke. *International Immunopharmacology*, *100*, 108073. doi:10.1016/j.intimp.2021.108073
- Zhou, X., Li, Y., Zhang, M., Hao, J., Gu, Q., Liu, H., Chen, W., Shi, Y., Dong,
 B., Zhang, Y., Li, C., & Li, G. (2019). Spectrum-Effect Relationship
 between UPLC Fingerprints and Antilung Cancer Effect of Si Jun Zi Tang. *Evidence-Based Complementary and Alternative Medicine*, 2019, 1–9.
 doi:10.1155/2019/7282681

- Zhou, Y., Li, X., Gong, W., Tian, J., Gao, X., Gao, L., Zhang, X., Du, G., & Qin, X. (2017). Protective effect of isoliquiritin against corticosterone-induced neurotoxicity in PC12 cells. *Food & Function*, *8*, 1235–1244. doi:10.1039/C6FO01503D
- Zhou, Xian, Seto, S. W., Chang, D., Kiat, H., Razmovski-Naumovski, V., Chan, K., & Bensoussan, A. (2016). Synergistic Effects of Chinese Herbal Medicine: A Comprehensive Review of Methodology and Current Research. *Frontiers in Pharmacology*, 7. doi:10.3389/fphar.2016.00201
- Zhu, H. (2014). Acupoints Initiate the Healing Process. *Medical Acupuncture*, 26, 264–270. doi:10.1089/acu.2014.1057
- Zhu, Y., Li, Y., Zhang, Q., Song, Y., Wang, L., & Zhu, Z. (2022). Interactions
 Between Intestinal Microbiota and Neural Mitochondria: A New
 Perspective on Communicating Pathway From Gut to Brain. *Frontiers in Microbiology*, *13*. doi:10.3389/fmicb.2022.798917
- Zhurinov, M. Zh., Miftakhova, A. F., Keyer, V., Shulgau, Z. T., Solodova, E. v., Kalykberdiyev, M. K., Abilmagzhanov, A. Z., Talgatov, E. T., Ait, S., & Shustov, A. v. (2023). Glycyrrhiza glabra L. Extracts and Other Therapeutics against SARS-CoV-2 in Central Eurasia: Available but Overlooked. *Molecules*, 28(16), 6142. doi:10.3390/molecules28166142
- Zielonka, J., & Kalyanaraman, B. (2010). Hydroethidine- and MitoSOX-derived red fluorescence is not a reliable indicator of intracellular superoxide formation: Another inconvenient truth. *Free*

RadicalBiologyandMedicine,48,983–1001.doi:10.1016/j.freeradbiomed.2010.01.028

- Ziemlewska, A., Zagórska-Dziok, M., & Nizioł-Łukaszewska, Z. (2021). Assessment of cytotoxicity and antioxidant properties of berry leaves as by-products with potential application in cosmetic and pharmaceutical products. *Scientific Reports*, *11*, 3240. doi:10.1038/s41598-021-82207-2
- Zinkel, B. (2019). What is NMR spectroscopy and How Does it Work? *Benchtop NMR Blog.* Retrieved from https://www.nanalysis.com/nmready-blog/2019/6/26/what-is-nmr-spectrog raphy-and-how-does-it-work (accessed 14 November 2020).
- Zisapel, N. (2018). New perspectives on the role of melatonin in human sleep, circadian rhythms and their regulation. *British Journal of Pharmacology*, *175*, 3190–3199. doi:10.1111/bph.14116
- Zoani, E. (2019). SOP No: PH-SOP-HHL-002. Organoleptic Sample Preparation Version 2. *Pukka Standard Operating Procedure*. Personal communication by email from Marion Mackonochie, 1 October 2020.
- Zore, G. B., Winston, U. B., Surwase, B. S., Meshram, N. S., Sangle, V. D.,
 Kulkarni, S. S., & Karuppayil, S. M. (2008). Chemoprofile and bioactivities of Taverniera cuneifolia (Roth) Arn.: A wild relative and possible substitute of Glycyrrhiza glabra L. *Phytomedicine*, *15*, 292–300. doi:10.1016/j.phymed.2007.01.006

- Zou, Y., Yang, R., Li, L., Xu, X., & Liang, S. (2023). Purinergic signaling: a potential therapeutic target for depression and chronic pain. *Purinergic Signalling*, *19*, 163–172. doi:10.1007/s11302-021-09801-x
- Zuo, J., Meng, T., Wang, Y., & Tang, W. (2023). A Review of the Antiviral Activities of Glycyrrhizic Acid, Glycyrrhetinic Acid and Glycyrrhetinic Acid Monoglucuronide. *Pharmaceuticals*, *16*, 641. doi:10.3390/ph16050641
- Zuo, L., & Wijegunawardana, D. (2021). 'Redox Role of ROS and Inflammation in Pulmonary Diseases' in Y.-X. Wang (ed.), *Lung Inflammation in Health and Disease*, vol 2. Cham: Springer Nature. pp.187–204. doi:10.1007/978-3-030-68748-9 11
- Zutz, A., Gompf, S., Schägger, H., & Tampé, R. (2009). Mitochondrial ABC proteins in health and disease. *Biochimica et Biophysica Acta (BBA) Bioenergetics*, *1787*, 681–690. doi:10.1016/j.bbabio.2009.02.009
- Zwerger, M., Zelger, J., & Ganzera, M. (2023). Separation of pyrrolizidine alkaloids in different Senecio species using ultra-high performance supercritical fluid chromatography. *Journal of Pharmaceutical and Biomedical Analysis*, 228, 115310. doi:10.1016/j.jpba.2023.115310